10-Q 1 d581127d10q.htm FORM 10-Q

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Form 10-Q

or

(Mark One)

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the Quarterly Period Ended September 30, 2013

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the transition period from to

Commission file number 1-12154

Waste Management, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 73-1309529 (I.R.S. Employer Identification No.)

1001 Fannin Suite 4000

Houston, Texas 77002 (Address of principal executive offices)

(713) 512-6200 (Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes [No ~

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes [No *

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer [Accelerated filer " Non-accelerated filer " Smaller reporting company "
(Do not check if a smaller reporting company)

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes " No [

The number of shares of Common Stock, \$0.01 par value, of the registrant outstanding at October 18, 2013 was 469,186,159 (excluding treasury shares of 161,096,302).

Item 1. FinancialStatements.		
WASTE MANAGEMENT, INC.		
CONDENSED CONSOLIDATED BALANCE SHEETS (In Millions, Except Share and Par Value Amounts)		
	September 30, 2013 (Unaudited)	December 31, 2012
ASSETS	(0)	
Current assets:		
Cash and cash equivalents	\$ 157	\$ 194
Accounts receivable, net of allowance for doubtful accounts of \$40 and \$45, respectively Other receivables	1,751 109	1,737 102
Parts and supplies	109	102
Deferred income taxes	84	76
Other assets	138	140
Total current assets	2,417	2,423
Property and equipment, net of accumulated depreciation and amortization of \$16,805 and \$16,112, respectively	12,621	12,651
Goodwill	6,593	6,291
Other intangible assets, net Investments in unconsolidated entities	569 656	397 667
Investments in unconsolidated entries Other assets	757	668
Total assets	\$ 23,613	\$ 23,097
	• 23,015	0 23,077
LIABILITIES AND EQUITY		
Current liabilities:		
Accounts payable	\$ 763	\$ 842
Accrued liabilities Deferred revenues	1,059 472	986 465
Current portion of long-term debt	472 568	405 743
Total current liabilities	2.862	3.036
Long-term debt, less current portion	9,491	9,173
Deferred income taxes	1,994	1,947
Landfill and environmental remediation liabilities	1,507	1,459
Other liabilities	744	807
Total liabilities	16,598	16,422
Commitments and contingencies		
Equity: Waste Management, Inc. stockholders' equity:		
Common stock, \$0.01 par value; 1,500,000,000 shares authorized; 630,282,461 shares issued	6	6
Additional paid-in capital	4,582	4,549
Retained earnings	7,066	6,879
Accumulated other comprehensive income	169	193
Treasury stock at cost, 161,100,606 and 166,062,235 shares, respectively	(5,115)	(5,273)
Total Waste Management, Inc. stockholders' equity	6,708	6,354
Noncontrolling interests	307	321
Total equity	7,015	6,675
Total liabilities and equity	\$ 23,613	\$ 23,097

PART I.

See notes to Condensed Consolidated Financial Statements.

WASTE MANAGEMENT, INC. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (In Millions, Except per Share Amounts) (Unaudited)

	Three Months Ended September 30,		Nine M Enc Septem	led
	2013	2012	2013	2012
Operating revenues	\$3,621	\$3,461	\$10,483	\$10,215
Costs and expenses:				
Operating	2,325	2,229	6,845	6,655
Selling, general and administrative	349	335	1,092	1,116
Depreciation and amortization	344	331	1,006	971
Restructuring	3	44	13	51
(Income) expense from divestitures, asset impairments and unusual items	23	22	38	55
	3,044	2,961	8,994	8,848
Income from operations	577	500	1,489	1,367
Other income (expense):				
Interest expense	(119)	(123)	(363)	(366)
Interest income		2	2	4
Equity in net losses of unconsolidated entities	(3)	(17)	(19)	(35)
Other, net	(3)	(14)	(12)	(16)
	(125)	(152)	(392)	(413)
Income before income taxes		348	1,097	954
Provision for income taxes	155	125	368	329
Consolidated net income	297	223	729	625
Less: Net income attributable to noncontrolling interests	6	9	26	32
Net income attributable to Waste Management, Inc.	\$ 291	\$ 214	\$ 703	\$ 593
Basic earnings per common share	\$ 0.62	\$ 0.46	\$ 1.50	\$ 1.28
Diluted earnings per common share	\$ 0.62	\$ 0.46	\$ 1.50	\$ 1.28
Cash dividends declared per common share	\$0.365	\$0.355	\$ 1.095	\$ 1.065

See notes to Condensed Consolidated Financial Statements.

WASTE MANAGEMENT, INC. CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (In Millions) (Unaudited)

	Three Months Ended September 30,		Nine Months Ended September 30,	
Consolidated net income	2013 \$297	2012 \$223	2013 \$729	2012 \$625
Other comprehensive income (loss), net of taxes:	<u>4277</u>	<u> </u>	<u> </u>	<u>4025</u>
Gains and losses on derivative instruments:				
Unrealized gains (losses), resulting from changes in fair value, net of tax expense (benefit) of \$(4), \$(8), \$6 and \$(16), respectively	(5)	(13)	10	(25)
Reclassification adjustment for (gains) losses included in net income, net of tax (expense) benefit of \$4, \$6, \$1 and \$6, respectively	7	10	2	10
	2	(3)	12	(15)
Unrealized gains (losses) on available-for-sale securities, net of tax expense (benefit) of \$0, \$0, \$0 and \$1, respectively	1	1	1	2
Foreign currency translation adjustments	32	39	(37)	41
Change in funded status of post-retirement benefit obligation, net of tax benefit of \$0, \$0, \$0 and \$0, respectively	_	_	_	
Other comprehensive income (loss), net of taxes	35	37	(24)	28
Comprehensive income	332	260	705	653
Less: Comprehensive income attributable to noncontrolling interests	6	9	26	32
Comprehensive income attributable to Waste Management, Inc.	\$326	\$251	\$679	\$621

See notes to Condensed Consolidated Financial Statements.

http://markets.on.nytimes.com//research/modules/company_topic/drawFiling.asp?docKey... 11/25/2013

WASTE MANAGEMENT, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In Millions) (Unaudited)

	Nine Months Ende	d September 30,
	2013	2012
Cash flows from operating activities:		
Consolidated net income	\$ 729	\$ 625
Adjustments to reconcile consolidated net income to net cash provided by operating activities:		
Depreciation and amortization	1,006	971
Deferred income tax (benefit) provision	33	12
Interest accretion on landfill liabilities	65	62
Interest accretion on and discount rate adjustments to environmental remediation liabilities and recovery assets	(6)	2
Provision for bad debts	27	39
Equity-based compensation expense	45	21
Excess tax benefits associated with equity-based transactions	(8)	(10)
Net gain from disposal of assets	(15)	(11)
Effect of (income) expense from divestitures, asset impairments and unusual items and other	49	69
Equity in net losses of unconsolidated entities, net of dividends	19	35
Change in operating assets and liabilities, net of effects of acquisitions and divestitures:		
Receivables	14	(112)
Other current assets	(9)	(45)
Other assets	(1)	102
Accounts payable and accrued liabilities	1	7
Deferred revenues and other liabilities	(91)	(49)
Net cash provided by operating activities	1,858	1,718
Cash flows from investing activities:		
Acquisitions of businesses, net of cash acquired	(698)	(178)
Capital expenditures	(824)	(1,132)
Proceeds from divestitures of businesses (net of cash divested) and other sales of assets	113	28
Net receipts from (deposits to) restricted trust and escrow accounts	58	15
Investments in unconsolidated entities	(29)	(61)
Other	(52)	(33)
Net cash used in investing activities	(1,432)	(1,361)
Cash flows from financing activities:		
New borrowings	1,218	685
Debt repayments	(1,244)	(473)
Cash dividends	(512)	(493)
Exercise of common stock options	116	39
Excess tax benefits associated with equity-based transactions	8	10
Distributions paid to noncontrolling interests	(40)	(31)
Other	(5)	44
Net cash used in financing activities	(459)	(219)
Effect of exchange rate changes on cash and cash equivalents	(4)	2
Increase (decrease) in cash and cash equivalents	(37)	140
Cash and cash equivalents at beginning of period	194	258
Cash and cash equivalents at end of period	\$ 157	\$ 398
Second and Constant Constant Constituted Firms and Constant		

See notes to Condensed Consolidated Financial Statements.

WASTE MANAGEMENT, INC. CONDENSED CONSOLIDATED STATEMENT OF CHANGES IN EQUITY (In Millions, Except Shares in Thousands) (Unaudited)

		Waste Management, Inc. Stockholders' Equity							
		Comm	on Stock	Additional		Accumulated Other	Treasu	y Stock	
	Total	Shares	Amounts	Paid-In Capital	Retained Earnings	Comprehensive Income (Loss)	Shares	Amounts	Noncontrolling Interests
Balance, December 31, 2012	\$6,675	630,282	\$ 6	\$ 4,549	\$ 6,879	\$ 193	(166,062)	\$ (5,273)	\$ 321
Consolidated net income	729			_	703	_		_	26
Other comprehensive income (loss), net of taxes	(24)	_	_	_	_	(24)	_	_	_
Cash dividends declared	(512)	_	_	_	(512)	_	_	_	_
Equity-based compensation transactions, including dividend equivalents, net of taxes	186	_	_	32	(4)	_	4,956	158	_
Distributions paid to noncontrolling interests	(40)	_	_	_	_	_	_	_	(40)
Other	1	_	_	1	_	_	5	_	_
Balance, September 30, 2013	\$7,015	630,282	\$ 6	\$ 4,582	\$ 7,066	\$ 169	(161,101)	\$ (5,115)	\$ 307

See notes to Condensed Consolidated Financial Statements.

1. Basis of Presentation

The financial statements presented in this report represent the consolidation of Waste Management, Inc., a Delaware corporation; Waste Management's wholly-owned and majority-owned subsidiaries; and certain variable interest entities for which Waste Management or its subsidiaries are the primary beneficiaries as described in Note 14. Waste Management is a holding company and all operations are conducted by its subsidiaries. When the terms "the Company," "us," out" are used in this document, those terms refer to Waste Management, Inc., its consolidated subsidiaries and consolidated variable interest entities. When we use the term "WM," we are referring only to Waste Management, Inc., the parent holding company.

We are North America's leading provider of comprehensive waste management environmental services. We partner with our residential, commercial, industrial and municipal customers and the communities we serve to manage and reduce waste at each stage from collection to disposal, while recovering valuable resources and creating clean, renewable energy. Our "Solid Waste" business is operated and managed locally by our subsidiaries that focus on distinct geographic areas and provides collection, transfer, recycling and resource recovery, and disposal services. Through our subsidiaries, we are also a leading developer, operator and owner of waste-to-energy and landfill gas-to-energy facilities in the United States.

We currently evaluate, oversee and manage the financial performance of our Solid Waste business subsidiaries through our 17 geographic Areas. Our Wheelabrator business provides waste-toenergy services and manages waste-to-energy facilities and independent power production plants. We also provide additional services that are not managed through our Solid Waste or Wheelabrator businesses, which are presented in this report as "Other." Additional information related to our segments can be found in Note 8.

The Condensed Consolidated Financial Statements as of September 30, 2013 and for the three and nine months ended September 30, 2013 and 2012 are unaudited. In the opinion of management, these financial statements include all adjustments, which, unless otherwise disclosed, are of a normal recurring nature, necessary for a fair presentation of the financial position, results of operations, comprehensive income, cash flows, and changes in equity for the periods presented. The results for interim periods are not necessarily indicative of results for the entire year. The financial statements included in our Annual Report on Form 10-K for the year ended December 31, 2012.

In preparing our financial statements, we make numerous estimates and assumptions that affect the accounting for and recognition and disclosure of assets, liabilities, equity, revenues and expenses. We must make these estimates and assumptions because certain information that we use is dependent on future events, cannot be calculated with precision from available data or simply cannot be calculated. In some cases, these estimates are difficult to determine, and we must exercise significant judgment. In preparing our financial statements, the most difficult, subjective and complex estimates and the assumptions that present the greatest amount of uncertainty relate to our accounting for landfills, environmental remediation liabilities, asset impairments, deferred income taxes and reserves associated with our insured and self-insured claims. Actual results could differ materially from the estimates and assumptions that we use in the preparation of our financial statements.

Adoption of New Accounting Pronouncement

Comprehensive Income — In February 2013, the Financial Accounting Standards Board issued amended authoritative guidance associated with comprehensive income, which requires companies to provide information about the amounts that are reclassified out of accumulated other comprehensive income by component. Additionally, companies are required to present significant amounts reclassified out of accumulated other comprehensive income by component. Additionally, companies are required to present significant amounts reclassified out of accumulated other comprehensive income by the respective line items of net income. The amendment to authoritative guidance associated with comprehensive income was effective for the Company on January 1, 2013. The adoption of this guidance did not have a material impact on our consolidated financial statements. We have presented the information required by this amendment in Note 12.



NOTES TO CONDENSED CONSOI							
	JIDATED FINA	NCIAL S	TATEMEN	NTS — (Contin	ued)		
eclassifications							
When necessary, reclassifications have been made to our prior period consolidated fin	nancial informatic	on in order	to conform	to the current y	ear presentation	n.	
andfill and Environmental Remediation Liabilities							
Liabilities for landfill and environmental remediation costs are presented in the table	below (in million	s):					
		Septemb	er 30, 2013			December 31, 2012	
	Landfill		onmental ediation	Total	Landfill	Environmental Remediation	Total
Current (in accrued liabilities)	\$ 99	\$	28	\$ 127	\$ 104	\$ 28	\$ 132
Long-term	1,299		208	1,507	1,234	225	1,459
	\$1,398	\$	236	\$1,634	\$1,338	\$ 253	\$1,591
The changes to landfill and environmental remediation liabilities for the year ended I)ecember 31 201	2 and the	nine months	ended Sentemb	per 30 2013 are	reflected in the tabl	
				· · · · · · · · · · · · · · · · · · ·			Environmental
						dfill	Remediation
December 31, 2011					\$1,	292	\$ 273
Obligations incurred and capitalized						58	(20)
Obligations settled Interest accretion						(87) 84	(30)
Revisions in cost estimates and interest rate assumptions						64 (8)	4
Acquisitions, divestitures and other adjustments						(1)	1
December 31, 2012					1	338	253
Obligations incurred and capitalized					1,	45	
Obligations settled						(53)	(15)
Interest accretion						65	3
Revisions in cost estimates and interest rate assumptions						_	(4)
Revisions in cost estimates and interest rate assumptions						3	(1)
Acquisitions, divestitures and other adjustments						2	(1)
					\$1,	398	\$ 236

At several of our landfills, we provide financial assurance by depositing cash into restricted trust funds or escrow accounts for purposes of settling final capping, closure, post-closure and environmental remediation obligations. Generally, these trust funds are established to comply with statutory requirements and operating agreements. See Note 14 for additional information related to these trusts.

3. Debt

The following table summarizes the major components of debt (in millions) at each balance sheet date and provides the maturities and interest rate ranges of each major category as of September 30, 2013

	September 30, 2013	December 31, 2012
\$2.25 billion revolving credit facility, maturing July 2018 (weighted average interest rate of 1.5% at September 30, 2013 and 1.4% at		
December 31, 2012)	\$ 140	\$ 400
Letter of credit facilities, maturing through June 2015	_	_
Canadian credit facility and term loan, maturing November 2017 (weighted average effective interest rate of 2.7% at September 30,		
2013 and 2.9% at December 31, 2012)	456	75
Senior notes maturing through 2039, interest rates ranging from 2.60% to 7.75% (weighted average interest rate of 5.7% at		
September 30, 2013 and December 31, 2012)	6,292	6,305
Tax-exempt bonds, maturing through 2045, fixed and variable interest rates ranging from 0.1% to 5.7% (weighted average interest rate		
of 2.4% at September 30, 2013 and 2.8% at December 31, 2012)	2,719	2,727
Capital leases and other, maturing through 2055, interest rates up to 12%	452	409
	10,059	9,916
Current portion of long-term debt	568	743
	\$ 9,491	\$ 9,173

Debt Classification

As of September 30, 2013, we had (i) \$510 million of debt maturing within the next 12 months, including \$350 million of 5.0% senior notes that mature in March 2014 and \$92 million of taxexempt bonds; (ii) short-term borrowings and advances outstanding under credit facilities with long-term maturities, including \$140 million of borrowings outstanding under the revolving credit facility and U.S.\$456 million of advances under our Canadian credit facility term loan; and (iii) \$857 million of tax-exempt borrowings subject to repricing within the next 12 months. Based on our intent and ability to refinance a portion of this debt on a long-term basis as of September 30, 2013, we have classified \$1.4 billion of this debt as long-term and the remaining \$568 million as.

Revolving Credit and Letter of Credit Facilities

As of September 30, 2013, we had an aggregate committed capacity of \$2.65 billion for letters of credit under various credit facilities. Our revolving credit facility is our primary source of letter of credit capacity. In July 2013, we anended and restated our revolving credit facility, increasing our total credit capacity to \$2.25 billion and extending the term through July 2018. As of September 30, 2013, we had an \$1.5 billion of letters of credit totals credit facility. We had an aggregate of \$1.5 billion of letters of credit totals credit facility. We had \$140 million of heuse letters of credit have been issued under our revolving credit facility. We had \$140 million of outstanding borrowings under our revolving credit facility as of September 30, 2013, leaving \$1.2 billion of unused and available capacity.

Significant Changes in Debt Balances

Revolving credit facility - During the nine months ended September 30, 2013, we have had net repayments under our revolving credit facility of \$260 million.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS - (Continued)

Canadian credit facility — We repaid U.S.\$75 million of net advances under our Canadian credit facility during the nine months ended September 30, 2013. Additionally, we borrowed C\$500 million, or U.S.\$470 million, under a term loan to fund our acquisition of RCI Environnement, Inc., which is discussed further in Note 9. Under the terms and conditions of our Canadian credit facility, interest on the term loan is based on a 1.4% spread from the applicable Canadian Dealer Offered Rate (CDOR). Our outstanding CDOR-based advances, which have terms of one month and three months, may be renewed under the terms of the facility through its maturity. We repaid U.S.\$29 million of advances under our term loan during the nine months ended September 30, 2013. Accounting for changes in the Canadian currency translation rate and the recognition of interest expense increased the carrying value of our Canadian borrowings by U.S.\$11 million and U.S.\$4 million, respectively, during the nine months ended September 30, 2013.

Tax-exempt bonds — During the nine months ended September 30, 2013, we repaid \$107 million of our tax-exempt bonds with cash. We issued \$100 million of tax-exempt bonds in August 2013. The proceeds from the issuance of the bonds were deposited directly into a trust fund and may only be used for the specific purpose for which the money was raised, which is generally to finance expenditures for landfill and recycling facility construction and development. Accordingly, the restricted funds provided by these financing activities have not been included in "New Borrowings" in our Condensed Consolidated Statement of Cash Flows.

Other — We had increases in our other obligations during the nine months ended September 30, 2013, primarily related to the deferred purchase price of (i) land needed to support a landfill expansion and (ii) Greenstar, LLC, which is discussed further in Note 9.

4. Derivative Instruments and Hedging Activities

The following table summarizes the fair values of derivative instruments recorded in our Condensed Consolidated Balance Sheet (in millions):

Derivatives Designated as Hedging Instruments	Balance Sheet Location	September 30, 2013	December 31, 2012
Electricity commodity derivatives	Current other assets	\$	\$ 1
Total derivative assets		\$	\$ 1
Interest rate derivatives	Current accrued liabilities	\$ 31	\$
Electricity commodity derivatives	Current accrued liabilities	3	5
Foreign currency derivatives	Current accrued liabilities	_	11
Interest rate derivatives	Long-term other liabilities		42
Total derivative liabilities		\$ 34	\$ 58

We have not offset fair value amounts recognized for our derivative instruments. For information related to the inputs used to measure our derivative assets and liabilities at fair value, refer to Note

Fair Value Hedges

13.

Interest Rate Swaps

In prior years, we used interest rate swaps to maintain a portion of our debt obligations at variable market interest rates. As of September 30, 2013 and December 31, 2012, we did not have any of these swaps outstanding. In April 2012, we elected to terminate our interest rate swaps related to the interest payments on \$1 billion of our senior notes and, upon termination, we received \$76 million in cash for their fair value plus accrued interest receivable. The terminated interest rate swaps were associated with senior notes with maturities between November 2012 and March 2018. The associated fair value adjustments to long-term debt are being amortized as a reduction to interest texpense over the remaining terms of the underlying debt using the effective interest method.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS - (Continued)

We designated our interest rate swaps as fair value hedges of our fixed-rate senior notes. Fair value hedge accounting for interest rate swap contracts increased the carrying value of our debt instruments by \$64 million as of September 30, 2013 and \$79 million as of December 31, 2012, representing the unamortized portion of our terminated swaps.

We recognize the impacts of (i) net periodic settlements of current interest on our active interest rate swaps, if any, and (ii) the amortization of previously terminated interest rate swap agreements as adjustments to interest expense. The following table summarizes the impact of periodic settlements of active swap agreements and the impact of terminated swap agreements on our results of operations (in millions):

	Three Mon	Three Months Ended		ths Ended	
	Septem	September 30,		September 30,	
Decrease to Interest Expense Due to Hedge Accounting for Interest Rate Swaps	2013	2012	2013	2012	
Periodic settlements of active swap agreements(a)	\$	\$	\$	\$ 7	
Terminated swap agreements	5	7	15	16	
	\$ 5	\$ 7	\$ 15	\$ 23	

(a) These amounts represent the net of our periodic variable-rate interest obligations and the swap counterparties' fixed-rate interest obligations. Our variable-rate obligations were based on a spread from the three-month LIBOR. These settlements have decreased due to our election to terminate our interest rate swap portfolio with a notional amount of \$1 billion in April 2012.

Cash Flow Hedges

Forward-Starting Interest Rate Swaps

In prior years, we entered into forward-starting interest rate swaps with a total notional value of \$525 million to hedge the risk of changes in semi-annual interest payments due to fluctuations in the forward ten-year LIBOR swap rate for anticipated fixed-rate debt issuances in 2011, 2012 and 2014. We designated these forward-starting interest rate swaps as cash flow hedges.

During the third quarter of 2012, \$200 million of these forward-starting interest rate swaps were terminated contemporaneously with the actual issuance of senior notes in September 2012, and we paid cash of \$59 million to settle the liabilities related to these swap agreements. The ineffectiveness recognized upon termination of these hedges was immaterial. As of September 30, 2013 and December 31, 2012, our "Accumulated other comprehensive income" included \$35 million and \$39 million, respectively, of after-tax deferred losses related to all previously terminated swaps, which are being amortized as an increase to interest expense over the ten-year life of the related senior note issuances using the effective interest method. As of September 30, 2013, \$7 million (on a pre-tax basis) is scheduled to be reclassified as an increase to interest expense over the next 12 months.

The active forward-starting interest rate swaps outstanding as of September 30, 2013 relate to an anticipated debt issuance to repay the \$350 million of 5.0% senior notes that mature in March 2014. As of September 30, 2013, the fair value of these active interest rate derivatives was comprised of \$31 million of current liabilities compared with \$42 million of long-term liabilities as of December 31, 2012.

Treasury Rate Locks

As of both September 30, 2013 and December 31, 2012, our "Accumulated other comprehensive income" included \$7 million of after-tax deferred losses associated with Treasury rate locks that had been executed in previous years in anticipation of senior note issuances. These deferred losses are reclassified as an increase to interest expense over the life of the related senior note issuances, which extend through 2032. As of September 30, 2013, \$1 million (on a pre-tax basis) is scheduled to be reclassified as an increase to interest expense over the next 12 months.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS - (Continued)

Foreign Currency Derivatives

We use foreign currency exchange rate derivatives to hedge our exposure to fluctuations in exchange rates for anticipated intercompany cash transactions between WM Holdings and one of its Canadian subsidiaries. As of September 30, 2013, we had foreign currency forward contracts outstanding for all of the anticipated cash flows associated with a debt arrangement between these wholly-owned subsidiaries. The hedged cash flows as of September 30, 2013 include C\$370 million of principal and C\$10 million of interest which are scheduled to be repaid on October 31, 2013. We designated these forward contracts as cash flow hedges. Gains or losses on the underlying hedged items attributable to foreign currency exchange risk are recognized in current earnings.

Electricity Commodity Derivatives

We use "receive fixed, pay variable" electricity commodity swaps to reduce the variability in our revenues and cash flows caused by fluctuations in the market prices for electricity. We hedged 628,800 megawatt hours, or approximately 20%, of Wheelabrator's full year 2012 merchant electricity sales, and the swaps executed through September 30, 2013 are expected to hedge about 1.75 million megawatt hours, or approximately 56%, of Wheelabrator's full year 2012 merchant electricity sales. For the three-month periods ended September 30, 2013 and 2012, we hedged 57% and 16%, respectively, of our merchant electricity sales. For the nine-month periods ended September 30, 2013 and 2012, we hedged 55% and 21%, respectively, of our merchant electricity sales.

There was no significant ineffectiveness associated with our cash flow hedges during the three and nine months ended September 30, 2013 or 2012. Refer to Note 12 for information regarding the impacts of our cash flow derivatives on our comprehensive income and results of operations.

Credit-Risk-Related Contingent Features

Our interest rate derivative instruments have in the past, and may in the future, contain provisions related to the Company's credit rating. These provisions generally provide that if the Company's credit rating were to fall to specified levels below investment grade, the counterparties have the ability to terminate the derivative agreements, resulting in settlement of all affected transactions. As of September 30, 2013 and December 31, 2012, we did not have any interest rate derivatives outstanding that contained these credit-risk-related features.

5. Income Taxes

Our effective income tax rate for the three and nine months ended September 30, 2013 was 34.3% and 33.6%, respectively, compared with 36.1% and 34.5%, respectively, for the comparable prior year periods. We evaluate our effective income tax rate at each interim period and adjust it as facts and circumstances warrant. The differences between federal income taxes computed at the federal statutory rate and reported income taxes for the reported periods were primarily due to the favorable impact of federal and state tax credits and tax audit settlements offset, in part, by the unfavorable impact of state and local income taxes and adjustments to our accruals and related deferred taxes due to the filing of our income tax runns.

Investment in Refined Coal Facility — In January 2011, we acquired a noncontrolling interest in a limited liability company established to invest in and manage a refined coal facility in North Dakota. The facility's refinement processes qualify for federal tax credits that are expected to be realized through 2019 in accordance with Section 45 of the Internal Revenue Code.

We account for our investment in this entity using the equity method of accounting, recognizing our share of the entity's results and other reductions in "Equity in net losses of unconsolidated entities," within our Condensed Consolidated Statements of Operations. During the three and nine months ended September 30, 2013, we recognized \$1 million and \$4 million of net losses resulting from our share of the entity's operating losses, respectively, and less than \$1 million and \$2 million during the three and nine months ended September 30, 2012, respectively. Our tax provision was reduced by \$5 million and \$14 million for the three and nine months

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS - (Continued)

ended September 30, 2013, respectively, and \$6 million and \$14 million for the three and nine months ended September 30, 2012, respectively, primarily as a result of tax credits realized from this investment. See Note 14 for additional information related to this investment.

Investment in Low-Income Housing Properties — In April 2010, we acquired a noncontrolling interest in a limited liability company established to invest in and manage low-income housing properties. The entity's low-income housing investments qualify for federal tax credits that are expected to be realized through 2020 in accordance with Section 42 of the Internal Revenue Code.

We account for our investment in this entity using the equity method of accounting. We recognize our share of the entity's results and reductions in the value of our investment in "Equity in net losses of unconsolidated entities," within our Condensed Consolidated Statements of Operations. The value of our investment decreases as the tax credits are generated and utilized. During the three and nine months ended September 30, 2013, we recognized \$6 million and \$18 million of losses relating to our equity investment in this entity, \$2 million and \$5 million of interest expense, and a reduction in our tax provision of \$11 million (including \$8 million (including \$19 million and \$5 million and \$18 million of losses relating to our equity investment in this entity, \$2 million and a reduction in our tax provision of \$11 million and \$18 million of tax credits, and \$28 million and \$18 million of accredits, respectively. See Note 14 for additional information related to this investment.

Recent Legislation — The American Taxpayer Relief Act of 2012 was signed into law on January 2, 2013 and includes an extension for one year of the bonus depreciation allowance. As a result, 50% of qualifying capital expenditures on property placed in service before January 1, 2014 can be depreciated immediately. The acceleration of deductions on 2013 qualifying capital expenditures resulting from the bonus depreciation provisions has no impact on our effective income tax rate for 2013 although it will reduce our cash taxes.

6. Earnings Per Share

Basic and diluted earnings per share were computed using the following common share data (shares in millions):

		Three Months Ended		onths ed
	Septemb	er 30,	September 30,	
	2013	2012	2013	2012
Number of common shares outstanding at end of period	469.2	463.9	469.2	463.9
Effect of using weighted average common shares outstanding	0.1	0.2	(1.5)	(0.6)
Weighted average basic common shares outstanding	469.3	464.1	467.7	463.3
Dilutive effect of equity-based compensation awards and other contingently issuable shares	1.9	0.8	1.5	0.9
Weighted average diluted common shares outstanding	471.2	464.9	469.2	464.2
Potentially issuable shares	12.9	16.5	12.9	16.5
Number of anti-dilutive potentially issuable shares excluded from diluted common shares outstanding	0.5	9.3	0.5	9.7

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS - (Continued)

7. Commitments and Contingencies

Financial Instruments — We have obtained letters of credit, surety bonds and insurance policies and have established trust funds and issued financial guarantees to support tax-exempt bonds, contracts, performance of landfill final capping, closure and post-closure requirements, environmental remediation, and other obligations. Letters of credit generally are supported by our revolving credit facility and other credit facilities established for that purpose. Surety bonds and insurance policies are supported by (i) a diverse group of third-party surety and insurance companies; (ii) wholly-owned insurance companies, the sole business of which is to issue surety bonds and/or insurance policies on our behalf.

Management does not expect that any claims against or draws on these instruments would have a material adverse effect on our consolidated financial statements. We have not experienced any unmanageable difficulty in obtaining the required financial assurance instruments for our current operations. In an ongoing effort to mitigate risks of future cost increases and reductions in available capacity, we continue to evaluate various options to access cost-effective sources of financial assurance.

Insurance — We carry insurance coverage for protection of our assets and operations from certain risks including automobile liability, general liability, real and personal property, workers' compensation, directors' and officers' liability, pollution legal liability and other coverages we believe are customary to the industry. Our exposure to loss for insurance claims is generally limited to the per incident deductible under the related insurance policy. Our exposure, however, could increase if our insurers are unable to meet their commitments on a timely basis.

We have retained a significant portion of the risks related to our automobile, general liability and workers' compensation claims programs. "General liability" refers to the self-insured portion of specific third party claims made against us that may be covered under our commercial General Liability Insurance Policy. For our self-insured retentions, the exposure for unpaid claims and associated expenses, including incurred but not reported losses, is based on an actuarial valuation and internal estimates. The accruals for these liabilities could be revised if future occurrences or loss development significantly differ from our assumptions. We do not expect the impact of any known casualty, property, environmental or other contingency to have a material impact on our financial condition, results of operations or cash flows.

Guarantees — In the ordinary course of our business, WM and WM Holdings enter into guarantee agreements associated with their subsidiaries' operations. Additionally, WM and WM Holdings have each guaranteed all of the senior debt of the other entity. No additional liabilities have been recorded for these intercompany guarantees because all of the underlying obligations are reflected in our Condensed Consolidated Balance Sheets.

We also have guaranteed the obligations and certain performance requirements of, and provided indemnification to, third parties in connection with both consolidated and unconsolidated entities. Guarantee agreements outstanding as of September 30, 2013 include (i) guarantees of unconsolidated entities' financial obligations maturing through 2020 for maximum future payments of \$10 million and (ii) agreements guaranteening certain market value losses for approximately 850 homeowners' properties adjacent to or near 21 of our landfills. Our indemnification obligations generally arise from divestitures and provide that we will be responsible for liabilities associated with our operations for events that occurred prior to the sale of the operations. Additionally, under certain of our acquisition agreements, we have provided for additional consideration to be paid to the sellers if established financial targets or other market conditions are achieved post-closing, and we have recognized liabilities or charge contingent obligations based on an estimate of the fair value of these contingencies at the time of acquisition. We do not currently believe that contingent obligations to provide indemnification or pay additional post-closing consideration in connection with our divestitures or acquisitions will have a material adverse effect on the Company's business, financial condition, results of operations or cash flows.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS - (Continued)

Environmental Matters — A significant portion of our operating costs and capital expenditures could be characterized as costs of environmental protection as we are subject to an array of laws and regulations, relating to the protection of the environment. Under current laws and regulations, we may have liabilities for environmental damage caused by our operations, or for damage caused by conditions that existed before we acquired a site. In addition to remediation activity required by state or local authorities, such liabilities include potentially responsible party, or PRP, investigations. The costs associated with these liabilities can include settlements, certain legal and consultant fees, as well as incremental internal and external costs directly associated with site investigation and clean-up.

Estimating our degree of responsibility for remediation is inherently difficult. We recognize and accrue for an estimated remediation liability when we determine that such liability is both probable and reasonably estimable. Determining the method and ultimate cost of remediation requires that a number of assumptions be made. There can sometimes be a range of reasonable estimates of the costs associated with the likely site remediation alternatives identified in the investigation of the extent of environmental impact. In these cases, we use the amount within the range that constitutes our best estimate. If no amount within a range appears to be a better estimate than any other, we use the amount that is the low end of such range. If we used the high ends of such ranges, our aggregate potential liability would be approximately \$190 million higher than the \$236 million recorded in the Condensed Consolidated Financial Statements as of September 30, 2013. Our ultimate responsibility may differ PRPs to contribute to the settlements of such liabilities, or other factors could require us to record additional liabilities. Our ongoing review of our remediation liabilities, in light of relevant internal and external facts and circumstances, could result in revisions to our accruals that could cause upward or downward adjustments to income from operations. These adjustments could be material in any given period.

As of September 30, 2013, we had been notified by the government that we are a PRP in connection with 78 locations listed on the EPA's Superfund National Priorities List, or NPL. Of the 78 sites at which claims have been made against us, 15 are sites we own. Each of the NPL sites we own was initially developed by others as a landfill disposal facility. At each of these facilities, we are working in conjuction with the government to evaluate or remediate identified site problems, and we have either agreed with other legally liable parties on an arrangement for sharing the costs of remediation or are working in working toward a cost-sharing agreement. We generally expect to receive any amounts due from other participating parties at or near the time that we make the remedial expenditures. The other 63 NPL sites, which we do not own, are at various procedural stages under the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, known as CERCLA or Superfund.

The majority of these proceedings involving NPL sites that we do not own are based on allegations that certain of our subsidiaries (or their predecessors) transported hazardous substances to the sites, often prior to our acquisition of these subsidiaries. CERCLA generally provides for liability for those parties owning, operating, transporting to or disposing at the sites. Proceedings arising under Superfund typically involve numerous waste generators and other waste transportation and disposal companies and seek to allocate or recover costs associated with site investigation and remediation, which costs could be substantial and could have a material adverse effect on our consolidated financial statements. At some of the sites at which we have been identified as a PRP, our liability is well defined as a consequence of a governmental decision and an agreement among liable parties as to the share each will pay for implementing that remedy. At other sites, where no remedy has been selected or the liable parties have been unable to agree on an appropriate allocation, our future costs are uncertain.

Item 103 of the SEC's Regulation S-K requires disclosure of certain environmental matters when a governmental authority is a party to the proceedings, or such proceedings are known to be contemplated, unless we reasonably believe that the matter will result in no monetary sanctions, or in monetary sanctions, exclusive of interest and costs, of less than \$100,000. The following matter is disclosed in accordance with that requirement.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS ---- (Continued)

We do not currently believe that the eventual outcome of such matter could have a material adverse effect on the Company's business, financial condition, results of operations or cash flows.

On December 22, 2011, the Harris County Attorney in Houston, Texas filed suit against McGinnes Industrial Maintenance Corporation ("MIMC"), WM and Waste Management of Texas, Inc., et. al, seeking civil penalties and attorneys' fees for alleged violations of the Texas Water Code and the Texas Health and Safety Code. The County's Original Petition pending in the District Court of Harris County, Texas alleges the mismanagement of certain waste pits that were operated from 1965 to 1966 by MIMC. In 1998, a predecessor of WM acquired the stock of the parent entity of MIMC.

Additionally, the United States Attorney's Office for the District of Hawaii has been conducting an investigation prompted by allegations of violations of the federal Clean Water Act involving discharge of stormwater at the Waimanalo Gulch Sanitary Landfill, located on Oahu, in connection with three major storm events in December 2010 and January 2011. No formal enforcement action has been brought against the Company. While we could potentially be subject to sanctions, including requirements to pay monetary penalties, in connection with a future proceeding that may arise from the investigation, a range of loss cannot currently be estimated because no proceeding has yet commenced and significant factual and legal issues remain. We are cooperating with the U.S. Attorney's Office.

Litigation — In October 2011 and January 2012, we were named as a defendant in a purported class action in the Circuit Court of Sarasota County, Florida and the Circuit Court of Lawrence County Alabama, respectively. These cases primarily pertain to our fuel and environmental charges included on our invoices, generally alleging that such charges were not properly disclosed, were unfair and were contrary to the customer service contracts. The law firm that filed these lawsuits had filed a purported class action in 2008 against subsidiaries of WM in Bullock County, Alabama, making similar allegations. The prior Alabama suit was removed to federal court, where the federal court ultimately dismissed the plaintiffs' national class action claims. The plaintiffs then elected to dismiss the case without prejudice. We will vigorously defend against these pending lawsuits. Given the inherent uncertainties of Iltigation, including the early stage of these cases, the unknown size of any potential class, and legal and factual issues in dispute, the outcome of these cases cannot be predicted and a range of loss cannot currently be estimated.

From time to time, we are also named as a defendant in personal injury and property damage lawsuits, including purported class actions, on the basis of having owned, operated or transported waste to a disposal facility that is alleged to have contaminated the environment or, in certain cases, on the basis of having conducted environmental remediation activities at sites. Some of the lawsuits may seek to have us pay the costs of monitoring of allegedly affected sites and health care examinations of allegedly affected persons for a substantial period of time, even where no actual damage is proven. While we believe we have meritorious defenses to these lawsuits, the ultimate resolution is often substantially uncertain due to the difficulty of determining the cause, extent and impact of alleged contamination (which may have occurred over a long period of time), the potential for successive groups of complainants to emerge, the diversity of the individual plaintiffs' circumstances, and the potential contribution or contractual conditions upon site closure or upon termination of the agreements. Compliance with these agreements is inherently subject to subjective determinations and may result in disputes, including litigation.

As a large company with operations across the United States and Canada, we are subject to various proceedings, lawsuits, disputes and claims arising in the ordinary course of our business. Many of these actions raise complex factual and legal issues and are subject to uncertainties. Actions filed against us include commercial, customer, and employment-related claims, including purported class action lawsuits related to our sales and marketing practices and our customer service agreements and purported class actions involving federal and state wage and hour and other laws. The plaintiffs in some actions seek unspecified damages or injunctive

relief, or both. These actions are in various procedural stages, and some are covered in part by insurance. We currently do not believe that the eventual outcome of any such actions could have a material adverse effect on the Company's business, financial condition, results of operations, or cash flows.

WM's charter and bylaws provide that WM shall indemnify against all liabilities and expenses, and upon request shall advance expenses to, any person who is subject to a pending or threatened proceeding because such person is a director or officer of the Company. Such indemnification is required to the maximum extent permitted under Delaware law. Accordingly, the director or officer must be company for any fees advanced if it is later determined that the director or officer and to note such fees advanced under Delaware law. Additionally, WM has entered into separate indemnification agreements with each of the members of its Board of Directors, its Chief Executive Officer and each of its executive vice presidents. The employment agreements between WM and its Chief Executive Officer and other executive and senior vice presidents also contain a direct contractual obligation of the Company to provide indemnification to the executive. The Company may incur substantial expenses in connection with the fulfillment of its advancement of costs and indemnification obligations in connection with actions or proceedings that may be brought against its former or current officers, directors and employees.

Multiemployer Defined Benefit Pension Plans — About 20% of our workforce is covered by collective bargaining agreements with various union locals across the United States and Canada. As a result of some of these agreements, certain of our subsidiaries are participating employers in a number of trustee-managed multiemployer defined benefit pension plans for the affected employees. In connection with our ongoing renegotiation of various collective bargaining agreements, we may discuss and negotiate for the complete or partial withdrawal from one or more of these pension plans. As a complete or partial withdrawal from a multiemployer pension plan may also occur if employees covered by a collective bargaining agreement to te decertify a union from continuing to represent them.

One of the most significant multiemployer pension plans in which we have participated is the Central States, Southeast and Southwest Areas Pension Plan ("Central States Pension Plan"). The Central States Pension Plan is in "critical status," as defined by the Pension Protection Act of 2006. Since 2008, certain of our affiliates have bargained to remove covered employees from the Central States Pension Plan, resulting in a series of withdrawals, and we have recognized charges to "Operating" expense associated with the withdrawal of certain bargaining units from the Central States Pension Plan and other underfunded multiemployer pension plans. In October 2011, employees at the last of our affiliates with active participants in the Central States Pension Plan voted to decertify the union that represented them, withdrawing themselves from the Central State Pension Plan.

We are still negotiating and litigating final resolutions of our withdrawal liability for certain previous withdrawals. Except in the case of our withdrawals from the Central States Pension Plan, we do not believe any additional liability above the charges we have already recognized for such previous withdrawals could be material to the Company's business, financial condition, liquidity, results of of the Central States Pension Plan. As a result, we do not anticipate that the final resolution of the Central States Pension Plan matter could be material to the Company's business, financial condition or liquidity; however, such loss could have a material adverse effect on our cash flows and, to a lesser extent, our results of operations, for a particular reporting period. Similarly, we also do not believe that any future withdrawals could have a material adverse effect on our cash flows and, to a lesser extent, our results of operations, for a particular reporting period. Similarly, we also do not believe that any future withdrawals could have a material adverse effect on our results of operations or cash flows and the financial condition or liquidity. However, such withdrawals could have a material adverse effect on our results of operations or cash flows for a particular reporting period, depending on the number of employees withdrawn in any future period and the financial condition of the time of such withdrawal(s).

Tax Matters — We are currently in the examination phase of IRS audits for the tax years 2012 and 2013 and expect these audits to be completed within the next six and 18 months, respectively. We participate in the IRS's

Compliance Assurance Process, which means we work with the IRS throughout the year in order to resolve any material issues prior to the filing of our annual tax return. We are also currently undergoing audits by various state and local jurisdictions for years that date back to 2000. We are not currently under audit in Canada and, due to the expiration of statutes of limitations, all tax years prior to 2009 are closed. In July 2011, we acquired Oakleaf Global Holdings ("Oakleaf"). Oakleaf is biget to potential IRS examinations for the years 2010 and 2011. Pursuant to the terms of our acquisition of Oakleaf", we are entitled to indemnification for Oakleaf's pre-acquisition period tax liabilities. We maintain a liability for uncertain tax positions, the balance of which management believes is adequate. Results of audit assessments by taxing authorities are not currently expected to have a material adverse impact on our results of operations or cash flows.

8. Segment and Related Information

Our senior management evaluates, oversees and manages the financial performance of our Solid Waste subsidiaries through our 17 geographic Areas. The 17 Areas constitute our operating segments and none of the Areas individually meet the quantitative criteria to be a separate reportable segment. We considered the similarities between our Areas, including the fact that our Solid Waste business is homogenous across geography with the same services offered across the Areas, and we concluded that aggregation of our Areas is appropriate for purposes of presenting our reportable segments. Accordingly, we have aggregated our 17 Areas into three tiers that we believe have similar economic characteristics and future prospects based in large part on a review of the Areas, including level of commercial and industrial activity; population density; service offering mix and disposal logistics, with no one factor being singularly determinative of an Area's current or future economic environment. As a result of our consideration of economic and other similarities, we have established the following three reportable segments for our Solid Waste business: Tier 1, which is comprised almost exclusively of Areas located in the Midwest and Northeast United States; and Tier 3, which encompasses all remaining Areas, including the Northwest and Mid-Atlantic regions of the United States and Eastern Canada. Our Wheelabrator business, which manages waste-to-energy facilities and independent power production plants, continues to be a separate reportable segment as it meets one of the quantitative disclosure thresholds. The operating segments and overseen through the 17 Areas or Wheelabrator are presented herein as "Other" as these operating segments do not meet the quantitative criteria to be aggregated with other operating segments and on ot meet the quantitative criteria to be separateled protein as the segments.

Summarized financial information concerning our reportable segments for the three and nine months ended September 30 is shown in the following table (in millions):

Three Months Ended:	Gross Operating Revenues	Intercompany Operating Revenues	Net Operating <u>Revenues</u>	Income from Operations
September 30, 2013 Solid Waste: Tier 1 Tier 2 Tier 3 Wheelabrator Other	\$ 898 1,670 940 214 <u>561</u> 4,283	\$ (140) (311) (157) (29) (25) (662)	\$ 758 1,359 783 185 536 3,621	\$ 224 347 148 39 (31) 727
Corporate and Other				(150)
Total	\$ 4,283	\$ (662)	\$ 3,621	\$ 577
September 30, 2012 Solid Waste: Tier 1 Tier 2 Tier 3 Wheelabrator	\$ 854 1,596 862 218	\$ (138) (290) (137) (30)	\$ 716 1,306 725 188	\$ 213 328 126 46
Other	553	(27)	526	40 (68)
Corporate and Other Total	4,083 \$ 4,083	(622) <u>\$ (622)</u>	3,461 \$ 3,461	645 (145) \$ 500
Nine Months Ended: September 20,0203 Solid Waste: Titer 1 Titer 2 Titer 3 Wheelabrator Other	\$ 2,632 4,827 2,637 634 1,649	\$ (417) (899) (424) (83) (73)	\$ 2,215 3,928 2,213 551 1,576	\$ 652 970 388 52 (99)
Corporate and Other	12,379	(1,896)	10,483	1,963 (474)
Total	\$ 12,379	\$ (1,896)	\$ 10,483	\$ 1,489
September 30, 2012 Solid Wast: Tier 1 Tier 2 Tier 3 Wheelabrator Other	\$ 2,531 4,698 2,523 631 <u>1,565</u> 11,948	\$ (386) (805) (381) (92) (69) (1,733)	\$ 2,145 3,893 2,142 539 1,496 10,215	\$ 641 946 370 68 (167) 1,858
Corporate and Other		(1,733)	10,215	(491)
Total	\$ 11,948	\$ (1,733)	\$ 10,215	\$ 1,367

Fluctuations in our operating results may be caused by many factors, including period-to-period changes in the relative contribution of revenue by each line of business, changes in commodity prices and by general

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS - (Continued)

economic conditions. In addition, our revenues and income from operations typically reflect seasonal patterns. Our operating revenues tend to be somewhat higher in the summer months, primarily due to the higher volume of construction and demolition waste. The volumes of industrial and residential waste in certain regions where we operate also tend to increase during the summer months. Our second and third quarter revenues and results of operations typically reflect these seasonal trends.

Additionally, certain destructive weather conditions that tend to occur during the second half of the year, such as the hurricanes that most often impact our operations in the Southern and Eastern U.S., can actually increase our revenues in the areas affected. While weather-related and other "one-time" occurrences can boost revenues through additional work for a limited time span, as a result of significant start-up costs and other factors, such revenue sometimes generates earnings at comparatively lower margins. Certain weather conditions, including severe winter storms, may result in the temporary suspension of our operations, which can significantly affect the operating results of the affected regions. The operating results of our first quarter also often reflect higher repair and maintenance expenses because we rely on the slower winter months, when waste flows are generally lower, to perform scheduled maintenance at our waste-to-energy facilities.

9. Acquisitions

Greenstar, LLC — On January 31, 2013, we paid \$170 million inclusive of certain adjustments, to acquire Greenstar, LLC ("Greenstar"). Pursuant to the sale and purchase agreement, up to an additional \$40 million is payable to the sellers during the period from 2014 to 2018; a portion of this consideration is contingent based on changes in certain recyclable commodity indexes. This additional consideration had a preliminary estimated fair value at closing of \$36 million. Greenstar was an operator of recycling and resource recovery facilities. This acquisition will provide the Company's customers with greater access to recycling solutions by supplementing the Company's extensive nationwide recycling network with the operations of one of the nation's largest private recyclers. For the three and nine months ended September 30, 2013, the Greenstar business recognized revenues of \$39 million and \$104 million, respectively, and net losses of \$2 million, respectively, which are included in our Condensed Consolidated Statements of Operations.

Goodwill of \$122 million was calculated as the excess of the consideration paid over the net assets recognized and represents the future economic benefits expected to arise from other assets acquired that could not be individually identified and separately recognized. Goodwill has been assigned predominantly to our Areas and, to a lesser extent, our recycling brokerage services, as they are expected to benefit from the synergies of the combination. Goodwill related to this acquisition is deductible for income tax purposes. Adjustments to the preliminary allocation of the purchase price of Greenstar were immaterial for the three and nine months ended September 30, 2013.

RCI Environmement, Inc. — On July 5, 2013, we paid C\$509 million, or \$481 million, to acquire substantially all of the assets of RCI Environmement, Inc. ("RCI"), the largest waste management company in Quebec, and certain related entities. Total consideration, inclusive of amounts for estimated working capital, was C\$515 million, or \$487 million. RCI provided collection, transfer, recycling and disposal operations throughout the Greater Montreal area. We acquired RCI to complement and expand the Company's existing assets and operations in Quebec. Since the acquisition date, the RCI business has recognized revenues of \$45 million and net income of \$7 million, which are included in our Condensed Consolidated Statement of Operations.

Goodwill of \$177 million was calculated as the excess of the consideration paid over the net assets recognized and represents the future economic benefits expected to arise from other assets acquired that could not be individually identified and separately recognized. Goodwill has been assigned to our Eastern Canada Area as it is expected to benefit from the synergies of the combination. A portion of goodwill related to this acquisition is deductible for income tax purposes in accordance with Canadian tax law.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The allocations of the purchase price for each acquisition are preliminary and subject to change based on the finalization of our detailed valuations. The preliminary purchase price allocations, as of the date acquired, were as follows (in millions):

	Greenstar	DCT
Accounts and other receivables	s 30	RCI \$ 32
Part and supplies	4	
Other current assets	2	_
Property and equipment	57	118
Goodwill	122	177
Other intangible assets	32	169
Accounts payable	(17)	_
Accrued liabilities	(12)	_
Deferred revenues	_	(4)
Landfill and environmental remediation liabilities	(2)	(1)
Deferred income taxes, net	—	—
Current portion of long-term debt	(4)	_
Long-term debt, less current portion	(1)	(4)
Other liabilities	(5)	
Total purchase price	\$ 206	\$487

The following table presents the preliminary allocations of the purchase price to other intangible assets (amounts in millions, except for amortization periods):

		Freenstar	RCI		
		Weighted Average Amortization		Weighted Average Amortization	
		Periods		Periods	
	Amount	(in Years)	Amount	(in Years)	
Customer relationships	\$—		\$162	15.0	
Supplier relationships	31	10.0	_	_	
Lease agreements	1	8.4	—	_	
Trade name		—	7	5.0	
Total other intangible assets subject to amortization	\$32	10.0	\$169	14.6	
The following pro forma consolidated results of operations have been prepared as if	the acquisitions of Greenstar an	d RCI occurred at January 1,	2012 (in millions, exce	pt per share amounts):	

	En	Three Months Ended September 30.		Ionths led ber 30,
	2013	2012	2013	2012
Operating revenues	\$ 3,623	\$ 3,548	\$ 10,585	\$ 10,493
Net income attributable to Waste Management, Inc.	291	208	704	584
Basic earnings per common share	0.62	0.45	1.51	1.26
Diluted earnings per common share	0.62	0.45	1.50	1.26

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS - (Continued)

Other — During the nine months ended September 30, 2013, we also paid approximately \$47 million primarily to acquire ten businesses related to our collection and energy services operations. During the nine months ended September 30, 2012, we paid \$94 million for interests in oil and gas producing properties through two transactions. The purchase price was allocated primarily to "Property and equipment." Additionally, during the nine months ended September 30, 2012, we acquired 21 other businesses related to our collection and recycling operations.

10. Restructuring

During the nine months ended September 30, 2013, we recognized a total of \$13 million of pre-tax restructuring charges, of which \$6 million was related to employee severance and benefit costs, including costs associated with our acquisition of Greenstar and our July 2012 restructuring. The remaining charges were primarily related to operating lease obligations for property that will no longer be utilized.

During the nine months ended September 30, 2012, we recognized a total of \$51 million of pre-tax restructuring charges, of which \$46 million was related to employee severance and benefit restructuring charges, primarily attributable to our July 2012 restructuring and including other charges associated with the reorganization of Oakleaf and certain other actions taken by the Company in 2012.

Through September 30, 2013, we had recognized charges of \$62 million related to employee severance and benefits associated with our restructuring efforts beginning in 2012 and we have paid approximately \$53 million of these costs. At September 30, 2013, we had approximately \$6 million of accrued employee severance related to our restructuring efforts, which will be paid through the end of 2014.

11. Asset Impairments and Unusual Items

(Income) expense from divestitures, asset impairments and unusual items

During the third quarter of 2013, we recognized charges of \$23 million, primarily related to (i) a \$12 million impairment charge to write down goodwill and indefinite-lived intangibles related to an investment in a waste diversion technology company, partially offset by a \$6 million benefit to noncontrolling interest and (ii) losses on divestitures primarily related to investments in oil and gas producing properties. During the nine months ended September 30, 2013, we also recognized additional charges of \$15 million, primarily related to (i) a \$14 million impairment charge in the second quarter of 2013 at a waste-to-energy facility as a result of projected operating losses, which caused us to write down the carrying value of the facility's property, plant and equipment to its estimated fair value; (ii) losses on divestitures in the first quarter of 2013 primarily related to investments in oil and gas producing properties and (iii) other charges to impair goodwill and write down the carrying value of assets to their estimated fair value; related to creating of our operations, all of which are individually immaterial. Partially offsetting these charges during 2013 were gains on divestitures including the sale of a transfer station in our Greater Mid-Atlantic Area during the first quarter of 2013.

During the third quarter of 2012, we recognized impairment charges aggregating \$22 million, relating in large part to (i) the impairment of a note receivable from an entity that we invested in that went into liquidation (see "Equity in net losses of unconsolidated entities" below), (ii) the impairment of certain related assets due to our decision not to use or develop those assets and (iii) the impairment of an oil and gas well as a result of projected operating losses that caused us to write-down the carrying value of the oil and gas well to its estimated fair value. During the nine months ended September 30, 2012, we also recognized additional impairment charges of \$33 million in the second quarter of 2012, largely attributable to two facilities in our medical waste services business as a result of projected operating losses at each of these facilities, which caused us to write down the carrying values of the facilities' operating permits and property, plant and equipment to their estimated fair values. These charges are included in our "Other" operations in Note 8.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Equity in net losses of unconsolidated entities

During the nine months ended September 30, 2012, we recognized a charge of \$10 million related to a payment we made under a guarantee on behalf of an unconsolidated entity that went into liquidation. This investment was accounted for under the equity method.

Other income (expense)

During the nine months ended September 30, 2013, we recognized impairment charges of \$11 million relating to other-than-temporary declines in the value of two investments in waste diversion technology companies accounted for under the cost method. Partially offsetting these charges was a \$4 million gain on the sale of a similar investment recognized in the second quarter of 2013. These net charges are recorded in "Other, net" in our Condensed Consolidated Statement of Operations.

During the nine months ended September 30, 2012, we recognized an impairment charge of \$14 million relating to an other-than-temporary decline in the value of another investment in a waste diversion technology company accounted for under the cost method. We wrote down the carrying value of our investment to its fair value based on other third-party investors' recent transactions in these securities, which are considered to be the best evidence of fair value currently available. This charge is recorded in "Other, net" in our Condensed Consolidated Statement of Operations.

12. Accumulated Other Comprehensive Income

The changes in the balances of each component of accumulated other comprehensive income, net of tax, which is included as a component of Waste Management, Inc. stockholders' equity, are as follows (in millions, with amounts in parentheses representing debits to accumulated other comprehensive income):

	Nine Months Ended September 30, 2013(a)				
		Unrealized			
		Gains and			
	Gains and	Losses on	Foreign	Post-	
	Losses on	Available-	Currency	Retirement	
	Derivative	for-Sale	Translation	Benefit	
	Instruments	Securities	Adjustments	Plans	Total
Balance, December 31, 2012	\$ (74)	\$ 4	\$ 276	\$ (13)	\$193
Other comprehensive income (loss) before reclassifications	10	1	(37)	_	(26)
Amounts reclassified from accumulated other comprehensive income	2				2
Net current period other comprehensive income (loss)	12	1	(37)		(24)
Balance, September 30, 2013	<u>\$ (62)</u>	<u>\$5</u>	<u>\$ 239</u>	<u>\$ (13)</u>	\$169
	22				

The amounts of other comprehensive income (loss) before reclassifications associated with our cash flow derivative instruments are as follows (in millions):

		Amount of Derivative Gain (Loss) Recognized in OCI (Effective Portion)				
	Three Months Ended			Nine Months Ended		
		September 30,		September 30,		
Derivatives Designated as Cash Flow Hedges	20	13	2012	2013	2012	
Forward-starting interest rate swaps	\$	(1) \$	(3)	\$ 11	\$ (28)	
Foreign currency derivatives		(8)	(15)	11	(14)	
Electricity commodity derivatives		_	(3)	(6)	1	
Total before tax	\$	(9) \$	(21)	\$ 16	\$ (41)	
Tax (expense) benefit		4	8	(6)	16	
Net of tax	\$	(5) \$	(13)	\$ 10	\$ (25)	

The significant amounts reclassified out of each component of accumulated other comprehensive income are as follows (in millions):

	Amount Reclassified from Accumulated Other Comprehensive Income(a)				
	Three M		Nine Months		
Details about Accumulated Other Comprehensive	End Septemi		Ended September 30,		Statement of
Income Components	2013	2012	2013	2012	Operations Classification
Gains and losses on cash flow hedges:					
Forward-starting interest rate swaps	\$ (2)	\$ (2)	\$ (6)	\$ (2)	Interest expense
Treasury rate locks	_	(2)	(1)	(6)	Interest expense
Foreign currency derivatives	(9)	(14)	10	(17)	Other, net
Electricity commodity derivatives	—	2	(6)	9	Operating revenues
	\$ (11)	\$ (16)	\$ (3)	\$ (16)	Total before tax
	4	6	1	6	Tax (expense) benefit
Total reclassifications for the period	<u>\$ (7</u>)	<u>\$ (10</u>)	<u>\$ (2</u>)	<u>\$ (10)</u>	Net of tax

(a) Amounts in parentheses represent debits to the statement of operations classification.

13. Fair Value Measurements

Assets and Liabilities Accounted for at Fair Value

Our assets and liabilities that are measured at fair value on a recurring basis include the following (in millions):

	Total	Quoted Prices in Active Markets (Level 1)	Fair Value Measurements at September 30, 2013 Using Significant Other Observable Inputs (Level 2)	Significant Unobservable Inputs (Level 3)(a)
Assets:	\$120	¢ 100	ŝ	<u>_</u>
Money market funds Fixed-income securities	\$120 37	\$ 120	\$ <u>-</u> 37	\$ —
Redeemable preferred stock	25	_	37	25
Total assets	<u>25</u> <u>\$182</u>	\$ 120	\$ 37	\$ 25
	3182	\$ 120	3 37	\$ 23
Liabilities:	¢ 21	s _	e 21	¢
Interest rate derivatives	\$ 31	\$ —	\$ 31	\$ _
Electricity commodity derivatives Foreign currency derivatives	3	_	3	_
Total liabilities	\$ 34	<u> </u>	\$ 34	<u> </u>
Total habilities	¢ 34	<u>ه </u>	3 54	<u>ه</u>
			Fair Value Measurements at December 31, 2012 Using	
	Total	Quoted Prices in Active Markets (Level 1)	Significant Other Observable Inputs (Level 2)	Significant Unobservable Inputs (Level 3)(a)
Assets:	\$127	\$ 127	s —	s —
Money market funds Fixed-income securities	\$127 37	\$ 127	\$ <u> </u>	\$ —
Redeemable preferred stock	25	_	37	25
Electricity commodity derivatives	1	_	1	
Total assets	\$190	\$ 127	\$ 38	\$ 25
Liabilities:				<u> </u>
Interest rate derivatives	\$ 42	s —	\$ 42	s —
Foreign currency derivatives	11	-	11	-
Electricity commodity derivatives	5	_	5	_
Total liabilities	\$ 58	\$	\$ 58	\$

(a) Level 3 investments have been measured based on third-party investors' recent or pending transactions in these securities, which are considered the best evidence of fair value currently available. When this evidence is not available, we use other valuation techniques as appropriate and available. These valuation methodologies may include transactions in similar instruments, discounted cash flow analyses, third-party appraisals or industry multiples and public comparables. There have not been any significant changes in the fair value of the redeemable preferred stock since our initial investment.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS - (Continued)

Fair Value of Debt

At September 30, 2013 and December 31, 2012, the carrying value of our debt was approximately \$10.1 billion and \$9.9 billion, respectively. The carrying value of our debt includes adjustments associated with fair value hedge accounting related to our interest rate swaps as discussed in Note 4.

The estimated fair value of our debt was approximately \$10.7 billion at September 30, 2013 and approximately \$11.3 billion at December 31, 2012. The estimated fair value of our senior notes is based on quoted market prices. The carrying value of remarketable debt and borrowings under our revolving credit facilities approximates fair value due to the short-term nature of the interest rates. The fair value of our other debt is estimated using discounted cash flow analysis, based on current market rates for similar types of instruments. The decrease in the fair value of our debt when comparing September 30, 2013 with December 31, 2012 is primarily related to recent increases in long-term interest rates, which have caused a decline in market prices for fixed-rate corporate debt securities.

Although we have determined the estimated fair value amounts using available market information and commonly accepted valuation methodologies, considerable judgment is required in interpreting market data to develop the estimates of fair value. Accordingly, our estimates are not necessarily indicative of the amounts that we, or holders of the instruments, could realize in a current market exchange. The use of different assumptions and/or estimation methodologies could have a material effect on the estimated fair values. The fair value estimates are based on Level 2 inputs of the fair value hierarchy available as of September 30, 2013 and December 31, 2012. These amounts have not been revalued since those dates, and current estimates of fair value could differ significantly from the amounts presented.

14. Variable Interest Entities

Following is a description of our financial interests in variable interest entities that we consider significant, including (i) those for which we have determined that we are the primary beneficiary of the entity and, therefore, have consolidated the entities into our financial statements; and (ii) those that represent a significant interest in an unconsolidated entity.

Consolidated Variable Interest Entities

Waste-to-Energy LLCs — In June 2000, two limited liability companies were established to purchase interests in existing leveraged lease financings at three waste-to-energy facilities that we lease, operate and maintain. We own a 0.5% interest in one of the LLCs ("LLC I") and a 0.25% interest in the second LLC ("LLC II"). John Hancock Life Insurance Company ("Hancock") owns 99.5% of LLC I and 99.75% of LLC I is owned by LLC I and the CIT Group ("CIT"). In 2000, Hancock and CIT made an initial investment of \$167 million in the LLCs, which was used to purchase the three waste-to-energy facilities and assume the seller's indebtedness. Under the LLC agreements, the LLCs shall be dissolved upon the occurrence of any of the following events: (i) a written decision of all members of the LLCs; (ii) December 31, 2063; (iii) a court's dissolution of the LLCs; or (iv) the LLCs ceasing to own any interest in the waste-to-energy facilities.

Income, losses and cash flows of the LLCs are allocated to the members based on their initial equity ownership percentages until Hancock and CIT achieve targeted returns on their initial capital investments in each respective LLC. All allocations made through September 30, 2013 have been based on initial equity ownership percentages as the target returns have not yet been achieved for either LLC. We currently expect Hancock and CIT to achieve their targeted return on LLC II in late 2014 and Hancock to achieve its targeted return on LLC II in early 2015. After the investors have achieved their targeted returns, the LLC agreements provide that we will receive 80% of the earnings of each of the LLCs and Hancock and CIT will be allocated the remaining 20%.

Our obligations associated with our interests in the LLCs are primarily related to the lease of the facilities. In addition to our minimum lease payment obligations, we are required to make cash payments to the LLCs for

²⁵

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS ---- (Continued)

differences between fair market rents and our minimum lease payments. These payments are subject to adjustment based on factors that include the fair market value of rents for the facilities and lease payments made through the re-measurement dates. In addition, we may also be required under certain circumstances to make capital contributions to the LLCs based on differences between the fair market value of the facilities and defined termination values as provided for in the underlying lease agreements, although we believe the likelihood of the occurrence of these circumstances is remote.

We have determined that we are the primary beneficiary of the LLCs and consolidate these entities in our Consolidated Financial Statements because (i) all of the equity owners of the LLCs are considered related parties for purposes of applying this accounting guidance; (ii) the equity owners share power over the significant activities of the LLCs; and (iii) we are the entity within the related party group whose activities are most closely associated with the LLCs.

As of September 30, 2013 and December 31, 2012, our Condensed Consolidated Balance Sheets included \$287 million and \$296 million, respectively, of net property and equipment associated with the LLCs' waste-to-energy facilities and \$247 million and \$245 million, respectively, in noncontrolling interests associated with Hancock's and CIT's interests in the LLCs. We recognized reductions in consolidated earnings of \$10 million and \$32 million for the three and nine months ended September 30, 2013, respectively, and \$9 million and \$34 million for the three and nine months ended September 30, 2012, respectively, for Hancock's and CIT's noncontrolling interests in the LLCs' earnings, which are included in our consolidated net income. The LLCs' earnings relate to the rental income generated from leasing the facilities to our subsidiaries, reduced by depreciation expense. The LLCs' rental income is eliminated in WM's consolidation.

Significant Unconsolidated Variable Interest Entities

Investment in U.K. Waste-to-Energy and Recycling Entity — In the first quarter of 2012, we formed a U.K. joint venture (the "JV"), together with a commercial waste management company ("Partner"), to develop, construct, operate and maintain a waste-to-energy and recycling facility in England. We own a 50% interest in the JV. The total cost of constructing this facility is expected to be £200 million, or \$324 million based on the exchange rate as of September 30, 2013. The JV will be funded primarily through lonsn from the joint venture partners and lonsn under the JV's credit facility agreements will be used for the development and construction of the facility. We are committed to provide up to £57 million, or \$92 million based on the exchange rate as of September 30, 2013, or the JV our actual commitment may be more or less depending on the actual cost of the facility. Through September 30, 2013, we had funded approximately £10 million, or \$17 million, through loans and less than \$1 million through equity contributions. These amounts are included in our Condensed Consolidated Balance Sheet as long-term "Other assets" and "Investments in unconsolidated entities," respectively. In addition to the funding commitments described above, the JV has entered into certain foreign currency and interest rate derivatives at the direction of the project, and accordingly, are not reflected in our "Equity in net losses of unconsolidated entities". We also have guaranteed the performance of certain management services for the project for which our maximum exposure is not material.

In addition, a wholly-owned subsidiary of WM will be responsible for constructing the waste-to-energy facility for the JV under a fixed-price construction contract. Once the facility is constructed, a majority-owned subsidiary of WM will be responsible for operating and maintaining the facility for the JV under a substantially fixed-price operating and maintenance contract. Under the operating and maintaining and contract we have guaranteed our ability to operate this facility at certain performance levels that we believe are achieved. We are solve able. We also will be jointly responsible, along with our Partner, for the performance of sales and marketing services for the JV through a 50%-owned unconsolidated entity. The fixed-price components of the above mentioned

contracts were established based on estimates of expected construction, operation and maintenance costs. However, we may not achieve the financial results anticipated and could incur losses if the actual costs differ from the costs established in the contracts. A range of our exposure to potential loss under these contracts cannot presently be estimated.

We determined that we are not the primary beneficiary of the JV, as all major decisions of the JV require either majority vote or unanimous consent of the directors (who are appointed in equal numbers by us and our Partner) or unanimous consent of the two shareholders of the JV. As such, our Partner shares equally in the power to direct the activities of the JV that most significantly impact its economic performance, including approval of the facility construction and operations and maintenance contract terms. Accordingly, we account for this investment under the equity method of accounting and on ot consolidate this entity.

Investment in Refined Coal Facility — In January 2011, we acquired a noncontrolling interest in a limited liability company established to invest in and manage a refined coal facility. Along with another equity investor, we support the operations of the entity in exchange for a pro-rata share of the tax credits it generates. Our initial consideration for this investment consisted of a cash payment of \$48 million. As of September 30, 2013 and December 31, 2012, our investment balance was \$27 million and \$19 million, respectively, representing our current maximum pre-tax exposure to loss. Under the terms and conditions of the transaction, we do not believe that we have any material exposure to loss. Required capital contributions commenced in the first quarter of 2013 and will continue through the expiration of the tax credits under Section 45 of the Internal Revenue Code, which occurs at the end of 2019. We are only obligated to make future contributions to the extent tax credits are generated. We determined that we are not the primary beneficiary of this entity as we do not have the power to individually direct the entity's activities. Accordingly, we account for this investment under the equity method of accounting and do not consolidate the entity. Additional information related to this investment is discussed in Note 5.

Investment in Low-Income Housing Properties — In April 2010, we acquired a noncontrolling interest in a limited liability company established to invest in and manage low-income housing properties. We support the operations of the entity in exchange for a pro-rata share of the tax credits it generates. Our target return on the investment is guaranteed, and therefore, we do not believe that we have any material exposure to loss. Our consideration for this investment totaled \$221 million, which was comprised of a \$215 million note payable and an initial cash payment of \$6 million. As of September 30, 2013 and December 31, 2012, our investment balance was \$135 million and \$153 million and \$153 million and \$152 million, respectively. We determined that we are not the primary beneficiary of this entity as we do not have the power to individually direct the entity's activities. Accordingly, we account for this investment under the equity method of accounting and do not consolidate the entity. Additional information related to this investment is discussed in Note 5.

Trusts for Final Capping, Closure, Post-Closure or Environmental Remediation Obligations — We have significant financial interests in trust funds that were created to settle certain of our final capping, closure, post-closure or environmental remediation obligations. Generally, we are the sole beneficiary of these restricted balances; however, certain of the funds have been established for the benefit of both the Company and the host community in which we operate. We have determined that these trust funds are variable interest entities; however, we are not the primary beneficiary of these entities because either (i) we do not have the power to direct the significant activities of (ii) power over the trusts' significant activities is shared.

We account for the trusts for which we are the sole beneficiary as long-term "Other assets" in our Condensed Consolidated Balance Sheet. We reflect our interests in the unrealized gains and losses on available-for-sale securities held by these trusts as a component of "Accumulated other comprehensive income." These trusts had a fair value of \$126 million at September 30, 2013 and \$125 million at December 31, 2012. Our interests in the trusts that have been established for the benefit of both the Company and the host community in which we operate are accounted for as investments in unconsolidated entities and receivables. These amounts are

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

recorded in "Other receivables," "Investments in unconsolidated entities" and long-term "Other assets" in our Condensed Consolidated Balance Sheet, as appropriate. Our investments and receivables related to these trusts had an aggregate carrying value of \$110 million as of both September 30, 2013 and December 31, 2012.

As the party with primary responsibility to fund the related final capping, closure, post-closure or environmental remediation activities, we are exposed to risk of loss as a result of potential changes in the fair value of the assets of the trust. The fair value of trust assets can fluctuate due to (i) changes in the market value of the investments held by the trusts and (ii) credit risk associated with trust receivables. Although we are exposed to changes in the fair value of the trust assets, we currently expect the trust funds to continue to meet the statutory requirements for which they were established.

15. Condensed Consolidating Financial Statements

WM Holdings has fully and unconditionally guaranteed all of WM's senior indebtedness. WM has fully and unconditionally guaranteed all of WM Holdings' senior indebtedness. None of WM's other subsidiaries have guaranteed any of WM's or WM Holdings' debt. As a result of these guarantee arrangements, we are required to present the following condensed consolidating financial information (in millions):

CONDENSED CONSOLIDATING BALANCE SHEETS

September 30, 2013 (Unaudited)

	(
	ASSETS WM	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Current assets:					
Cash and cash equivalents	s —	\$ —	\$ 157	\$	\$ 157
Other current assets	·	6	2,254		2,260
		6	2,411		2,417
Property and equipment, net	_	_	12,621	_	12,621
Investments in and advances to affiliates	12,802	16,790	4,174	(33,766)	12,021
Other assets	44	10,750	8,519	(55,700)	8,575
				¢ (22.7(c))	
Total assets	<u>\$12,846</u>	\$16,808	\$ 27,725	\$ (33,766)	\$ 23,613
	LIABILITIES AND EQUITY				
Current liabilities:	LIABILITIES AND EQUILI				
Current portion of long-term debt	\$ 275	¢	\$ 293	¢	\$ 568
Accounts payable and other current liabilities	\$ 275 94	پ س ح	2,195	ф —	2,294
Accounts payable and other current nabilities					
	369	5	2,488	_	2,862
Long-term debt, less current portion	5,769	449	3,273	_	9,491
Other liabilities			4,245		4,245
Total liabilities	6,138	454	10,006	_	16,598
Equity:					
Stockholders' equity	6,708	16,354	17,412	(33,766)	6,708
Noncontrolling interests	_	_	307		307
	6,708	16,354	17,719	(33,766)	7,015
Total liabilities and equity	\$12,846	\$16,808	\$ 27,725	\$ (33,766)	\$ 23,613
roun nuonnes une equity	\$12,840	\$10,000	φ 21,125	φ (33,700)	φ 25,015

CONDENSED CONSOLIDATING BALANCE SHEETS (Continued)

December 31, 2012

	ASSETS WM	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Current assets:					
Cash and cash equivalents	\$ 60	\$ —	\$ 134	\$ —	\$ 194
Other current assets		7	2,222		2,229
	60	7	2,356	_	2,423
Property and equipment, net	—	_	12,651	_	12,651
Investments in and advances to affiliates	12,686	16,697	3,437	(32,820)	_
Other assets	45	12	7,966		8,023
Total assets	\$12,791	\$16,716	\$ 26,410	\$ (32,820)	\$ 23,097
	LIABILITIES AND EQUITY				
Current liabilities:					
Current portion of long-term debt	\$ 400	\$ —	\$ 343	\$ —	\$ 743
Accounts payable and other current liabilities	77	13	2,203		2,293
	477	13	2,546		3,036
Long-term debt, less current portion	5,918	449	2,806	_	9,173
Other liabilities	42	_	4,171	_	4,213
Total liabilities	6,437	462	9,523		16,422
Equity:					
Stockholders' equity	6,354	16,254	16,566	(32,820)	6,354
Noncontrolling interests	—	_	321	_	321
	6,354	16,254	16,887	(32,820)	6,675
Total liabilities and equity	\$12,791	\$16,716	\$ 26,410	\$ (32,820)	\$ 23,097

CONDENSED CONSOLIDATING STATEMENTS OF OPERATIONS

Three Months Ended September 30, 2013 (Unaudited)

Operating revenues Costs and expenses Income from operations	<u>WM</u> \$	WM Holdings \$	Non-Guarantor Subsidiaries \$ 3,621 3,044 577	Eliminations \$	Consolidated \$ 3,621 3,044 577
Other income (expense): Interest income (expense) Equity in earnings of subsidiaries, net of taxes	(88) 344	(8) 349	(23)	(693)	(119)
Other, net	256	341	(6) (29)	(693)	(6) (125)
Income before income taxes	256	341	548	(693)	452
Provision for (benefit from) income taxes	(35)	(3)	193		155
Consolidated net income	291	344	355	(693)	297
Less: Net income attributable to noncontrolling interests			6		6
Net income attributable to Waste Management, Inc.	\$291	\$ 344	\$ 349	\$ (693)	\$ 291

Three Months Ended September 30, 2012 (Unaudited)

Operating revenues Costs and expenses Income from operations Other income (expense):	<u>wm</u> \$	WM Holdings \$	Non-Guarantor Subsidiaries 3,461 2,961 500	Eliminations \$	Consolidated \$ 3,461 2,961 500
Interest income (expense) Equity in earnings of subsidiaries, net of taxes Other, net	(91) 269 	$ \begin{array}{r} (8)\\ 274\\ \underline{}\\ 266\\ \end{array} $	(22) (31) (53)	(543) (543)	$(121) \\ (31) \\ (152)$
Income before income taxes Provision for (benefit from) income taxes Consolidated net income Less: Net income attributable to noncontrolling interests Net income attributable to Waste Management, Inc.	$ \begin{array}{r} \overline{178} \\ \underline{(36)} \\ 214 \\ \underline{\overline{3214}} \\ \overline{3214} \\ 21 \end{array} $		447 164 283 9 \$ 274	(543) (543) (543) (543) (543)	$ \begin{array}{r} 348 \\ 125 \\ 223 \\ 9 \\ $ 214 \\ \end{array} $

CONDENSED CONSOLIDATING STATEMENTS OF OPERATIONS (Continued)

Nine Months Ended September 30, 2013 (Unaudited)

Operating revenues Costs and expenses Income from operations	<u>WM</u> \$		Subsidiaries \$ 10,483 8,993 1,490	Eliminations \$	Consolidated \$ 10,483 8,994 1,489
Other income (expense):					
Interest income (expense)	(268)	(24)	(69)	—	(361)
Equity in earnings of subsidiaries, net of taxes	865	881	_	(1,746)	_
Other, net			(31)		(31)
	597	857	(100)	(1,746)	(392)
Income before income taxes	597	856	1,390	(1,746)	1,097
Provision for (benefit from) income taxes	(106)	(9)	483	_	368
Consolidated net income	703	865	907	(1,746)	729
Less: Net income attributable to noncontrolling interests	_	_	26	_	26
Net income attributable to Waste Management, Inc.	\$ 703	\$ 865	\$ 881	\$ (1,746)	\$ 703

Nine Months Ended September 30, 2012 (Unaudited)

Operating revenues Costs and expenses Income from operations Other income (expense): Interest income (expense) Equity in earnings of subsidiaries, net of taxes Other, net	<u>wм</u> <u>-</u> (267) 755	WM Holdings \$ (24) 770	Non-Guarantor <u>Subsidiaries</u> \$ 10,215 <u>8,848</u> <u>1,367</u> (71) (51)	Eliminations \$ (1,525)	Consolidated \$ 10,215 8,848 1,367 (362) (51)
Income before income taxes Provision for (benefit from) income taxes	488 488 (105)		(51) (122) 1,245 443	(1,525) (1,525)	(51) (413) 954 329
Consolidated net income Less: Net income attributable to noncontrolling interests Net income attributable to Waste Management, Inc.	593 	755 <u>\$</u> 755	802 32 \$ 770	(1,525) <u>(1,525)</u>	625 32 \$ 593

CONDENSED CONSOLIDATING STATEMENTS OF COMPREHENSIVE INCOME

(Unaudited)

	WM	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
<u>Three Months Ended September 30, 2013</u> Comprehensive income Less: Comprehensive income attributable to noncontrolling interests Comprehensive income attributable to Waste Management, Inc. <u>Three Months Ended September 30, 2012</u>	\$292 	\$ 344 <u> </u>	\$ 389 <u>6</u> <u>\$ 383</u>	\$ (693) <u>\$ (693</u>)	$ \begin{array}{r} \$ 332 \\ $
Comprehensive income Less: Comprehensive income attributable to noncontrolling interests Comprehensive income attributable to Waste Management, Inc.	\$215 \$215	\$ 269 	\$ 319 9 <u>\$ 310</u>	\$ (543) <u> </u>	
Nine Months Ended September 30, 2013	WM	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
<u>Nine Months Ended September 30, 2013</u> Comprehensive income Less: Comprehensive income attributable to noncontrolling interests Comprehensive income attributable to Waste Management, Inc. Nine Months Ended September 30, 2012	<u>wm</u> \$714 <u></u> \$714			Eliminations \$ (1,746) \$ (1,746)	Consolidated \$ 705 26 \$ 679

CONDENSED CONSOLIDATING STATEMENTS OF CASH FLOWS

Nine Months Ended September 30, 2013 (Unaudited)

	(,				
	WM	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Cash flows from operating activities:					
Consolidated net income	\$ 703	\$ 865	\$ 907	\$ (1,746)	\$ 729
Equity in earnings of subsidiaries, net of taxes	(865)	(881)		1,746	
Other adjustments	(19)	(8)	1,156		1,129
Net cash provided by (used in) operating activities	(181)	(24)	2,063		1,858
Cash flows from investing activities:					
Acquisitions of businesses, net of cash acquired	—	—	(698)	—	(698)
Capital expenditures	—	—	(824)	—	(824)
Proceeds from divestitures of businesses (net of cash divested) and other sales of assets	_	_	113	_	113
Net receipts from restricted trust and escrow accounts and other, net			(23)		(23)
Net cash provided by (used in) investing activities	—	—	(1,432)	—	(1,432)
Cash flows from financing activities:					
New borrowings	200	_	1,018	_	1,218
Debt repayments	(460)	—	(784)	—	(1,244)
Cash dividends	(512)	—	—	—	(512)
Exercise of common stock options	116	—	—	—	116
Distributions paid to noncontrolling interests and other	12	_	(49)	-	(37)
(Increase) decrease in intercompany and investments, net	765	24	(789)		
Net cash provided by (used in) financing activities	121	24	(604)		(459)
Effect of exchange rate changes on cash and cash equivalents			(4)		(4)
Increase (decrease) in cash and cash equivalents	(60)		23		(37)
Cash and cash equivalents at beginning of period	60		134	_	194
Cash and cash equivalents at end of period	\$	<u>\$ </u>	\$ 157	\$	\$ 157
	34				

CONDENSED CONSOLIDATING STATEMENTS OF CASH FLOWS (Continued)

Nine Months Ended September 30, 2012 (Unaudited)

Cash flows from operating activities:	WM	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Consolidated net income	\$ 593	\$ 755	\$ 802	\$ (1,525)	\$ 625
Equity in earnings of subsidiaries, net of taxes	(755)	(770)	_	1,525	
Other adjustments	75	(8)	1,026		1,093
Net cash provided by (used in) operating activities	(87)	(23)	1,828		1,718
Cash flows from investing activities:					
Acquisitions of businesses, net of cash acquired	_	_	(178)	_	(178)
Capital expenditures	—	—	(1,132)	—	(1,132)
Proceeds from divestitures of businesses (net of cash divested) and other sales of assets	_	_	28	—	28
Net receipts from restricted trust and escrow accounts and other, net			(79)		(79)
Net cash provided by (used in) investing activities			(1,361)		(1,361)
Cash flows from financing activities:					
New borrowings	645	—	40	—	685
Debt repayments	(335)	—	(138)	—	(473)
Cash dividends	(493)	_	-	-	(493)
Exercise of common stock options	39	-		-	39
Distributions paid to noncontrolling interests and other	14		9	—	23
(Increase) decrease in intercompany and investments, net	368	23	(391)		(210)
Net cash provided by (used in) financing activities	238	23	(480)		(219)
Effect of exchange rate changes on cash and cash equivalents			2		2
Increase (decrease) in cash and cash equivalents	151	-	(11)	_	140
Cash and cash equivalents at beginning of period	119		139		258
Cash and cash equivalents at end of period	<u>\$ 270</u>	<u>\$ </u>	<u>\$ 128</u>	<u>\$ </u>	\$ 398
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

The following discussion should be read in conjunction with the Condensed Consolidated Financial Statements and notes thereto included under Item 1 and our Consolidated Financial Statements and notes thereto and related Management's Discussion and Analysis of Financial Condition and Results of Operations included in our Annual Report on Form 10-K for the year ended December 31, 2012. In an effort to keep our stockholders and the public informed about our business, we may make "forward-looking statements." Forward-looking statements usually relate to future events and

anticipated revenues, each flows or other aspects of our operating results. Forward-looking statements are often identified by the words, "will," "may," "should," "continue," "anticipated revenues, each flows or other aspects of our operating results. Forward-looking statements are often identified by the words, "will," "may," "should," "continue,"

- % projections about accounting and finances;
- % plans and objectives for the future;
- % projections or estimates about assumptions relating to our performance; or
- % our opinions, views or beliefs about the effects of current or future events, circumstances or performance.

You should view these statements with caution. These statements are not guarantees of future performance, circumstances or events. They are based on the facts and circumstances known to us as of the date the statements are made. All aspects of our business are subject to uncertainties, risks and other influences, many of which we do not control. Any of these factors, either alone or taken together, could have a material adverse effect on us and could change whether any forward-looking statement ultimately turns out to be true. Additionally, we assume no obligation to update any forward-looking statement as a result of future events, circumstances or developments.

Some of the risks that we believe could affect our financial statements for 2013 and beyond and that could cause actual results to be materially different from those that may be set forth in forward-looking statements made by the Company include the following:

- competition may negatively affect our profitability or cash flows, our pricing strategy may have negative effects on volumes, and inability to execute our pricing strategy in order to retain and attract customers may negatively affect our average yield on collection and disposal business;
- we may fail in implementing our cost saving, optimization and growth initiatives and overall business strategy, which could adversely impact our financial performance and growth, and implementation of our initiatives and strategy may have associated negative consequences, such as fragmentation of efforts and oversight, increased indebtedness, asset impairments, business disruption, employee distraction, and regulatory issues;
- % regulations may negatively impact our business by, among other things, restricting our operations, increasing costs of operations or requiring additional capital expenditures;
- solution possible changes in our estimates of costs for site remediation requirements, final capping, closure and post-closure obligations, compliance and regulatory developments may increase our expenses;
- certain materials processed by our recycling operations are subject to significant commodity price fluctuations, as are methane gas, electricity and other energy-related products marketed and sold by our landfill gas recovery, waste-to-energy and independent power production plant operations; fluctuations in commodity prices may have negative effects on our operating results;
- % increasing customer preference for alternatives to traditional disposal, government mandates requiring recycling and prohibiting disposal of certain types of waste, and overall reduction of waste generated could continue to have a negative effect on volumes of waste going to landfills and waste-to-energy facilities;
- & developments in technology could trigger a fundamental change in the waste management industry, as waste streams are increasingly viewed as a resource, which may adversely impact volumes at our landfills and waste-to-energy facilities and our profitability;

Form 10-Q

- our existing and proposed service offerings to customers may require that we develop or license, and protect, new technologies; and our inability to obtain or protect new technologies could impact our services to customers and development of new revenue sources;
- we are investing in emerging technologies to provide disposal alternatives, which may divert capital investment away from our traditional business operations; such emerging technologies may not perform as intended or may experience other difficulties or delays that prevent us from realizing a return on our investment;
- adverse publicity (whether or not justified) relating to activities by our operations, employees or agents could tarnish our reputation and reduce the value of our brand;
- there is a risk of incurring significant environmental liabilities in the use, treatment, storage, transfer and disposal of waste materials; any substantial liability for environmental damage could have a material adverse effect on our financial condition and cash flows;
- % weak economic conditions may negatively affect the volumes of waste generated;
- some of our customers, including governmental entities, have suffered financial difficulties that could affect our business and operating results, due to their credit risk and the impact of the municipal debt market on remarketing of our tax-exempt bonds;
- % if we are unable to obtain and maintain permits needed to open, operate, and/or expand our facilities, our results of operations will be negatively impacted;
- % fuel price increases or fuel supply shortages may increase our expenses and restrict our ability to operate;
- problems with the operation of current information technology or the development and deployment of new information systems could decrease our efficiencies and increase our costs;
- a cybersecurity incident could negatively impact our business and our relationships with customers;
- efforts by labor unions to organize our employees may increase operating expenses and we may be unable to negotiate acceptable collective bargaining agreements with those who have chosen to be represented by unions, which could lead to labor disruptions, including strikes and lock-outs, which could adversely affect our results of operations and cash flows;
- ‰ we could face significant liability for withdrawal from multiemployer pension plans;
- % we are subject to operational and safety risks, including the risk of personal injury to employees and others;
- 🐒 increased costs for financial assurance or the inadequacy of our insurance coverage could negatively impact our liquidity and increase our liabilities;
- possible charges as a result of shut-down operations, uncompleted development or expansion projects or other events may negatively affect earnings;
- we may reduce or suspend capital expenditures, acquisition activity, dividend declarations or share repurchases if we suffer a significant reduction in cash flows;
- we may be unable to incur future indebtedness to support our growth and development plans on terms we deem acceptable or to refinance our debt obligations, including near-term maturities, on acceptable terms and higher interest rates and market conditions may increase our expense;
- % climate change legislation, including possible limits on carbon emissions, may negatively impact our results of operations by increasing expenses;
- weather conditions and one-time special projects cause our results to fluctuate, and harsh weather or natural disasters may cause us to temporarily suspend operations; our stock price may be negatively impacted by interim variations in our results;
- % we could be subject to significant fines and penalties, and our reputation could be adversely affected, if our business, or third parties with whom we have relationships, were to fail to comply with United States or foreign laws or regulations;

- % we could fail to achieve the financial results anticipated from the construction of new international waste-to-energy facilities;
- & negative outcomes of litigation or threatened litigation or governmental proceedings may increase our costs, limit our ability to conduct or expand our operations, or limit our ability to execute our business plans and strategies; and
 - & the adoption of new accounting standards or interpretations may cause fluctuations in reported quarterly results of operations or adversely impact our reported results of operations.

General

Our principal executive offices are located at 1001 Fannin Street, Suite 4000, Houston, Texas 77002. Our telephone number at that address is (713) 512-6200. Our website address is www.wm.com. Our annual reports on Form 10-K, quarterly reports on Form 10-Q and current reports on Form 8-K are all available, free of charge, on our website as soon as practicable after we file the reports with the SEC. Our stock is traded on the New York Stock Exchange under the symbol "WM."

We are North America's leading provider of comprehensive waste management environmental services. We partner with our residential, commercial, industrial and municipal customers and the communities we serve to manage and reduce waste at each stage from collection to disposal, while recovering valuable resources and creating clean, renewable energy. Our "Solid Waste" business is operated and managed locally by our subsidiaries that focus on distinct geographic areas and provides collection, transfer, recycling and resource recovery, and disposal services. Through our subsidiaries, we are also a leading developer, operator and owner of waste-to-energy facilities in the United States.

Overview

Every day, Waste Management is helping industries, communities and individuals reduce, reuse and remove waste better through sound sustainability strategies. We are developing and implementing new ways to handle and extract value from waste, while maintaining our day-to-day focus on collecting and handling our customers' waste efficiently and responsibly. Our employees are committed to delivering environmental performance — our mission is to maximize resource value, while minimizing environmental impact, so that both our economy and our environment can thrive. Drawing on our resources and experience, we actively pursue projects and initiatives that benefit the waste industry, the customers and communities we serve and the environment.

The Company also remains committed to providing long-term value to our stockholders by successfully executing on our strategic goals of knowing and servicing the customer better than anyone else, extracting more value from the materials we handle, and optimizing our business. In pursuit of these long-term goals, we have sharpened our focus on the following key priorities:

- % Pursue revenue growth through customer-focused segmentation, pricing discipline and strategic acquisitions;
- & Continually emphasize cost control and investment in technology and systems that enhance the efficiency of our operations; and
- & Invest in emerging technologies that offer alternatives to traditional disposal and generate additional value from the waste streams we manage.

We believe that execution of our strategy through these key priorities will drive continued growth and leadership in a dynamic industry, as customers increasingly seek non-traditional waste management solutions.

Highlights of our financial results for the current quarter include:

Revenues of \$3,621 million compared with \$3,461 million in the third quarter of 2012, an increase of \$160 million, or 4.6%. This increase in revenues is primarily attributable to (i) revenue from acquisitions, driven in large part by our acquisitions of Greenstar and RCI, which increased revenues by \$39 million and \$41 million, respectively, and (ii) positive revenue growth from yield on our collection and disposal operations of \$68 million, or 2.3%. These increases were partially offset by lower volumes, which decreased our revenues by \$22 million;

- Operating expenses of \$2,325 million, or 64.2% of revenues, compared with \$2,229 million, or 64.4% of revenues, in the third quarter of 2012. This increase of \$96 million is largely due to (i) our acquisition of Greenstar, which increased operating expenses by \$37 million, and was primarily related to cost of goods sold and to a lesser extent, labor and related benefits and other categories; (ii) higher labor and related benefits due to merit increases and higher incentive compensation costs attributed to higher anticipated payouts and (iii) higher costs from the acquired RCI operations, primarily subcontractor costs. The increases attributable to Greenstar and RCI were offset by related revenue increases noted above;
- Selling, general and administrative expenses of \$349 million, or 9.6% of revenues, compared with \$335 million, or 9.7% of revenues, in the third quarter of 2012. This increase of \$14 million is driven mainly by higher incentive compensation costs attributable to higher anticipated payouts, offset, in part, by our restructuring efforts and cost control initiatives and the collection of reserved receivables in Puerto Rico;
- Income from operations of \$577 million, or 15.9% of revenues, compared with \$500 million, or 14.4% of revenues, in the third quarter of 2012; and
- 8 Net income attributable to Waste Management, Inc. of \$291 million, or \$0.62 per diluted share, as compared with \$214 million, or \$0.46 per diluted share in the third quarter of 2012.
- The comparability of our diluted earnings per share has been affected by the following items that occurred in the third quarter of 2013:
- The recognition of pre-tax charges aggregating \$23 million comprised of (i) \$16 million related to impairments, primarily attributable to an investment in a waste diversion technology company, which was partially offset by a \$6 million benefit to noncontrolling interests, and (ii) \$7 million of losses on divestitures, primarily related to oil and gas producing properties. These items had a negative impact of \$0.02 on our diluted earnings per share; and
- The recognition of pre-tax charges aggregating \$8 million primarily associated with the partial withdrawal from an underfunded multiemployer pension plan and, to a lesser extent, other restructuring charges. These items had a negative impact of \$0.01 on our diluted earnings per share.

Our third quarter of 2012 results were affected by the following:

- Pre-tax costs aggregating \$47 million related to our July 2012 restructuring and integration costs associated with our acquisition of Oakleaf, which had a negative impact of \$0.06 on our diluted earnings per share;
- The recognition of pre-tax impairment charges of \$45 million, primarily associated with certain of our investments in unconsolidated entities and related assets, which had an unfavorable impact of \$0.08 on our diluted earnings per share; and
- The recognition of a pre-tax charge of \$6 million resulting from a union labor dispute in the Pacific Northwest Area, which had a negative impact of \$0.01 on our diluted earnings per share.

Free Cash Flow

As is our practice, we are presenting free cash flow, which is a non-GAAP measure of liquidity, in our disclosures because we use this measure in the evaluation and management of our business. We define free cash flow as net cash provided by operating activities, less capital expenditures, plus proceeds from divestitures of businesses (net of cash divested) and other sales of assets. We believe it is indicative of our ability to pay our quarterly dividends, repurchase common stock, fund acquisitions and other investments and, in the absence of refinancings, to repay our debt obligations. Free cash flow is not intended to replace "Net cash provided by operating activities," which is the most comparable U.S. GAAP measure. However, we believe free cash flow gives investors useful insight into how we view our liquidity. Nevertheless, the use of free cash flow as a liquidity measure has material limitations because it excludes certain expenditures that are required or that we have committed to, such as declared dividend payments and debt service requirements.



Our calculation of free cash flow and reconciliation to "Net cash provided by operating activities," is shown in the table below (in millions), and may not be the same as similarly-titled measures presented by other companies:

	Three Months Ended September 30,			Aonths ded ıber 30,
	2013	2012	2013	2012
Net cash provided by operating activities	\$ 736	\$ 574	\$1,858	\$ 1,718
Capital expenditures	(323)	(402)	(824)	(1,132)
Proceeds from divestitures of businesses (net of cash divested) and other sales of assets	39	8	113	28
Free cash flow	\$ 452	\$ 180	\$1,147	\$ 614

When comparing our cash flow from operating activities for the three and nine months ended September 30, 2013 to the comparable periods in 2012, the increase was due primarily to higher cash earnings, the payment of \$59 million to settle the liabilities associated with the termination of forward-starting swaps in September 2012 and the favorable impacts of working capital changes. The year-overyear increase for the nine months also included the favorable impact of \$86 million resulting from significantly lower annual incentive plan payments made in the first quarter of 2012, largely offset by the favorable cash receipt of \$72 million resulting from the termination of interest rate swaps in April 2012.

The decrease in capital expenditures when comparing the first nine months of 2013 with the prior year period can generally be attributed to increased focus on capital spending management and the impact of timing differences associated with cash payments for the previous years' fourth quarter capital spending.

Acquisitions

Greenstar, LLC — On January 31, 2013, we paid \$170 million inclusive of certain adjustments, to acquire Greenstar, LLC ("Greenstar"). Pursuant to the sale and purchase agreement, up to an additional \$40 million is payable to the sellers during the period from 2014 to 2018; a portion of this consideration is contingent based on changes in certain recyclable commodity indexes. This additional consideration had a preliminary estimated fair value at closing of \$36 million. Greenstar was an operator of recycling and resource recovery facilities. This acquisition will provide the Company's customers with greater access to recycling solutions by supplementing the Company's extensive nationwide recycling network with the operations of one of the nation's largest private recyclers. For the three and nine months ended September 30, 2013, the Greenstar business recognized revenues of \$39 million and \$104 million, respectively, and net losses of \$2 million, respectively, which are included in our Condensed Consolidated Statements of Operations.

RCI Environmement, Inc. — On July 5, 2013, we paid C\$509 million, or \$481 million, to acquire substantially all of the assets of RCI Environmement, Inc. ("RCI"), the largest waste management company in Quebec, and certain related entities. Total consideration, inclusive of amounts for estimated working capital, was C\$515 million, or \$487 million. RCI provided collection, transfer, recycling and disposal operations throughout the Greater Montreal area. We acquired RCI to complement and expand the Company's existing assets and operations in Quebec. Since the acquisition date, the RCI business has recognized revenues of \$45 million and net income of \$7 million, which are included in our Condensed Consolidated Statement of Operations.

Adoption of New Accounting Pronouncement

Comprehensive Income — In February 2013, the Financial Accounting Standards Board issued amended authoritative guidance associated with comprehensive income, which requires companies to provide information about the amounts that are reclassified out of accumulated other comprehensive income by component. Additionally, companies are required to present significant amounts reclassified out of accumulated other comprehensive income by component. Additionally, companies are required to present significant amounts reclassified out of accumulated other comprehensive income by the respective line items of net income. The amendment to authoritative guidance associated with comprehensive income was effective for the Company on January 1, 2013. The adoption of this

guidance did not have a material impact on our consolidated financial statements. We have presented the information required by this amendment in Note 12 to the Condensed Consolidated Financial Statements.

Critical Accounting Estimates and Assumptions

In preparing our financial statements, we make numerous estimates and assumptions that affect the accounting for and recognition and disclosure of assets, liabilities, equity, revenues and expenses. We must make these estimates and assumptions because certain information that we use is dependent on future events, cannot be calculated with precision from available data or simply cannot be calculated. In some cases, these estimates are difficult to determine, and we must exercise significant judgment. In preparing our financial statements, the most difficult, subjective and complex estimates and the assumptions that present the greatest amount of uncertainty relate to our accounting for landfills, environmental remediation liabilities, asset impairments, deferred income taxes and reserves associated with our insured and self-insured claims, as described in Item 7 of our Annual Report on Form 10-K for the year ended December 31, 2012. Actual results could differ materially from the estimates and assumptions that we use in the preparation of our financial statements.

We perform our annual impairment test of goodwill balances using a measurement date of October 1. The October 1. 2012 impairment test indicated that the estimated fair value of our Wheelabrator reporting unit exceeded its carrying value by approximately 10% and the fair value of our Eastern Canada Area reporting unit exceeded its carrying value by approximately 5%. These impairment tests required us to make numerous estimates and assumptions. We will be performing our 2013 annual impairment tests as of October 1, 2015. In performing this test, we will again be making numerous estimates and assumptions. Given the subjective nature of these impairment tests, even slight differences in estimates and assumptions candue the fair value of the Wheelabrator reporting unit, Eastern Canada Area reporting unit or both are below their carrying values which could result in an impairment of goodwill being recorded in the fourth quarter of 2013.

Results of Operations

Operating Revenues

We evaluate, oversee and manage the financial performance of our Solid Waste business subsidiaries through our 17 geographic Areas. Our Wheelabrator business provides waste-to-energy services and manages waste-to-energy facilities and independent power production plants. We also provide additional services that are not managed through our Solid Waste or Wheelabrator businesses, including WM Sustainability Business Services, recycling brokerage services, electronic recycling services, in-plant services, landfill gas-to-energy services, integrated medical waste services, expanded service offerings and solutions. Part of our expansion of services offering fluorescent bub and universal waste mail-back through our LampTracker® program and portable self-storage services. In addition, we have made investments that involve the acquisition and development of interests in oil and gas producing properties. These operations are presented as "Other" in the table below. Shown below (in millions) is the contribution to revenues during each period:

	Three	Three Months		
	En	ded	En	ded
	Septen	iber 30,	Septen	iber 30,
	2013	2012	2013	2012
Solid Waste:				
Tier 1:	\$ 898	\$ 854	\$ 2,632	\$ 2,531
Tier 2:	1,670	1,596	4,827	4,698
Tier 3:	940	862	2,637	2,523
Solid Waste	3,508	3,312	10,096	9,752
Wheelabrator	214	218	634	631
Other	561	553	1,649	1,565
Intercompany	(662)	(622)	(1,896)	(1,733)
Total	\$ 3,621	\$3,461	\$ 10,483	\$ 10,215

The mix of operating revenues from our major lines of business is reflected in the table below (in millions):

	En	Three Months Ended September 30,		Ionths ded ber 30,
	2013	2012	2013	2012
Commercial	\$ 871	\$ 851	\$ 2,566	\$ 2,571
Residential	659	644	1,956	1,932
Industrial	589	553	1,659	1,591
Other	74	71	200	205
Total collection	2,193	2,119	6,381	6,299
Landfill	741	705	2,085	1,996
Transfer	352	332	997	961
Wheelabrator	214	218	634	631
Recycling	367	316	1,091	1,030
Other	416	393	1,191	1,031
Intercompany	(662)	(622)	(1,896)	(1,733)
Total	\$ 3,621	\$ 3,461	\$ 10,483	\$ 10,215

The following table provides details associated with the period-to-period change in revenues (dollars in millions) along with an explanation of the significant components of the current period changes:

	C Thr	eriod-to-Period Change for the ee Months Ended September 30, 2013 vs. 2012	C Nin S	eriod-to-Period Change for the e Months Ended September 30, 2013 vs. 2012
		As a % of Total		As a % of Total
	Amount	Company(a)	Amount	Company(a)
Average yield(b)	\$ 90	2.6%	\$ 137	1.3%
Volume	(22)	(0.6)	(56)	(0.5)
Internal revenue growth	68	2.0	81	0.8
Acquisitions	103	3.0	204	2.0
Divestitures	(2)	(0.1)	(4)	_
Foreign currency translation	(9)	(0.3)	(13)	(0.2)
	<u>\$ 160</u>	4.6%	<u>\$ 268</u>	2.6%

(a) Calculated by dividing the amount of current period increase or decrease by the prior period's total Company revenue adjusted to exclude the impacts of divestitures for the current period (\$3,460 million and \$10,212 million for the three- and nine-month periods, respectively).

(b) The amounts reported herein represent the changes in our revenue attributable to average yield for the total Company. We analyze the changes in average yield in terms of related business revenues in order to differentiate the changes in yield attributable to our pricing strategies from the changes that are caused by market-driven price changes in commodities. The following table summarizes changes in revenues from average yield on a related-business basis (dollars in millions):

	Chan Three M Septe	-to-Period ge for the onths Ended mber 30, vs. 2012	Chang Nine Mo Septe	-to-Period ge for the nths Ended mber 30, vs. 2012
	Amount	As a % of Related Business(i)	Amount	As a % of Related Business(i)
Average yield:				
Collection, landfill and transfer	\$69	2.5%	\$171	2.1%
Waste-to-energy disposal(ii)	<u>(1)</u>	(0.9)	(6)	(1.8)
Collection and disposal(ii)	68	2.3	165	1.9
Recycling commodities	5	1.6	(80)	(7.7)
Electricity(ii)	7	9.7	17	8.7
Fuel surcharges and mandated fees	10	6.0	35	7.3
Total	\$90	2.6%	\$137	1.3%

(i) Calculated by dividing the increase or decrease for the current period by the prior period's related business revenue, adjusted to exclude the impacts of divestitures for the current period. The table below summarizes the related business revenues for the three and nine months ended September 30, 2012 adjusted to exclude the impacts of divestitures (in millions):

	Denominator			
	Three Months Ended September 30		1	e Months Ended tember 30
Related business revenues:				
Collection, landfill and transfer	\$	2,794	\$	8,180
Waste-to-energy disposal		110		325
Collection and disposal		2,904		8,505
Recycling commodities		316		1,034
Electricity		72		195
Fuel surcharges and mandated fees		168		478
Total Company	\$	3,460	\$	10,212

(ii) Average revenue growth from yield for "Collection and disposal" excludes all electricity-related revenues generated by our Wheelabrator business and our landfill gas-to-energy operations, which are reported as "Electricity" revenues.

Our revenues increased \$160 million, or 4.6%, for the three months ended September 30, 2013 as compared with the prior year period and \$268 million, or 2.6%, for the nine months ended September 30, 2013 as compared with the prior year period. During the three- and nine-month periods ended September 30, 2013, our revenue growth has been driven by: (i) revenues from acquisitions, particularly the acquisition of Greenstar, which increased consolidated revenues by \$39 million and \$104 million for the three and nine months ended September 30, 2013, respectively; and the acquisition of RCI in July 2013, which contributed \$41 million to our revenue increase; (ii) increased revenue growth from our collection and disposal average yield; (iii) higher revenues provided by our fuel surcharge program and (iv) higher electricity prices at our merchant waste-to-

energy facilities. Offsetting these revenue increases, for the current year periods, were revenue declines due to lower volumes, foreign currency translation from our Canadian operations and lower recyclable commodity prices, although we did see slightly higher year-over-year recyclable commodity prices for the three-month period.

The following provides further details associated with our period-to-period change in revenues.

Average yield

Collection and disposal average yield — This measure reflects the effect on our revenue from the pricing activities of our collection, transfer, landfill and waste-to-energy disposal operations, exclusive of volume changes. Revenue growth from collection and disposal average yield includes not only base rate changes and environmental and service fee increases, but also (i) certain average price changes related to the overall mix of services, which are due to both the types of services provided and the geographic locations where our services are provided; (ii) changes in average price from new and lost business and (iii) price decreases to retain customers.

Our revenue growth from yield on our collection and disposal lines of business was \$68 million, or 2.3%, and \$165 million, or 1.9%, for the three and nine months ended September 30, 2013, respectively, which demonstrates our commitment to our pricing strategies. During 2013, we have executed aggressive price increases on our existing business, decreased the dollar impact of rollbacks associated with those price increases and improved pricing on our new business. This increase in revenue from yield was driven primarily by our collection operations with growth of \$66 million and \$163 million for the three and nine months ended September 30, 2013, respectively. We experienced growth in all three of our principal collection lines of business in 2013 for the three and nine months ended September 30, 2013, as follows:

	Average Y	ield
	Three Months	Nine Months
	Ended	Ended
	September 30	September 30
Commercial	3.4%	2.8%
Industrial	5.5%	4.6%
Residential	2.0%	1.8%

Part of the year-over-year revenue growth from yield is attributable to the new regulatory cost recovery fee that we instituted in April 2013 to help us recover a portion of the significant regulatory costs and fees, such as host fees and disposal taxes, which have not been recouped by our pricing programs. This new fee contributed approximately \$18 million and \$26 million to our revenue growth for the three and nine months ended September 30, 2013, respectively. Additionally, revenue growth from yield in our industrial line of business was aided by our continued expansion in the Energy Service business, which typically has higher average rates due to extended transportation distances, special waste handling costs and higher disposal costs. With respect to our residential line of business, we are focused on bidding on new contracts that improve our yield performance and increase our overall returns. Our effort to increase yield in our residential line of business is a challenge due principally to a very competitive environment. A high percentage of our residential business is in municipal franchise markets, and many municipalities are facing significant budget challenges, which results in very competitive bid processes as we rebid contracts and try to win new contracts. Finally, yield growth from our landfill and transfer station operations increased slightly in the current quarter. Improving yield in our landfill business has proved to be a challenge, due in part to excess disposal capacity that exists in many of the markets in which we own or operate landfills.

The \$1 million and \$6 million revenue decline from yield in our waste-to-energy disposal line of business for the three and nine months ended September 30, 2013, respectively, was due to the expiration and renegotiation of two long-term waste-to-energy disposal contracts in South Florida. The negative impact from the renegotiated contracts will continue through the first half of 2014.

Revenues from our environmental fee, which are included in average revenue growth from yield on collection and disposal, increased \$1 million and \$3 million for the three and nine months ended September 30,

2013, respectively. These revenues were \$89 million and \$260 million for the three and nine months ended September 30, 2013, respectively, as compared with \$88 million and \$257 million for the three and nine months ended September 30, 2013, respectively. Additionally, as mentioned above, we instituted a new regulatory cost recovery fee in April 2013 that contributed approximately \$18 million and \$26 million to revenue growth for the three and nine months ended September 30, 2013, respectively.

Recycling commodities — During the third quarter of 2013, we saw higher year-over-year recyclable commodity prices, which resulted in a \$5 million revenue increase, due principally to commodity prices in the third quarter of 2012 being at the lowest levels since late 2009. Decreases in the prices of the recycling commodities we sold resulted in revenue declines of \$80 million for the nine months ended September 30, 2013 with an estimated negative impact on income from operations of approximately \$24 million. During the first nine months of 2013, our year-over-year commodity prices declined about 7% as compared with the same prior year period.

Fuel surcharges and mandated fees — These revenues, which are predominantly generated by our fuel surcharge program increased by \$10 million and \$32 million for the three and nine months ended September 30, 2013, respectively. These revenues fluctuate in response to changes in the national average prices for diesel fuel on which our surcharge is based. During 2013, the fuel surcharge revenues increased in response to an increase in the factor applied to diesel fuel prices in order to calculate the surcharge. The mandated fees included in this line item are primarily related to pass-through fees and taxes assessed by various state, county and municipal governmental agencies at our landfills and transfer stations.

Volume — Changes in our volume caused our revenue to decrease \$22 million, or 0.6%, and \$56 million, or 0.5%, for the three and nine months ended September 30, 2013, respectively. We continue to experience revenue declines in our collection business, principally due to lower volumes in our commercial collection, and to a lesser extent, our residential line of business, which we attribute to the effects of pricing, competition and diversion of waste by customers. Additionally, for the nine months ended September 30, 2013, we experienced revenue declines due to lower volumes associated with the loss of certain strategic accounts including certain large retail mall customers primarily in the second quarter of 2013. Revenue growth from higher landfill volumes, primarily in our municipal solid waste business, partially offset these revenue decreases.

Acquisitions and divestitures — Revenues increased \$103 million and \$204 million for the three and nine months ended September 30, 2013, respectively, due to acquisitions. This increase was principally associated with the acquisition of Greenstar, which is reported in our "Recycling" line of business, and the acquisition of RCI in July 2013, which is reported primarily in our "Collection" line of business.

Operating Expenses

Our operating expenses increased by \$96 million, or 4.3%, and \$190 million, or 2.9%, for the three and nine months ended September 30, 2013, respectively, as compared with the same periods in 2012. Our operating expenses were 64.2% and 64.4% of revenues for the third quarter of 2013 and 2012, respectively, and 65.3% and 65.1% for the nine months ended September 30, 2013 and 2012, respectively. The increase in our operating expenses during the three and nine months ended September 30, 2013, respectively, can largely be attributed to the following:

Recycling operations — In January 2013, we acquired Greenstar, an operator of recycling and resource recovery facilities. The acquisition increased operating expenses by \$37 million and \$100 million for the three and nine months ended September 30, 2013, respectively, primarily increasing cost of goods sold and, to a lesser extent, labor and related benefits and other categories. Also contributing to the increases for the reported periods were the impacts of higher processing costs associated with outbound quality control. The increases in operating expenses associated with Greenstar were offset by increased revenues resulting from the acquisition in both the three and nine months ended September 30, 2013.

The following table summarizes the major components of our operating expenses, which include the impact of foreign currency translation, for the three- and nine-month periods ended September 30 (dollars in millions):

	September 30, Period				iber 30,), Period		
	2013	2012	Cha		2013	2012	Char	ıge
Labor and related benefits	\$ 641	\$ 595	\$46	7.7%	\$ 1,874	\$ 1,793	\$ 81	4.5%
Transfer and disposal costs	252	244	8	3.3	734	712	22	3.1
Maintenance and repairs	294	284	10	3.5	897	876	21	2.4
Subcontractor costs	306	314	(8)	(2.5)	879	889	(10)	(1.1)
Cost of goods sold	260	224	36	16.1	754	706	48	6.8
Fuel	154	161	(7)	(4.3)	456	485	(29)	(6.0)
Disposal and franchise fees and taxes	171	163	8	4.9	490	467	23	4.9
Landfill operating costs	62	58	4	6.9	174	168	6	3.6
Risk management	57	52	5	9.6	181	170	11	6.5
Other	128	134	(6)	(4.5)	406	389	17	4.4
	\$ 2,325	\$ 2,229	\$96	4.3%	\$ 6,845	\$ 6,655	\$190	2.9%

Other significant changes in our operating expenses are discussed below.

- & Labor and related benefits Significant items affecting the comparability of expenses for the periods presented include:
 - % higher wages due to merit increases effective in the second quarter of 2013 and the Greenstar acquisition;
 - % higher incentive compensation expense in current year periods due to higher anticipated payouts;
 - % increased contract labor in the current year periods principally attributed to the recycling line of business;
 - & non-cash charges incurred during the third quarter of 2013 and the second quarter of 2012 as a result of our partial withdrawals from underfunded multiemployer pension plans; and
 - % lower headcount, exclusive of acquisitions, in the current year periods.
- & Maintenance and repairs The increase was driven by (i) the Greenstar acquisition; (ii) fleet labor costs and (iii) planned maintenance costs in the third quarter of 2013 at our waste-to-energy facilities.
- Subcontractor costs The decrease in subcontractor costs was largely driven by lower costs in our WM Sustainability Business Services due to lower volumes offset, in part, by higher costs associated with special waste projects in several of our Areas and the acquired RCI operations.
- Cost of goods sold The increase in cost of goods sold is attributable, in part, to higher customer rebates resulting from higher volumes in our recycling commodity business driven primarily by the Greenstar acquisition. Higher market prices for recyclable commodities in the current quarter as compared with the third quarter of 2012 also contributed to the increase in cost of goods sold. Lower commodity prices during the nine month period ending September 30, 2013 partially offset these increases.
- *Fuel* The decrease in fuel expense when compared to the prior year period was due to (i) a retroactive CNG fuel excise tax credit recognized in the first quarter of 2013; (ii) reduced fuel purchases due to reduced collection volumes; (iii) lower costs as we convert our fleet to CNG vehicles and (iv) lower fuel prices.
- Disposal and franchise fees and taxes The increase in costs is attributable to (i) higher disposal fees and taxes due to higher landfill volumes and (ii) higher municipal franchise fees relating to the collection line of business in certain of our Areas. A disposal surcharge at one of our waste-to-energy facilities in the second quarter of 2013 contributed to the increased costs in the nine month period ended September 30, 2013.

Nine Months

- Landfill operating costs The increases in landfill operating costs when compared to the prior periods was driven in large part by higher leachate costs caused by increased precipitation in several of our Areas. The increase in the nine month period ended September 30, 2013, was offset in large part by a favorable adjustment due to an increase in U.S. Treasury rates used to discount the present value of our environmental remediation obligations and recovery assets.
- Risk management Increases in the third quarter 2013 were driven by higher fleet insurance costs and accident claims while increases in the nine months ended September 30, 2013 were driven mainly by higher workers' compensation claims.
- Other Decreases in the third quarter of 2013 compared to the prior year period are attributable to: (i) adjustments to contingent consideration associated with the Greenstar acquisition, which is based on changes in certain recyclable commodity indexes and (ii) strike related costs incurred in the prior year period. Increases in the nine-month period ended September 30, 2013 were the result of (i) telecommunications, driven by our initiative to equip our fleet with onboard computers; (ii) utilities; (iii) property taxes; and (iv) rentals. These increases were partially offset by the third quarter decreases noted above.

Three Months

Selling, General and Administrative

Our selling, general and administrative expenses increased by \$14 million, or 4.2% and decreased \$24 million, or 2.2%, when comparing the three and nine months ended September 30, 2013 with the comparable prior year periods. As a percentage of revenue, our selling, general and administrative expenses decreased from 9.7% for the third quarter of 2012 to 9.6% for the third quarter of 2013, and decreased from 10.9% for the nine months ended September 30, 2012 to 10.4% for the nine months ended September 30, 2013.

The following table summarizes the major components of our selling, general and administrative expenses for the three- and nine-month periods ended September 30 (dollars in millions):

	Enc	Ended		eriod-to-		Ended		d-to-		
	Septem	ber 30,	Period		September 30,		er 30, Perio			
	2013	2012	Change		Change		2013	2012	Cha	inge
Labor and related benefits	\$232	\$187	\$ 45	24.1%	\$ 693	\$ 643	\$ 50	7.8%		
Professional fees	29	35	(6)	(17.1)	95	122	(27)	(22.1)		
Provision for bad debts	2	13	(11)	(84.6)	28	42	(14)	(33.3)		
Other	86	100	(14)	(14.0)	276	309	(33)	(10.7)		
	\$ 349	\$ 335	\$ 14	4.2%	\$ 1,092	\$ 1,116	\$(24)	(2.2)%		

Labor and related benefits - Factors affecting the current year change in our labor and related benefits costs include:

- 8 Higher incentive compensation costs of \$45 million and \$73 million for the three- and nine-month periods ended September 30, 2013, respectively, due to higher anticipated payouts.
- Higher non-cash compensation charges of \$5 million and \$27 million for the three- and nine-month periods ended September 30, 2013, respectively, attributable to our long-term incentive plan due primarily to the recognition of expense for the payout of performance share units granted in 2010, which was approved in the first quarter of 2013. Expense associated with these awards had been reversed in the second quarter of 2012 when it no longer appeared probable that threshold performance would be achieved.
- Costs savings realized from our July 2012 restructuring of approximately \$7 million and \$43 million for the three- and nine-month periods ended September 30, 2013, respectively.
- & Lower payroll tax expense related to the lower annual incentive plan payout to participants in 2013 as compared to 2012.

Professional fees — In 2013, our professional fees decreased primarily due to (i) consulting fees incurred during 2012 associated with the start-up phase of our cost savings programs, (ii) higher costs in 2012 related to international development of our waste-to-energy business and (iii) higher legal fees in 2012.

Provision for bad debts — Our provision for bad debts decreased as a result of the collection of certain fully reserved receivables related to our Puerto Rico operations. Additionally, many of the billing delay issues we experienced throughout 2012 with certain of our strategic account customers have been resolved, favorably affecting our year-over-year bad debt comparisons.

Other — During 2013, controllable costs associated with advertising, travel and entertainment, building and equipment costs, computer costs, and seminars and education have declined primarily as a result of our July 2012 restructuring and continued focus on cost-control initiatives. Additionally, we experienced higher 2012 litigation settlement costs, principally noted in the three-month comparison.

Depreciation and Amortization

The following table summarizes the components of our depreciation and amortization expense for the three- and nine-month periods ended September 30 (dollars in millions):

	Three Months									
	Ended Period-t		Ended Period-to-		Ene		Perio	d-to-		
	Septem	uber 30,	Per	Period		Period Septer		September 30,		iod
	2013	2013 2012 Change		2013	2012	Change				
Depreciation of tangible property and equipment	\$ 216	\$ 207	\$9	4.3%	\$ 638	\$ 622	\$ 16	2.6%		
Amortization of landfill airspace	107	107		_	310	297	13	4.4		
Amortization of intangible assets	21	17	4	23.5	58	52	6	11.5		
	\$ 344	\$ 331	\$ 13	3.9%	\$ 1,006	\$ 971	\$ 35	3.6%		

Restructuring

During the nine months ended September 30, 2013, we recognized a total of \$13 million of pre-tax restructuring charges, of which \$6 million was related to employee severance and benefit costs, including costs associated with our acquisition of Greenstar and our July 2012 restructuring. The remaining charges were primarily related to operating lease obligations for property that will no longer be utilized.

During the nine months ended September 30, 2012, we recognized a total of \$51 million of pre-tax restructuring charges, of which \$46 million was related to employee severance and benefit restructuring charges, primarily attributable to our July 2012 restructuring and including other charges associated with the reorganization of Oakleaf and certain other actions taken by the Company in 2012.

(Income) Expense from Divestitures, Asset Impairments and Unusual Items

During the third quarter of 2013, we recognized charges of \$23 million, primarily related to (i) a \$12 million impairment charge to write down goodwill and indefinite-lived intangibles related to an investment in a waste diversion technology company, partially offset by a \$6 million benefit to noncontrolling interest and (ii) losses on divestitures primarily related to investments in oil and gas producing properties. During the nine months ended September 30, 2013, we also recognized additional charges of \$15 million, primarily related to (i) a \$14 million impairment charge in the second quarter of 2013 at a waste-to-energy facility as a result of projected operating losses, which caused us to write down the carrying value of the facility's property, plant and equipment to its estimated fair value; (ii) losses on divestitures in the first quarter of 2013 primarily related to investments in oil and gas producing properties and (iii) other charges to impair goodwill and write down the carrying value of the scale set to their estimated fair values; related to certain of our operations, all of which are individually immaterial. Partially offsetting these charges during 2013 were gains on divestitures including the sale of a transfer station in our Greater Mid-Atlantic Area during the first quarter of 2013.

During the third quarter of 2012, we recognized impairment charges aggregating \$22 million, relating in large part to (i) the impairment of a note receivable from an entity that we invested in that went into liquidation

(see "Equity in net losses of unconsolidated entities" below), (ii) the impairment of certain related assets due to our decision not to use or develop those assets and (iii) the impairment of an oil and gas well as a result of projected operating losses that caused us to write-down the carrying value of the oil and gas well to its estimated fair value. During the nine months ended September 30, 2012, we also recognized additional impairment charges of \$33 million in the second quarter of 2012, largely attributable to two facilities in our medical waste services business as a result of projected operating losses at each of these facilities, which caused us to write down the carrying values of the facilities' operating permits and property, plant and equipment to their estimated fair values. These charges are included in our "Other" operations in Note 8 to the Condensed Consolidated Financial Statements.

We review long-lived assets for impairment whenever events or changes in circumstances indicate that a long-lived asset might not be recoverable. Additionally, at least annually as of October 1, and more frequently if warranted, we assess our goodwill for impairment.

Income from Operations

The following table summarizes income from operations for the three- and nine-month periods ended September 30 (dollars in millions):

	Three 1 		Period-to- Period Change		Period		Period		Period		Period		Period		Nine M Ended Sep 2013		Period-to- Period Change	
Solid Waste:																		
Tier 1	\$ 224	\$ 213	\$11	5.2%	\$ 652	\$ 641	\$ 11	1.7%										
Tier 2	347	328	19	5.8	970	946	24	2.5										
Tier 3	148	126	22	17.5	388	370	18	4.9										
Solid Waste	719	667	52	7.8	2,010	1,957	53	2.7										
Wheelabrator	39	46	(7)	(15.2)	52	68	(16)	(23.5)										
Other	(31)	(68)	37	(54.4)	(99)	(167)	68	(40.7)										
Corporate and Other.	(150)	(145)	(5)	3.4	(474)	(491)	17	(3.5)										
Total	\$ 577	\$ 500	\$77	15.4%	\$ 1,489	\$ 1,367	\$122	8.9%										

Items affecting all components of our results of operations during the three and nine months ended September 30, 2013 include (i) restructuring charges recognized in the third quarter of 2012 associated with our July 2012 restructuring; (ii) subsequent benefits realized as a result of our July 2012 restructuring and ongoing cost containment efforts and (iii) increased labor costs due to merit increases effective in the second quarter of 2013 and higher expected year-over-year incentive compensation payouts. Also affecting our components, excluding Wheelabrator, was the reclassification of employees to *Solid Waste* from *Other* and *Corporate and Other*.

Solid Waste — The most significant items affecting the results of operations of our Solid Waste business during the three and nine months ended September 30, 2013 as compared with the prior year periods are summarized below:

Our base business excluding recycling benefited from (i) internal revenue growth, principally in our collection and landfill lines of business and (ii) increased fuel cost recovery. These favorable variances were partially offset by net cost increases mainly driven by transfer and disposal costs.

Results from our recycling business were lower compared to prior year periods due primarily to: (i) higher processing costs driven in part by increased outbound quality control; (ii) operating losses related to the acquired operations of Greenstar and (iii) lower prices for commodities, although we did see slightly higher year-over-year prices in the current quarter.

Other items affecting the reported periods include:

- % the RCI acquisition;
- % reduced fuel expense due to a retroactive CNG excise fuel tax credit recognized in the first quarter of 2013; and
- % decreased bad debt expense due primarily to collection of reserved receivables.

Wheelabrator — The decrease in income from operations of our Wheelabrator business for the three and nine months ended September 30, 2013 as compared to 2012 was largely driven by (i) maintenance and repair costs, primarily due to differences in the timing and scope of planned maintenance activities; (ii) higher labor costs; (iii) a disposal surcharge at one of our waste-to-energy facilities and (iv) a \$14 million impairment charge recognized in the second quarter of 2013.

- Other Significant items affecting the comparability of results for the periods presented include:
- & impairment charges incurred during 2012 in our medical waste service business and investments in waste diversion technology companies;
- % improved results from our organics and medical waste service businesses;
- & year-to-date losses from our efforts to integrate our strategic accounts business with Oakleaf, including the loss of certain strategic accounts. However, in the current quarter, we have experienced year-over-year improvements as a result of our system and process enhancements; and
- a favorable adjustment to contingent consideration associated with the Greenstar acquisition which is based on changes in certain recyclable commodity indexes offset by higher administrative and restructuring costs associated with the acquired operations.
- Corporate and Other Significant items affecting the comparability of expenses for the periods presented include:
- & decreased professional fees primarily due to higher consulting fees incurred during 2012 in connection with the start-up phase of our cost savings programs; and
- & a favorable adjustment to our estimated environmental remediation obligations and recovery assets in the second quarter of 2013 due to increases in U.S. Treasury rates.

Interest Expense

Our interest expense was \$119 million and \$363 million during the three and nine months ended September 30, 2013, respectively, compared with \$123 million and \$366 million during the three and nine months ended September 30, 2012, respectively. Our interest expense has declined on a year-over-year basis in spite of a slight increase in our average debt balances. This is primarily attributable to interest expense reductions that have been achieved by (i) remarketing tax-exempt debt at lower rates, (ii) issuing new debt at lower fixed interest rates than debt repaid upon scheduled maturities and (iii) the amendment and restatement of our revolving credit facility.

Equity in Net Losses of Unconsolidated Entities

Our "Equity in net losses of unconsolidated entities" is primarily related to our noncontrolling interests in two limited liability companies established to invest in and manage low-income housing properties and a refined coal facility, as well as noncontrolling investments made to support our strategic initiatives and unconsolidated trusts for final capping, closure, post-closure or environmental obligations. The tax impacts realized as a result of our investments in low-income housing properties and the refined coal facility are discussed below in *Provision for Income Taxes*. Refer to Notes 5 and 14 to the Condensed Consolidated Financial Statements for more information related to these investments. The decreases in 2013 are primarily attributable to a \$10 million charge recognized in the third quarter of 2012 related to a payment we made under a guarantee on behalf of an unconsolidated entity that went into liquidation. This investment was accounted for under the equity method.

Other, net

We recognized other, net expense of \$3 million and \$12 million for the three and nine months ended September 30, 2013, respectively, compared with other, net expense of \$14 million and \$16 million for the three and nine months ended September 30, 2012, respectively. The expense in 2013 was impacted by impairment charges of \$11 million in the first quarter of 2013 relating to other-thantemporary declines in the value of two investments in waste diversion technology companies accounted for under the cost method. Partially offsetting these charges was a \$4 million gain on the sale of a similar investment recognized in the second quarter of 2013. The expense in 2012 was impacted by an impairment charge of \$14 million in the third quarter of 2012 related to an other-than-temporary decline in the value of another investment in a waste diversion technology company accounted for under the cost method. The remaining expenses recognized during the reported periods are primarily related to the impact of foreign currency translation.

Provision for Income Taxes

We recorded a provision for income taxes of \$155 million during the third quarter of 2013, representing an effective income tax rate of 34.3%, compared with a provision for income taxes of \$125 million during the third quarter of 2012, representing an effective income tax rate of 36.1%. Our effective income tax rate for the nine months ended September 30, 2013 was 33.6% compared with 34.5% for the nine months ended September 30, 2012. The increase in our provision for income taxes when comparing the three months ended September 30, 2013 with the prior year is primarily due to an increase in pre-tax income.

Our investments in low-income housing properties and the refined coal facility reduced our provision for income taxes by \$11 million and \$5 million, respectively, for the three months ended September 30, 2013. These investments reduced our provision for income taxes by \$11 million and \$6 million, respectively, for the nine months ended September 30, 2013. These investments reduced our provision for income taxes by \$11 million and \$6 million, respectively, for the nine months ended September 30, 2013. These investments reduced our provision for income taxes by \$11 million and \$6 million, respectively, for the three months ended by \$27 million and \$14 million, respectively, for the nine months ended September 30, 2012. Refer to Note 5 to the Condensed Consolidated Financial Statements for more information related to these investments.

Noncontrolling Interests

Net income attributable to noncontrolling interests was \$6 million and \$26 million for the three and nine months ended September 30, 2013, respectively, and \$9 million and \$32 million for the three and nine months ended September 30, 2012, respectively. These amounts are principally related to third parties' equity interests in two limited liability companies that own three waste-to-energy facilities operated by our Wheelabrator business. Refer to Note 14 to the Condensed Consolidated Financial Statements for information related to the consolidation of these variable interest entities. The decreases in 2013 are primarily attributable to the recognition of a \$6 million benefit associated with the \$12 million impairment charge discussed above in (*Income) Expense from Divestitures, Asset Impairments and Unusual Items*.

Liquidity and Capital Resources

Summary of Cash and Cash Equivalents, Restricted Trust and Escrow Accounts and Debt Obligations

The following is a summary of our cash and cash equivalents, restricted trust and escrow accounts and debt balances as of September 30, 2013 and December 31, 2012 (dollars in millions):

	September 30, 2013	December 31, 2012
Cash and cash equivalents	<u>\$ 157</u>	<u>\$ 194</u>
Restricted trust and escrow accounts:		
Final capping, closure, post-closure and environmental remediation funds	\$ 126	\$ 125
Tax-exempt bond funds	41	1
Other	12	12
Total restricted trust and escrow accounts	\$ 179	\$ 138
Debt:	<u> </u>	
Current portion	\$ 568	\$ 743
Long-term portion	9,491	9,173
Total debt	\$ 10,059	\$ 9,916
Increase in carrying value of debt due to hedge accounting for interest rate swaps	\$ 64	\$ 79

As of September 30, 2013, we had (i) \$510 million of debt maturing within the next 12 months, including \$350 million of 5.0% senior notes that mature in March 2014 and \$92 million of taxexempt bonds; (ii) short-term borrowings and advances outstanding under credit facilities with long-term maturities, including \$140 million of borrowings outstanding under the revolving credit facility and U.S.\$456 million of advances under our Canadian credit facility term loan; and (iii) \$857 million of tax-exempt borrowings subject to repricing within the next 12 months. Based on our intent and ability to refinance a portion of this debt on a long-term basis as of September 30, 2013, we have classified \$1.4 billion of this debt as long-term and the remaining \$568 million as.

In July 2013, we amended and restated our \$2.0 billion revolving credit facility, increasing our total credit capacity to \$2.25 billion and extending the term through July 2018.

Summary of Cash Flow Activity

The following is a summary of our cash flows for the nine-month periods ended September 30 (in millions):

	Nine Mo	nths
	Ende	d
	Septembe	er 30,
	2013	2012
Net cash provided by operating activities	\$ 1,858	\$ 1,718
Net cash used in investing activities	\$ 1,858 \$(1,432)	\$ 1,718 \$(1,361)
Net cash used in financing activities	\$ (459)	\$ (219)

Net Cash Provided by Operating Activities — We generated \$1,858 million of cash flow from operating activities during the nine-month period ended September 30, 2013, compared with \$1,718 million during the nine-month period ended September 30, 2012. The \$140 million increase was primarily driven by the items described below:

& Increased earnings — Our income from operations, excluding depreciation and amortization, impairments and restructuring costs increased by \$102 million, on a year-over-year basis.

- Decreased bonus payments Payments for our annual incentive plan are made in the first quarter of the year based on prior year performance. Our cash flow from operating activities was favorably impacted by approximately \$86 million on a year-over-year basis as the annual incentive plan payments made in the first quarter of 2013 were significantly lower than the payments made in the first quarter of 2012.
- Forward-starting swaps During the third quarter of 2012, forward starting interest rate swaps associated with anticipated fixed rate debt issuances were terminated contemporaneously with the issuance of senior notes and we paid cash of \$59 million. This cash payment has been classified as a change in "Other liabilities" within "Net cash provided by operating activities" in the Condensed Consolidated Statement of Cash Flows.

These were partially offset by:

- Increased income tax payments Cash paid for income taxes, net of excess tax benefits associated with equity-based transactions, was approximately \$165 million higher on a year-over-year basis due to higher pre-tax earnings and the timing of estimated tax payments.
- Termination of interest rate swaps In April 2012, we elected to terminate our \$1 billion interest rate swap portfolio associated with senior notes that were scheduled to mature from November 2012 through March 2018. Upon termination of the swaps, we received \$72 million in cash for their fair value. The cash proceeds received from the termination of interest rate swap agreements have been classified as a change in "Other assets" within "Net cash provided by operating activities" in the Condensed Consolidated Statement of Cash Flows.

Our cash flow from operating activities is impacted on a year-over-year basis by changes in our working capital accounts. After consideration of the items mentioned above, the impacts of working capital changes were favorable during the reported period. Although our working capital changes may vary from year to year, they are typically driven by changes in accounts receivable, which are affected by both revenue changes and timing of payments received, and changes in accounts payable, and other accruals which are affected by both cost changes and timing of payments.

Net Cash Used in Investing Activities — The most significant items included in our investing cash flows for the nine-month periods ended September 30, 2013 and 2012 are summarized below:

- Capital expenditures We used \$824 million during the first nine months of 2013 for capital expenditures compared with \$1,132 million in the first nine months of 2012, a decrease of \$308 million. The decrease can generally be attributed to increased focus on capital spending management and the impact of timing differences associated with cash payments for the previous years' fourth quarter capital spending that was paid in the first quarter of 2013 compared with approximately \$244 million of our fourth quarter 2012 spending was paid in cash in the first quarter of 2013 compared with approximately \$244 million of our fourth quarter 2012.
- Acquisitions Our spending on acquisitions was \$698 million in the first nine months of 2013 compared with \$178 million in the first nine months of 2012. In 2013, our acquisitions consisted primarily of the recycling operations of Greenstar, for which we paid \$170 million, and substantially all of the assets of RCI, for which we paid \$481 million. The remainder of our 2013 acquisitions related to collection and energy services operations. In 2012, our acquisitions consisted primarily of interests in oil and gas producing properties through two transactions, for which we paid \$94 million. The remainder of our 2012 acquisitions related to collection and recycling operations.

Net Cash Used in Financing Activities — During the first nine months of 2013, net cash used in financing activities was \$459 million, compared with \$219 million during the comparable prior year period. The most significant items affecting the comparison of our financing cash flows for the nine-month periods ended September 30, 2013 and 2012 are summarized below:

& Net debt borrowings (repayments) — The following summarizes our cash borrowings and debt repayments during each period (in millions):

	E	Nine Months Ended	
		September 30,	
	2013	2012	
Borrowings:			
Revolving credit facility(a)	\$ 200	\$	
Canadian credit facility(a)	940	115	
Senior Notes	_	495	
Capital leases and other debt	78	75	
	\$ 1,218	\$ 685	
Repayments:			
Revolving credit facility(a)	\$ (460)	\$(150)	
Canadian credit facility(a)	(574)	(142)	
Tax-exempt bonds	(107)	(86)	
Capital leases and other debt	(103)	(95)	
-	\$(1,244)	\$(473)	
Net (repayments) borrowings	\$ (26)	\$ 212	

(a) Due to the short-term maturities of the borrowings under these credit facilities, we have reported certain of these cash flows on a net basis.

Refer to Note 3 to the Condensed Consolidated Financial Statements for additional information related to our debt borrowings and repayments.

- Dividend payments We paid \$512 million in cash dividends in the first nine months of 2013 compared with \$493 million in the first nine months of 2012. The increase in dividend payments is due to our quarterly per share dividend declared increasing from \$0.355 in 2012 to \$0.365 in 2013 and increased shares outstanding.
- Proceeds from the exercise of common stock options We received \$116 million for the exercise of stock options in the first nine months of 2013 compared with \$39 million in the first nine months of 2012. The increase in exercised stock options is primarily due to the increase in the Company's stock price combined with approaching expirations of stock options in the first nine months of 2013.

Liquidity Impacts of Income Tax Items

Recent Legislation — The American Taxpayer Relief Act of 2012 was signed into law on January 2, 2013 and includes an extension for one year of the bonus depreciation allowance. As a result, 50% of qualifying capital expenditures on property placed in service before January 1, 2014 can be depreciated immediately. The acceleration of deductions on 2013 qualifying capital expenditures resulting from the bonus depreciation provisions has no impact on our effective income tax rate for 2013, but reduces our cash taxes in the periods in which the deductions are taken.

The acceleration of depreciation deductions related to qualifying capital expenditures in 2012 decreased our full year 2012 cash taxes by approximately \$90 million and, based on our current forecast of 2013 capital expenditures, we estimate a reduction in our full year 2013 cash taxes of approximately \$105 million related to

qualifying capital expenditures in 2013. However, taking accelerated deductions results in increased cash taxes in subsequent periods when the deductions for these capital expenditures would have otherwise been taken. On a net basis, after taking into account the effect of all applicable years' bonus depreciation programs, the deductions taken in previous years from acceleration programs more than offset the benefits received in 2012 and results in an immaterial benefit in 2013.

In addition, as a result of the application in 2012 of the depreciation safe harbor provisions of the Income Tax Regulations, our estimated tax payments in 2013 were higher in the first nine months of the year than the comparable prior year period. We also expect our full year tax payments to be approximately \$150 million higher than the tax payments made in 2012.

Uncertain Tax Positions — We have liabilities associated with unrecognized tax benefits and related interest. These liabilities are primarily included as a component of long-term "Other liabilities" in our Condensed Consolidated Balance Sheet because the Company generally does not anticipate that settlement of the liabilities will require payment of cash within the next 12 months. We are not able to reasonably estimate when we would make any cash payments required to settle these liabilities, but we do not believe that the ultimate settlement of our obligations will materially affect our liquidity. We anticipate that approximately \$7 million of liabilities for uncertain tax positions, including accrued interest, and \$3 million of related deferred tax assets may be reversed within the next 12 months. The anticipate reversals are primarily related to state tax items, none of which are material, and are expected to result from audit settlements or the expiration of the applicable statute of limitations period.

Off-Balance Sheet Arrangements

We have financial interests in unconsolidated variable interest entities as discussed in Note 14 to the Condensed Consolidated Financial Statements. Additionally, we are party to guarantee arrangements with unconsolidated entities as discussed in the *Guarantees* section of Note 7 to the Condensed Consolidated Financial Statements. These arrangements have not materially affected our financial position, results of operations or liquidity during the nine months ended September 30, 2013, nor are they expected to have a material impact on our future financial position, results of operations or liquidity.

Seasonal Trends

Our operating revenues tend to be somewhat higher in summer months, primarily due to higher volume of construction and demolition waste. The volumes of industrial and residential waste in certain regions where we operate also tend to increase during the summer months. Our second and third quarter revenues and results of operations typically reflect these seasonal trends.

Additionally, certain destructive weather conditions that tend to occur during the second half of the year, such as the hurricanes that most often impact our operations in the Southern and Eastern U.S., can actually increase our revenues in the areas affected. While weather-related and other "one-time" occurrences can boost revenues through additional work for a limited time span, as a result of significant start-up costs and other factors, such revenue sometimes generates earnings at comparatively lower margins. Certain weather conditions, including severe winter storms, may result in the temporary suspension of our operations, which can significantly affect the operating results of the affected regions. The operating results of our first quarter also often reflect higher repair and maintenance expenses because we rely on the slower winter months, when waste flows are generally lower, to perform scheduled maintenance at our waste-to-energy facilities.

Inflation

While inflationary increases in costs, including the cost of diesel fuel, have affected our income from operations margins in recent years, we believe that inflation generally has not had, and in the near future is not expected to have, any material adverse effect on our results of operations. However, a portion of our collection revenues are generated under long-term agreements with price adjustments based on various indexes intended to measure inflation. Additionally, management's estimates associated with inflation have had, and will continue to have, an impact on our accounting for landfill and environmental remediation liabilities.



Item 3. Quantitative and Qualitative Disclosures About Market Risk.

Information about market risks as of September 30, 2013, does not differ materially from that discussed under Item 7A in our Annual Report on Form 10-K for the year ended December 31, 2012.

Item 4. Controls and Procedures.

Effectiveness of Controls and Procedures

Our management, with the participation of our principal executive and financial officers, has evaluated the effectiveness of our disclosure controls and procedures in ensuring that the information required to be disclosed in reports that we file or submit under the Securities Exchange Act of 1934, as amended, is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, including ensuring that such information is accumulated and communicated to management (including the principal executive and financial officers) as appropriate to allow timely decisions regarding required disclosure. Based on such evaluation, our principal executive and financial officers have concluded that such disclosure controls and procedures were effective as of September 30, 2013 (the end of the period covered by this Quarterly Report on Form 10-Q).

Changes in Internal Control over Financial Reporting

Management, together with our CEO and CFO, evaluated the changes in our internal control over financial reporting during the quarter ended September 30, 2013. We determined that there were no changes in our internal control over financial reporting during the quarter ended September 30, 2013 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II.

Item 1. Legal Proceedings.

Information regarding our legal proceedings can be found under the "Environmental Matters" and "Litigation" sections of Note 7, Commitments and Contingencies, to the Condensed Consolidated Financial Statements.

Item 1A. Risk Factors.

There have been no material changes from risk factors previously disclosed in our Annual Report on Form 10-K for the year ended December 31, 2012 in response to Item 1A to Part I of Form 10-K.

Item 4. Mine Safety Disclosures.

Information concerning mine safety and other regulatory matters required by Section 1503(a) of the Dodd-Frank Wall Street Reform and Consumer Protection Act and Item 104 of Regulation S-K is included in Exhibit 95 to this quarterly report.

Item 6.	Exhibits.	
Exhibit <u>No.</u> 10.1	_	Description \$2.25 Billion Second Amended and Restated Revolving Credit Agreement dated as of July 26, 2013 by and among Waste Management, Inc. and Waste Management Holdings, Inc. and certain banks party thereto, Bank of America, N.A., as Administrative Agent, JPMorgan Chase Bank, N.A. and Barclays Bank PLC, as syndication agents, BNP Paribas, Citibank, N.A., Deutsche Bank AG New York Branch, The Bank of Tokyo-Mistubishi UFJ, Ltd., The Royal Bank of Scotland plc, U.S. Bank National Association and Wells Fargo Bank, National Association, as co-documentation agents and J.P. Morgan Sccurities LLC, Merrill Lynch, Pierce, Fenner & Smith Incorporated, and Barclays Bank PLC, as lead arrangers and joint bookrunners. [incorporated by reference to Exhibit 10.1 to Current Report on Form 8-K filed July 30, 2013]
10.2	_	CDN\$650 Million Credit Facilities Credit Agreement by and among Waste Management of Canada Corporation and WM Quebec Inc., as borrowers, Waste Management, Inc. and Waste Management Holdings, Inc., as guarantors, The Bank of Nova Scotia, as administrative agent, JP Morgan Chase Bank, N.A., Bank of America, N.A. and PNC Bank, National Association, as co-syndication agents, the Bank of Nova Scotia, J.P. Morgan Securities LLC, Merrill, Lynch, Pierce, Fenner & Smith Incorporated and PNC Capital Markets LLC, as joint lead arrangers and joint bookrunners and the Lenders from time to time party thereto.
10.3	_	First Amendment Agreement to CDN\$650 Credit Facilities Credit Agreement by and among Waste Management of Canada Corporation and WM Quebec Inc., as borrowers, Waste Management, Inc. and Waste Management Holdings, Inc., as guarantors, the Lenders from time to time party thereto, and The Bank of Nova Scotia, as administrative agent.
31.1	_	Certification Pursuant to Rules 13a-14(a) and 15d-14(a) under the Securities Exchange Act of 1934, as amended, of David P. Steiner, President and Chief Executive Officer.
31.2	_	Certification Pursuant to Rules 13a-14(a) and 15d-14(a) under the Securities Exchange Act of 1934, as amended, of James C. Fish, Jr., Executive Vice President and Chief Financial Officer.
32.1	_	Certification Pursuant to 18 U.S.C. §1350 of David P. Steiner, President and Chief Executive Officer.
32.2	_	Certification Pursuant to 18 U.S.C. \$1350 of James C. Fish, Jr., Executive Vice President and Chief Financial Officer.

Form 10-Q

Exhibit No.			Description
95	-	Mine Safety Disclosures.	
101.INS	—	XBRL Instance Document.	
101.SCH	_	XBRL Taxonomy Extension Schema Document.	
101.CAL	_	XBRL Taxonomy Extension Calculation Linkbase Document.	
101.DEF	_	XBRL Taxonomy Extension Definition Linkbase Document.	
101.LAB	_	XBRL Taxonomy Extension Labels Linkbase Document.	
101.PRE	_	XBRL Taxonomy Extension Presentation Linkbase Document.	

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

WASTE MANAGEMENT, INC.

By: /s/ JAMES C. FISH, JR.

James C. Fish, Jr. Executive Vice President and Chief Financial Officer (Principal Financial Officer)

WASTE MANAGEMENT, INC.

Don P. Carpenter Vice President and Chief Accounting Officer (Principal Accounting Officer)

Date: October 29, 2013

59

EXHIBIT 10.2

WASTE MANAGEMENT OF CANADA CORPORATION WM QUÉBEC INC.

as Borrowers

and

WASTE MANAGEMENT, INC.

WASTE MANAGEMENT HOLDINGS, INC.

as Guarantors

and

THE BANK OF NOVA SCOTIA as Administrative Agent

and

JPMORGAN CHASE BANK, N.A., BANK OF AMERICA, N.A. and PNC BANK, NATIONAL ASSOCIATION as Co-Syndication Agents

THE BANK OF NOVA SCOTIA, J.P. MORGAN SECURITIES LLC, MERRILL LYNCH, PIERCE, FENNER & SMITH INCORPORATED AND PNC CAPITAL MARKETS LLC

as Joint Lead Arrangers and Joint Bookrunners

and

THE LENDERS FROM TIME TO TIME PARTY TO THIS AGREEMENT as Lenders

C\$650,000,000 CREDIT FACILITIES

CREDIT AGREEMENT

DATED AS OF 7 NOVEMBER 2012

BORDEN LADNER GERVAIS LLP

http://markets.on.nytimes.com//research/modules/company_topic/drawFiling.asp?docKey... 11/25/2013

TABLE OF CONTENTS

ARTICLE 1	INTERPRETATION
1.1	Definitions
1.2	Construction
1.3	Certain Rules of Interpretation
1.4	Accounting Terms
1.5	Existing Credit Agreement
ARTICLE 2	THE CREDITS
2.1	Amounts and Availment Options
2.2	Re-borrowing
2.3	Use of the Credits
2.4	Term and Repayment
2.5	Interest Rates, Fees and Commissions
2.6	Standby Fee
2.7	Agency, Assignment and Other Fees
2.8	Exchange Rate Fluctuations
ARTICLE 3	SECURITY
3.1	Security
ARTICLE 4	DISBURSEMENT CONDITIONS
4.1	Conditions Precedent to Initial Advance
4.2	Conditions Precedent to all Advances
ARTICLE 5	REPRESENTATIONS AND WARRANTIES
5.1	Representations and Warranties
5.2	Survival of Representations and Warranties
ARTICLE 6	COVENANTS
6.1	Financial Covenants
6.2	Positive Covenants
6.3	Reporting Requirements
PERIODIC F	INANCIAL REPORTS

6.4 Negative Covenants

(i)

ARTICLE 7 DEFAULT Default

- Acceleration and Termination of Rights
- 7.1 7.2 7.3 Payment of L/Cs and B/As
- 7.4 Remedies
- 7.5 Interest After Stay of Proceedings
- 7.6 Saving Perform Obligations 7.7
- 7.8 7.9 Third Parties Remedies Cumulative
- 7.10 Set-Off or Compensation
- 7.11 Direct Payments

ARTICLE 8 THE AGENT AND THE LENDERS Authorization of Agent

- 8.1 8.2 Rights as a Lender
- Exculpatory Provisions Reliance by Agent 8.3
- 8.4
- Delegation of Duties Administration of the Credits 8.5
- 8.6 8.7 Acknowledgments, Representations and Covenants of Lenders
- 8.8 Defaulting Lenders
- 8.9 Reference Lenders
- 8.10 Provisions Operative Between Lenders and Agent Only

ARTICLE 9 DETAILS REGARDING ADVANCES, PAYMENTS, INTEREST AND FEES 9.1 Lenders' Obligations Relating to L/Cs and Revolving Credit 9.2 Adjustment of Applicable Percentages for Revolving Credit

- 9.3
- Exceptions Regarding the Revolving Credit Evidence of Indebtedness 9.4
- 9.5 Calculation and Other Matters Regarding Interest and Fees
- 9.6 Conversions, Rollovers, Renewals, Repayments and Reductions
- 9.7 Notice of Advances and Payments
- 9.8 Size and Term of Advances Payment of B/As, LIBOR Advances and L/Cs
- 9.9
- 9.10 9.11 Co-ordination of Prime Rate, Base Rate, B/A and LIBOR Advances
- Execution of B/As
- 9.12 Funding of B/As
- 9.13 Other B/A Provisions 9.14
- Issuance and Use of L/Cs Reimbursement Obligation L/Cs 9.15
- 9.16 Failure of Lender to Fund
- Payments by the Borrowers 9.17
- 9.18 Payments by Agent

9.19	Increased Costs, Etc.
9.20	Taxes
9.21	Mitigation Obligations; Replacement of Lenders
9.22	Illegality
9.23	Inability to Determine Rates, Etc.
	ADDITIONAL LENDERS, SUCCESSORS AND ASSIGNS
10.1	Successors and Assigns
10.2	Assignments by Lenders
10.3	Register
10.4	Participations
	MISCELLANEOUS PROVISIONS
11.1	Severability, Etc.
11.2	Amendment, Supplement or Waiver
11.3	Governing Law and Agent for Service
11.4	Due Diligence by Lenders
11.5	Ситтепсу
11.6	Liability of Lenders
11.7	Currency Indemnity
11.8	Expenses and Indemnity
11.9	Address for Notice
11.10	Time of the Essence
11.11	Further Assurances
11.12	Term of Agreement
11.13	Payments on Business Day
11.14	Counterparts and Facsimile
11.15	Waiver of Jury Trial and Consequential Damages
11.16	Whole Agreement
11.17 11.18	This Agreement to Govern Language
11.18	Language Limitation Periods
11.19	Eminator renots Services Provided and Conflicts of Interest
11.20	Date of Agreement
11.21	
SCHEDULE A	A – FORM OF NOTICE OF ADVANCE OR PAYMENT
SCHEDULE I	3 – FORM OF COMPLIANCE CERTIFICATE
SCHEDULE (C – FORM OF ASSIGNMENT AGREEMENT
SCHEDULE I	
SCHEDULE I	E – LENDERS AND THEIR APPLICABLE PERCENTAGES

(iii)

CREDIT AGREEMENT dated as of 7 November 2012.

BETWEEN:

WASTE MANAGEMENT OF CANADA CORPORATION WM QUÉBEC INC.

as Borrowers

- and -

WASTE MANAGEMENT, INC. WASTE MANAGEMENT HOLDINGS, INC. as Guarantors

- and -

THE LENDERS FROM TIME TO TIME PARTY TO THIS AGREEMENT as Lenders

- and -

THE BANK OF NOVA SCOTIA as Administrative Agent

RECITALS:

- A. Certain of the parties to this Agreement are also parties to a credit agreement dated as of November 30, 2005 as amended by an amendment agreement dated as of December 21, 2007 (as amended, the "Existing Credit Agreement").
- B. By commitment letter dated July 23, 2012 and accepted on July 25, 2012, The Bank of Nova Scotia provided certain commitments to the Borrowers in connection with the establishment in favour of the Borrowers of certain new senior credit facilities, namely, a revolving operating credit in the maximum amount of C\$150,000,000 and a non-revolving term credit in the maximum amount of C\$500,000,000 or, in each case, the Equivalent Amount in US Dollars, subject to certain conditions.
- C. The Parties are entering into this Agreement to provide for the terms of the new senior credit facilities.

FOR VALUE RECEIVED, and intending to be legally bound by this Agreement, the Parties agree as follows:

ARTICLE 1 INTERPRETATION

1.1 Definitions

In this Agreement, unless something in the subject matter or context is inconsistent therewith:

- (1) "Accountants" means Ernst & Young LLP or other independent auditors reasonably acceptable to the Lenders.
- (2) "Advance" means an availment of a Credit by the Borrowers by way of Prime Rate Advance, Base Rate Advance, B/A, B/A Equivalent Loan, L/C or LIBOR Advance, including overdrafts under the Swing Line Tranche, deemed advances and conversions, renewals and rollovers of existing Advances. Any reference to the amount of Advances is a reference to the sum of all outstanding Prime Rate Advances, Base Rate Advances, including overdrafts under the Swing Line Tranche, the face amount of all outstanding B/As, the aggregate amount payable at the end of the term of all outstanding B/A Equivalent Loans, the undrawn amount of all outstanding L/Cs and the amount of any Advance for which the Borrowers have failed to provide for payment under Section 9.9.
- (3) "Advance Date" means the date, which shall be a Business Day, of any Advance.
- (4) "Affiliate" means, with respect to a specified Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified.
- (5) "Agent" or "Administrative Agent" means The Bank of Nova Scotia in its capacity as administrative agent for the Lenders, and any successor administrative agent appointed in accordance with this Agreement.
- (6) "Agreement", "hereof", "herein", "herein", "herein", "herein" or similar expressions mean this Agreement, the Recitals hereto and any Schedules hereto, as amended, supplemented, restated and replaced from time to time in accordance with the provisions hereof, and not any particular Article, Section or other portion hereof.
- (7) "Agreement of Purchase and Sale" means the asset purchase agreement dated July 26, 2012 entered into between WMQ, WMI, RCI Environnement Inc., Location P.S.M. Inc., Gestion Environnementale Nord-Sud Inc. and Fiducie Remdev in connection with the Transaction.
- (8) "Applicable Law" means (a) any domestic or foreign statute, law (including common and civil law), treaty, code, ordinance, rule, regulation, restriction or by-law (zoning or otherwise); (b) any judgment, order, writ, injunction, decision, ruling, decree or award; (c) any regulatory policy, practice, request, guideline or directive; or (d) any franchise, licence, qualification, authorization, consent, exemption, waiver, right, permit or other approval of any Governmental Authority, binding on or affecting the Person referred to in the context in which the term is used or binding on or affecting the Property of that Person, in each case whether or not having the force of law.

- 2 -

- (9) "Applicable Percentage" means with respect to any Lender, the percentage of the total Commitments represented by such Lender's Commitment. If the Commitments have terminated or expired, the Applicable Percentages shall be the percentage of the total outstanding Advances represented by such Lender's outstanding Advances. The Applicable Percentage of each Lender as of the date of this Agreement is the percentage calculated based on the amounts set out in Schedule E to this Agreement, which shall be amended and distributed to all parties by the Agent from time to time as Applicable Percentages change in accordance with this Agreement.
- (10) "Assignment Agreement" means an assignment agreement substantially in the form of Schedule C or any other form approved by the Agent.
- (11) "B/A" means a depository bill as defined in the Depository Bills and Notes Act (Canada) in Canadian Dollars that is in the form of an order signed by the Borrowers and accepted by a Lender pursuant to this Agreement or, for Lenders not participating in clearing services contemplated in that Act, a draft or bill of exchange in Canadian Dollars that is drawn by the Borrowers and accepted by a Lender pursuant to this Agreement. For this purpose, orders or drafts that become depository bills, drafts and bills of exchange are sometimes collectively referred to as "orders" in this Agreement.
- (12) "B/A Discount Proceeds" means, in respect of any B/A, the amount that is calculated on the applicable Advance Date in accordance with Section 9.5(5).
- (13) "B/A Discount Rate" means:
 - (a) with respect to any B/A accepted by a Lender named on Schedule I to the Bank Act (Canada), the greater of:
 - (i) the average rate that appears on the Reuters screen CDOR page at or about 10:00 a.m. on the applicable Advance Date, for bankers' acceptances having an identical maturity date to the maturity date of that B/A; or
 - (ii) the rate determined by the Agent as being the arithmetic average (rounded upward to the nearest multiple of 0.01%) of the discount rates of the Schedule I Reference Lenders, determined in accordance with normal market practice at or about 10:00 a.m. on the applicable Advance Date, for bankers' acceptances having a comparable face amount and identical maturity date to the face amount and maturity date of that B/A; and
 - (b) with respect to any B/A accepted by any other Lender, the rate determined in accordance with item (a) above plus 0.07% per annum.
- (14) **"B/A Equivalent Loan"** is defined in Section 9.12(3).

- 3 -

- (15) "B/A Fee" means the fee payable with respect to a B/A that is calculated in accordance with Section 9.5(4).
- (16) "Balance Sheet Date" means December 31, 2011.

- (a) the average of the annual rates of interest established by each Schedule I Reference Lender as its reference rate for that day for commercial loans made by it in Canada in US Dollars;
- (b) the interest rate per annum equal to 0.50% per annum above (i) the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers as published for the day (or, if the day is not a Business Day, for the first preceding Business Day) by the Federal Reserve Bank of New York or, (ii) if that rate is not published for that day by the Federal Reserve Bank of New York, the average of the quotations for that day for those transactions received by the Agent from three federal funds brokers of recognized standing; and
- (c) LIBOR for a period of 30 days plus 1.00% per annum.
- (18) "Base Rate Advance" means an Advance in US Dollars bearing interest based on the Base Rate, and includes availments that are deemed to be Base Rate Advances under this Agreement.
- (19) "Borrowers" means WMOCC and WMQ, on a joint and several basis, and their respective successors and permitted assigns.
- (20) "Branch of Account" means GWS-Loan Administration & Agency Operations of the Agent located at 720 King Street West, 2nd Floor, Global Wholesale Services, Toronto, Ontario, M5V 2T3, or such other branch or branches as may be designated by the Agent from time to time.
- (21) "Business Day" means a day of the year, except a Saturday or a Sunday, on which:
 - (a) the Agent is open for normal banking business at its principal offices in Toronto, Ontario and New York, U.S.A.; and
 - (b) with respect to notices, determinations, payments or advances relating to LIBOR Advances, the Agent is open for normal banking business at its principal offices in New York, U.S.A., London, England and Toronto, Ontario;

except that, if banks are open in some but not all of these locations on a particular day and the Agent determines that the closing of its offices on that day will not adversely affect completion of relevant transactions in accordance with customary banking market and trading practices, the Agent may, on reasonable notice to the Borrowers and the Lenders, specify that particular day to be a Business Day. Any reference in the Provisions to a "business day" shall be interpreted as referring to a Business Day.

- 4 -

^{(17) &}quot;Base Rate" means, on any day, the greatest of:

- (22) "Canadian Dollar", "Cdn. Dollars", "Cdn. \$" and "\$" mean the lawful money of Canada.
- (23) "Capital Leases" means leases under which a Person is the lessee or obligor and the discounted future rental payment obligations under which are required to be capitalized on the consolidated balance sheet of the lessee or obligor in accordance with GAAP.
- (24) "CDOR Rate" means, on any date, with respect to any B/A, the simple average of the rates shown on the display referred to as the "CDOR Page" (or any display substituted therefor) on Reuters Domestic Money Service (or any successor source from time to time) with respect to the banks and other financial institutions named in such display at or about 10:00 a.m. (Toronto time) on such date for banker's acceptances having an identical maturity date to the maturity date of such B/A, as determined by the Agent, or if such day is not a Business Day, then on the immediately preceding Business Day; provided, however, that if such rates are not available, then the CDOR Rate for any day shall be calculated as the average of the bid rates (rounded upwards to the nearest 1/16th of 1%) quoted by each of the Schedule I Reference Lenders for its own banker's acceptances for the applicable period as of 10:00 a.m. (Toronto time) on such day, as determined by the Agent, or if such day is not a Business Day, then on the immediately preceding Business Day.
- (25) "CERCLA" has the meaning defined in Section 5.1(15).
- (26) "Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any Applicable Law, (b) any change in any Applicable Law or in the administration, interpretation or application thereof by any Governmental Authority, or (c) the making or issuance of any Applicable Law by any Governmental Authority (it being acknowledged that (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (y) all requests, rules, guidelines or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law").
- (27) "Closing Date" means 7 November 2012 or such other day as may be agreed to by the parties.
- (28) "Collateral" means cash, a bank draft or a letter of credit issued by a Canadian chartered bank, all in a form satisfactory to the Agent, acting reasonably.
- (29) "Commitment" means, for each Lender from time to time, the agreement to make Advances to the Borrowers and to purchase participations in Advances in accordance with Sections 9.1(1) and 9.1(2) in the Lender's Applicable Percentage of the maximum amount of each Credit and, where the context requires, the maximum amount of Advances that the Lender has agreed to make. Each Lender's Commitment may change from time to time or be cancelled in accordance with this Agreement.

- 5 -

- (30) "Compliance Certificate" means a certificate in the form of Schedule B signed by a senior officer of each of the Borrowers and WMI.
- (31) "Consolidated Tangible Assets" means all assets of WMI determined on a consolidated basis in accordance with GAAP less the sum of:
 - (a) the total book value of all assets of WMI on a consolidated basis properly classified as intangible assets under GAAP, including such items as goodwill, the purchase price of acquired assets in excess of the fair market value thereof, trademarks, trade names, service marks, customer lists, brand names, copyrights, patents and licenses, and rights with respect to the foregoing; plus
 - (b) all amounts representing any write-up in the book value of any assets of WMI on a consolidated basis resulting from a revaluation thereof subsequent to December 31, 2010.
- (32) "Consolidated Total Interest Expense" means, for any period, the aggregate amount of interest expense required by GAAP to be paid or (without duplication) accrued during such period on all Debt of WMI on a consolidated basis outstanding during all or any part of such period, including capitalized interest expense for such period, the amortization of debt discounts and the amortization of fees payable in connection with the incurrence of Debt.
- (33) "Consolidated Net Income (or Deficit)" means the consolidated net income (or deficit) of WMI, after deduction of all expenses, taxes, and other proper charges, determined in accordance with GAAP.
- (34) "Constating Documents" means, with respect to any Person, its articles or certificate of incorporation, amendment, amalgamation, continuance or association, memorandum of association, by-laws, declaration of trust, trust indenture, partnership agreement, limited liability company agreement or other similar document, as applicable, and all unanimous shareholder agreements, other shareholder agreements, voting trust agreements and similar arrangements applicable to the Person's capital stock, all as amended, supplemented, restated or replaced from time to time.
- (35) "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "Controlleg" and "Controlleg" have corresponding meanings.
- (36) "Credits" means the Revolving Credit and the Term Credit, and "Credit" means either of them.

- 6 -

(37) "Debt" means collectively, without duplication, whether classified as indebtedness, an investment or otherwise on the Obligor's balance sheet, (a) all indebtedness for borrowed money (including the face amount of all bankers' acceptances), (b) all obligations for the deferred purchase price of property or services (other than trade payables incurred in the ordinary course of business which either (i) are not overdue by more than 90 days, or (ii) are being disputed in good faith and for which adequate reserves have been established in accordance with GAAP), (c) all obligations evidenced by notes, bonds, debentures or other similar debt instruments, (d) all obligations created or arising under any conditional sale or other title retention agreement with respect to property acquired (even though the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such property), (e) all obligations, liabilities and indebtedness under Capital Leases, (f) all obligations, liabilities or indebtedness arising from the making of a drawing under surety, performance bonds, or any other bonding arrangement, (g) Guarantees of any Debt for others referred to in clauses (a) through (f) above, and (h) all Debt of others referred to in clauses (a) through (f) above, such (b) are secured or supported by (or for which the holder of such Debt has an existing right, contingent or otherwise, to be secured or supported by any Encumbrance on the property or assets of any Obligor or any Subsidiary, even though the owner of the property has not assumed or become liable, contractually or otherwise, for the payment of such Debt, provided that if a Permitted Receivables Transaction is outstanding and is accounted for as a sale of accounts receivable under generally accepted accounting principles, Debt shall also include the additional Debt, determined on a consolidated basis, which would have been outstanding had such Permitted Receivables Transaction been accounted fo

(38) **"Defaulting Lender**" means any Lender, as determined by the Agent, that:

- (a) has failed to fully fund its share of any Advance or fulfill its obligations under Section 9.1 within three Business Days of the date it is required to do so under this Agreement unless such Lender notifies the Agent and the Borrowers in writing that such failure is the result of such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied;
- (b) has notified the Borrowers, the Agent or any other Lender in writing that it does not intend to comply with any of its funding obligations under this Agreement (including Section 9.1), has issued financial statements containing a "going concern" or similar qualification or indicating a potential inability to comply with funding obligations, or has made a public statement to the effect that it does not intend or is unable to comply with its funding obligations under this Agreement or under other agreements in which it commits to extend credit unless such written notice relates to such Lender's obligation to fund an Advance hereunder and states that such position is based on such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing or public statement) cannot been satisfied;

- 7 -

- (c) has failed, within three Business Days after request by the Agent, to confirm that it will comply with its funding obligations under this Agreement (including Section 9.1);
- (d) has otherwise failed to pay over to the Agent or any other Lender any other amount required to be paid by it under this Agreement within three Business Days of the date when due, unless payment is the subject of a good faith dispute;
- (e) has become or is insolvent, is deemed to be insolvent, or is Controlled by a Person that has become or is insolvent or deemed to be insolvent; or
- (f) has itself or is Controlled by a Person that has (i) become the subject of a bankruptcy or insolvency proceeding, (ii) had a receiver, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or custodian, appointed for it, or (iii) taken any action in furtherance of, or indicating its consent to, approval of or acquiescence in any such proceeding or appointment.
- (39) "Designated Account" means, in respect of any Advance, the account or accounts maintained by the Borrowers at the Agent's West Metro Commercial Banking Centre, 2 Robert Speck Parkway, Mississauga, Ontario L4Z 1H8 that the Borrowers designate in their notice requesting an Advance.
- (40) "Disclosure Documents" means the financial statements referred to in Section 5.1(4)(a), and filings made by any Obligor with the Securities and Exchange Commission that were publicly available prior to the date of this Agreement.
- (41) "Distribution" means the declaration or payment of any dividend or other return on equity on or in respect of any shares of any class of capital stock, any partnership interests or any membership interests of any Person (other than dividends or other such returns payable solely in shares of capital stock, partnership interests or membership units of such Person, as the case may be); the purchase, redemption, or other returned any shares of any class of capital stock, partnership interests or membership units of such Person, directly or indirectly through a Subsidiary or otherwise; the return of equity capital by any Person to its shareholders, partners or members as such; or any other distribution on or in respect of any shares of any class of capital stock, partners as such; or any other distribution on or in respect of any shares of any class of capital stock, partners or membership units of such Person.
- (42) "Drawdown Date" means the date, which shall be a Business Day, of any Advance.
- (43) "EBIT" means, for any period, the Consolidated Net Income (or Deficit) of WMI on a consolidated basis plus, without duplication, the sum of (a) interest expense, (b) equity in losses (earnings) of unconsolidated entities, (c) income taxes, (d) non-cash write-downs or write-offs of assets, including non-cash losses on the sale of assets outside the ordinary course of business and (e) EBIT of the businesses acquired by WMI or any of its Subsidiaries (through asset purchases or otherwise) (each an "Acquired Business") or the Subsidiaries acquired or formed since the beginning of such period (each a "New Subsidiary") provided that a statement identifying all such Acquired Businesses and the

- 8 -

EBIT of such Acquired Businesses is delivered to the Lenders with the Compliance Certificate for such period, all to the extent that each of items (a) through (d) was deducted in determining Consolidated Net Income (or Deficit) in the relevant period, minus non-cash extraordinary gains on the sale of assets outside the ordinary course of business to the extent included in Consolidated Net Income (or Deficit).

- (44) "EBITDA" means, for a period, EBIT plus (a) depreciation expense, and (b) amortization expense to the extent the same would be included in the calculation of Consolidated Net Income (or Deficit) for such period, determined in accordance with GAAP.
- (45) "Eligible Assignee" means any Person (other than a natural person, any Obligor or any Affiliate of an Obligor), in respect of which any consent that is required by Section 10.2(1) has been obtained.
- (46) "Employee Benefit Plan" means any employee benefit plan within the meaning of \$3(3) of ERISA maintained or contributed by WMI, any of its Subsidiaries, or any ERISA Affiliate, other than a Multiemployer Plan.
- (47) "Encumbrance" means, with respect to any asset, (a) any mortgage, deed of trust, lien (statutory or otherwise), pledge, hypothecation, encumbrance, charge, security interest, assignment, deposit arrangement or other restriction in, on or of such asset, (b) the interest of a vendor or a lessor under any conditional sale agreement, capital lease or title retention agreement (or any financing lease having substantially the same economic effect as any of the foregoing) relating to such asset, and (c) in the case of securities, any purchase option, call or similar right of a third party with respect to such securities.
- (48) "Environmental Laws" has the meaning defined in Section 5.1(15)(a).
- (49) "Equivalent Amount" means, with respect to an amount in one currency, the amount in another currency that could be purchased by the amount in the first currency determined by reference to the Exchange Rate at the time of determination.
- (50) "ERISA" means the Employee Retirement Income Security Act of 1974, as amended and in effect from time to time.
- (51) **"ERISA Affiliate"** means any Person which is treated as a single employer, member of a controlled group, or under common control with WMI or any of its Subsidiaries under \$412, \$412, or \$430 of the Code.
- (52) "ERISA Reportable Event" means a reportable event within the meaning of §4043 of ERISA and the regulations promulgated thereunder with respect to a Guaranteed Pension Plan as to which the requirement of notice has not been waived.
- (53) "Event of Default" means any of the events or circumstances described in Section 7.1.
- (54) "Exchange Rate" means on any day, for the purpose of calculations under this Agreement, the amount of Canadian Dollars into which another currency may be converted, or vice versa, using the Bank of Canada noon spot rate for converting the one

- 9 -
currency into the other on that day or if that day is not a Business Day, the preceding Business Day, or if such rate is not so published by the Bank of Canada for any such day, then at the mid rate (i.e. the average of the Agent's spot buying and selling rates) quoted by the Agent at the Branch of Account at approximately noon (Toronto time) on that day in accordance with its normal practice for the applicable currency conversion in the wholesale market, or if that day is not a Business Day, the preceding Business Day.

- (55) "Excluded Taxes" means any income or capital Tax now or hereafter imposed, levied, collected, withheld or assessed on a Lender by any applicable Governmental Authority in Canada or any other jurisdiction in which that Lender is subject to Tax as a result of the Lender: (a) having a permanent establishment in such jurisdiction, (b) being organized under the laws of such jurisdiction, or (d) not dealing at arm's length with an Obligor or any other; build be not in such jurisdiction, or (d) not dealing at arm's length with an Obligor or any other; but does not include any sales, goods or services Tax payable under the laws of any such jurisdiction with respect to any goods or services made available by a Lender to the Borrowers under this Agreement or any withholding tax.
- (56) "Existing Credit Agreement" is defined in the first recital to this Agreement.
- (57) "FASB ASC" means the Accounting Standards Codification of the Financial Accounting Standards Board
- (58) "Fee Letter" means the confidential fee letter dated July 23, 2012 from The Bank of Nova Scotia to the Borrowers and accepted by the Borrowers on July 25, 2012.
- (59) "GAAP" " means, when used in this Agreement, whether directly or indirectly through reference to a capitalized term used therein, means (a) principles that are consistent with the principles promulgated or adopted by the Financial Accounting Standards Board (U.S.) and its predecessors, in effect for the fiscal year ended on the December 31, 2010, and (b) to the extent consistent with such principles, the accounting practice of WMI reflected in its financial statements for the year ended on December 31, 2010; provided, that, with respect to any financial statements prepared after the December 31, 2010; und (b) to the extent consistent with such principles, the accounting practice of WMI reflected in its financial statements for the year ended on December 31, 2010; provided, that, with respect to any financial statements prepared after the December 31, 2010; provided, further, that in each case referred to in this definition of "GAAP" a certified public accountant would, insofar as the use of such accounting principles is pertinent, be in a position to deliver an unqualified opinion (other than a qualification regarding changes in generally accepted accounting principles) as to financial statements in which such principles have been properly applied. Notwithstanding the foregoing, for purposes of determining compliance with any covenant (including the computation of any financial covenant) contained herein, "Debt" of WMI and its subsidiaries shall be deemed to be carried at 100% of the outstanding principal amount thereof, and the effects of FASB ASC 825 and FASB ASC 470-20 on financial liabilities shall be disregarded.
- (60) "Governmental Authority" means the government of Canada or any other nation, or of any political subdivision thereof, whether provincial, state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising

- 10 -

executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government, including any supra-national bodies (such as the European Union or the European Central Bank) and any group or body charged with setting financial accounting or regulatory capital rules or standards (including, without limitation, the Financial Accounting Standards Board, the Bank for International Settlements or the Basel Committee on Banking Supervision or any successor or similar authority to any of the foregoing).

- (61) "Guarantee" means any obligation, contingent or otherwise, of a Person guaranteeing or having the economic effect of guaranteeing any Debt or other obligation of any other Person (the "primary obligor") in any manner, whether directly or indirectly, and including any obligation of the guarantor, direct or indirect, (a) to purchase or pay (or advance or supply funds for the purchase or payment of) such Debt or other obligation or to purchase (or to advance or supply funds for the purchase of) any security for the payment thereof, (b) to purchase or lease property, securities or services for the purpose of assuring the owner of such Debt or other obligation of the payment thereof, (c) to maintain working capital, equity capital or any other financial statement condition or liquidity of the primary obligor to pay such Debt or other obligation, or (d) as an account party in respect of any letter of credit or letter of guarantee shall not include endorsements for collection or deposit in the ordinary course of business.
- (62) "Guaranteed Pension Plan" means any employee pension benefit plan within the meaning of \$3(2) of ERISA maintained or contributed to by WMI, its Subsidiaries or any ERISA Affiliate (or pursuant to which any such Person accrued an obligation to make contributions at any time during the preceding five plan years) the benefits of which are guaranteed on termination in full or in part by the PBGC pursuant to Title IV of ERISA, other than a Multiemployer Plan.
- (63) "Guarantors" means each of Waste Management, Inc., a Delaware corporation, and Waste Management Holdings, Inc., a Delaware corporation, and each other Person which delivers a guarantee hereunder, and becomes a party hereto, from time to time.
- (64) "Hazardous Substances" has the meaning defined in Section 5.1(15)(b).
- (65) "Indemnified Taxes" means Taxes other than Excluded Taxes.
- (66) "Interbank Reference Rate" means, in respect of any currency, the interest rate expressed as a percentage per annum that is determined by the Agent at any time in accordance with banking industry rules on interbank compensation for use when calculating interest due by it or owing to it arising from correction of errors in transactions in that currency between it and other banks.
- (67) "Interest Payment Date" means the 21st day of each calendar month.
- (68) "Interest Period" has the same meaning as LIBOR Period.

- 11 -

(69) "Interim Balance Sheet Date" means September 30, 2012.

- (70) "Investment" means all expenditures made by a Person and all liabilities incurred (contingently or otherwise) by a Person for the acquisition of stock of (other than the stock of Subsidiaries), or Debt of, or for loans, advances, capital contributions or transfers of property to, or in respect of any Guarantees or other commitments as described under Debt, or obligations of, any other Person, including without limitation, the funding of any captive insurance company (other than loans, advances, capital contributions or transfers of property to any Subsidiary or variable interest entities consolidated in accordance with FASB ASC 810, or Guarantees with respect to Debt of any Subsidiary or variable interest entities consolidated in accordance with FASB ASC 810). In determining the aggregate amount of Investments outstanding at any particular time: (a) the amount of any Investment represented by a Guarantee shall be taken at not less than the principal amount of the obligations guaranteed and still outstanding, (b) there shall be included as an Investment all interest accrued with respect to Debt constituting an Investment unless and until such interest is paid, (c) there shall be deducted in respect of each such Investment any amount received as a return of capital (but only by partial or full repurchase, redemption, retirement, repayment, liquidating dividend or liquidating distribution), (d) there shall not be deducted in respect of any Investment any amounts received as earnings on such Investment, whether as dividends, interest or otherwise, except that accrued interest included as provided in the foregoing clause (b) may be deducted when paid; and (e) there shall not be deducted from the aggregate amount of Investments any decrease in the value thereof.
- (71) "Issuing Bank" means The Bank of Nova Scotia as the issuer of Letters of Credit on the basis that it is "fronting" for the other Lenders and not on the basis that it is the attorney of the other Lenders to sign Letters of Credit on their behalf. If The Bank of Nova Scotia resigns as Issuing Bank, a successor shall be agreed to by the Borrower and the Required Lenders, acting reasonably, and the new Issuing Bank.
- (72) "L/C" or "Letter of Credit" means a standby letter of credit, letter of guarantee or commercial letter of credit in a form satisfactory to the Issuing Bank, issued by the Issuing Bank at the request of the Borrowers in favour of a third Person to secure the payment or performance of an obligation to the third Person.
- (73) "L/C Fee" means the fee payable with respect to an L/C that is calculated in accordance with Section 9.5(6).
- (74) "Lenders" means each of the Persons listed on Schedule E and other lenders that from time to time become Lenders in accordance with the terms of this Agreement and "Lender" means any one of the Lenders. However, references to the Lenders in the context of the Agent or others holding Security or other rights or documents for the benefit or on behalf of the Lenders shall be interpreted as including other Persons to whom Other Guaranteed Obligations are owed.

- 12 -

- (75) "LIBOR" means, for any LIBOR Period and LIBOR Advance, the greatest of:
 - (a) the rate expressed as a percentage per annum for deposits in US Dollars in the London interbank market for a period equal to the LIBOR Period and in an amount approximately equal to the amount of the LIBOR Advance, that appears on the Reuters LIBOR 01 Page (or any successor source from time to time) as of 11:00 a.m. (London time) two Business Days before the first day of the LIBOR Period; and
 - (b) the interest rate expressed as a percentage per annum determined by the Agent to be its cost of funds in the London interbank market for US Dollars for a period equal to the LIBOR Period and in an amount approximately equal to the amount of the LIBOR Advance, at 11:00 a.m. (London time) two Business Days before the first day of the LIBOR Period.
- (76) "LIBOR Advance" or "LIBO Rate Loan" means an Advance in US Dollars bearing interest based on LIBOR.
- (77) "LIBOR Period" means the period selected by the Borrowers for a LIBOR Advance.
- (78) "Loan Documents" means this Agreement, the Security, the Fee Letter, and all other documents relating to the Credits, or either of them.
- (79) "Material Adverse Effect" means a material adverse effect on (a) the business, operations, or financial condition of the Obligors, taken as a whole, (b) the ability of either Guarantor to perform its obligations under any Loan Document to which it is a party, or (c) the rights of, or remedies or benefits available to, the Agent or any of the Lenders under any Loan Document.
- (80) "Maturity Date" means 7 November 2017.
- (81) "Moody's" means Moody's Investors Service, or any successor to it.
- (82) "Multiemployer Plan" means any multiemployer plan within the meaning of §3(37) of ERISA maintained or contributed to by WMI, any of its Subsidiaries, or any ERISA Affiliate (or pursuant to which any such Person accrued an obligation to make contributions at any time during the preceding five plan years).
- (83) "Non-B/A Lender" is defined in Section 9.12(3).
- (84) "Obligations" means all debts, liabilities and obligations of the Borrowers to the Lenders under or in connection with this Agreement, whether present or future, direct or indirect, absolute or contingent, matured or not, at any time owing or remaining unpaid by the Borrowers to the Lenders in any currency under or in connection with this Agreement, whether arising from dealings between the Lenders and the Borrowers or from other dealings or proceedings by which the Lenders may be or become in any manner whatever creditors of the Borrowers under or in connection with this Agreement, and whether incurred by the Borrowers alone or with another or others and

- 13 -

whether as principal or surety, and all interest, fees, commissions and legal and other costs, charges and expenses owing or remaining unpaid by the Borrowers to the Lenders in any currency under or in connection with this Agreement. In this definition, "Lenders" means the Lenders, or any of them. For certainty, the Other Guaranteed Obligations are not Obligations.

- (85) "Obligors" means the Borrowers and each of the Guarantors, and "Obligor" means any of them.
- (86) "Other Guaranteed Obligations" means the present and future debts, liabilities and obligations of the Borrowers or either of them to any Lender or its affiliates that arise under or in connection with, (i) derivatives or other hedging arrangements, and (ii) cash management arrangements, entered into with such Lender or its affiliates that are permitted under this Agreement, which Other Guaranteed Obligations shall rank pari passu with the Credits.
- (87) "Parties" means collectively the Borrowers, the other Obligors, the Lenders and the Agent.
- (88) "PBGC" means the Pension Benefit Guaranty Corporation created by §4002 of ERISA.
- (89) "Pending Event of Default" means an event which would constitute an Event of Default hereunder, except for satisfaction of any requirement for giving of notice, lapse of time, or both, or other condition subsequent.
- (90) "Permitted Encumbrances" means, any of the following:
 - Encumbrances for taxes not due or that are being contested in good faith by appropriate proceedings and for which adequate reserves with respect thereto have been set aside as required by GAAP;
 - (b) carriers', warehousemen's, maritime, mechanics', materialmen's, repairmen's other like Encumbrances arising in the ordinary course of business that are being contested in good faith by appropriate proceedings and for which adequate reserves with respect thereto have been set aside as required by GAAP;
 - (c) pledges and deposits made in the ordinary course of business in compliance with workmen's compensation, unemployment insurance and other social security laws or regulations;
 - (d) Encumbrances to secure the performance of bids, trade contracts (other than for Debt), leases (other than Capital Leases), statutory obligations, surety and appeal bonds, suretyship, performance and landfill closure bonds and other obligations of a like nature incurred in the ordinary course of business;
 - (e) zoning restrictions, easements, rights-of-way, restrictions on use of property and other similar encumbrances incurred in the ordinary course of business which, in the aggregate, are not substantial in amount and do not materially detract from the value of the property subject thereto or interfere with the ordinary conduct of the business of WMI or any of its Subsidiaries;

- 14 -

- (f) the Encumbrances on Schedule 1.1 of the U.S. Credit Agreement (a copy of which is attached hereto as Annex 1.1) securing the obligations listed on such Schedule and any replacement Encumbrance securing any renewal, extension or refunding of such obligations if the amount secured by such renewal, extension or refunding Encumbrance shall not exceed the amount of the outstanding obligations secured by the Encumbrance being replaced at the time of such renewal, extension or refunding premiums and fees, related to such renewal, extension or refunding) and if such replacement Encumbrance shall be limited to substantially the same property that secured the Encumbrance so replaced;
- (g) legal or equitable encumbrances deemed to exist by reason of the existence of any litigation or other legal proceeding or arising out of a judgment or award with respect to which an appeal is being prosecuted in good faith by appropriate action and with respect to which adequate reserves are being maintained and, in the case of judgment liens, execution thereon is stayed;
- (h) rights reserved or vested in any municipality or governmental, statutory or public authority to control or regulate any property of WMI or any Subsidiary, or to use such property in a manner that does not materially impair the use of such property for the purposes for which it is held by WMI or such Subsidiary;
- (i) any obligations or duties affecting the property of WMI or any of its Subsidiaries to any municipality, governmental, statutory or public authority with respect to any franchise, grant, license or permit;
- (j) Encumbrances filed in connection with sales of receivables by any of the Subsidiaries of WMI (other than Waste Management Holdings, Inc.) to a wholly-owned special purpose financing Subsidiary for purposes of perfecting such sales, provided that no third party has any rights with respect to such Encumbrances or any assets subject thereto;
- (k) any interest or title of a lessor under any sale-leaseback transaction entered into by WMI or any Subsidiary conveying only the assets so leased back to the extent the related Debt is permitted under Section 6.4(1);
- (I) Encumbrances created or deemed to be created under Permitted Receivables Transactions at any time provided such Encumbrances do not extend to any property or assets other then the trade receivables sold pursuant to such Permitted Receivables Transactions, interests in the goods or products (including returned goods and products), if any, relating to the sales giving rise to such trade receivables; any security interests or other Encumbrances and property subject thereto (other than on any leases or related lease payment rights or receivables between WMI and any of its Subsidiaries, as lessors or sublessors) from time to

- 15 -

- time purporting to secure the payment by the obligors of such trade receivables (together with any financing statements authorized by such obligors describing the collateral securing such trade receivables) pursuant to such Permitted Receivables Transactions; and
- (m) Encumbrances securing other Debt, provided that the aggregate amount of all liabilities, including any Debt, of WMI and its Subsidiaries secured by all Encumbrances permitted in subsections (k), (l) and (m), when added (without duplication) to the aggregate amount of Debt of WMI's Subsidiaries permitted under Section 6.4(1)(b) and Debt with respect to Permitted Receivables Transactions, shall not exceed 15% of Consolidated Tangible Assets at any time.
- (91) "Permitted Receivables Transaction" means any sale or sales of, and/or securitization of, any accounts receivable of WMI and/or any of its Subsidiaries (the "Receivables") pursuant to which (a) WMI and its Subsidiaries realize aggregate net proceeds of not more than \$750,000,000 at any one time outstanding, including, without limitation, any revolving purchase(s) of Receivables where the maximum aggregate uncollected purchase price (exclusive of any deferred purchase price) for such Receivables at any time outstanding does not exceed \$750,000,000, and (b) which Receivables shall not be discounted more 25%.
- (92) "Person" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.
- (93) "Prime Rate" means, on any day, the greater of:
 - (a) the average of the annual rates of interest established by each of the Schedule I Reference Lenders as its reference rate for that day for commercial loans made by it in Canada in Canadian Dollars; and
 - (b) the CDOR Rate for one-month Canadian Dollar bankers' acceptances on that day plus 1.00% per annum.
- (94) "Prime Rate Advance" means an Advance in Canadian Dollars bearing interest based on the Prime Rate and includes availments that are deemed to be Prime Rate Advances under this Agreement.
- (95) "RCRA" has the meaning defined in Section 5.1(15).
- (96) "Real Property" means all real property heretofore, now, or hereafter owned, operated or leased by WMI or any of its Subsidiaries.
- (97) "Register" has the meaning defined in Section 10.3.
- (98) "Release" shall have the meaning specified in CERCLA and the term "Disposal" (or "Disposed") shall have the meaning specified in RCRA and regulations promulgated thereunder provided that in the event either CERCLA or RCRA is amended so as to broaden the meaning of any term defined thereby, such broader meaning shall apply as of

- 16 -

the effective date of such amendment and provided further, to the extent that the laws of Canada or a state, province, territory or other political subdivision thereof wherein the property lies establish a meaning for "Release" or "Disposal" which is broader than specified in either CERCLA, or RCRA, such broader meaning shall apply to WMI's or any of its Subsidiaries' activities in that state, province, territory or political subdivision.

- (99) "Relevant Rating" means, as of any date of determination, the ratings as determined by S&P and Moody's of WMI's non-credit enhanced, senior unsecured long-term debt and in circumstances when the ratings are not the same level (in the grid set forth in Section 2.5(1)), then the higher of the two ratings shall apply, provided however that if the higher rating is more than one level higher than the lower rating, the Relevant Rating shall be set at one level below the higher rating.
- (100) "Required Lenders" means a Lender or Lenders holding, in the aggregate, a minimum of 50.1% of the amount of the Commitments (or the outstanding Advances if the Commitments have terminated including after the occurrence of any Default), excluding in all cases Commitments or Advances held by any Obligor or any Affiliate or Related Party of any Obligor.
- (101) "Revolving Credit" means the revolving credit of up to C\$150,000,000 or the Equivalent Amount in US Dollars in favour of the Borrowers that is established pursuant to Section 2.1(1), unless the context otherwise requires, includes the Swing Line Tranche.
- (102) "Revolving Lenders" means the Lenders who have provided Commitments relating to the Revolving Credit as specified in Schedule E.
- (103) "S&P" means Standard & Poor's, a division of The McGraw-Hill Companies, Inc., or any successor to it.
- (104) "Schedule I Reference Lenders" means Lenders that are banks named on Schedule I of the Bank Act (Canada) and that have been designated as or deemed to be Schedule I Reference Lenders pursuant to Section 8.9.
- (105) "Security" means the guarantees held from time to time by or on behalf of the Agent and the Lenders supporting or intended to support, *inter alia*, repayment of any of the Obligations, including, without limitation, the Existing Guarantees and the New Guarantees.
- (106) "Standby Fee" has the meaning defined in Section 2.5(5).
- (107) "Subsidiary" means any Person of which the designated parent shall at any time own directly or indirectly through one or more subsidiaries at least a majority of the outstanding capital stock or other interests entitled to vote generally and whose financial results are required to be consolidated with the financial results of the designated parent in accordance with GAAP.
- (108) "Swap Contract" means all obligations in respect of interest rate, currency or commodity exchange, forward, swap, or futures contracts or similar transactions or arrangements entered into to protect or hedge any of the Obligors against interest rate, exchange rate or commodity price risks or exposure, or to lower or diversify their funding costs.

- 17 -

- (109) "Swing Line Lender" means the Lender that, subject to Section 9.1(2), makes Advances under the Swing Line Tranche. As of the date of this Agreement, the Swing Line Lender is The Bank of Nova Scotia. The Borrowers may from time to time with the consent of the Agent and the proposed replacement Lender, designate another Revolving Lender to replace The Bank of Nova Scotia as the Swing Line Lender, in which case the Parties shall make such arrangements as are necessary to ensure that all amounts owing under or in connection with the Swing Line Tranche to the outgoing Swing Line Lender are promptly paid in full.
- (110) "Swing Line Tranche" means the carveout of the Revolving Credit under which overdrafts are available as described in Section 2.1(1).
- (11) **"Taxes"** means all taxes, levies, imposts, stamp taxes, duties, deductions, withholdings and similar governmental impositions payable, levied, collected, withheld or assessed as of the date of this Agreement or at any time in the future and all interest, charges and penalties in respect thereof, and **"Tax"** shall have a corresponding meaning.
- (112) "Term Credit" means the credit of up to C\$500,000,000 or the Equivalent Amount in US Dollars in favour of the Borrowers that is established pursuant to Section 2.1(2).
- (113) "Term Lenders" means the Lenders who have provided Commitments relating to the Term Credit as specified in Schedule E.
- (114) "Total Debt" means the sum, without duplication, of all (1) Debt of WMI on a consolidated basis, under subsections (a) through (h) of the definition of "Debt" (provided, however, that Debt with respect to Permitted Receivables Transactions shall not be included in such calculation), plus (2) non-contingent reimbursement obligations of WMI and its Subsidiaries with respect to drawings under any L/Cs.
- (115) "Transaction" means the acquisition of the assets of RCI Environnement Inc., Location P.S.M. Inc. and Gestion Environnementale Nord-Sud Inc. by WM Québec as described more particularly in the Agreement of Purchase and Sale.
- (116) **"U.S. Credit Agreement**" means the U.S. \$2,000,000,000 amended and restated revolving credit agreement dated as of May 9, 2011 by and among WMI, as borrower, Waste Management Holdings, Inc., as guarantor, various banks party thereto from time to time, as lenders, Bank of America, N.A., as administrative agent and others.
- (117) "US Dollars" and "U.S. \$" means lawful monies of the United States of America.
- (118) "WMI" means Waste Management, Inc., a Delaware corporation.
- (119) "WMOCC" means Waste Management of Canada Corporation, a Nova Scotia unlimited liability company.

- 18 -

(120) "WMQ" means WM Québec Inc., a corporation incorporated under the laws of Canada.

1.2 Construction

This Agreement has been negotiated by each party with the benefit of legal representation and any rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not apply to the construction or interpretation of this Agreement.

1.3 Certain Rules of Interpretation

In this Agreement:

- (a) the division into sections and other subdivisions thereof and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Agreement; and
- (b) unless otherwise specified or the context otherwise requires:
 - (i) references to any Section or Schedule are references to the Section of, or Schedule to, this Agreement;
 - (ii) "including" or "includes" means "including (or includes) but not limited to" and shall not be construed to limit any general statement preceding it to the specific or similar items or matters immediately following it;
 - (iii) references to contracts, agreements or instruments, unless otherwise specified, are deemed to include all present and future amendments, supplements, restatements or replacements to or of such contracts, agreements or instruments, provided that such amendments, supplements, restatements or replacements to or of such contracts, agreements or instruments have been, if applicable, approved or consented to and otherwise made in accordance with the provisions of this Agreement;
 - (iv) references to any legislation, statutory instrument or regulation or a section or other provision thereof, unless otherwise specified, is a reference to the legislation, statutory instrument, regulation, section or other provision as amended, restated or re-enacted from time to time;
 - (v) references to any thing includes the whole or any part of that thing and a reference to a group of things or Persons includes each thing or Person in that group;
 - (vi) references to a Person includes that Person's successors and assigns;
 - (vii) all references to specific times are references to Toronto time; and
 - (viii) words in the singular include the plural and vice-versa and words in one gender include all genders.

1.4 Accounting Terms

In this Agreement, unless specified otherwise, each accounting term has the meaning assigned to it under GAAP.

1.5 Existing Credit Agreement

The Credits are in addition to the credit provided to WMOCC under the Existing Credit Agreement which remains in full, force and effect, unamended by this Agreement.

ARTICLE 2 THE CREDITS

2.1 Amounts and Availment Options

- (1) Upon and subject to the terms and conditions of this Agreement, the Revolving Lenders severally (not jointly and not jointly and severally) agree to provide to the Borrowers a credit facility referred to as the Revolving Credit for the use of the Borrowers in the aggregate amount of up to \$150,000,000 (provided that each Revolving Lender's obligation hereunder shall be limited to its respective Applicable Percentage of the Revolving Credit).
- (2) Subject to the terms and conditions of this Agreement, the Term Lenders severally (not jointly and not jointly and severally) agree to provide to the Borrowers a credit facility referred to as the Term Credit for the use of the Borrowers in the aggregate amount of up to \$500,000,000 or the Equivalent Amount in US Dollars (provided that each Term Lender's obligation shall be limited to its respective Applicable Percentage of the Term Credit).
- (3) The principal amount of the Credits may be increased by up to an aggregate of \$100,000,000 to be allocated between the Credits as agreed between the Borrowers and the Agent, on one or more occasions if (a) one or more existing Lenders agree with the Borrowers in their sole discretion to increase their Commitments and confirm their agreement in a form satisfactory to the Agent or (b) one or more other financial institutions identified by the Borrowers and/or the Agent and acceptable to the Agent, the Issuing Lender and the Swing Line Lender, provide Commitments in a form satisfactory to the Agent. In that case, the Agent shall notify the Lenders and the Obligors of the increased amount of each of the Credits and shall provide them with a revised version of Schedule E.

(4) At the option of the Borrowers, the Revolving Credit (other than the Swing Line Tranche) may be used by:

- (a) requesting the Revolving Lenders to make Prime Rate Advances, Base Rate Advances and/or LIBOR Advances;
- (b) presenting orders to the Revolving Lenders for acceptance as B/As and/or requesting the Revolving Lenders to make B/A Equivalent Loans; and/or

- 20 -

- (c) requesting that L/Cs in Canadian Dollars or US Dollars be issued by the Issuing Bank on behalf of all the Revolving Lenders except that the aggregate face amount of L/Cs outstanding under the Revolving Credit at any time shall not exceed \$50,000,000, measured at the time of issuance or renewal of any L/C.
- (5) At the option of the Borrowers, they may obtain Advances in an aggregate amount of up to \$20,000,000 or the Equivalent Amount in US Dollars by incurring overdrafts in their Canadian Dollar and US Dollar accounts with the Swing Line Lender, which shall be deemed to be Prime Rate Advances and Base Rate Advances, respectively, by the Swing Line Lender. The Swing Line Lender shall not, however, be required to make any such Advance if doing so would result in the sum of the Advances made by the Swing Line Lender under the Revolving Credit (other than the Swing Line Tranche) plus the Advances made by the Swing Line Lender under the Revolving Credit (without reference to the Swing Line Tranche).
- (6) The Term Credit may be drawn only in a single Advance only up to 7 August 2013, at which time the undrawn amount of the Term Credit shall be permanently cancelled. The Borrowers may cancel the Term Credit at any time following the Closing Date prior to the date which is nine months following the Closing Date upon written notice to the Agent. Other than the Standby Fee (as defined in Section 2.6), no other fee, penalty or other amount shall be owed in connection with cancellation of the Term Credit.
- (7) At the option of the Borrowers, the Term Credit may be used by:
 - (a) requesting the Term Lenders to make Prime Rate Advances, Base Rate Advances and/or LIBOR Advances; and/or
 - (b) presenting orders to the Term Lenders for acceptance as B/As and/or requesting the Term Lenders to make B/A Equivalent Loans.

2.2 Re-borrowing

- (1) The Revolving Credit is a revolving credit and the principal amount of any Advance under the Revolving Credit that is repaid may be re-borrowed, if the Borrowers are otherwise entitled to an Advance under the Revolving Credit.
- (2) The Term Credit is a non-revolving credit and the principal amount of any Advance under the Term Credit that is repaid may not be re-borrowed.

2.3 Use of the Credits

(1) The Revolving Credit may be used to, (a) finance the general corporate requirements of the Borrowers and their Subsidiaries, (b) at the option of the Borrowers, refinance WMOCC's existing indebtedness owing under the Existing Credit Agreement, and (c) to finance the implementation of the Transaction including, without limitation, payment of the purchase price, payment of any ancillary costs and expenses relating thereto and funding of any immediate working capital requirements in connection therewith, subject to the other terms of this Agreement.

- 21 -

(2) The Term Credit may only be used to finance the implementation of the Transaction including, without limitation, payment of the purchase price, payment of any ancillary costs and expenses relating thereto and funding of any immediate working capital requirements in connection therewith.

2.4 Term and Repayment

- (1) The Revolving Credit shall be repaid in full and cancelled on or before the Maturity Date.
- (2) The Term Credit shall be repaid in full and cancelled on or before the Maturity Date.
- (3) Subject to the terms hereof, the Swing Line Lender may, at its option, request that the aggregate outstanding Advances under the Swing Line Tranche be reduced by way of Advances under the Revolving Credit. The Agent, in consultation with the Swing Line Lender, shall notify the other Revolving Lenders of the Advances they are required to make based on their respective Applicable Percentages. On the day of receipt of that notice, each of the other Revolving Lenders shall disburse their respective Advances to the Swing Line Lender, which shall threupon be deemed to be Prime Rate Advances and Base Rate Advances applicable, in each case under the Revolving Credit. No Revolving Lender shall, however, be required to make an Advance under this Section 2.4 (3) that would result in the total Advances made by the Lender under the Revolving Credit ender's Applicable Percentage of the Revolving Credit.
- (4) If at any time the Borrowers request an Advance under the Revolving Credit (other than the Swing Line Tranche) and the Advance would result in the sum of the Advances made by the Swing Line Lender under the Revolving Credit (other than the Swing Line Tranche) plus the Advances made by the Swing Line Lender the Swing Line Tranche) plus the Advances made by the Swing Line Lender the Swing Line Tranche exceed the Swing Line Tranche) plus the Advances made by the Swing Line Lender to mach exceed the Revolving Credit (other than the Swing Line Tranche) plus the Advances made by the Swing Line Lender to apy the proceeds of the requested Advance to the Swing Line Lender to the swing Line Tranche to the extent of the excess, and the Agent shall do so. In addition, and notwithstanding anything to the contrary in this Agreement, (i) if an Event of Default occurs and is continuing, or (ii) if the Swing Line Lender so requires from time to time, and there are then outstanding any Advances under the Swing Line Tranche, effective on the day of notice to that effect to the other Revolving Lenders from the Swing Line Tranche in the currencies in which they were made, including accrued and unpaid interest in respect thereof. On the day of receipt of such notice, each of the other Revolving Lenders shall disburse to the Swing Line Lender its respective Applicable Percentage thereof, which shall thereupon be deemed to be Prime Rate Advances (to the extent that the Advances under the Swing Line Tranche were in Canadian Dollars) and Base Rate Advances (to the extent that the Advances under the Swing Line Tranche were in Canadian Dollars) and Base Rate Advances (to the extent that the Advances under the Swing Line Tranche were in Canadian Dollars) and Base Rate Advances (to the extent that the Advances under the Swing Line Tranche were in Canadian Dollars) and Base Rate Advances (to the extent that

- 22 -

Revolving Lender shall, however, be required to make an Advance under this Section 2.4(4) that would result in the total Advances made by the Lender under the Revolving Credit exceeding the Lender's Applicable Percentage of the Revolving Credit.

2.5 Interest Rates, Fees and Commissions

(1) Interest rates on Prime Rate Advances, Base Rate Advances and LIBOR Advances and the rates for calculation of B/A Fees and L/C Fees shall be determined and adjusted based on the Relevant Rating as follows:

			Prime Rate	
			and Base Rate	B/A Fees/
Relevant Rating	Standby Fee	L/C Fees	Advances	LIBOR Advances
Greater than or equal to A-/A3	0.225%	1.125%	0.125%	1.125%
BBB+/Baa1	0.25%	1.25%	0.25%	1.25%
BBB/Baa2	0.28%	1.40%	0.40%	1.40%
BBB-/Baa3	0.33%	1.65%	0.65%	1.65%
Less than or equal to BB+/ Ba1	0.43%	2.15%	1.15%	2.15%

All figures in the table above represent percent per annum. Each of the amounts specified above, except the Standby Fee, shall be increased by 2% per annum if an Event of Default has occurred and is continuing. If an Event of Default is disclosed by a Compliance Certificate, any increase shall be applied beginning as of the end of the fiscal period to which the Compliance Certificate relates.

- (2) Any increase or decrease in the interest rates and fees resulting from a change in the Relevant Rating shall be effective on the day that the Relevant Rating changes. WMI shall immediately notify the Agent of any change in the Relevant Rating, or in the rating of S&P or Moody's composing the Relevant Rating.
- (3) Interest shall acrue and be payable on Prime Rate Advances and Base Rate Advances at the Prime Rate or the Base Rate, respectively, plus the relevant figure shown under "Prime Rate and Base Rate Advances" in the table in Section 2.5(1). Interest shall acrue and be payable on LIBOR Advances at LIBOR plus the relevant figure shown under "LIBOR Advances" in that table. The rate for calculation of B/A Fees and L/C Fees shall be the relevant figure shown under "B/A Fees" and "L/C Fees", respectively, in that table. In addition, a fronting fee at a rate of 0.25% per annum on the amount of each L/C shall be payable to the Issuing Bank for its own account.

- 23 -

- (4) The Borrowers shall pay interest and fees at the applicable rate specified in Section 2.5(1) to the Agent at the Branch of Account on Advances outstanding from time to time, except that the B/A Fee for any B/A shall be paid by each Lender deducting the B/A Fee from the proceeds of the B/A remitted to the Agent pursuant to Section 9.10(4). The Borrowers shall pay interest on Prime Rate Advances and Base Rate Advances on the 22nd day of each month. The Borrowers shall pay interest on each LIBOR Advance on the last day of the applicable LIBOR Period and, if the LIBOR Period is longer than three months, every three months after the date of the relevant LIBOR Advance. The Borrowers shall pay L/C Fees and the fronting fee applicable to L/Cs quarterly in arrears on the third Business Day after the end of each calendar quarter.
- (5) The Agent shall distribute interest and fees for the Revolving Credit (other than the Swing Line Tranche) to the Revolving Lenders based on their respective Applicable Percentages, except that the fronting fee for L/Cs shall be paid to the Issuing Bank for its own account. Interest for the Swing Line Tranche shall be paid to the Swing Line Lender for its own account. The Agent shall distribute interest and fees for the Term Credit to the Term Lenders based on their respective Applicable Percentages.

2.6 Standby Fee

The Borrowers shall pay a standby fee (the "Standby Fee") on the daily unadvanced portion of the Revolving Credit and, until the date of the first Advance thereunder or its cancellation, the Term Credit, at a rate that shall be adjusted based on the Relevant Rating and that shall be as specified under "Standby Fee" in the table in Section 2.5(1). The Standby Fee shall be payable quarterly in arrears on the third Business Day after the end of each quarter, with the first payment to be made on the third Business Day after the end of the first quarter following the Closing Date. On termination of all Commitments under the Revolving Credit, the Borrowers shall also pay any accrued but unpaid Standby Fees.

2.7 Agency, Assignment and Other Fees

The Borrowers shall, concurrently with the execution of this Agreement, pay the non-refundable upfront and other fees to the Agent in accordance with the Fee Letter. The Borrowers shall also pay agency and other fees to the Agent from time to time in accordance with the Fee Letter.

2.8 Exchange Rate Fluctuations

If fluctuations in rates of exchange in effect between US Dollars and Canadian Dollars cause the amount of Advances (expressed in Canadian Dollars) under a Credit to exceed the maximum amount of that Credit permitted in this Agreement by five (5%) percent or more for a period of five (5) consecutive Business Days, the Borrowers shall promptly pay the Lenders the amount by which the maximum amount of that Credit is exceeded. If, on the date of any Advance under a Credit (whether by rollover, renewal, conversion or otherwise), the amount of Advances (expressed in Canadian Dollars) under that Credit exceeds the maximum amount of that Credit because of fluctuations in rates of exchange, the Borrowers shall promptly pay the Lenders the excess and shall not be entitled to any Advance that would result in the amount of that Credit being exceeded.

- 24 -

ARTICLE 3 SECURITY

3.1 Security

(1) The security shall comprise the, (i) unlimited and unconditional guarantees of the Obligations and the Other Guaranteed Obligations in favour of the Agent and the Lenders by each of the Guarantors, and (ii) the unlimited and unconditional guarantees of the Other Guaranteed Obligations of the other Borrower in favour of the Agent and the Lenders by each of the Borrowers, each of which are to be delivered pursuant to this Agreement.

ARTICLE 4 DISBURSEMENT CONDITIONS

4.1 Conditions Precedent to Initial Advance

The conditions precedent specified in this Section 4.1 must be satisfied at or before the time of the first Advance under this Agreement, unless waived by all Lenders. Where delivery of documents is referred to, the documents must be delivered to the Agent, for and on behalf of the Lenders, the documents must be in form and substance satisfactory to the Lenders, duly executed by all parties and in full force and effect, and all matters disclosed by the documents must be satisfactory to the Lenders.

(1) Documents

- (a) The Agent must have received duly executed copies of this Agreement and the Fee Letter.
- (b) The Agent must have received duly executed copies of the Security and the other Loan Documents.

(2) Other Debt and Liens

- (a) The Agent must have received a true copy of the U.S. Credit Agreement together with all amendments thereto up to the date of this Agreement;
- (b) The Agent must have received all statements, postponements and acknowledgements that are reasonably required in respect of other Encumbrances affecting the Property of the Obligors to confirm that those Encumbrances are Permitted Encumbrances.

(3) <u>Financial Information</u>

- (a) The Agent must have received summary projected information for WMOCC.
- (b) The Agent must have received a Compliance Certificate from the Borrowers for the trailing twelve month period ended as of the Interim Balance Sheet Date.

- 25 -

(5)

(4) Corporate and Other Information

- (a) The Agent must have received a certificate of each Obligor attaching copies of its Constating Documents, a list of its officers and directors with occupations of all directors, specimens of the signatures of those officers or directors who are executing Loan Documents on its behalf, copies of the corporate proceedings taken to authorize it to execute, deliver and perform its obligations under the Loan Documents, and other corporate and "know your client" information that the Agent or any Lender may reasonably require.
- (b) The Agent must have received a certificate of status, compliance, good standing or equivalent for each Obligor for its jurisdiction of incorporation and for each jurisdiction where it carries on business or where registrations or filings in relation to the Security given by that Obligor have been effected, except for any jurisdiction where certificates of that kind are not customarily issued by a Governmental Authority.
- (c) The Agent must have received evidence that the delivery of Loan Documents will not contravene applicable law governing financial assistance or other similar law that affects the Loan Documents.
- (d) Each Obligor must have complied with Section 6.3(2).
- Opinions The Agent must have received the opinions of:
- (a) Borden Ladner Gervais LLP, counsel to the Agent and the Lenders in relation to the Loan Documents which are governed by Ontario law.
- (b) In-house counsel to the Obligors addressed to the Agent, the Lenders and Borden Ladner Gervais LLP in relation to, among other things, the Obligors and the enforceability of the Loan Documents governed by Ontario law and such other matters as the Agent may reasonably require.
- (c) The opinion of local counsel to the Borrowers, addressed to the Agent, the Lenders and Borden Ladner Gervais LLP in relation to, among other things, the existence of each of the Borrowers, its corporate power and authority and the due authorization, execution and delivery of the Loan Documents and such other matters as the Agent may reasonably require.
- (6) <u>Other Matters</u> The following other conditions must be satisfied:
 - (a) All fees and expenses payable under the Loan Documents and the Fee Letter (including upfront fees, agency fees, and reasonable legal fees and expenses of the Lenders' counsel invoiced prior to the Closing Date) shall have been paid.
 - (b) The Lenders must be satisfied that there has not occurred and does not exist since December 31, 2011 any event or circumstance that has, or could reasonably be expected to have, a Material Adverse Effect; and

- 26 -

(c) The Agent must have received any other documents that the Lenders may reasonably require.

4.2 Conditions Precedent to an Advance under the Term Credit

In addition to the other conditions precedent specified in this Agreement, the obligation of the Lenders to make any Advance under the Term Credit is subject to the conditions precedent specified in this Section 4.2, unless waived by all the Lenders. Where delivery of documents is referred to, the documents must be delivered to the Agent, for and on behalf of the Lenders, the documents must be in form and substance reasonably satisfactory to the Lenders, duly executed by all parties and in full force and effect, and all matters disclosed by the documents must be satisfactory to the Lenders.

- (a) The Lenders must have been provided with all relevant information and documentation reasonably requested in connection with the Transaction, including, without limitation, the Agreement of Purchase and Sale and copies of material third party consents and approvals.
- (b) The Lenders must be satisfied that the Transaction has been completed substantially in compliance with the terms of the Agreement of Purchase and Sale.
- (c) The Lenders must be satisfied that there is no pending or threatened dispute that seeks to adjourn, delay, enjoin, prohibit or impose material limitations on any aspect of the Transaction or that has had, or could reasonably be expected to have, a Material Adverse Effect.
- (d) The Agent must have received other documents that the Lenders may reasonably require.

4.3 Conditions Precedent to all Advances

In addition to the other conditions precedent specified in this Agreement, the obligation of the Lenders to make any Advance is subject to the following conditions precedent:

- (a) No Pending Event of Default or Event of Default has occurred and is continuing on the Advance Date, or would result from making the Advance.
- (b) The Agent must have received timely notice as required under Section 9.7.
- (c) The representations and warranties made in Section 5.1 of this Agreement, except those expressly stated to be made as of a specific date, must be true and correct on and as of the Advance Date with the same force and effect as if the representations and warranties had been made on and as of the Advance Date.
- (d) All other terms and conditions of this Agreement on which an Advance may be obtained must be fulfilled.

- 27 -

ARTICLE 5 REPRESENTATIONS AND WARRANTIES

5.1 Representations and Warranties

Each Obligor represents and warrants to the Lenders as specified in this Section 5.1.

- (1) Corporate Authority.
 - (a) Each of the Obligors (i) is duly organized, validly existing and in good standing under the laws of its jurisdiction of formation, (ii) has all requisite corporate power to own its property and conduct its business as now conducted and as presently contemplated, and (iii) is in good standing and is duly authorized to do business in each jurisdiction in which its property or business as presently conducted or contemplated makes such qualification necessary, except where a failure to be so qualified could not reasonably be expected to have a Material Adverse Effect.
 - (b) The execution, delivery and performance of its Loan Documents and the transactions contemplated hereby and thereby (i) are within the corporate authority of each of the Obligors, (ii) have been duly authorized by all necessary corporate proceedings on the part of each of the Obligors, (iii) do not conflict with or result in any breach or contravention of any provision of law, statute, rule or regulation to which any of the Obligors is subject, (iv) do not contravene any judgment, order, writ, injunction, license or permit applicable to any Obligor so as to have a Material Adverse Effect, and (v) do not conflict with any provision of the Constating Documents of any of the Obligors, except for those conflicts with any such agreement or instrument which could not reasonably be expected to have a Material Adverse Effect.
 - (c) The execution, delivery and performance of the Loan Documents by each of the Obligors will result in valid and legally binding obligations of each of the Obligors enforceable against each in accordance with the respective terms and provisions hereof and thereof, except as enforceability is limited by bankruptcy, insolvency, reorganization, moratorium or other laws relating to or affecting generally the enforcement of creditors' rights generally and general principles of equity.
- (2) Governmental and Other Approvals. The execution, delivery and performance of the Loan Documents by each of the Obligors and the consummation by each of the Obligors of the transactions contemplated hereby and thereby do not require any approval or consent of, or filing with, any governmental agency or authority or other third party other than those already obtained and those required after the date hereof in connection with the Obligors' performance of the covenants contained in this Agreement.

- 28 -

(3) Title to Properties; Leases. WMI and its Subsidiaries own all of the assets reflected in the consolidated balance sheet as at the Interim Balance Sheet Date or acquired since that date (except property and assets (a) operated under capital leases (b) sold or otherwise disposed of in the ordinary course of business since that date), or (c) consolidated in accordance with variable entity guidance in FASB ASC 810) subject to no Encumbrances except Permitted Encumbrances.

(4) Financial Statements; Solvency.

- (a) There have been furnished to the Lenders consolidated balance sheets of WMI dated the Balance Sheet Date and consolidated statements of operations for the fiscal periods then ended, certified by Accountants. In addition, there have been furnished to the Lenders consolidated balance sheets of WMI and its Subsidiaries (including the Borrowers) dated the Interim Balance Sheet Date and the related consolidated statements of operations for the fiscal quarter ending on the Interim Balance Sheet Date. All said balance with GAAP (but, in the case of any of such financial statements which are unaudited, only to the extent GAAP is applicable to interim innavial statements which are unaudited, only to the extent GAAP is applicable to interim unaudited reports), and fairly present, in all material respects, the financial condition of WMI on a consolidated basis as at the close of business on the dates thereof and the results of operations for the periods then ended, subject, in the case of unaudited interim financial statements, to changes resulting from audit and normal year-end adjustments and to the absence of complete footnotes. There are no contingent liabilities of WMI and its Subsidiaries involving material amounts, known to the officers of any of the Obligors, which have not been disclosed in said balance sheets and the related notes thereto or otherwise in writing to the Lenders.
- (b) Each of the Obligors on a consolidated basis (both before and after giving effect to the transactions contemplated by this Agreement) is solvent (i.e., it has assets having a fair value in excess of the amount required to pay its probable liabilities on its existing debts as they become absolute and matured) and has, and expects to have, the ability to pay its debts from time to time incurred in connection therewith as such debts mature.
- (5) No Material Changes, Etc. Since the Balance Sheet Date, there have been no material adverse changes in the consolidated financial condition, business, assets or liabilities (contingent or otherwise) of WMI and its Subsidiaries, taken as a whole, other than changes in the ordinary course of business which have not had a Material Adverse Effect.
- (6) Franchises, Patents, Copyrights, Etc. Each of the Obligors possess all franchises, patents, copyrights, trademarks, trade names, licenses and permits, and rights in respect of the foregoing, adequate for the conduct of their business substantially as now conducted (other than those the absence of which would not have a Material Adverse Effect) without known conflict with any rights of others other than a conflict which would not have a Material Adverse Effect.

- 29 -

- (7) Litigation. Except as disclosed in the Disclosure Documents, there are no actions, suits, proceedings or investigations of any kind pending or, to the knowledge of the Obligors, threatened against the Obligors before any court, tribunal or administrative agency or board which, either in any case or in the aggregate, could reasonably be expected to have a Material Adverse Effect.
- (8) No Materially Adverse Contracts, Etc. None of the Obligors is subject to any restriction in its Constating Documents, corporate or other legal restriction, or any judgment, decree, order, rule or regulation which in the judgment of such Obligor's officers has or could reasonably be expected in the future to have a Material Adverse Effect. None of the Obligors is a party to any contract or agreement which in the judgment of such Obligor's officers has or could reasonably be expected to have any Material Adverse Effect, except as otherwise reflected in adequate reserves as required by GAAP.
- (9) Compliance With Other Instruments, Laws, Etc. None of the Obligor is (a) violating any provision of its Constating Documents, or (b) violating any agreement or instrument to which any of them may be subject or by which any of them or any of their properties may be bound or any decree, order, judgment, or any statute, license, rule or regulation, in a manner which could (in the case of such agreements or such instruments) reasonably be expected to result in a Material Adverse Effect.
- (10) Tax Status. Each of the Obligors have filed all federal, state, provincial and territorial income and all other tax returns, reports and declarations (or obtained extensions with respect thereto) required by applicable law to be filed by them (unless and only to the extent that such Obligor has set aside on its books provisions reasonably adequate for the payment of all unpaid and unreported taxes as required by GAAP); and have paid all taxes and other governmental assessments and charges (other than taxes, assessments and other governmental charges imposed by jurisdictions other than the United States, Canada or any political subdivision thereof which in the aggregate are not material to the financial condition, business or assets of any Obligor on an individual basis or of WMI on a consolidated basis) that are material in amount, shown or determined to be due on such returns, reports and declarations, except those being contested in good faith; and, as required by GAAP, have set aside on their books provisions reasonably adequate for the payment of all taxes for periods subsequent to the periods to which such returns, reports or declarations apply. Except to the extent contested in the manner permitted in the preceding sentence, there are no unpaid taxes in any material amount claimed by the taxing authority of any jurisdiction to be due and owing by any Obligor, nor do the officers of any Obligor know of any basis for any such claim.
- (11) No Event of Default. No Pending Event of Default or Event of Default has occurred hereunder and is continuing
- (12) Investment Company Act. None of the Obligors is a "registered investment company", or an "affiliated company" or a "principal underwriter" of a "registered investment company", as such terms are defined in the Investment Company Act of 1940.

- 30 -

(13) Absence of Financing Statements, Etc. Except as permitted by Section 6.4(1), there is no Debt senior to the Obligations, and except for Permitted Encumbrances, there are no Encumbrances, or any effective financing statement, security agreement, chattel mortgage, real estate mortgage or other document filed, registered or recorded with any filing records, registry, or other public office, which purports to cover, affect or give notice of any present or possible future Encumbrances on any assets or property of any Obligor or right thereunder.

(14) Employee Benefit Plans.

- (a) In General. Each Employee Benefit Plan has been maintained and operated in material compliance with the provisions of ERISA and, to the extent applicable, the Code, including but not limited to the provisions thereunder respecting prohibited transactions. Promptly upon the request of any Lender or the Agent, the Borrowers will furnish to the Administrative Agent the most recently completed annual report, Form 5500, with all required attachments, and actuarial statement required to be submitted under \$103(d) of ERISA, with respect to each Guaranteed Pension Plan.
- (b) <u>Terminability of Welfare Plans</u>. Under each Employee Benefit Plan which is an employee welfare benefit plan within the meaning of \$3(1) or \$3(2)(B) of ERISA, no benefits are due unless the event giving rise to the benefit entitlement occurs prior to plan termination (except as required by Title 1, Part 6 of ERISA). WMI or an ERISA Affiliate, as appropriate, may terminate each such employee welfare benefit plan at any time (or at any time subsequent to the expiration of any applicable bargaining agreement) in the discretion of WMI or such ERISA Affiliate without material liability to any Person.
- (c) <u>Guaranteed Pension Plans.</u> Each contribution required to be made to a Guaranteed Pension Plan, whether required to be made to avoid the incurrence of an accumulated funding deficiency, the notice or lien provisions of \$303(k) of ERISA, or otherwise, has been timely made. No waiver of an accumulated funding deficiency or extension of amortization periods has been inceived with respect to any Guaranteed Pension Plan. No liability to the PBGC (other than required insurance premiums, all of which have been paid) has been incurred by any Obligor or any ERISA Affiliate with respect to any Guaranteed Pension Plan No liability to the PBGC (other than required has not been any ERISA Reportable Event, or any other event or condition which presents a material risk of termination of any Guaranteed Pension Plan by the PBGC. Other than with respect to the Terminated Plans, based on the latest valuation of each Guaranteed Pension Plan to the dates of this representation), and on the actuarial methods and assumptions employed for that valuation, each Guaranteed Pension Plan is in compliance with the minimum funding standards as set forth in \$302 of ERISA and is not subject to any restrictions concerning (i) providing shutdown or similar benefits, (ii) amendments to increase benefits, (iii) paying lump sums or (iv) continuing to accrue benefits, as described by the Pension Protection Act of 2006.

- 31 -

- (d) <u>Multiemployer Plans</u>. Except for liabilities that have been discharged prior to the Effective Date or as to which accruals have been made in accordance with GAAP prior to the Effective Date as reflected in the Disclosure Documents, neither any Obligor nor any ERISA Affiliate has incurred any material liability (including secondary liability) to any Multiemployer Plan as a result of a complete or partial withdrawal from such Multiemployer Plan under \$4201 of ERISA or as a result of a sale of assets described in \$4204 of ERISA. Neither any Obligor nor any ERISA Affiliate has been notified that any Multiemployer Plan is in reorganization or insolvent under and within the meaning of \$4241 or \$4245 of ERISA or that any Multiemployer Plan intends to terminate or has been terminated under \$4041A of ERISA.
- (15) Environmental Compliance. Each of the Obligors have taken all steps that they have deemed reasonably necessary to investigate the past and present condition and usage of its Real Property and the operations conducted by it and, based upon such diligent investigation, have determined that, except as set forth in the Disclosure Documents:
 - (a) None of the Obligors, nor any operator of their properties, is in violation, or alleged violation, of any judgment, decree, order, law, permit, license, rule or regulation pertaining to environmental matters, including without limitation, those arising under the Resource Conservation and Recovery Act, ("RCRA") the Comprehensive Environmental Response, Compensation and Liability Act of 1980 as amended ("CERCLA"), the Superfund Amendments and Reauthorization Act of 1986, the Federal Clean Water Act, the Federal Clean Air Act, the Toxic Substances Control Act, or any applicable international, federal, state, provincial, territorial or local statute, regulation, ordinance, order or decree relating to health, safety, waste transportation or disposal, or the environment (the "Environmental Laws"), which violation, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.
 - (b) Except with respect to any such matters that could not reasonably be expected to have a Material Adverse Effect, none of the Obligors has received notice from any third party including, without limitation: any federal, state, provincial, territorial or local governmental authority, (i) that any one of them has been identified by the United States Environmental Protection Agency ("EPA") as a potentially responsible party under CERCLA with respect to a site listed on the National Priorities List, 40 C.F.R. Part 300 Appendix B, (ii) that any hazardous waste, as defined by 42 U.S.C. §6903(5), any hazardous substances as defined by 42 U.S.C. §6901(14), any pollutant or contaminant as defined by 42 U.S.C. §6003(3) or any toxic substance, oil or hazardous materials or other chemicals or substances regulated by any Environmental Laws, excluding household hazardous waste ("Hazardous Substances"), which any one of them has generated, transported or disposed of, has been found at any site at which a federal, state, provincial, territorial or local agency or other third party has conducted or has ordered that an Obligor conduct a remedial investigation, removal or other response action pursuant to any Environmental Law, or (iii) that it is or shall be a named party to any claim, action, cause of action, complaint, legal or administrative proceeding arising out of any third party's incurrence of costs, expenses, losses or damages of any kind whatsoever in connection with the Release of Hazardous Substances.

- 32 -

- (c) Except for those occurrences or situations that could not reasonably be expected to have a Material Adverse Effect (i) no portion of the Real Property or other assets of the Obligors has been used for the handling, processing, storage or disposal of Hazardous Substances except in accordance with applicable Environmental Laws, (ii) in the course of any activities conducted by the Obligors or their respective operators of the Real Property or other assets of the Obligors, no Hazardous Substances have been generated or are being used on such properties except in accordance with applicable Environmental Laws, (iii) there have been no unpermitted Releases of Hazardous Substances on unpermitted Releases of the Cobligors, and (iv) any Hazardous Substances that have been generated on the Real Property or other assets of the Obligors, and (iv) any Hazardous Substances that have been generated on the Real Property or other assets of the Obligors have been transported offsite only by carriers having an identification number issued by the EPA or other relevant Governmental Authority, treated or disposed of only by treatment or disposal facilities maintaining valid permits as required under applicable Environmental Laws.
- (16) Disclosure. No representation or warranty made by any Obligor in this Agreement or in any agreement, instrument, document, certificate, or financial statement furnished to the Lenders or the Agent by or on behalf of or at the request of the Borrowers and any other Obligor in connection with any of the transactions contemplated by the Loan Documents contains any untrue statement of a material fact or omits to state a material fact necessary in order to make the statements contained therein, taken as a whole, not misleading in light of the circumstances in which they are made.
- (17) Permits and Governmental Authority. All permits (other than those the absence of which could not reasonably be expected to have a Material Adverse Effect) required for the construction and operation of all landfills currently owned or operated by WMI or the other Obligors have been obtained and remain in full force and effect and are not subject to any appeals or further proceedings or to any unsatisfied conditions that may allow material modification or revocation. None of the Obligors, to the knowledge of any such Obligor, or the holder of such permits, is in violation of any such permits, except for any violation which could not reasonably be expected to have a Material Adverse Effect.
- (18) Margin Stock. No Obligor is engaged in the business of extending credit for the purpose of purchasing or carrying margin stock (within the meaning of Regulation U issued by the Board of Governors of the Federal Reserve System), and no proceeds of any Loans will be used to purchase or carry any margin stock or to extend credit to others for the purpose of purchasing or carrying any margin stock in violation of Regulations T, U or X of the Board of Governors of the Federal Reserve System.

- 33 -

5.2 Survival of Representations and Warranties

The representations and warranties made in this Agreement or any other Loan Document shall survive the execution of this Agreement and all other Loan Documents until such time as all of the Obligations and Other Guaranteed Obligations have been paid in full. No investigation by or on behalf of the Agent or Lenders at any time shall have the effect of waiving, diminishing the scope or otherwise affecting any representation or warranty made in any Loan Document. Unless expressly stated to be made as of a specific date, they shall be deemed to be repeated as of the date of each Advance and as of the date of delivery of each Compliance Certificate. The Lenders shall be deemed to have relied upon all representations and warranties at each time they make an Advance as a condition of making an Advance and continuing to extend the Credits.

ARTICLE 6 COVENANTS

6.1 Financial Covenants

WMI shall comply with the following covenants:

- (a) Interest Coverage Ratio. As of the end of any fiscal quarter of WMI, WMI will not permit the ratio of (a) EBIT for the four fiscal quarters then ending to (b) Consolidated Total Interest Expense for such period to be less than 2.75:1.00;
- (b) Total Debt to EBITDA. As of the end of any fiscal quarter of WMI, WMI will not permit the ratio of (a) Total Debt to (b) EBITDA for the four fiscal quarters then ending to exceed 3.50:1.00.

6.2 Positive Covenants

Each Obligor shall perform the covenants specified in this Section 6.2.

- (1) Punctual Payment. Each of the Borrowers shall duly and punctually pay and perform its indebtedness, liabilities and obligations under this Agreement and under the other Loan Documents to which it is a party and each Guarantor shall duly and punctually pay and perform (or cause to be paid and performed) its indebtedness, liabilities and obligations under this Agreement and under the other Loan Documents to which it is a party, in each case, at the times and places and in the manner required by the terms hereof and thereof.
- (2) Chief Place of Business. The Borrowers' place of business is located at 117 Wentworth Court, Brampton, Ontario L6T 5L4. The Borrowers will give 30 days' prior written notice to the Agent of any change in their place of business.
- (3) Records and Accounts. Each of the Obligors will keep true and accurate records and books of account in which full, true and correct entries will be made in accordance with GAAP and with the requirements of all regulatory authorities and maintain adequate accounts and reserves for all taxes (including income taxes), depreciation, depletion, obsolescence and amortization of its properties, all other roomproper reserves.



- (4) Existence and Conduct of Business. Each of the Obligors will do or cause to be done all things necessary to preserve and keep in full force and effect its existence, rights and franchises; and effect and maintain its foreign qualifications (except where the failure to do so could not reasonably be expected to have a Material Adverse Effect), licensing, domestication or authorization, except as any of the foregoing may be terminated by its board of directors in the exercise of its reasonable judgment; provided that such termination could not reasonably be expected to have a Material Adverse Effect. None of the Obligors will become obligated under any contract or binding arrangement which, at the time it was entered into, could reasonably be expected to have a Material Adverse Effect. Each of the Obligors will continue to engage primarily in any of the businesses now conducted by it and in related, complementary or supplemental businesses, and any additional businesses acquired pursuant to the terms of Section 6.4(4).
- (5) Maintenance of Properties. Each of the Obligors will cause all material properties used or useful in the conduct of its businesses to be maintained and kept in good condition, repair and working order (ordinary wear and tear excepted) and supplied with all necessary equipment and cause to be made all necessary repairs, renewals, replacements, betterments and improvements thereof, all as in its judgment may be necessary so that the businesses carried on in connection therewith may be properly and advantageously conducted at all times; provided, however, that nothing in this Section 6.2(6) shall prevent any Obligor from discontinuing the operation and maintenance of any of its properties if such discontinuance is, in its judgment, desirable in the conduct of its business and could not reasonably be expected to have a Material Adverse Effect.
- (6) Insurance. Each Obligor shall maintain or cause to be maintained insurance of the kinds, covering the risks (other than risks arising out of or in any way connected with personal liability of any officers and directors thereof) and in the relative proportionate amounts usually carried by reasonable and prudent companies conducting businesses similar to that of the Obligors, in amounts substantially similar to the existing coverage maintained by the Obligors. Such insurance shall be with financially sound and reputable insurance companies (including captive insurance companies), funds or underwriters, or may be pursuant to self-insurance plans. In addition, the Obligors will furnish from time to time, upon the Agent's request, summary of the insurance coverage of the Obligors, which summary shall be in form and substance satisfactory to the Agent and, if requested by the Agent, will furnish to the Agent copies of the applicable policies.
- (7) Taxes. Each Obligor will duly pay and discharge, or cause to be paid and discharged, before the same shall become overdue, all taxes, assessments and other governmental charges imposed upon it and its real properties, sales and activities, or any part thereof, or upon the income or profits therefrom, as well as all claims for labour, materials, or supplies, which if unpaid might by law become an Encumbrances upon any of its property; provided, however, that any such tax, assessment, charge, levy or claim need not be paid if the failure to do so (either individually, or in the aggregate for all such failures) could not reasonably be expected to have a Material Adverse Effect and the

- 35 -

validity or amount thereof shall currently be contested in good faith by appropriate proceedings and if such Obligor shall have set aside on its books adequate reserves with respect thereto as required by GAAP; and provided, further, that each Obligor will pay all such taxes, assessments, charges, levies or claims prior to the foreclosure on any Encumbrance which may have attached as security therefor.

- (8) Inspection of Properties, Books and Contracts. Each Obligor will permit the Agent or any Lender or any of their respective designated representatives, upon reasonable notice, to visit and inspect any of its properties, to examine its books of account or contracts (and to make copies thereof and extracts therefrom), and to discuss its affairs, finances and accounts with, and to be advised as to the same by, its officers, all at such times and intervals as may be reasonably requested.
- (9) Compliance with Laws, Contracts, Licenses and Permits; Maintenance of Material Licenses and Permits. Each Obligor will (a) comply with the provisions of its Constating Documents, (b) comply with all agreements and instruments by which it or any of its properties may be bound except where non-compliance could not reasonably be expected to have a Material Adverse Effect, (c) comply with all applicable laws and regulations (including Environmental Laws), decrees, orders, judgments, licenses and permits, including, without limitation, all environmental permits ("Applicable Requirements"), except where non-compliance with such Applicable Requirements could not reasonably be expected to have a Material Adverse Effect, (d) maintain all operating permits for all landfills now owned or hereafter acquired, except where the failure to do so could not reasonably be expected to have a Material Adverse Effect, (d) maintain all operations what end y atticensed disposal facilities operating, to such Obligor's knowledge, in compliance with Environmental Laws, except where the failure to do so could not reasonably be expected to have a Material Adverse Effect, and (e) dispose of hazardous waste only at licensed disposal facilities operating, to such Obligor's knowledge, in compliance with Environmental Laws, except where the failure to do so could not reasonably be expected to have a Material Adverse Effect. If at any time any authorization, consent, approval, permit or license from any officer, agency or instrumentality of any government shall become necessary or required in order that any Obligor may fulfill any of its obligations hereunder or under any other Loan Document, such Obligor will immediately take or cause to be taken all reasonable steps within its power to obtain such authorization, consent, approval, permit or license and furnish the Agent with evidence thereof.
- (10) Environmental Indemnification. Each of the Obligors covenants and agrees that it will indemnify and hold the Lenders and the Agent and their respective affiliates, and each of the representatives, agents and officers of each of the foregoing, harmless from and against any and all claims, expense, damage, loss or liability incurred by the Lenders or the Agent (including all reasonable costs of legal representation incurred by the Lenders or the Agent) relating to (a) any Release or threatened Release of Hazardous Substances on any of its Real Property, (b) any violation of any Environmental Laws or other Applicable Requirements with respect to conditions of the Real Property or other assets of the Obligors, or the operations conduced thereon, or (c) the investigation or remediation of offsite locations at which any of the Obligors, or their predecessors, are alleged to have directly or indirectly disposed of Hazardous Substances. It is expressly acknowledged by the Obligors that this covenant of indemnification shall survive the payment of the Obligations and termination of the Credit and shall inure to the benefit of the Lenders, the Agent and their affiliates, successors and assigns.

- 36 -

- (11) **Further Assurances.** Each of the Obligors will cooperate with the Agent and execute such further instruments and documents as the Agent shall reasonably request to carry out to the Required Lenders' satisfaction the transactions contemplated by this Agreement.
- (12) Notice of Potential Claims or Litigation. WMI shall deliver to the Agent written notice of the initiation of any action, claim, complaint, investigation or any other notice of dispute or litigation against any Obligor that could reasonably be expected to have a Material Adverse Effect, or which questions the validity or enforceability of any Loan Document, together with a copy of each such complaint or other notice received by such Obligor if requested by the Agent within 30 days of receipt thereof or of the determination that such action could reasonably be expected to have a Material Adverse Effect, whichever occurs later (and such Obligor will make such determination in each case as promptly as practicable).
- (13) Notice of Certain Events Concerning Environmental Claims. WMI shall promptly, and in any event within ten Business Days of the Obligor obtaining knowledge thereof, notify the Agent of any of the following events:
 - (a) any Obligor obtaining knowledge of any violation of any Environmental Law regarding its Real Property or operations which violation could reasonably be expected to have a Material Adverse Effect;
 - (b) any Obligor obtaining knowledge of any potential or known Release, or threat of Release, of any Hazardous Substance at, from, or into any Real Property which could reasonably be expected to have a Material Adverse Effect;
 - (c) any Obligor receiving any notice of any material violation of any Environmental Law or of any Release or threatened Release of Hazardous Substance, including a notice or claim of liability or potential responsibility from any third party (including any Governmental Authority) and including notice of any formal inquiry, proceeding, demand, investigation or other action with regard to (i) any Obligor's or other Person's operation of the Real Property of such Obligor, (ii) contamination on, from, or into the Real Property, or (iii) investigation or remediation of offsite locations at which any Obligor, or its predecessors, are alleged to have directly or indirectly disposed of Hazardous Substances, if any thereof could reasonably be expected to have a Material Adverse Effect; or
 - (d) any Obligor obtaining knowledge that any expense or loss has been incurred by any Governmental Authority in connection with the assessment, containment, removal or remediation of any Hazardous Substance with respect to which any Obligor has been alleged to be liable by such Governmental Authority or for which an Encumbrance may be imposed on any Real Property by such Governmental Authority, if any thereof could reasonably be expected to have a Material Adverse Effect.

- 37 -

(14) Notice of Default. The Borrowers will promptly notify the Agent in writing of the occurrence of any Pending Event of Default or Event of Default. If any Person shall give any notice or take any other action in respect of a claimed default (whether or not constituting an Event of Default) under this Agreement or any other note, evidence of indebtedness, indenture or other obligation evidencing indebtedness in excess of U.S. \$75,000,000 as to which any Obligor is a party or obligor, whether as principal or surety, such Obligor shall promptly upon obtaining actual knowledge thereof give written notice thereof to the Agent, describing the notice of action and the nature of the claimed default.

(16) Certain Transactions. Except as disclosed in the Disclosure Documents, and except for arm's length transactions pursuant to which any Obligor makes payments in the ordinary course of business, none of such Obligor's officers, directors, or employees (or any affiliate of such officers, directors or employees) are presently or shall be a party to any transaction with the Borrowers or any Guarantor (other than for services as employees, officers and directors), including any contract, agreement or other arrangement providing for the furnishing of services to or by, providing for rental of real or personal property to or from, or otherwise requiring payments to or from any officer, director or such employee or, to the knowledge of the Borrowers or any Guarantor, any corporation, partnership, trust or other entity in which any officer, director, or any such employee has a substantial interest or is an officer, trustee or partner.

6.3 Reporting Requirements

During the term of this Agreement, WMI and the Borrowers shall deliver or cause the delivery of the periodic reports specified below and shall give notices in the circumstances specified below, or cause notices to be given. All financial statements and other reports shall be prepared in accordance with GAAP applied on a consistent basis.

(1) Periodic Financial Reports

- (a) WMI shall, as soon as practicable and in any event within 60 days of the end of each of its fiscal quarters (excluding the fourth fiscal quarter), cause to be prepared and delivered to the Agent, its interim unaudited consolidated financial statements as at the end of such quarter; provided, however, if such information is filed with the Securities and Exchange Commission and publicly available on the Securities and Exchange Commission's website, such filing shall be deemed delivery to the Agent.
- (b) WMI shall, as soon as practicable and in any event within 100 days after the end of each of its fiscal years, prepare and deliver to the Agent its consolidated annual financial statements, audited by the Accountants; provided, however, if such information is filed with the Securities and Exchange Commission and publicly available on the Securities and Exchange Commission's website, such filing shall be deemed delivery to the Agent.

- 38 -

⁽¹⁵⁾ Use of Proceeds. The proceeds of the Advances shall be used as set forth in Section 2.3.

- (c) WMOCC shall, as soon as practicable and in any event within 100 days after the end of each of its fiscal years, deliver to the Agent its internally prepared consolidated financial information reasonably acceptable to the Lenders.
- (d) WMI shall, concurrently with the delivery of its quarterly financial statements and annual financial statements, provide the Agent with a Compliance Certificate.
- (e) WMI shall provide to the Agent, promptly following the filing or mailing thereof, copies of all material of a financial nature filed by WMI with the Securities and Exchange Commission or sent to WMI's stockholders generally; provided, however, if such information is filed with the Securities and Exchange Commission and publicly available on the Securities and Exchange Commission's website, such filing shall be deemed delivery to the Agent.
- (f) The Obligors shall promptly provide the Agent with all other information, reports and certificates reasonably requested by the Lenders from time to time concerning the business, financial condition and property of the Borrowers and each other Obligor.
- (2) Know Your Client Matters The Obligors shall promptly provide all information, including information concerning their direct and indirect holders of equity interests and other Persons exercising control over it, and its and their respective directors and officers, and including supporting documentation and other evidence, as may reasonably be requested by the Agent, any Lender, or any prospective assignee or participant of a Lender, in order to comply with the requesting person's policies and procedures relating to Applicable Law regarding anti-more Jaundering, anti-terrorist financing, government sanction and "know your client" matters, including the *Proceeds of Crime (Money Laundering) and Terrorist Financing Act* (Canada) and the *United Nations Act* (Canada) or any similar Canadian legislation, together with all rules, regulations and interpretations thereunder or related thereto including, without limitation, the *Regulations Act* (Nations *Resolutions as propression of Terrorism and the United Nations Act*).

6.4 Negative Covenants

No Obligor shall do any of the things specified in this Section 6.4.

(1) Restrictions on Indebtedness. None of the Obligors (other than WMI) will create, incur, assume, or be or remain liable, contingently or otherwise, with respect to any Debt, or become or be responsible in any manner (whether by agreement to purchase any obligations, stock, assets, goods or services, or to supply or advance any funds, assets, goods or services or otherwise) with respect to any Debt, in each case, of any other Person (other than WMI or any of its Subsidiaries), other than:

- 39 -

- (a) Debt of WMI's Subsidiaries listed in Schedule 8.1(a) of the U.S. Credit Agreement (a copy of which is attached hereto as Annex 8.1(a)) and any extension, renewal or refinancing of such Debt, provided that the terms and conditions of any such extensions, renewals or refinancings do not increase the relative priority of the original Debt and provided, further, that such extended, renewed or refinanced Debt does not in the aggregate exceed the U.S. Dollar amount of the original Debt; and
- (b) other Debt of WMI's Subsidiaries (other than Waste Management Holdings, Inc.) provided that the aggregate amount of all such Debt under this Section 6.4(1)(b), when added (without duplication) to the aggregate outstanding amount of secured Debt of WMI and its Subsidiaries under (k), (l) and (m) of the definition of "Permitted Encumbrances" and Debt with respect to Permitted Receivables Transactions, shall not exceed 15% of Consolidated Tangible Assets.

(2) Restrictions on Encumbrances.

- (a) None of the Obligors will create or incur or suffer to be created or incurred or to exist any Encumbrance of any kind upon any property or assets of any character, whether now owned or hereafter acquired, or upon the income or profits therefrom; or transfer any of such property or assets or the income or profits therefrom for the purpose of subjecting the same to the payment of Debt or performance of any other obligation in priority to payment of its general creditors; or acquire, or agree or have an option to acquire, any property or assets upon conditional sale or other title retention or purchase money security agreement, device or arrangement; or suffer to exist for a period of more than 30 days after the same shall have been incurred any Debt or claim or demand against it which if unpaid might by law or upon bankruptcy or insolvency, or othervise, be given any priority whatsoever over its general creditors; or sell, assign, pledge or otherwise transfer any accounts, contract rights, general intangibles or chatel paper, with or without recourse, except for Permitted Encumbrances.
- (b) Each Obligor covenants and agrees that if any of them shall create or incur any Encumbrance upon any of their respective property or assets, whether now owned or hereafter acquired, other than Permitted Encumbrances (unless prior written consent shall have been obtained from the Lenders), the Obligors will make or cause to be made effective provision whereby the Obligations will be secured by such Encumbrance equally and ratably with any and all other Debt thereby secured so long as such other Debt; provided that the covenants of the Obligors contained in this sentence shall only be in effect for so long as the Obligors thall be similarly obligated under any other Debt; provided, further, that an Event of Default shall occur for so long as such other Debt becomes secured notwithstanding any actions taken by the Obligor to ratably secure the Obligations hereunder.

- 40 -

(3) Restrictions on Investments. Except to the extent provided in Section 6.4(4), none of the Obligors may make or permit to exist or to remain outstanding any Investment, other than Investments in cash equivalents unless both before and after giving effect thereto (a) WMI and its Subsidiaries are in compliance with the covenants set forth herein, and (ii) there does not exist an Event of Default or Pending Event of

(4) Mergers, Consolidations, Sales

- (a) No Obligor shall be a party to any merger, consolidation or exchange of stock unless WMI shall be the surviving entity with respect to any such transaction to which WMI is a party and Waste Management Holdings, Inc. shall be the survivor of any merger with any other Obligor or a Subsidiary shall be the surviving entity (and continue to be a Subsidiary) with respect to any such transactions to which one or more Obligors is a party (and the conditions set forth below are satisfied), or purchase or otherwise acquire all or substantially all of the assets or stock of any class of, or any partnership, membership or joint venture or other interest in, any other Person except as otherwise acquire all or substantially all of the assets or stock of any class of, or joint venture or other interest in, any other Person except as otherwise acquire or other interest in, any Person if the following conditions have been met: (i) the proposed transaction will not otherwise create an Event of Default or a Pending Event of Default hereunder; and (ii) the business to be acquired predominantly involves (A) the collection, transfer, hauling, disposal or recycling of solid waste or thermal soil remediation, or (B) other lines of businesses currently engaged in, or related, associated, complementary on supplementary thereto, whether from an operational, business financial, technical or administrative standpoint; provided that an Obligor may purchase or otherwise acquire all or substantially all of the assets of sock of any class of, or partnership, membership or joint venture or other interest in, any Persons in unrelated businesses, not to exceed a total aggregate amount of \$400,000,000 during the term of this Agreement. Notwithstanding anything herein to the contrary, the ability of the Subsidiaries of WMI to incur any Debt in connection with any transaction permitted pursuant to this Section 6.4(4) shall be governed Section 6.4(1). For the avoidance of doubt, this section shall apply against the total aggregate cap of \$400,
- (b) No Obligor shall sell, transfer, convey or lease any assets or group of assets, including the sale or transfer of any property owned by such Obligor in order then or thereafter to lease such property or lease other property which such Obligor intends to use for substantially the same purpose as the property being sold or transferred, or sell or assign, with or without recourse, any receivables, except (i) transfers of real or personal property among Subsidiaries of WMI, (ii) so long as no Default or Event of Default has occurred and is continuing, or would result

- 41 -

therefrom, sales of assets or pursuant to sale-leaseback transaction; provided that any net cash proceeds from any such sale or sale-leaseback shall within 180 days, either be used to pay down outstanding Obligations under either this Agreement or the U.S. Credit Agreement or be reinvested by such Person in assets of the business of the Obligors, used for working capital, invested in Investments in accordance with the provisions of Section 6.4(3) or used for general corporate purposes, (iii) sales of accounts receivable (and contract rights, general intangibles or chattel paper related thereto) more than sixty (60) days past due sold or assigned in the ordinary course of collecting past due accounts, or (iv) pursuant to a Permitted Receivables Transaction.

(5) Restricted Distributions and Redemptions. None of the Obligors will (a) declare or pay any Distributions, or (b) redeem, convert, retire or otherwise acquire shares of any class of its capital stock (other than in connection with a merger permitted by Section 6.4(4) hereof or conversion into another form of equity of any preferred shares of WMI existing as of the Closing Date pursuant to the terms thereof), unless at the time of such Distribution or redemption no Event of Default or Pending Event of Default exists or would be created hereunder. Notwithstanding the above, any Subsidiary may make Distributions to WMI and WMI agrees that no Obligor will enter into any agreement restricting Distributions from such Obligor to the Borrower.

(6) Employee Benefit Plans. No Obligor or any ERISA Affiliate will:

- (a) engage in any "prohibited transaction" within the meaning of \$406 of ERISA or \$4975 of the Code which could result in a material liability for WMI on a consolidated basis; or
- (b) permit any Guaranteed Pension Plan to be in "at risk" status or subject to the notice and lien provisions described in \$303 of ERISA, whether or not a minimum funding waiver has been granted; or
- (c) fail to contribute to any Guaranteed Pension Plan to an extent which, or terminate any Guaranteed Pension Plan in a manner which, could result in the imposition of a lien or encumbrance on the assets of any Obligor pursuant to \$303 or \$4068 or ERISA; or
- (d) permit or take any action which would result in the aggregate benefit liabilities (within the meaning of \$4001 of ERISA), other than with respect to The Waste Management, Inc. Pension Plan and The Waste Management of Alameda County, Inc. Retirement Plan, of all Guaranteed Pension Plans exceeding the value of the aggregate assets of such Guaranteed Pension Plans, disregarding for this purpose the benefit liabilities and assets of any such Guaranteed Pension Plan with assets in excess of benefit liabilities.

The Obligors will (i) promptly upon the request of any Lender or the Agent, furnish to the Lenders a copy of the most recent actuarial statement required to be submitted under \$103(d) of ERISA and Annual Report, Form 5500, with all required attachments, in

- 42 -

respect of each Guaranteed Pension Plan, and (ii) promptly upon receipt or dispatch, furnish to the Lenders any notice, report or demand sent or received in respect of a Guaranteed Pension Plan under §§302, 303, 4041, 4042, 4043, 4063, 4065, 4066 and 4068 of ERISA, or in respect of a Multiemployer Plan, under §§4041A, 4202, 4219, 4242 or 4245 of ERISA.

(7) Perform Obligations. If an Event of Default has occurred and is continuing and any Obligor has failed to perform any of its covenants or agreements in the Loan Documents, the Required Lender, may, but shall be under no obligation to, instruct the Agent on behalf of the Lenders to perform any such covenants or agreements in any manner deemed fit by the Required Lenders without thereby waiving any rights to enforce the Loan Documents. The reasonable expenses (including any legal costs) paid by the Agent and/or the Lenders in respect of the foregoing shall be secured by the Security.

ARTICLE 7 DEFAULT

7.1 Default

Each of the following events shall constitute an Event of Default under this Agreement:

- (a) the Borrowers fail to pay, whether by acceleration or otherwise, any amount of principal (including any amount relating to a B/A or an L/C) when due; or
- (b) the Borrowers fail to pay any amount of interest, fees, commissions or other Obligations (other than amounts on account of principal) when due, and such failure continues for five Business Days after the date of such default; or
- (c) there occurs a breach of any of the covenants in Section 6.1 or Section 6.4; or
- (d) any Obligor makes any representation or warranty in any Loan Document, or in any written statement or certificate made or delivered pursuant to this Agreement which shall prove to have been false in any material respect upon the date when made or deemed to be made; or
- (e) there is a breach of any covenant, condition or other provision of any Loan Document (other than a breach which is specifically dealt with elsewhere in this Section 7.1), by any party thereto other than the Agent or the Lenders, and such breach, if capable of being remedied, is not corrected or otherwise remedied within 30 days after the Agent or any Lender gives written notice thereof to the Borrowers; or
- (f) any Obligor shall fail to pay when due, or within any applicable period of grace, any Debt or obligations under Swap Contracts in an aggregate amount greater than U.S. \$75,000,000, or fail to observe or perform any material term, covenant or agreement contained in any one or more agreements by which it is bound, evidencing or securing any Debt or obligations under Swap Contracts in an aggregate amount greater than U.S. \$75,000,000 for such period of time as would permit, or would have permitted (assuming the giving of appropriate notice if required) the holder or holders thereof or of any obligations issued thereunder to accelerate the maturity thereof or terminate its commitment with respect thereto; or

- (g) any Obligor makes an assignment for the benefit of creditors, or admits in writing its inability to pay or generally fails to pay its debts as they mature or become due, or petitions or applies for the appointment of a trustee or other custodian, liquidator or receiver of any Obligor, or of any substantial part of the assets of any Obligor or commences any case or other proceeding relating to any Obligor under any bankruptcy, reorganization, arrangement, insolvency, readjustment of debt, dissolution or liquidation or similar law of any jurisdiction, now or hereafter in effect, or takes any action to authorize or in furtherance of any of the foregoing, or if any such petition or application is filed or any such case or other proceeding is commenced against any Obligor or any Obligor indicates its approval thereof, consent thereto or acquiescence therein; or
- (h) if a decree or order is entered appointing any such trustee, custodian, liquidator or receiver or adjudicating any Obligor bankrupt or insolvent, or approving a petition in any such case or other proceeding, or a decree or order for relief is entered in respect of any Obligor in an involuntary case under the bankruptcy laws of any jurisdiction as now or hereafter constituted; or
- (i) if there shall remain in force, undischarged, unsatisfied and unstayed, for more than 30 days, whether or not consecutive, any final judgment against any Obligor which, with other outstanding final judgments against any Obligor, exceeds in the aggregate U.S. \$50,000,000 after taking into account any undisputed insurance coverage; or
- (j) if, with respect to any Guaranteed Pension Plan, an ERISA Reportable Event shall have occurred and the Lenders shall have determined in their reasonable discretion that such event reasonably could be expected to result in liability of the Borrowers or any Subsidiary to the PBGC or such Plan in an aggregate amount exceeding \$50,000,000 and such event in the circumstances occurring reasonably could constitute grounds for the partial or complete termination of such Plan by the PBGC or for the appointment by the appropriate United States District Court of a trustee to administer such Plan; or a trustee shall have been appointed by the appropriate United States District Court to administer such Plan; or the PBGC shall have instituted proceedings to terminate such Plan;
- (k) if any of the Loan Documents shall be cancelled, terminated, revoked or rescinded otherwise than in accordance with the terms thereof or with the express prior written agreement, consent or approval of the Lenders, or any action at law, suit or in equity or other legal proceeding to cancel, revoke or rescind any of the Loan Documents shall be commenced by or on behalf of any Obligor, or any of their respective stockholders, or any court or any other governmental or regulatory authority or agency of competent jurisdiction shall make a determination that, or issue a judgment, order, decree or ruling to the effect that, any one or more of the Loan Documents is illegal, invalid or unenforceable in accordance with the terms thereof;

- 44 -

- (1) if any person or group of persons (within the meaning of Section 13 or 14 of the Securities Exchange Act of 1934, as amended) shall have acquired beneficial ownership (within the meaning of Rule 13d-3 promulgated by the Securities and Exchange Commission under said Act) of 25% or more of the outstanding shares of common voting stock of WMI; or during any period of twelve consecutive calendar months, individuals who were directors of WMI on the first day of such period (together with any new directors whose election by such board or whose nomination for election by the shareholders of WMI was approved by a vote of a majority of the directors still in office who were either directors at the beginning of such period or whose election or nomination for election was previously so approved) shall cease to constitute a majority of the board of directors of WMI;
- (m) either of the Borrowers ceases to be directly or indirectly wholly-owned by WMI; or
- (n) the occurrence of any Event of Default (as such term is defined in the U.S. Credit Agreement as at the date of this Agreement) under the U.S. Credit Agreement.

7.2 Acceleration and Termination of Rights

- (1) If any Event of Default occurs, the Lenders shall be under no further obligation to make Advances and the Required Lenders may instruct the Agent to give notice to the Borrowers (a) declaring the Lenders' obligations to make Advances to be terminated, in which case they shall terminate immediately, and/or (b) declaring the Obligations or any of them to be due and payable, in which case they shall be immediately due and payable without presentment, demand, protest or further notice of any kind, all of which are expressly waived by the Borrowers.
- (2) Notwithstanding the preceding paragraph, if any Obligor becomes a bankrupt (voluntarily or involuntarily), or institutes any proceeding seeking liquidation, dissolution, arrangement, winding-up, relief of debtors or from creditors or the appointment of a receiver or trustee over any material part of its property or analogous proceeding in any jurisdiction, then without prejudice to the other rights of the Lenders as a result of any such event, without any notice or action of any kind by the Agent or any Lender, and without presentment, demand or protest, the Lenders' obligation to make Advances shall immediately terminate, the Obligations shall immediately become due and payable and the Borrowers shall be obligated to deposit forthwith with the Agent for the Lenders' benefit Collateral equal to the full face amount at maturity of all L/Cs and B/As then outstanding for its account

- 45 -
7.3 Payment of L/Cs and B/As

- (1) Immediately on any Obligations becoming due and payable under Section 7.2, the Borrowers shall, without necessity of further act or evidence, be unconditionally obligated to immediately deposit with the Agent for the Lenders' benefit Cash Collateral equal to the full face amount of all L/Cs and B/As then outstanding for its account and the Borrowers hereby unconditionally promise and agree to do so. The amounts required to be deposited as Cash Collateral form part of the Obligations. Each of the Borrowers authorizes the Lenders, or any of them, to debit its accounts with the amount required to pay outstanding L/Cs, and to pay outstanding B/As including any that are held by the Lenders, or any of them, in their own right at maturity. Amounts paid to the Agent in respect of B/As and L/Cs shall be applied against, and shall reduce, *pro rata* among the Lenders, to the extent of the amounts paid to the Agent in respect of B/As and L/Cs, respectively, the obligations of the Borrowers to pay amounts then or subsequently payable under B/As and L/Cs, respectively, at the times amounts become payable thereunder.
- (2) The Borrowers shall be entitled to receive interest on cash held as Cash Collateral in accordance with Section 9.5(9).

7.4 Remedies

On the occurrence of any event by which any of the Obligations become due and payable under Section 7.2, the Security shall become immediately enforceable and the Required Lenders may instruct the Agent to take any action or proceedings on behalf of the Lenders and in compliance with applicable Requirements of Law that the Required Lenders in their sole discretion deem expedient to enforce the Security, all without any additional notice, presentment, demand, protest or other formality, all of which are expressly waived by the Obligors.

7.5 Interest After Stay of Proceedings

If a stay of proceedings is obtained or ordered in respect of either of the Borrowers under the provisions of the *Bankruptcy and Insolvency Act* (Canada) or the *Companies' Creditors Arrangement Act* (Canada), then without prejudice to the Lenders' rights to contest the stay, the Borrowers agree to continue to pay interest and fees on all amounts due to the Lenders in accordance with this Agreement. Each of the Borrowers acknowledge that permitting the Borrowers to continue to use the proceeds of Advances after a stay is obtained or ordered constitutes valuable consideration in the same way that permitting the Borrowers to use leased premises constitutes valuable consideration.

7.6 Saving

Neither the Agent nor any Lender shall be under any obligation to the Obligors or any other Person to realize any collateral that is subject to the Security or enforce all or any part of the Security or to allow any of the collateral to be sold, dealt with or otherwise disposed of. Neither the Agent nor any Lender shall be responsible or liable to the Obligors or any other Person for any loss or damage upon the realization or enforcement of, the failure to realize or enforce the Security or any part of it, the failure to allow any of the collateral to be sold, dealt with or otherwise disposed of or any act or or omission on their respective parts or on the part of any director, officer, agent, servant or adviser in connection with any of the foregoing, except that the Agent or a Lender may be responsible or liable for any loss or damage arising from its wilful misconduct or gross negligence.

7.7 Perform Obligations

If an Event of Default has occurred and is continuing and if any Obligor has failed to perform any of its covenants or agreements in the Loan Documents, the Required Lenders may, but shall be under no obligation to, instruct the Agent on behalf of the Lenders to perform any of those covenants or agreements in any manner deemed fit by the Required Lenders without waiving any rights to enforce the Loan Documents as a result. The reasonable expenses (including any legal costs) paid by the Agent and/or the Lenders in respect of the foregoing shall be secured by the Security.

7.8 Third Parties

No Person dealing with any Lender, the Agent or any agent of the Lenders is required to determine (a) whether the Security has become enforceable, the powers that the Lenders, the Agent or their agents are purporting to exercise have become exercisable, or any Obligations remain outstanding, (b) as to the necessity or expediency of the stipulations and conditions subject to which any sale shall be made, (c) otherwise as to the propriety or regularity of any sale or other disposition or any other dealing with any collateral, or (d) how any payment to the Lenders or the Agent.

7.9 Remedies Cumulative

The rights and remedies of the Lenders under the Loan Documents are cumulative and are in addition to and not in substitution for any rights or remedies provided by Applicable Law. Any single or partial exercise by the Lenders of any right or remedy for a default or breach of any agreement, term, covenant or condition in this Agreement shall not be deemed to be a waiver of or to alter, affect, or prejudice any other right or remedy or other rights or remedies to which the Lenders may be lawfully entitled for the same default or breach. Any waiver by the Lenders of the strict observance, performance or compliance with any agreement, term, covenant or condition in this Agreement and any indulgence granted by the Lenders shall be deemed not to be a waiver of any subsequent default.

7.10 Set-Off or Compensation

During the continuance of an Event of Default, any deposits or other sums credited by or due from any Lender to any Borrower and any securities or other property of any Borrower in the possession of such Lender may be applied to or set-off against the payment of the Obligations and any and all other liabilities, direct, or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, of the Borrowers to the Lenders or the Agent.

- 47 -

7.11 Direct Payments

- (1) If any Lender, by exercising any right of set-off or counterclaim or otherwise, obtains any payment or other reduction that might result in that Lender receiving payment or other reduction of a proportion of the aggregate amount of its Advances and accrued interest on them or other Obligations and Other Secured Obligations greater than its pro rata share of them as provided in this Agreement, then the Lender receiving the payment or other reduction shall (a) notify the Agent of that fact, and (b) purchase (for cash at face value) participations in the Advances and other Obligations and Other Secured Obligations owing to the other Lenders, or make other adjustments as shall be equitable, so that the benefit of all those payments shall be shared by the Lenders and others to whom Other Secured Obligations are owed rateably in accordance with the aggregate amount of Obligations and Other Secured Obligations owing to them, provided that:
 - (a) if any participation is purchased and all or any portion of the payment giving rise to it is recovered, the participation shall be rescinded and the purchase price restored to the extent of the recovery, without interest,
 - (b) the provisions of this Section shall not be construed to apply to (i) any payment made by any Obligor pursuant to and in accordance with the express terms of this Agreement or (ii) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Advances to any assignee or participant, other than to any Obligor or any Affiliate of an Obligor (as to which the provisions of this Section shall apply); and
 - (c) the provisions of this Section shall not be construed to apply to (i) any payment (including a payment in respect of Other Secured Obligations) made while no Event of Default has occurred and is continuing in respect of obligations of the Borrowers to a Lender that do not arise under or in connection with the Loan Documents, (ii) any payment made in respect of an obligation that is secured by a Permitted Lien (other than the Security) or that is otherwise entitled to priority over the Borrowers' obligations under or in connection with the Loan Documents, (iii) any reduction arising from an amount owing to an Obligor upon the termination of Derivatives entered into between the Obligors and a Lender except for a net amount available after the termination of all Derivatives entered into between the Obligors and that Lender and the set-off of resulting amounts owing by the Obligors, or (iv) any payment to which a Lender is entitled as a result of any form of credit protection obtained by that Lender.
- (2) The Obligors consent to the foregoing and agree, to the extent they may effectively do so under Applicable Law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against each Obligor rights of set-off and counterclaim and similar rights of Lenders with respect to that participation as fully as if the Lender were a direct creditor of each Obligor in the amount of the participation.

- 48 -

ARTICLE 8 THE AGENT AND THE LENDERS

8.1 Authorization of Agent

(1) Each of the Lenders hereby irrevocably appoints The Bank of Nova Scotia as the Agent to act on its behalf as the Agent under the Loan Documents and authorizes the Agent to take the actions on its behalf and to exercise the powers that are delegated to the Agent by the terms of the Loan Documents, together with actions and powers that are reasonably incidental to them.

(2) Without limiting section 8.1(1), each Lender grants to the Agent:

- a power of attorney for the purposes of Applicable Laws in respect of the Security, to sign documents comprising the Security from time to time as the party accepting the grant of the Security; and
 - (b) the right to delegate its authority as attorney to any other Person, whether or not an officer or employee of the Agent.

8.2 Rights as a Lender

The Person serving as the Agent shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Agent in its individual capacity. That Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with any Obligor or any Affiliate of an Obligor as if the Person were not the Agent and without any duty to account to the Lenders.

8.3 Exculpatory Provisions

- (1) The Agent shall not have any duties or obligations except those expressly specified in the Loan Documents. Without limiting the generality of the foregoing, the Agent:
 - (a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing;
 - (b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated by the Loan Documents that the Agent is required to exercise as directed in writing by the Required Lenders (or another number or percentage of the Lenders that is expressly provided for in the Loan Documents), but the Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Agent to liability or that is contrary to any Loan Document or Applicable Law; and

- (c) shall not, except as expressly specified in the Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrowers or any of their Affiliates that is communicated to or obtained by the person serving as the Agent or any of its Affiliates in any capacity.
- (2) The Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or another number or percentage of the Lenders that is necessary, or that the Agent believes in good faith is necessary, under the provisions of the Loan Documents) or (ii) in the absence of its own gross negligence or wilful misconduct. The Agent shall be deemed not to have knowledge of any Default unless and until notice describing the Default is given to the Agent by the Borrowers or a Lender.
- (3) Except as otherwise expressly specified in this Agreement, the Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with any Loan Document, (ii) the contents of any certificate, report or other document delivered under or in connection with any Loan Document, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions in any Loan Document or the occurrence of any Default, (iv) the validity, enforceability, effectiveness or genuineness of any Loan Document, on any other agreement, instrument or document or (v) the satisfaction of any condition specified in any Loan Document, other than to confirm receipt of items expressly required to be delivered to the Agent

8.4 Reliance by Agent

The Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying on it. In determining compliance with any condition to the making of an Advance, including the issuance of an L/C, that by its terms must be fulfilled to the satisfaction of a Lender, the Agent may presume that the condition is satisfactory to the Lender unless the Agent has received notice to the contrary from the Lender before the making of the Advance or the issuance of the L/C. The Agent may consult with legal counsel (who may be counsel for the Borrowers), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with their advice.

8.5 Delegation of Duties

The Agent may perform any and all of its duties and exercise its rights and powers under any Loan Document by or through any one or more sub-agents appointed by the Agent from among the Lenders (including the Person serving as Agent) and their respective Affiliates. The Agent and any sub-agent may perform any and all of its duties and

- 50 -

exercise its rights and powers by or through their respective Related Parties. The provisions of this Article and other provisions of this Agreement for the benefit of the Agent shall apply to any sub-agent and to the Related Parties of the Agent and any sub-agent, and shall apply to their respective activities in connection with the syndication of the Credits as well as activities as Agent.

8.6 Administration of the Credits

- (1) Any reference in this Agreement to the Agent means, where the Agent is also a Lender, the agency department of that Lender specifically responsible for acting as Agent under and in connection with this Agreement. In acting as Agent, the agency department will be treated as a separate entity from any other department or division of the Lender in question. Without limiting the foregoing, the Agent shall not be deemed to have notice of a document or information received by any other department or division of that Lender, nor will that Lender be deemed to have notice of a document or information received by the Agent.
- (2) Unless otherwise specified in this Agreement, the Agent shall perform the following duties:
 - (a) take delivery of each Lender's Applicable Percentage of an Advance and make all Advances hereunder in accordance with the procedures set forth in Article 9;
 - (b) use reasonable efforts to collect promptly all sums due and payable by the Borrowers pursuant to this Agreement;
 - (c) make all payments to the Lenders in accordance with the provisions hereof;
 - (d) hold the Security on behalf of the Lenders;
 - (e) hold all legal documents relating to the Credit, maintain complete and accurate records showing all Advances made by the Lenders, all remittances and payments made by the Agent to the Lenders and all fees or any other sums received by the Agent and, except for accounts, records and documents relating to the fees payable by the Borrowers to the Agent in its capacity as Agent under the Agency Fee Letter or the Arrangers under the Fee Letter, allow each Lender and their respective advisors to examine such accounts, records and documents at their own expense, and provide any Lender, upon reasonable notice, with such copies thereof or information contained therein as such Lender may reasonably require from time to time at the Lender's expense;
 - (f) except as otherwise specifically provided for in this Agreement, promptly advise each Lender upon receipt of each notice and deliver to each Lender, promptly upon receipt, all other written communications furnished by any Obligor to the Agent on behalf of the Lenders pursuant to this Agreement, including copies of financial reports and certificates which are to be furnished to the Agent; and

- 51 -

- (g) upon learning of same, promptly advise each Lender in writing of the occurrence of an Event of Default or Pending Event of Default or the occurrence of any event, condition or circumstance which would or could reasonably be expected to have a Material Adverse Effect or of any material adverse information coming to the attention of the Agent (using reasonable efforts) relative to the Security or of the occurrence of any material adverse change in the financial condition or property of any Obligor, provided that, except as aforesaid, the Agent shall be under no duty or obligation whatsoever to provide any notice to the Lenders and further provided that each Lender hereby agrees to notify the Agent of Default of which it may become aware.
- (3) The Agent may take the following actions only with the prior consent of the Required Lenders, subject to Sections 8.6(4), 8.6(5) and 8.6(6) or unless otherwise expressly specified in this Agreement:
 - (a) subject to Section 9.7(4), exercise any and all rights of approval conferred upon the Lenders by this Agreement;
 - (b) give written notice to any Obligor in respect of any matter in respect of which notice may be required, permitted, necessary or desirable in accordance with or pursuant to this Agreement, promptly after receiving the consent of the Required Lenders, except that the Agent shall, without direction from the Lenders, immediately give the Borrowers notice of any payment that is due or overdue under the terms of this Agreement unless the Agent considers that it should request the direction of the Required Lenders, in which case the Agent shall promptly request that direction;
 - (c) amend, modify or waive any of the terms of this Agreement, including waiver of an Event of Default or Pending Event of Default, if such action is not otherwise provided for in Section 9.7(4);
 - (d) declare an Event of Default or take, or cause to be taken by the Agent, action to enforce performance of the Obligations and to realize upon the Security, including the appointment of a receiver, the exercise of powers of distress, lease or sale given by the Security or by law and the taking of foreclosure proceedings and/or the pursuit of any other legal remedy necessary;
 - (e) decide to accelerate the amounts outstanding under the Credit; and
 - (f) pay, or instruct the Agent to pay insurance premiums, Taxes and any other sums as may be reasonably required to protect the interests of the Lenders.
- (4) The Agent may take the following actions only with the unanimous consent of the Lenders that are directly affected, unless otherwise specified in this Agreement:
 - (a) amend, modify, discharge, terminate or waive any of the terms of the Security;

- 52 -

- (b) amend, modify, discharge, terminate or waive any of the terms of this Agreement or the Security if such amendment, modification, discharge, termination or waiver would increase the amount of the Credit, amend the purpose of the Credit, reduce the interest rates and similar charges applicable to the Credit, reduce the fees payable with respect to the Credit, extend any date fixed for payment of principal, interest or any other amount relating to the Credit or extend the term of the Credit;
- (c) waive payment of any amounts outstanding under the Credit; and
- (d) amend the definition of "Required Lenders" or this Section 9.7(4).
- (5) Except as expressly contemplated in this Agreement, no Lender's Commitment or Applicable Percentage may be amended without the consent of that Lender. In addition, no amendment, modification or waiver affecting the rights or obligations of the Agent, the Issuing Bank or the Swing Line Lender may be made without their respective consents and no amendment, modification or waiver affecting rights or obligations under the Loan Documents relating to Other Guaranteed Obligations may be made without the consent of the Lenders who are or whose Affiliates are affected as holders of Other Guaranteed Obligations. No amendment to the definition of "Defaulting Lender" or Section 8.8 may be made without the consent of the Agent, the Issuing Bank and the Swing Line Lender. An amendment, modification or waiver affecting the Agency Fee Letter shall be made only by the parties to it.
- (6) Notwithstanding Sections 8.6(3), 8.6(4) and 8.6(5), the Agent may, without the consent of any Lender, make amendments to the Loan Documents that are for the sole purpose of curing any ambiguity, defect or inconsistency that is clerical or not Material, but shall immediately notify the Lenders if it does so. The Agent may also, without the consent of any Lender, discharge any Security and/or release any Obligor other than the Borrowers to the extent necessary to allow completion of any sale or other disposition of Property (including Equity Interests of an Obligor) or any corporate reorganization permitted by this Agreement or in respect of which the Lenders or Required Lenders have, as applicable, consented or waived the requirements of this Agreement.
- (7) As between the Obligors, on the one hand, and the Agent and the Lenders, on the other hand:
 - (a) all statements, certificates, consents and other documents that the Agent purports to deliver on behalf of the Lenders or the Required Lenders shall be binding on each Lender, and the Obligors shall not be required to ascertain or confirm the authority of the Agent in delivering the documents;
 - (b) all certificates, statements, notices and other documents that are delivered by the Obligors to the Agent in accordance with this Agreement shall be deemed to have been duly delivered to each Lender; and
 - (c) all payments that are delivered by the Borrowers to the Agent in accordance with this Agreement shall be deemed to have been received by each Lender entitled to payment.

- 53 -

(8) Except in its own right as a Lender, the Agent shall not be required to advance its own funds for any purpose, and in particular, shall not be required to pay with its own funds insurance premiums, taxes or public utility charges or the cost of repairs or maintenance with respect to the Property that is the subject matter of the Security, nor shall it be required to pay with its own funds the fees of solicitors, counsel, auditors, experts or agents engaged by it as permitted by the Loan Documents.

8.7 Acknowledgments, Representations and Covenants of Lenders

- (1) Each Lender represents and warrants that it has full power and authority, and has taken all action necessary, to execute and deliver this Agreement, to consummate the transactions contemplated by this Agreement and to be a Lender.
- (2) Each Lender acknowledges and confirms that if the Agent does not receive payment in accordance with this Agreement, it shall not be the Agent's obligation to maintain the Credits in good standing, nor shall any Lender have recourse to the Agent in respect of any amounts owing to the Lender under this Agreement.
- (3) Each Lender acknowledges and agrees that its obligation to advance its Applicable Percentage of Advances in accordance with the terms of this Agreement is independent and in no way related to the obligation of any other Lender.
- (4) Each Lender acknowledges receipt of a copy of this Agreement, the Security (to the extent that the Security has been delivered) that exists at the date of this Agreement, and acknowledges that it is satisfied with the form and content of those documents.

8.8 Defaulting Lenders

- (1) Notwithstanding any other provision of this Agreement, if any Lender becomes a Defaulting Lender, then the provisions of this Section 8.8 shall apply until the Agent, the Borrowers, the Issuing Bank and the Swing Line Lender all agree that the Defaulting Lender has remedied all matters that caused it to be a Defaulting Lender.
- (2) Any Standby Fee shall cease to accrue on the Defaulting Lender's unadvanced portion of any Credit.
- (3) The Defaulting Lender shall not be entitled to exercise any right of consent under Sections 8.6(3) and 8.6(4) and its Commitment shall not be included in determining whether the Lenders or the Required Lenders have provided any consent under those Sections. However, the Defaulting Lender shall be entitled to exercise its right of consent in respect of any matter that expressly requires its consent and any matter that requires the consent of all Lenders if it would be affected differently than the other Lenders.
- (4) If the Defaulting Lender is a Revolving Lender, the Borrowers' right to receive Advances of the Defaulting Lender's unadvanced Commitment under the Revolving Credit shall be suspended and, provided that no Event of Default or Pending Event of Default has occurred and is continuing, the participation of the other Revolving Lenders in Advances under the Revolving Credit shall be re-adjusted without regard to the unadvanced

- 54 -

Commitment of the Defaulting Lender but without increasing the overall Commitments of the other Revolving Lenders to the Revolving Credit. If the unadvanced Commitments of the other Revolving Lenders would not be sufficient to cover their obligations and the obligations of the Defaulting Lender under Section 9.1 then the Borrowers shall repay Advances under the Swing Line Tranche and/or provide cash Collateral to the Issuing Bank to secure L/Cs to the extent necessary to cover the deficiency.

- (5) If the Borrowers provide cash Collateral to the Issuing Bank to secure L/Cs, the Borrowers shall not be required to pay an L/C Fee for the account of the Defaulting Lender in respect of the amount for which it has provided cash Collateral. If the obligation of the Defaulting Lender regarding L/Cs under Section 9.1 is borne by the other Revolving Lenders as a result of Section 8.8(4), then the other Revolving Lenders shall be entitled to receive any L/C Fee that would otherwise have been payable to the Defaulting Lender.
- (6) The Agent may, without prejudice to the other rights of the Lenders, make adjustments to the payments to a Defaulting Lender under this Agreement as necessary to compensate the other Lenders and the Agent for the Defaulting Lender's failure to make any payment or fulfill any other obligation under this Agreement.

8.9 Reference Lenders

- (1) If more than one Lender is a bank named on Schedule I of the *Bank Act* (Canada), the Agent shall be a Schedule I Reference Lender and the Borrowers shall designate a different Lender named on Schedule I to be a Schedule I Reference Lender for the purpose of providing quotations to the Agent to be used in determining rates as required by this Agreement.
- (2) If any Schedule I Reference Lender ceases to be a Lender, the person that originally designated that Schedule I Reference Lender shall have the right to designate in a timely manner another Lender that is a named on Schedule I of the *Bank Act* (Canada) as a replacement Reference Lender, failing which the applicable rate shall be determined on the basis of the quotation provided by the notice from the remaining Schedule I Reference Lender.
- (3) If only one Lender is a bank named on Schedule I of the Bank Act (Canada), that Lender shall be deemed to be the Schedule I Reference Lender and any applicable rate shall be determined on the basis of the quotation provided by that Lender.

8.10 No Other Duties etc.

Anything herein to the contrary notwithstanding, none of the Joint Bookrunners, Joint Lead Arrangers or Co-Syndication Agents listed on the cover page hereof shall have any powers, duties or responsibilities under this Agreement or any of the other Loan Documents, except in its capacity, as applicable, as the Agent, a Lender or an Issuing Bank hereunder.

- 55 -

8.11 Provisions Operative Between Lenders and Agent Only

Except for Sections 8.6(7), 8.7(1), 8.7(3), 8.8, 8.9 and this Section 8.11, the provisions of this Article relating to the rights and obligations of the Lenders and the Agent *inter se* shall be operative as between the Lenders and the Agent only, and the Obligors shall not have any rights under or be entitled to rely for any purpose on those provisions.

ARTICLE 9 DETAILS REGARDING ADVANCES, PAYMENTS, INTEREST AND FEES

9.1 Lenders' Obligations Relating to L/Cs and Revolving Credit

- (1) Notwithstanding that L/Cs are issued by the Issuing Bank, it is the intention of the Parties that the ultimate credit risk and exposure of any Revolving Lender be in accordance with its overall Applicable Percentage of the Revolving Credit. Each Revolving Lender shall immediately indemnify the Issuing Bank for that Revolving Lender's Applicable Percentage of any payment made by the Issuing Bank in respect of an L/C for which the Issuing Bank is not immediately reimbursed by the Borrowers, and shall do all things, including purchases of participations in Advances made by the Issuing Bank, that are necessary to ensure that result. Any such action on the part of the Lenders shall be binding on the Borrowers.
- (2) Each Lender acknowledges and agrees that its obligations under Section 9.1(1) in respect of L/Cs and under Sections 2.4(3) and 2.4(4) in respect of Advances under the Swing Line Tranche are absolute, unconditional and irrevocable and shall not be affected by any circumstance whatsoever, including any amendment, renewal or extension of any L/C, the occurrence and continuance of a Default or the reduction or termination of its Commitment, and that any payment it is required to make pursuant to its obligations shall be made without any offset, abatement, withholding or reduction whatsoever. If for any reason a payment to the Swing Line Lender under Sections 2.4(4) or 2.4(1) in respect of Advances under the Swing Line Tranche cannot be made or must be rescinded, each Lender shall immediately indemnify the Swing Line Lender for that Lender's Applicable Percentage of any such payment, and shall do all such things, including purchase of participations in Advances made by the Swing Line Lender, as shall be required to ensure that result.

9.2 Adjustment of Applicable Percentages for Revolving Credit

While the Swing Line Lender is the sole Lender making Advances under the Swing Line Tranche, its participation in Advances and payments (including standby fees) under the Revolving Credit shall be reduced and the participation of the other Revolving Lenders under the Revolving Credit shall be correspondingly increased as specified in Schedule E. The Agent may revise those adjustments from time to time, having regard to the amounts outstanding or committed under the Swing Line Tranche and the Main Tranche, so that the aggregate of the Advances made by the Swing Line Lender under the Revolving Credit reflects as closely as reasonably possible its overall Applicable Percentage of the Revolving Credit. The Agent shall amend Schedule E and distribute it

- 56 -

to all Parties to reflect the adjustment to the Applicable Percentages. The Agent may also, and shall at the request of the Swing Line Lender, require the Borrowers to obtain Advances under the Revolving Credit and use the proceeds to repay the Swing Line Tranche, if doing so would result in the aggregate of the Advances made by the Swing Line Lender under the Revolving Credit reflecting more closely its overall Applicable Percentage of the Revolving Credit.

9.3 Exceptions Regarding the Revolving Credit

- (1) All Advances under the Swing Line Tranche shall be made solely by the Swing Line Lender and records concerning Advances under the Swing Line Tranche shall be maintained solely by the Swing Line Lender. All payments of principal, interest, fees and other amounts relating to the Swing Line Tranche shall be made solely to the Swing Line Lender. Any notices by the Borrowers in connection with the Swing Line Tranche shall be made to the Swing Line Lender. Notice and minimum amount requirements for Advances shall not apply to Advances by way of overdraft under the Swing Line Tranche.
- (2) In connection with Advances by way of overdraft under the Swing Line Tranche, the Swing Line Lender shall, at the close of business daily, ascertain the position or net position of the account or accounts agreed on by the Borrowers and the Swing Line Lender from time to time. If a position or net position is a debit in favour of the Swing Line Lender and if the Borrowers are entitled to an Advance, the debit shall be deemed to be a Prime Rate Advance (in the case of Canadian Dollars) or Base Rate Advance (in the case of US Dollars) under the Swing Line Tranche in the amount of the debit. If a position or net position is a credit in favour of the Borrowers, the credit shall be deemed to be a repayment of Prime Rate Advances (in the case of Canadian Dollars) or Base Rate Advances (in the case of US Dollars) under the Swing Line Tranche in the amount of the credit.

9.4 Evidence of Indebtedness

The Agent shall maintain records concerning the Obligations and each Lender shall maintain records concerning those Advances it has made. The records maintained by the Agent, by the Issuing Bank relating to L/Cs and by the Swing Line Lender relating to the Swing Line Tranche, shall constitute *prima facie* evidence of the Obligations and all related details. The failure of the Agent or any Lender to correctly record any detail relating to an Advance shall not, however, adversely affect the obligation of the Borrowers to pay any of the Obligations in accordance with this Agreement.

Any Lender may request that any Advances made by it be evidenced by a promissory note. In such event, the Borrowers shall prepare, execute and deliver to such Lender a promissory note payable to the order of such Lender (or, if requested by such Lender, to such Lender and its registered assigns) and in a form approved by the Agent. Thereafter, the Advances evidenced by such promissory note and interest thereon shall at all times (including after assignment in accordance with this Agreement) be represented by one or more promissory notes in such form payable to the order of the payee named therein (or, if such promissory note is a registered note, to such payee and its registered assigns).

- 57 -

9.5 Calculation and Other Matters Regarding Interest and Fees

- (1) All interest on Prime Rate Advances, Base Rate Advances and LIBOR Advances shall accrue from day to day and shall be payable in arrears, calculated on the actual number of days elapsed from and including the date of Advance or the previous date on which interest was due in accordance with Section 2.5, as the case may be, to but excluding the date on which interest is due. If interest is not paid on the date it is due, the principal amount shall continue to bear interest at the rate that is applicable to the particular type of Advance from time to time in accordance with Section 2.5(1), both before and after maturity, default and judgment, and overdue interest shall bear interest at the same rate, compounded monthly, and be payable on demand. Notwithstanding the immediately preceding sentence, upon the expiry of the LIBOR Period applicable to any LIBOR Advance, the principal amount and any overdue interest with respect to that LIBOR Advance shall bear interest calculated at the rates applicable to Base Rate Advances.
- (2) Interest and fees shall be calculated on the basis of a calendar year unless otherwise specified. Interest calculated with reference to LIBOR shall be calculated on the basis of a year of 360 days and the B/A Discount Rate shall be calculated on the basis of a year of 365 days. Any rate that is calculated with reference to a period (the "deemed interest period") that is less than the actual number of days in the calendar year of calculation is, for the purposes of the *Interest Act* (Canada), equivalent to a rate based on a calendar year calculated by multiplying that rate of interest by the actual number of days in the calendar year of calculation and dividing by the number of days in the calendar year calculated in the satis of a feet with reference to a period. All calculations of interest and fees under the Loan Documents shall be made on the basis of the nominal rates described in this Agreement and not on the basis of effective yearly rates or on any other basis that gives effect to the principle of deemed reinvestment. The Parties acknowledge that there is a material difference between the stated nominal rates and effective yearly rates taking into account reinvestment, and that they are capable of making the calculations required to determine effective yearly rates.
- (3) In this Section 9.5, any reference to a "calendar year" means the calendar year in which the period for which the calculation in question falls. If the period falls in two calendar years, one of which is a leap year, the calculation shall be done separately for the parts of the period that fall in each calendar year and the calculated amounts for each period shall be added.
- (4) The B/A Fee for a B/A is calculated by multiplying the face amount of the B/A by the rate for calculation of the B/A Fee specified in Section 2.5, and multiplying the result by a fraction, the numerator of which is the term of the B/A and the denominator of which is the number of days in the calendar year.
- (5) The B/A Discount Proceeds for a B/A are equal to the face amount of that B/A multiplied by the price and shall be rounded to the nearest full cent, with one-half of one cent being rounded up. The price is calculated by dividing one by the sum of one plus the product

- 58 -

of (i) the B/A Discount Rate applicable to that B/A expressed as a decimal fraction, multiplied by (ii) a fraction, the numerator of which is the term in days of that B/A and the denominator of which is 365. The price shall be rounded to the nearest multiple of 0.001%.

- (6) The L/C Fee for an L/C is calculated by multiplying the face amount of the L/C by the rate for calculation of the L/C Fee specified in Section 2.5, and multiplying the result by a fraction, the numerator of which is the number of days elapsed in the period from and including the issuance or renewal of the L/C or the previous date on which the L/C Fee was due, as the case may be, to but excluding the date on which the L/C Fee is due or the L/C is drawn or cancelled, as the case may be, and the denominator of which is the number of days in the calculated in the same manner using the rate specified in Section 2.5(3).
- (7) The standby fee shall be calculated daily on the undrawn amount of the applicable Credit at the rate for calculation of the standby fee specified in Section 2.5, beginning on the date of this Agreement, and each payment shall cover the period from and including the date of this Agreement or the previous date on which the standby fee was due in accordance with Section 2.5, as the case may be, to but excluding the date on which the standby fee is due.
- (8) If an Obligor fails to pay when due any amount payable under any Loan Document for which interest is not otherwise provided in this Agreement or another relevant Loan Document, that Obligor shall, on demand, pay interest on the overdue amount to the Agent from and including the due date up to but excluding the date of actual payment, both before and after demand, default or judgment, at the rate of interest determined from time to time in accordance with Section 2.5(1) that is applicable to Prime Rate Advances (if the amount is denominated in Canadian Dollars) or Base Rate Advances (if the amount is denominated in US Dollars), in each case compounded monthly.
- (9) If the Borrowers deposit cash as cash Collateral pursuant to a requirement under this Agreement, the Agent or the Lender or Lenders holding the cash shall pay the Borrowers interest on the cash while it continues to be held as cash Collateral at the rate offered by the Person holding the cash from time to time for deposits in the relevant currency of comparable size and term.
- (10) The Parties intend to comply with Applicable Law relating to usury. Notwithstanding any other provision of this Agreement or any other Loan Document, in no event shall any Loan Document require the payment or permit the collection of interest or other amounts in an amount or at a rate in excess of the amount or rate that is permitted by Applicable Law or in an amount or at a rate that would result in the receipt by the Lenders or the Agent of interest at a criminal rate, as the terms "interest" and "criminal rate" are defined under the *Criminal Code* (Canada). Where more than one Applicable Law applies to any Obligor, that Obligor shall not be obliged to make payment in an amount or at a rate at elevest permitted by Applicable Law applies to any Obligor, that Obligor shall not be obliged to make payment or any provision of any Loan Document would result in exceeding the highest rate or amount permitted by Applicable Law for the collection or charging of interest, the

- 59 -

(2)

obligation to be fulfilled shall be reduced to reflect the highest permitted rate or amount. If from any circumstance the Agent or the Lenders shall ever receive anything of value as interest or deemed interest under any Loan Document that would result in exceeding the highest lawful rate or amount of interest permitted by Applicable Law, the amount that would be excessive interest shall be applied to the reduction of the principal amount of the relevant Credit, and not to the payment of interest, or if the excessive interest exceeds the unpaid principal balance of the relevant Credit, the amount exceeding the unpaid balance shall be refineded to the Borrowers. In determining whether or not the interest pay due and ray specified contingency exceeds the highest lawful rate, the Obligors, the Agent and the Lenders shall, to the maximum extent permitted by Applicable Law, (i) characterize any non-principal payment as an expense, fee or premium rather than as interest, (ii) exclude voluntary prepayments and their effects, (iii) amortize, prorate, allocate interest between portions of the Obligations to the applicable Law, and/or (iv) allocate interest between portions of the Obligations to the end that no portion shall bear interest at a rate greater than that permitted by Applicable Law, the determined in accordance with generally accepted actuarial practices and principles and if there is any dispute, the determination of a Fellow of the Canadian Institute of Actuaries appointed by the Agent shall be conclusive.

9.6 Conversions, Rollovers, Renewals, Repayments and Reductions

(1) Subject to the other terms of this Agreement, the Borrowers may from time to time:

- (a) convert all or any part of the outstanding amount of any LIBOR Advance into a Base Rate Advance in the same principal amount, or vice versa;
- (b) convert all or any part of the outstanding amount of any Advance by way of B/As into a Prime Rate Advance in the same principal amount, or vice versa;
- (c) rollover all or any part of the outstanding amount of any LIBOR Advance at the end of the LIBOR Period as a new LIBOR Advance or rollover all or any part of the outstanding amount of any Advance by way of B/As as a new Advance by way of B/As;
- (d) renew an L/C at its maturity date by extending the maturity date; and
- (e) in circumstances not mentioned in items (a) to (d) immediately above, concurrently repay one Advance and obtain a different type of Advance.
- Subject to giving notice required by Section 9.7, the Borrowers may from time to time repay Advances outstanding under the Revolving Credit without penalty, except that:
 - (a) The Borrowers must prepay all Lenders under the Revolving Credit pro rata and must prepay the same Advance made by each applicable Lender.

- 60 -

- (b) LIBOR Advances may not be paid before the end of the applicable LIBOR Periods unless the Borrowers indemnify the relevant Lenders for any loss or expense that the Lenders incur as a result, including any breakage costs.
- (c) No B/A may be paid before its maturity date. The Borrowers may provide Cash Collateral to the Lenders that have accepted B/As in an amount equal to the face amount of any or all outstanding B/As if the Borrowers concurrently repay all corresponding BA Equivalent Loans.

No voluntary payment by the Borrowers shall affect their obligation to make all other principal payments required by this Agreement in full when due, until the Revolving Credit is repaid in full and cancelled.

- (3) The Borrower may from time to time cancel the Revolving Credit, or permanently reduce the committed amount of the Revolving Credit by an amount that must be a minimum of \$5,000,000 and a whole multiple of \$1,000,000. The Borrower shall do so by giving not less than three Business Days' irrevocable notice to the Agent and paying all accrued and unpaid standby fees on the amount cancelled or reduced to the effective date of cancellation or reduction. Any reduction shall be applied *pro rata* to the respective Commitments of the Lenders under the Revolving Credit. The Borrower shall have no right to any reinstatement of any previously committed amount of the Revolving Credit after any cancellation or reduction.
- (4) Subject to giving notice required by Section 9.7, the Borrowers may from time to time repay Advances outstanding under the Term Credit without penalty, except that:
 - (a) The Borrowers must prepay all Lenders under the Term Credit pro rata and must prepay the same Advance made by each applicable Lender.
 - (b) LIBOR Advances may not be paid before the end of the applicable LIBOR Periods unless the Borrowers indemnify the relevant Lenders for any loss or expense that the Lenders incur as a result, including any breakage costs.
 - (c) No B/A may be paid before its maturity date. The Borrowers may provide Cash Collateral to the Lenders that have accepted B/As in an amount equal to the face amount of any or all outstanding B/As if the Borrowers concurrently repay all corresponding BA Equivalent Loans.

No voluntary payment by the Borrowers shall affect their obligation to make all other principal payments required by this Agreement in full when due, until the Term Credit is repaid in full and cancelled.

9.7 Notice of Advances and Payments

(1) The Borrowers shall give the Agent irrevocable notice, in the form attached as Schedule A, of any request for any Advance to it. The Borrowers shall also give the Agent irrevocable notice in the same form of any payment by it of any Advance (whether resulting from repayment, prepayment, rollover or conversion).

- 61 -

- (2) Notice for B/As shall be given not later than the second Business Day before the Advance Date or date of payment. Notice for LIBOR Advances shall be given not later than the third Business Day before the Advance Date or date of payment. Notice for an Advance by way of L/C shall be given not later than the third Business Day before the Advance Date or at such earlier time as the Issuing Bank may reasonably require so that it has sufficient time to review the proposed form of L/C. Notice for a Prime Rate Advance or Base Rate Advance shall be given on or before the Business Day before the Advance base or date of payment.
- (3) Notices shall be given not later than 11:00 a.m. on the date for notice. Payments (except those being made solely from the proceeds of rollovers and conversions) must be made before 11 a.m. on the date for payment. If a notice or payment is not given or made by those times, it shall be deemed to have been given or made on the next Business Day, unless all Lenders affected by the late notice or payment agree, in their sole discretion, to accept a notice or payment at a later time as being effective on the date it is given or made.
- (4) Without limiting section 9 of the Provisions, the Borrowers shall indemnify the Agent and the Lenders for all costs that they incur if the Borrowers give notice requesting an Advance or notice of a payment and subsequently fails to complete the Advance or payment or the conditions of the Advance are not satisfied before the time specified in Section 9.10(5) on the proposed Advance Date.

9.8 Size and Term of Advances

- (1) Each Prime Rate Advance or Base Rate Advance shall be in an aggregate minimum amount of \$1,000,000 or US \$1,000,000, respectively and in a whole multiple of \$100,000 or US \$100,000, respectively.
- (2) Each Advance of B/As shall be in an aggregate minimum amount of \$5,000,000 and in a whole multiple of \$1,000. In its notice requesting an Advance of B/As, the Borrowers shall select a term of one, two, three or six months to apply to the Advance.
- (3) Each LIBOR Advance shall be in a minimum amount of US \$5,000,000 and a whole multiple of US \$500,000. In its notice requesting a LIBOR Advance, the Borrowers shall select a LIBOR Period of one, two, three or six months to apply to any particular LIBOR Advance.
- (4) Terms of B/As and LIBOR Periods of lengths other than those specified in Sections 9.8(2) and 9.8(3) shall also be available at the discretion of the Lenders from time to time and the Agent may, in circumstances of market disruption or illiquidity, restrict the term or maturity dates of B/As and/or LIBOR Advances. There shall not at any time be B/As and/or LIBOR Advances outstanding with more than six different maturity dates. No B/A may mature and no LIBOR Period may end on a date that is not a Business Day, after the maturity date of the applicable Credit, or after a date on which the applicable Credit is required to be reduced if that would adversely affect the Borrowers' ability to make the reduction.

- 62 -

(5) Each L/C issued under this Agreement shall have a term that is not more than one year after its issuance date or renewal date (which may extend beyond the then-current maturity date of the Credit under which the L/C is issued), but may provide for automatic renewal of its term for successive periods of up to one year each as long as the Issuing Bank has the right to avoid automatic renewal by giving notice to the beneficiary of the L/C before the extension becomes effective. An L/C may otherwise be renewed by the Borrowers subject to complying with the terms of this Agreement applicable to an Advance by way of L/C. On the cancellation of the applicable Credit, the Borrowers shall arrange for all outstanding L/Cs to be returned to the Issuing Bank in an amount sufficient to fully secure all outstanding L/Cs, all L/C Fees and all fronting fees for the remainder of their respective terms, in which case the Cosh Collateral shall be held by the Issuing Bank in place of the Security.

9.9 Payment of B/As, LIBOR Advances and L/Cs

- (1) The Borrowers shall provide for the following by giving notice under Section 9.7 requesting a rollover or conversion if the Borrowers are otherwise entitled to an Advance, or by delivery of payment:
 - (a) payment to the Agent at the Branch of Account of the full face amount of each B/A for value on the date of its maturity;
 - (b) payment to the Agent at the Branch of Account of the amount of each LIBOR Advance for value on the last day of the applicable LIBOR Period; and
 - (c) payment to the Issuing Bank at the office where an L/C was issued of each amount payable to the beneficiary of the L/C for value on the date on which the amount is payable.
- (2) If the Borrowers fail to provide for payment in accordance with Section 9.9(1), and if the Borrowers are otherwise entitled to an Advance but for failing to give notice under Section 9.7 due to inadvertence, the Agent may deem a Prime Rate Advance to have been made in the case of failure to provide for a LIBOR Advance to have been made in the case of failure to provide for a LIBOR Advance or US Dollar L/C, in each case under the Credit under which the B/A, LIBOR Advance or L/C was originally advanced. If the Agent deems an Advance to have been made, the Agent shall immediately give notice to the Borrowers and the Lenders and, in the case of a deemed Advance relating to an L/C, the Lenders shall remit their respective shares of the Advance to the Agent for forwarding to the Issuing Bank.
- (3) If the Borrowers fail to provide for payment in accordance with Section 9.9(1) and if the Agent does not deem an Advance to have been made under Section 9.9(2), then without limiting the Lenders' other rights, the Borrowers shall pay interest on the amount for which it has not provided for payment at a rate of interest per annum equal to the rate applicable to Prime Rate Advances (in the case of a B/A or Canadian Dollar L/C) or Base Rate Advances (in the case of a LIBOR Advance or US Dollar L/C) as determined in

- 63 -

accordance with Section 2.5(1), compounded monthly. Interest shall be calculated from and including the date on which payment was to have been provided for, up to but excluding the date the payment and all interest, both before and after demand, default and judgment, is provided for by the Borrowers.

9.10 Co-ordination of Prime Rate, Base Rate, B/A and LIBOR Advances

- (1) The Agent shall advise each Lender of its receipt of a notice pursuant to Section 9.7 requesting a Prime Rate, Base Rate or LIBOR Advance or Advance of B/As on the day that notice is received from the Borrowers and shall, as soon as possible, advise each Lender of that Lender's share of the Advance. Each Lender's share shall be based on its Applicable Percentage, but if the Agent determines that a Lender's Applicable Percentage would result in its share of an Advance not being a whole multiple of \$1,000 or US \$1,000, as the case may be, the Agent may increase or reduce the amount to be advanced by that Lender in the Agent's sole discretion to the extent necessary to make the amount a whole multiple.
- (2) In advising a Lender of the amount it is to deliver to the Agent in respect of any Advance, the Agent shall allow for deduction by each Lender of the applicable B/A Fee in connection with an Advance by way of B/As and may also net other amounts payable in the same currency by the Borrowers to the Agent for the account of that Lender on the Advance Date.
- (3) The LIBOR Period applicable to a LIBOR Advance and the maturity date for an Advance of B/As shall be identical for each Lender.
- (4) Each Lender shall deliver its share of the Advance to the Agent not later than 11:00 a.m. on the Advance Date, for value on that date
- (5) If the Agent determines that all the conditions precedent to an Advance specified in this Agreement have been met, it shall advance to the Borrowers the amount delivered by each Lender by crediting the Designated Account before 2:00 p.m. (Toronto time) on the Advance Date, but if the conditions precedent to the Advance are not met by 2:00 p.m. (Toronto time) on the Advance Date, but if the conditions precedent to the Advance are not met by 2:00 p.m. (Toronto time) on the Advance Date, the Agent shall return the funds to the Lenders or invest them in an overnight investment as orally instructed by each Lender until the Advance is made. The Agent is not required to advance any amount to the Borrowers that the Agent has not actually received from a Lender.
- (6) Any difference between the actual proceeds of a newly issued B/A and the amount required to pay a maturing B/A that is being rolled over or the amount required to pay a Prime Rate Advance that is being converted to B/As, any difference between the actual proceeds of an Advance and the amount required to repay any Advance that is concurrently being repaid and any difference between the actual proceeds of an Advance and the amount required to the amount required to the amount required to the amount required to be tween the actual proceeds of an Advance and the amount required to the amount required to the Agent to make, shall be paid by the Borrowers to the Agent from its own resources by 11:00 a.m. (Toronto time) on the Advance Date or may be advanced as a Prime Rate Advance or Base Rate Advance under the Swing Line Tranche if the Borrowers are otherwise entitled to an Advance under the Swing Line Tranche.

- 64 -

9.11 Execution of B/As

- (1) To facilitate the acceptance of B/As under this Agreement, the Borrowers appoint each Lender as its attorney to sign and endorse on its behalf, as and when considered necessary by the Lender, an appropriate number of orders in the form prescribed by that Lender.
- (2) Each Lender may, at its option, execute any order in handwriting or by the facsimile or mechanical signature of any of its authorized officers, and the Lenders are authorized to accept or pay, as the case may be, any order of the Borrowers that purports to bear such a signature notwithstanding that the signatory has ceased to be an authorized officer of the Lender. Any such order or B/A shall be as valid as if the individual were an authorized officer at the date of issue of the order or B/A.
- (3) Any order or B/A signed by a Lender as attorney for the Borrowers, whether signed in handwriting or by facsimile or mechanical signature may be dealt with by the Agent or any Lender to all intents and purposes and shall bind the Borrowers as if duly signed and issued by the Borrowers.
- (4) The receipt by the Agent of a notice requesting an Advance by way of B/As shall be each Lender's sufficient authority to execute, and each Lender shall, subject to the terms and conditions of this Agreement, execute orders in accordance with that request and the advice of the Agent given pursuant to Section 9.10. The executed orders shall be deemed to have been presented for acceptance.

9.12 Funding of B/As

- (1) It shall be the responsibility of each Lender to arrange, in accordance with normal market practice, for the sale on each Advance Date of the B/As issued by the Borrowers and to be accepted by that Lender, failing which the provisions of this Agreement relating to Non B/A Lenders shall apply.
- (2) Notwithstanding any other provision of this Agreement, the amount to be transferred by a Lender to the Agent in connection with any B/A accepted by that Lender shall be determined by the B/A Discount Proceeds calculated with respect to the B/A rather than the actual proceeds of any sale of that B/A. Accordingly, in respect of any particular B/A accepted by it, a Lender (a) shall be entitled to retain for its own account the amount, if any, by which any actual proceeds of sale exceed the calculated B/A Discount Proceeds with respect to the B/A, and (b) shall be required to pay out of its own funds the amount, if any, by which the actual proceeds of sale are less than the calculated B/A Discount Proceeds.
- (3) Whenever the Borrowers request an Advance that includes B/As, each Lender that is not permitted by Applicable Law or by customary market practice to accept B/As or for any other reason elects by notice to the Agent from time to time not to do so (a "Non B/A Lender") shall, in lieu of accepting its pro rata amount of B/As, make available to the

- 65 -

Borrowers on the Advance Date a loan (a "B/A Equivalent Loan") in Canadian Dollars and in an amount equal to the B/A Discount Proceeds of the B/As that the Non B/A Lender would otherwise have accepted, less the B/A Fee that would otherwise have been applicable. The B/A Equivalent Loan shall have a term equal to the term of the B/As that the Non B/A Lender would otherwise have accepted. All provisions of this Agreement applicable to B/As and Lenders that accept B/As shall apply *mutatis mutandis* to B/A Equivalent Loans and Non B/A Lenders.

9.13 Other B/A Provisions

- (1) The Borrowers shall not claim from a Lender any days of grace for the payment at maturity of any B/A accepted by the Lender pursuant to this Agreement. The Borrowers waive any defence to payment that might otherwise exist if for any reason a B/A is held at maturity by a Lender in its own right, and the doctrine of merger shall not apply to any B/A that is at any time held by a Lender in its own right.
- (2) Any executed orders to be used as B/As shall be held by a Lender in safekeeping with the same degree of care as if they were the Lender's own Property, and shall be kept at the place at which executed orders are ordinarily held by the Lender.
- (3) The obligations of the Borrowers with respect to B/As under this Agreement shall be unconditional and irrevocable, and shall be paid strictly in accordance with the terms of this Agreement under all circumstances, including the following circumstances:
 - (a) any lack of validity or enforceability of any order accepted by a Lender as a B/A; or
 - (b) the existence of any claim, set-off, defence or other right that the Borrowers may have at any time against the holder of a B/A, a Lender or any other Person or entity, whether in connection with this Agreement or otherwise.
- (4) The Borrowers shall not enter into any agreement or arrangement of any kind with any Person to whom B/As have been delivered by which the Borrowers undertake to replace the B/As on a continuing basis with other B/As, nor shall the Borrowers directly or indirectly take, use or provide B/As as security for loans or advances from any other Person.

9.14 Issuance and Use of L/Cs

(1) A request for an Advance by way of L/C shall be made by the Borrowers in accordance with Section 9.7, except that a copy of the request shall be sent directly to the Issuing Bank. The Agent shall promptly notify the Lenders of the receipt of a request for an L/C, but L/Cs shall only be issued by the Issuing Bank as fronting bank for all Revolving Lenders. A request shall include the details of the L/C to be issued. The Issuing Bank shall promptly notify the Borrowers on the Advance Date or as soon afterwards as the Issuing Bank is satisfied with the form of L/C to be issued.

- 66 -

(2) The Borrowers shall not directly or indirectly use or provide an L/C as security for Debt owing to any other Person.

9.15 Reimbursement Obligation - L/Cs

- (1) Notwithstanding Section 9.9 or any other provision of this Agreement, the Borrowers' obligations to reimburse any payment to a beneficiary of an L/C shall be absolute and unqualified, but without prejudice to the Borrowers' right, after making full reimbursement, to claim damages from the Issuing Bank for matters arising from the Issuing Bank's wilful misconduct or gross negligence. The Borrowers' obligation shall not be reduced by any demand or other request for payment of an L/C (a "Demand") that is paid or acted on in good faith and in conformity with Applicable Laws or applicable commercial customs or practices being invalid, insufficient, fraudulent or forged, nor shall the Borrowers' obligation be subject to any defence or be affected by any right of set-off, counter-claim or recoupment that the Borrowers may have now or in the future against the beneficiary, the Issuing Bank or any other Person for any reason whatsoever, including the fact that a Demand or Demands (if applicable) aggregating up to the amount of the L/C were paid notwithstanding any contrary instructions from the Borrowers and shall not place of any event including the commencement of legal proceedings to prohibit payment of a Demand. Any action, inaction or omission taken or suffered by the Issuing Bank or any other Person under or in connection with an L/C or any Demand, if in good faith and in conformity with Applicable Laws or applicable customs or practices shall be binding on the Borrowers and shall not place the Issuing Bank or any other Person under any resulting liability to the Borrowers. Without limiting the foregoing, the Issuing Bank may receive, accept or pay as complying with the terms of the L/C, any Demand that is otherwise in order that may be signed by, or issued to, any administrator, executor, trustee in bankruptcy, receiver or other person or entity acting as the representative or in place of, the beneficiary. The Borrowers shall not state any steps, issue any other Person to honour or pay any Demand.
- (2) If any L/C remains outstanding after the maturity date or other cancellation of the Revolving Credit, the Borrowers' obligations under Section 9.15(1) shall continue, notwithstanding the delivery of Cash Collateral.

9.16 Failure of Lender to Fund

(1) Unless the Agent has received notice from a Lender before the proposed date of any Advance that the Lender will not make available to the Agent its share of the Advance, the Agent may assume that the Lender has made its share available on that date in accordance with the provisions of this Agreement concerning funding by Lenders and may, in reliance upon that assumption, make available to the Borrowers a corresponding amount. In that event, if a Lender has not in fact made its share of the applicable

- 67 -

Advance available to the Agent, then the applicable Lender shall pay to the Agent immediately on demand that corresponding amount, with interest for each day from and including the date the amount is made available to the Borrowers to but excluding the date of payment to the Agent, at a rate determined by the Agent in accordance with prevailing banking industry practice on interbank compensation. If the Lender pays that amount to the Agent, then the amount shall constitute the Lender's Advance. If the Lender does not do so immediately, the Borrowers shall pay to the Agent immediately on demand the corresponding amount with interest at the interest rate applicable to the Advance in question. Any payment by the Borrowers shall be without prejudice to any claim the Borrowers may have against a Lender that has failed to make its payment to the Agent.

(2) Notwithstanding Section 9.16(1), if any Lender fails to deliver its share of any Advance to the Agent (that Lender being the "Non-Funding Lender"), the Agent shall immediately give notice of that failure by the Non-Funding Lender to the Borrowers and the other Lenders and indicate to the other Lenders that any Lender (individually a "Contributing Lender" and collectively the "Contributing Lenders"), may make available to the Agent all or any portion of the Non-Funding Lender's share of that Advance in place of the Non-Funding Lender, but in no way shall any other Lender or the Agent be obliged to do so. A Contributing Lender shall make funds available to the Agent as soon as possible for delivery by the Agent to the Borrowers. If more than one Contributing Lender gives notice that it is prepared to make funds available and the aggregate of the funds that the Contributing Lender's relative Contributing Lender shall make, then each Contributing Lender shall make available aportion of that Advance by the Advance that the Non-Funding Lender shall make, then each Contributing Lender shall make available aportion of that Advance by the Agent to the Borrowers. The funder shall make shall be available to the Agent as available to the Agent as available aportion of that Advance by the Non-Funding Lender's relative Commitments to the applicable Credit. If a Contributing Lender shall make the place of a Non-Funding Lender shall pay the Contributing Lender, immediately on demand, the amount advanced on its behalf together with interest at the rate applicable to that Advance to the date of payment, and the Non-Funding Lender shall make the Borrowers. The failure of any Lender to deliver is Applicable Percentage of any Advance to the Agent as required shall not relieve any other Lender of its obligation to deliver is Applicable Percentage of any Advance to the Agent as required.

9.17 Payments by the Borrowers

- (1) Except as otherwise specified in this Agreement, all payments made by or on behalf of the Borrowers shall be made to and received by the Agent and shall be distributed by the Agent to the Lenders as soon as possible upon receipt by the Agent. Except as otherwise provided in this Agreement (including Section 9.18), the Agent shall distribute:
 - (a) payments of interest in accordance with each Lender's Applicable Percentage of the relevant Credit;
 - (b) repayments of principal in accordance with each Lender's Applicable Percentage of the relevant Credit; or

- 68 -

- (c) all other payments received by the Agent, including amounts received on the realization of Security, in accordance with each Lender's Applicable Percentage of the relevant Credit, except that no Lender shall receive proceeds of realization in excess of the Obligations owing to it.
- (2) If the Agent does not distribute a Lender's share of a payment made by the Borrowers to that Lender for value on the day that payment is made or deemed to have been made to the Agent, the Agent shall pay to the Lender on demand an amount equal to the product of (a) the Interbank Reference Rate per annum multiplied by (b) the Lender's share of the amount received by the Agent from the Borrowers and not distributed to the Lender, multiplied by (c) a fraction, the numerator of which is the number of days that have elapsed from and including the date or freeeipt of the payment by the Agent to but excluding the date on which the payment is made by the Agent to the Lender and the denominator of which is 365. The Agent shall be entitled to withhold any Tax applicable to any payment as required by Applicable Law.
- (3) The Agent may debit accounts, credits and other balances maintained by the Borrowers from time to time with the Agent or its Affiliates to facilitate or otherwise obtain payment of interest, fees and other Obligations owing by the Borrowers.

9.18 Payments by Agent

For greater certainty, the following provisions shall apply to payments made by the Agent to the Lenders:

- (a) The Agent shall be under no obligation to make any payment to any Lender (whether in respect of principal, interest, fees or otherwise) until an amount in respect of that payment has been received by the Agent from the Borrowers.
- (b) If the Agent receives less than the full amount of any payment of principal, interest, fees or other amount owing by the Borrowers under this Agreement, the Agent shall have no obligation to remit to each Lender any amount except that Lender's Applicable Percentage of the amount actually received by the Agent.
- (c) If any Lender advances more or less than its Applicable Percentage of a Credit, that Lender's entitlement to that payment shall be increased or reduced, as the case may be, in proportion to the amount actually advanced by that Lender.
- (d) Except as specified in any applicable Assignment, if a Lender's share of an Advance has been advanced, or a Lender's Commitment has been outstanding, for less than the full period to which any payment (except a payment of principal) by the Borrowers relates, that Lender's entitlement to that payment shall be reduced in proportion to the length of time the Lender's share of the relevant Advance or the Lender's Commitment, as the case may be, has actually been outstanding.
- (e) The Agent acting reasonably and in good faith shall, after consultation with the Lenders in the case of any dispute, determine in all cases the amount of all payments to which each Lender is entitled and that determination shall, in the absence of manifest error, be binding and conclusive.

- 69 -

On request, the Agent shall deliver a statement detailing any of the payments to the Lenders.

9.19 Increased Costs, Etc.

- If any Change in Law from time to time shall:
 - (a) impose, modify or deem applicable any reserve, special deposit, compulsory loan, insurance charge or similar requirement against assets of, deposits with or for the account of, or credit extended or participated in by, any Lender;
 - (b) subject any Lender to any Tax of any kind whatsoever with respect to this Agreement or any Advance made by it, or change the basis of taxation of payments to the Lender in respect of this Agreement or any Advance, except for Indemnified Taxes or Other Taxes covered by Section 9.20 and the imposition, or any change in the rate, of any Excluded Tax payable by that Lender; or
 - (c) impose on any Lender or any applicable interbank market any other condition, cost or expense affecting this Agreement or any Advance made by that Lender;

and the result of any of the foregoing shall be to increase the cost to the Lender of making or maintaining any Advance (or of maintaining its obligation to make any Advance), or to reduce the amount of any sum received or receivable by the Lender under this Agreement (whether of principal, interest or any other amount), then upon request of the Lender from time to time the Borrowers will pay to that Lender an additional amount or amounts that will compensate the Lender for the additional costs incurred or reduction suffered.

- (2) If any Lender determines in its sole and absolute discretion that any Change in Law affecting the Lender or any lending office of the Lender or its holding company (or other Controlling Person), if any, regarding capital requirements has or would have the effect of reducing the rate of return on the Lender's capital or on the capital of its holding company (or other Controlling Person), if any, as a consequence of this Agreement, the Commitments of the Lender or any Advance made by the Lender, to a level below that which the Lender or its holding company (or other Controlling Person) could have achieved but for that Change in Law (taking into consideration the Lender's policies and the policies of its holding company (or other Controlling Person) with respect to capital adequacy, each from time to time), then from time to time the Borrowers will pay to that Lender an additional amount or amounts that will compensate Lender or its holding company (or other Controlling Person) for the reduction suffered.
- (3) A certificate of a Lender specifying the amount or amounts necessary to compensate the Lender or its holding company (or other Controlling Person), as the case may be, as specified in this Section, including reasonable detail of the basis of calculation of the amount or amounts determined based on methods of averaging and attribution in its sole and absolute discretion, and delivered to the Borrowers shall be conclusive absent manifest error. The Borrowers shall pay the Lender the amount shown as due on any such certificate within 10 days after receipt of the certificate.

- 70 -

(4) Failure or delay on the part of any Lender to demand compensation pursuant to this Section shall not constitute a waiver of the Lender's right to demand compensation, except that the Borrowers shall not be required to compensate a Lender pursuant to this Section for any increased costs incurred or reductions suffered more than nine months before the date that the Lender notifies the Borrowers of the Change in Law giving rise to the increased costs or reductions and of the Lender's intention to claim compensation, unless the Change in Law giving rise to the increased costs or reductions is retroactive, in which case the nine month period referred to above shall be extended to include the period of retroactive effect.

9.20 Taxes

- (1) If any Obligor, the Agent, or any Lender is required by Applicable Law to deduct or pay any Indemnified Taxes (including any Other Taxes) in respect of any payment by or on account of any obligation of an Obligor under any Loan Document, then (i) the sum payable shall be increased by that Obligor when payable as necessary so that after making or allowing for all required deductions and payments (including deductions and payments applicable to additional sums payable under this Section) the Agent or Lender, as the case may be, receives an amount equal to the sum it would have received had no deductions or payments been required, (ii) the Obligor shall make any deductions required to be made by it under Applicable Law and (iii) the Obligor shall timely pay the full amount required to the relevant Governmental Authority in accordance with Applicable Law.
- (2) Without limiting the provisions of Section 9.20(1), the Borrowers shall timely pay any Other Taxes to the relevant Governmental Authority in accordance with Applicable Law.
- (3) The Borrowers shall indemnify the Agent and each Lender, within 10 days after demand, for the full amount of any Indemnified Taxes or Other Taxes (including Indemnified Taxes or Other Taxes) imposed or asserted on or attributable to amounts payable under this Section) paid by the Agent or that Lender and any penalties, interest and reasonable expenses arising from or with respect to the Indemnified Taxes or Other Taxes, whether or not they were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of the payment or liability delivered to the Borrowers by a Lender (with a copy to the Agent), or by the Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.
- (4) As soon as practicable after any payment of Indemnified Taxes or Other Taxes by an Obligor to a Governmental Authority, the Obligor shall deliver to the Agent the original or a certified copy of a receipt issued by the Governmental Authority evidencing payment, a copy of the return reporting the payment or other evidence of the payment reasonably satisfactory to the Agent.

- 71 -

- (5) Any Foreign Lender that is entitled to an exemption from or reduction of withholding tax under the law of the jurisdiction in which each of the Borrowers is resident for tax purposes, or any treaty to which that jurisdiction is a party, with respect to payments under any Loan Document shall, at the request of the Borrowers, deliver to the Borrowers (with a copy to the Agent), at the ime or times prescribed by Applicable Law or reasonably requested by the Borrowers or the Agent, all properly completed and excuted documentation prescribed by Applicable Law or reasonably requested by the Borrowers or the Agent, all properly completed and executed documentation prescribed by Applicable Law or reasonably requested by the Borrowers or the Agent that will permit the payments to be made without withholding or at a reduced rate of withholding. In addition, (a) any Lender, if requested by the Borrowers or the Agent, shall deliver other documentation prescribed by Applicable Law or reasonably requested by the Borrowers or the Agent that will neable the Borrowers or the Agent to determine whether or not that Lender is subject to withholding or information reporting requirements, and (b) any Lender shall notify the Borrowers and the Agent in writing within five Business Days of ceasing to be, or to be deemed to be, resident in Canada for purposes of Part XIII of the Income Tax Act (Canada) or any successor provision.
- (6) If the Agent or a Lender determines, in its sole discretion, that it has received a refund of any Taxes or Other Taxes as to which it has been indemnified by the Borrowers or with respect to which an Obligor has paid additional amounts pursuant to this Section or that, because of the payment of those Taxes or Other Taxes, it has benefited from a reduction in Excluded Taxes otherwise payable by it, it shall pay to the Borrowers or the Obligor, as applicable, an amount equal to the refund or reduction (but only to the extent of indemnity payments made, or additional amounts paid, by the Borrowers or Obligor under this Section with respect to the Taxes giving rise to the refund or reduction), net of all out-of-pocket expenses of the Agent or Lender, as the case may be, and without interest (other than any net after Tax interest paid by the relevant Governmental Authority with respect to the refund). The Borrowers or Obligor, as applicable, upon the request of the Agent or Lender, agrees to repay the amount paid over to the Borrowers or Obligor (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) to the Agent or Lender if the Agent or Lender is required to repay the refund or reduction to the Governmental Authority. This Section shall not be construed to require the Agent or any Lender to make available refund or reduction.

9.21 Mitigation Obligations; Replacement of Lenders

(1) If any Lender requests compensation under Section 9.19, or requires the Borrowers to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 9.20, then the Lender shall use reasonable efforts to designate a different lending office for funding or booking its Advances or to assign its rights and obligations under this Agreement to another of its offices, branches or Affiliates, if, in the judgment of the Lender, doing so (i) would eliminate or reduce amounts payable pursuant to Section 9.19 or 9.20, as the case may be, in the future and (ii) would not subject the Lender to any unreimbursed cost or expense and would not otherwise be disadvantageous to the Lender. Each of the Borrowers hereby agrees to pay all reasonable costs and expenses incurred by any Lender in connection with any such designation or assignment.

- 72 -

- (2) If any Lender requests compensation under Section 9.19, if the Borrowers are required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 9.20, if any Lender's obligations are suspended pursuant to Section 9.22 or if any Lender becomes a Defaulting Lender, then the Borrowers may, at their sole expense and effort, upon 10 days' notice to that Lender and the Agent, require the Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 10.2), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an assignee that will assume those obligations (which assignee may be another Lender, if a Lender accepts the assignment), provided that:
 - (a) in the case of an assignment, the Borrowers pay the Agent the assignment fee specified in Section 10.2(1)(g);
 - (b) the Lender receives payment of an amount equal to the outstanding principal of its Advances, accrued interest, accrued fees and all other amounts payable to it under the Loan Documents (including any breakage costs and amounts required to be paid under this Agreement as a result of prepayment to a Lender) from any assignee and/or the Borrowers;
 - (c) in the case of any assignment resulting from a claim for compensation under Section 9.19 or payments required to be made pursuant to Section 9.20, the assignment will result in a reduction in future compensation or payments; and
 - (d) any assignment does not conflict with Applicable Law.
- (3) A Lender shall not be required to make any such assignment or delegation or accept repayment if, before completion, as a result of a waiver by the Lender or otherwise, the circumstances entitling the Borrowers to require the assignment and delegation or repayment cease to apply.

9.22 Illegality

If any Lender determines that any Applicable Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender or its applicable lending office to make or maintain any Advance (or to maintain its obligation to make any Advance), or to determine or charge interest rates based upon any particular rate, then, on notice by that Lender to the Borrowers through the Agent, any obligation of the Lender with respect to the activity that is unlawful shall be suspended until the Lender notifies the Agent and the Borrowers shall appen demand from that Lender (with a copy to the Agent), prepay or, if conversion would avoid the activity that is unlawful, convert any Advances, or take any necessary steps with respect to any L/C in order to avoid the activity that is unlawful. Upon any such prepayment or conversion, the Borrowers shall also pay accrued interest on the amount so prepaid or converted. Each Lender agrees to designate a different lending office if doing so will avoid the need for that notice and will not, in the good faith judgment of the Lender, otherwise be materially disadvantageous to the Lender.

- 73 -

9.23 Inability to Determine Rates, Etc.

Notwithstanding Section 9.12(1), if the Required Lenders determine that for any reason a market for B/As does not exist at any time, or the Lenders cannot for other reasons, after reasonable efforts, readily sell B/As or perform their other obligations under this Agreement with respect to B/As, the Agent will promptly notify the Borrowers and each Lender. Thereafter, the Borrowers' right to request acceptance of B/As shall be and remain suspended until the Required Lenders determine and the Agent notifies the Borrowers and each Lender that any condition causing the original determination no longer exists. If the Required Lenders determine that for any reason adequate and reasonable means do not exist for determining LIBOR for any requested LIBOR Period with respect to a proposed LIBOR Advance, or that LIBOR for any requested LIBOR Period with respect to a proposed LIBOR Advance does not adequately and fairly reflect the cost to those Lenders of funding that Advance, the Agent will promptly notify the Borrowers and each Lender. Thereafter, the obligation of the Lenders to make or maintain LIBOR Advances shall be suspended until the Agent (upon the instruction of the Required Lenders) revokes the notice. Upon receipt of that notice, the Borrowers may revoke any pending request for a borrowing, conversion or continuation of LIBOR Advances or, failing that, will be deemed to have converted its request into a request for a borrowing of Base Rate Advances in the amount specified in the request.

ARTICLE 10 ADDITIONAL LENDERS, SUCCESSORS AND ASSIGNS

10.1 Successors and Assigns

The provisions of this Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and assigns permitted by this Agreement, except that no Obligor may assign or otherwise transfer any of its rights or obligations under any Loan Document without the prior written consent of the Agent and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations under this Agreement except (i) to an Eligible Assignee in accordance with the provisions of Section 10.2, or (ii) by way of participation in accordance with the provisions of Section 10.4, (and any other attempted assignment or transfer by any Party shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the Parties, their respective successors and permitted assigns, sub-agents contemplated by this Agreement, Participants to the extent provided in Section 10.4 and, to the extent expressly contemplated by this Agreement, the Related Parties of each of the Agent, any sub-agent and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.

- 74 -

10.2 Assignments by Lenders

- (1) Any Lender may at any time assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Advances at the time owing to it), provided that:
 - (a) except if an Event of Default has occurred and is continuing or in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and the Advances at the time owing to it or in the case of an assignment to a Lender or an Affiliate of a Lender, the aggregate amount of the Commitment being assigned (which for this purpose includes Advances outstanding under the Commitment) or, if the applicable Commitment is not then in effect, the principal outstanding balance of the Advances of the assignment (determined as of the date the Assignment with respect to the assignment is delivered to the Agent or, if "Trade Date" is specified in the Assignment, as of the Trade Date) shall not be less than \$5,000,000, in the case of any assignment in respect of the Revolving Credit, or \$1,000,000, in the case of any assignment in respect of the Term Credit, unless each of the Agent and, so long as no Default has occurred and is continuing, the Borrowers otherwise consent to a lower amount (each consent not to be unreasonably withheld or delayed);
 - (b) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement;
 - (c) any assignment of a Commitment relating to a Credit under which L/Cs may be issued must be approved by the Issuing Bank in its sole discretion unless the Person that is the proposed assignee is itself already a Lender with a Commitment under that Credit;
 - (d) any assignment must be approved by the Agent (which is not to be unreasonably withheld or delayed) unless:
 - (i) in the case of an assignment of a Commitment relating to the Revolving Credit, the proposed assignee is itself already a Revolving Lender;
 - (ii) no Event of Default has occurred and is continuing, the assignment is of a Commitment relating to the Term Credit, and the Term Credit is fully advanced; or
 - (iii) the proposed assignee is a bank whose senior, unsecured, non-credit enhanced, long term debt is rated at least A3, A- or A low by at least two of Moody's, S&P and Dominion Bond Rating Service Limited, respectively;
 - (e) any assignment of a Commitment relating to the Revolving Credit must be approved by the Swing Line Lender in its sole discretion;

- 75 -

- (f) any assignment must be approved by the Borrowers (which is not to be unreasonably withheld or delayed) unless the proposed assignee is itself already a Lender with the same type of Commitment or is an Affiliate of the Assignor or an Event of Default or Pending Event of Default has occurred and is continuing; and
- (g) the parties to each assignment shall execute and deliver to the Agent an Assignment, together with a processing and recordation fee of \$3,500 and the Eligible Assignee, if it is not a Lender, shall deliver any administrative questionnaire required by the Agent.
- (2) Subject to acceptance and recording by the Agent pursuant to Section 10.3, from and after the effective date specified in the Assignment, the Eligible Assignee shall be a Party and, to the extent of the interest assigned by the Assignment, have the rights and obligations of a Lender under this Agreement and the other Loan Documents, including the Security, and the assigning Lender's rights and obligations under this Agreement (and, in the case of an Assignment covering all of the assigning Lender's rights and obligations under this Agreement (and, in the case of an Assignment covering all of the assigning Lender's rights and obligations under this Agreement (and, in the case of an Assignment covering all of the assigning Lender's rights and obligations under this Agreement, it shall continue to be liable for any breach by it of this Agreement, with respect to facts and circumstances occurring before the effective date of the assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this Section 10.2, shall be treated for purposes of this Agreement as a sale by that Lender of a participation in those rights and obligations in accordance with Section 10.4. Any payment by an assignee to an assigning Lender in connection with an assignment or transfer shall not be or be deemed to be a repayment by the Borrowers.

10.3 Register

The Agent shall maintain at one of its offices in Toronto, Ontario a copy of the Assignment delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amounts of the Advances owing to, each Lender from time to time (the "Register"). The entries in the Register shall be conclusive, absent manifest error, and the Borrowers, the Agent and the Lenders may treat each Person whose name is recorded in the Register as a Lender for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrowers and any Lender, at any reasonable time and from time to time upon reasonable prior notice.

10.4 Participations

(1) Any Lender may at any time, without the consent of, or notice to, the Borrowers or the Agent, sell participations to any Person (other than a natural person, an Obligor or any Affiliate of an Obligor) (each, a "Participant") in all or a portion of that Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Advances owing to it). However, (i) the Lender's obligations under this Agreement shall remain unchanged, (ii) the Lender shall remain solely responsible to the

- 76 -

other Parties for the performance of its obligations and (iii) the Borrowers, the Agent and the other Lenders shall continue to deal solely and directly with that Lender in connection with its rights and obligations under this Agreement. Any payment by a Participant to a Lender in connection with a sale of a participation shall not be or be deemed to be a repayment by the Borrowers or a new Advance to the Borrowers.

- (2) Subject to Section 10.4(3), each of the Borrowers agrees that each Participant shall be entitled to the benefits of Sections 9.19 and 9.20 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to Section 10.2. To the extent permitted by Applicable Law, each Participant also shall be entitled to the benefits of Section 7.10 as though it were a Lender, provided the Participant agrees to be subject to Section 7.11 as though it were a Lender.
- (3) A Participant shall not be entitled to receive any greater payment under Sections 9.19 and 9.20 than the applicable Lender would have been entitled to receive with respect to the participation sold to that Participant, unless the sale of the participation to the Participant is made with the Borrowers' prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of Section 9.20 unless the Borrowers are notified of the participation sold to that Participant agrees, for the benefit of the Borrowers, to comply with that Section as though it were a Lender.

ARTICLE 11 MISCELLANEOUS PROVISIONS

11.1 Severability, Etc.

If, in any jurisdiction, any provision of any Loan Document or its application to any circumstance is restricted, prohibited or unenforceable, that provision shall, as to that jurisdiction, be ineffective only to the extent of that restriction, prohibition or unenforceability without invalidating the remaining provisions of the affected Loan Document, without affecting the validity or enforceability of that provision in any other jurisdiction and, if applicable, without affecting its application to other circumstances.

11.2 Amendment, Supplement or Waiver

No amendment, supplement or waiver of any provision of any Loan Document, nor any consent to any departure by an Obligor from any provision, shall in any event be effective unless it is in writing, makes express reference to the affected provision and is signed by the Agent and the applicable Lenders, or by the Agent with any approval of any applicable Lenders, all as required by Sections 8.6(3), 8.6(4), 8.6(5) and 8.6(6), as applicable. It shall be effective only in the specific instance and for the specific purpose for which it is given. No waiver or act or omission of the Agent or any Lender shall extend to or be taken in any way to affect any subsequent Default or other matter or their resulting rights.

- 77 -

11.3 Governing Law and Agent for Service

(1) Each of the Loan Documents, except for those that expressly provide otherwise, shall be conclusively deemed to be a contract made under, and shall for all purposes be governed by and construed in accordance with, the laws of the Province of Ontario and the laws of Canada applicable in Ontario. Each Obligor irrevocably and unconditionally submits, for itself and its Property, to the exclusive jurisdiction of the courts of the Province of Ontario and any appellate court from any of those courts, in any action or proceeding arising out of or relating to this Agreement or any other Loan Document, or for recognition or enforcement of any judgment, and each Party irrevocably and unconditionally agrees that all claims in respect of any such action or proceeding may be heard and determined in those courts. Each Party agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other Loan Document shall affect any right that the Agent or any Lender may otherwise have to bring any action or proceeding relating to this Agreement or any other Loan Document shall affect any right that the Agent or any cached by Applicable Law, any objection that it may now or in the future have to the laying of venue of any action or proceeding arising out of or relating to this Agreement or any other Loan Document forum to the maintenance of such a action or proceeding in any such action. Each Party hereby irrevocably avievs, to the fullest extent permitted by Applicable Law, any objection that it may now or in the future have to the laying of venue of any action or proceeding arising out of or relating to this Agreement or any other Province of Ontario. Each Party hereby irrevocably waives, to the fullest extent permitted by Applicable Law, the defence of an inconvenient forum to the maintenance of such an action or proceeding in any such court.

11.4 Due Diligence by Lenders

Each Obligor agrees that none of the Agent or the Lenders shall have any liability to it in relation to any due diligence investigations conducted by any of them in connection with the transactions contemplated by the Loan Documents or be under any obligation whatsoever to disclose to it any information received or facts disclosed by any investigations. Each Obligor also agrees that it is not relying, will not rely, and will not rely, and will not be deemed, in any respect whatsoever, to have relied upon the facts received by and information disclosed to any of the Agent or the Lenders under or in connection with due diligence investigations.

11.5 Currency

All payments made under this Agreement shall be made in the currency in which the obligation requiring payment arose. Unless the context otherwise requires, all amounts expressed in this Agreement in terms of money shall refer to Canadian Dollars. Except as otherwise expressly provided in this Agreement, wherever this Agreement contemplates or requires the calculation of the equivalent in one currency of an amount expressed in another currency for a purpose that does not involve the actual purchase of currency, the calculation shall be made on the basis of the Exchange Rate at the effective time of the calculation.

- 78 -

11.6 Liability of Lenders

The liability of the Lenders in respect of all matters relating to this Agreement and the other Loan Documents is several and not joint or joint and several. Without limiting that statement, the obligations of the Lenders to make Advances are limited to their respective Applicable Percentages of any Advance that is requested, and, in the aggregate, to their respective Applicable Percentages of the total amount of each Credit.

11.7 Currency Indemnity

If a judgment or order is rendered by any court or tribunal for the payment of any amount owing to the Agent or any Lender under any Loan Document or for the payment of damages in respect of any breach of any Loan Document, or under or in respect of a judgment or order of another court or tribunal for the payment of those amounts or damages, and the judgment or order is expressed in a currency ("the **Judgment Currency**") except the currency payable under the relevant Loan Document ("**the Agreed Currency**"), the party against whom the judgment or order is made shall indemnify and hold the Agent and the Lenders harmless against any deficiency in terms of the Agreed Currency for the purposes of the judgment or order, and (b) the actual rate of exchange at which the Agreed Currency is converted into the Judgment Currency for the purposes of the judgment or order, and (b) the actual rate of exchange at which the Agreed Currency with the amounts of the Judgment Currency actually received by the Agent or Lender on the date of receipt. The indemnify in this Section shall constitute a separate and independent obligation from the other obligations of the Obligors under the Loan Documents, shall apply irrespective of any indugence granted by the Agent or any Lender and shall be secured by the Security.

11.8 Expenses and Indemnity

(1) The Borrowers shall pay (i) all reasonable out-of-pocket expenses incurred by the Agent and its Affiliates, including the reasonable fees, charges and disbursements of counsel for the Agent, in connection with the syndication of the Credits, the preparation, negotiation, execution, delivery and administration of the Loan Documents or any amendments, modifications or waivers of their provisions (whether or not the transactions contemplated by them are consummated), (ii) all reasonable out-of-pocket expenses incurred by the Issuing Bank in connection with the sisuance, amendment, renewal or extension of any LC or any demand for payment under an LC and (iii) all reasonable out-of-pocket expenses incurred by the Agent or any Lender, including the reasonable fees, charges and disbursements of counsel, in connection with the enforcement or protection of its rights in connection with the Loan Documents, including its rights under this Section, or in connection with Advances, including those incurred during any workout, restructuring or negotiations in respect of the Obligations but excluding any expenses incurred in connection with the assignment or participation of an interest in the Credits following the initial Advance under this Agreement. Without limiting the foregoing, the Borrowers shall promptly pay all invoices issued by legal counsel to the Agent and the Agent may debit any account maintained by the Borrowers with it to pay invoices.

- 79 -

- (2) The Borrowers shall indemnify the Agent (and its sub-agents), each Lender and each Related Party of any of the foregoing Persons (each of whom is called an "Indemnitee") against, and hold each Indemnites harmless from, any and all losses, claims, damages, liabilities and related expenses, including the fees, charges and disbursements of any counsel for any Indemnitee by any Indemnitee by any Obligor or other Person (except an Indemnitee) arising out of, in connection with, or as a result of (i) the execution or delivery of any Loan Document or any agreement or instrument contemplated by it, the performance or non-performance by the Parties of their respective obligations under any Loan Document or the consummation or non-consummation of the transactions contemplated by the Loan Documents, (ii) any Advance or the use or proposed use of its proceeds (including any refusal by the Essing Bank to honour a demand for payment under an L/C if the documents presented in connection with the demand do not strictly comply with the terms of the L/C), (iii) any actual or alleged presence or release, spill, leakage, emission, deposit, discharge, leaching, migration or disposition of any Hazardous Material on or from any Property owned or operated by any Obligor, or any related remedial action taken by the Agent or Lender or any breach of Applicable Law with respect thereto that is related in any way to any Obligor (including the assertion of any Lien under Applicable Law), or (iv) any actual or prospective claim, liftgation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brough thy a third party or by an Obligor and regardless of whether any Indemnitee is a party to it, provided that the indemnity shall not, as to any Indemnitee, be available to the extent that its losses, claims, damages, liabilities or related expenses (x) are determined by a court of competent jurisdiction by final and non-appealable judgment to have resulted from its gross
- (3) All amounts due under this Section 11.8 shall be payable promptly after demand. A certificate of the Agent or a Lender specifying the amount or amounts owing to the Agent, Lender or a subagent or Related Party, as the case may be, as specified in this Section, including reasonable detail of the basis of calculation of the amount or amounts, and delivered to the Borrowers shall be conclusive absent manifest error.

11.9 Address for Notice

(1) Except in the case of notices and other communications expressly permitted to be given by telephone (and except as provided in Section 11.9(3)), all notices and other communications provided for in this Agreement shall be in writing and shall be delivered by hand or overright courier service, mailed by certified or registered mail or sent by telecopier to the addresses or telecopier numbers specified on Schedule D or, if to a Lender, to it at its address or telecopier number specified in the Register or, if to an Obligor other than the Borrowers, in care of the Borrowers. Those addresses shall apply to notices under all Loan Documents, unless otherwise expressly provided.

- 80 -

- (2) Notices sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received; notices sent by telecopier shall be deemed to have been given when sent (except that, if not given on a Business Day between 9:00 a.m. and 5:00 p.m. local time where the recipient is located, shall be deemed to have been given at 9:00 a.m. on the next Business Day for the recipient). Notices delivered through electronic communications to the extent provided in Section 11.9(3), shall be effective as provided in that Section.
- (3) Notices and other communications to the Lenders may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by the Agent, except that the foregoing shall not apply to notices to any Lender of Advances to be made if the Lender has notified the Agent that it is incapable of receiving notices relating to Advances by electronic communications. The Agent or the Borrowers may, in its discretion, agree to accept notices and other communications to it by electronic communications pursuant to procedures approved by it, but approval of those procedures may be limited to particular notices or communications.
- (4) Unless the Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), except that if a notice or other communication is not sent during the normal business hours of the recipient, the notice or communication shall be deemed to have been sent at the opening of business on the next Business Day for the recipient, and (ii) notices or communication sposted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient at its e-mail address as described in the foregoing clause (i) of notification that the notice or communication is available and identifying the website address.
- (5) Any Party may change its address or telecopier number for notices and other communications by notice to the other Parties.

11.10 Time of the Essence

Time is of the essence of this Agreement.

11.11 Further Assurances

The Obligors shall, at the request of the Agent, promptly do, execute, deliver or cause to be done, executed or delivered all further acts, documents and matters that may, in the reasonable opinion of the Agent (or the Required Lenders or Lenders, as applicable), be necessary or desirable in order to fully perform and carry out the purpose and intent of the Loan Documents.

- 81 -
11.12 Term of Agreement

Except as otherwise provided in this Agreement, it shall remain in full force and effect until the indefeasible payment and performance in full in cash of all of the Obligations and the termination of all Commitments. The obligations of the Obligors in Sections 9.15, 9.19, 9.20, 11.7 and 11.8 shall continue for the benefit of those to whom the obligations are owed notwithstanding the termination of this Agreement, the Commitments or any particular person's role as Obligor, Agent or Lender.

11.13 Payments on Business Day

Whenever any payment or performance under the Loan Documents would otherwise be due on a day other than a Business Day, such payment shall be made on the following Business Day, unless the following Business Day is in a different calendar month, in which case the payment shall be made on the preceding Business Day.

11.14 Counterparts and Facsimile

This Agreement may be executed in any number of counterparts, each of which when executed and delivered shall be deemed to be an original, and such counterparts together shall constitute one and the same agreement. For the purposes of this Section, the delivery of a facsimile copy of an executed counterpart of this Agreement shall be deemed to be valid execution and delivery of this Agreement.

11.15 Waiver of Jury Trial and Consequential Damages

- (1) Each party hereto hereby waives, to the fullest extent permitted by applicable law, any right it may have to a trial by jury in any legal proceeding directly or indirectly arising out of or relating to this the Loan Documents, the transactions contemplated thereby or any course of conduct, course of dealing, statements (whether oral or written) or actions of any party (whether based on contract, tort or any other theory).
- (2) No party shall assert, and each party hereby waives, to the fullest extent permitted by applicable law, any claim against any other party on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, the Loan Documents, the transactions contemplated thereby or any course of conduct, course of dealing, statements (whether oral or written) or actions of any party (whether based on contract, tort or any other theory).
- (3) Each Obligor acknowledges and agrees that none of the Agent or the Lenders shall have any liability to them in relation to any due diligence investigations conducted by any of them in connection with the transactions contemplated hereby or be under any obligation whatsoever to disclose to them any information received or facts disclosed by any such investigations. Each Obligor further acknowledges and agrees that it is not relying, will not rely, and will not be deemed, in any respect whatsoever, to have relied upon the facts received by and information disclosed to any of the Agent or the Lenders under or in connection with such due diligence investigations.



(4) Each party hereto (a) certifies that no representative, agent or attorney of any other party has represented, expressly or otherwise, that such other party would not, in the event of litigation, seek to enforce the foregoing provisions, and (b) acknowledges that it and the other parties hereto have been induced to enter into this Agreement by, among other things, the waivers, acknowledgments and certifications in this Section.

11.16 Whole Agreement

Except in relation to matters contemplated by the other Loan Documents, this Agreement constitutes the whole and entire agreement between the parties hereto concerning the matters addressed in this Agreement, and cancels and supersedes any prior agreements, undertakings, declarations, commitments or representations, written or verbal, in respect thereof.

11.17 This Agreement to Govern

If there is any conflict or inconsistency between the terms of this Agreement and the terms of any other Loan Document (except any Intercreditor Agreement, which shall prevail as against this Agreement), the provisions of this Agreement shall govern to the extent necessary to remove the conflict or inconsistency.

11.18 USA PATRIOT Act.

Each Lender that is subject to the Act (as hereinafter defined) and the Agent (for itself and not on behalf of any Lender) hereby notifies the Borrowers that pursuant to the requirements of the USA PATRIOT Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Act"), it is required to obtain, verify and record information that identifies the Borrowers, which information includes the name and address of each Borrower and other information that will allow such Lender or the Agent, as applicable, to identify such Borrower in accordance with the Act. Each Borrower shall, promptly following a request by the Agent or any Lender, provide all documentation and other information that the Agent or such Lender requests in order to comply with its ongoing obligations under applicable. "Know your customer" and anti-money laundering rules and regulations, including the Act.

11.19 Language

The Loan Documents have been negotiated in English and will be or have been executed in the English language. Les soussignés ont expressément demandé que ce document soit rédigé en langue anglaise. All paper writings given or delivered pursuant to this Agreement and the other Loan Documents shall, if requested by the Agent, be in the English language or, if not, shall be accompanied by a certified English translation thereof. The English language version of any document shall, absent manifest error, control the meaning and interpretation of the matters set forth therein.

- 83 -

11.20 Limitation Periods

To the extent that any limitation period applies to any claim for payment of obligations or remedy for enforcement of obligations under any Loan Document, the Obligors agree that:

- (a) any limitation period applying to a Loan Document expressed to be payable on demand shall not begin before an express demand for payment of the relevant obligations is made in writing by the Agent to the relevant Obligor;
- (b) any applicable limitation period shall begin afresh upon any payment or other acknowledgment by any Obligor of its relevant obligations; and
- (c) each Loan Document is a "business agreement" as defined in the Limitations Act, 2002 (Ontario) if that Act applies to it.

11.21 Services Provided and Conflicts of Interest

- (1) The Agent and Lenders shall not be responsible for providing or arranging services to the Obligors except providing the Credits. The services of the Agent and Lenders do not include the provision of general corporate finance advisory services. The responsibility of the Agent and Lenders is solely contractual in nature and they do not owe the Obligors any fiduciary duty as a result of the Loan Documents.
- (2) The Obligors acknowledge that the Agent, the Lenders and/or one or more of their respective Affiliates may now and in the future provide debt financing, equity capital or other services (including financial advisory services) to other Persons with whom the Obligors may have conflicting interests. Subject to the provisions of this Agreement, the Agent, the Lenders and their respective Affiliates will not disclose confidential information obtained from the Obligors in connection with the performance of services for others. Similarly, the Agent, the Lenders and their respective Affiliates have no obligation to disclose the existence of or use for the Obligors' benefit confidential information that they have obtained or may obtain from any other Person.
- (3) The Obligors also acknowledge that the Agent, the Lenders and their respective Affiliates engage in securities trading and brokerage activities as well as providing investment banking and other financial services. In the ordinary course of business, the Agent, the Lenders and their respective Affiliates may provide investment banking and other financial services to other Persons with which the Obligors may have commercial or other relationships, and/or acquire, hold or sell, for their own accounts and the accounts of customers, the Obligors' equity, debt and other securities and financial instruments (including bank loans and other obligations) and those of other Persons with which the Obligors may have commercial or other relationships. All rights in respect of securities and financial instruments held by the Agent, the Lenders and their respective Affiliates or their respective customers, including any voting rights, will be exercised by the holder of the rights in its sole discretion.

- 84 -

- (4) The Agent, the Lenders and their respective Affiliates are not responsible to provide the Obligors with advice relating to legal, regulatory, accounting or tax matters. The Obligors acknowledge that (a) they have relied and will continue to rely on the advice of their own legal, regulatory, accounting and tax advisors for all matters relating to the Credits and otherwise and (b) they have not received or relied upon advice from the Agent, the Lenders and their respective Affiliates or advisors regarding legal, regulatory, accounting or tax matters.
- (5) The Obligors acknowledge that no client or near-client relationship has been or will be established between the Obligors and any legal counsel to the Agent or Lenders as a result of their representation of the Agent or Lenders, including by reason of any confidential information regarding any Obligor being provided to legal counsel to the Agent or Lenders or by reason of any Obligor paying or reimbursing the Agent or Lenders for fees, charges or disbursements of legal counsel to the Agent or Lenders. The Obligors also acknowledge that legal counsel to the Agent or I Lenders or by reason of any Confidential information regarding any Obligor being provided to legal counsel to the Agent or Lenders or by reason of any Confidential on the prevented from (a) continuing to act for the Agent and Lenders in connection with the Credits and the Loan Documents, including any enforcement of the Security, for any reason including any Client or near-client relationship that may exist now or in the future between legal counsel to the Agent or Lenders and any Obligor, or (b) acting for any other Person whose interests conflict with the interests of any Obligors that they have received as a result of any presumption, that counsel has provided the other Person, to the detriment of the Obligors, with confidential information regarding the Obligors that they have received as a result of acting as legal counsel to the Agent or Lenders.

11.22 Date of Agreement

This Agreement may be referred to as being dated 7 November 2012 or as of 7 November 2012, notwithstanding the actual date of execution.

[SIGNATURE PAGES FOLLOW]

- 85 -

IN WITNESS OF WHICH, the Parties have duly executed this Agreement.

WASTE MANAGEMENT OF CANADA CORPORATION

By: /s/ Devina A. Rankin Devina A. Rankin Vice President & Treasurer

By: /s/ Don P. Carpenter Don P. Carpenter Vice President, Chief Financial Officer & Controller

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S1 -

WM QUÉBEC INC.

By: /s/ Devina A. Rankin Devina A. Rankin Vice President & Treasurer

By: /s/ Don P. Carpenter Don P. Carpenter Vice President, Chief Financial Officer & Controller

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S2 -

WASTE MANAGEMENT, INC.

By: /s/ Devina A. Rankin Devina A. Rankin Vice President & Treasurer

By: /s/ Don P. Carpenter Don P. Carpenter Vice President & Chief Accounting Officer

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S3 -

WASTE MANAGEMENT HOLDINGS, INC.

By: /s/ Devina A. Rankin Devina A. Rankin Vice President & Treasurer

By: /s/ Don P. Carpenter Don P. Carpenter Vice President, Chief Financial Officer and Controller

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S4 -

THE BANK OF NOVA SCOTIA, as Lender

By: /s/ Matt Macdonald Name: Matt Macdonald Title: Director, Commercial Banking

By: /s/ Phil Damecour Name: Phil Damecour Title: Director, Credit Solutions

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S5 -

COMERICA BANK, as Lender

By: /s/ Lad Perenyi Name: Lad Perenyi Title: Vice President

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S6 -

SUMITOMO MITSUI BANKING CORPORATION OF CANADA, as Lender

By: /s/ E.R. Langley Name: E.R. Langley Title: Senior Vice President

By: ______Name: Title:

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S7 -

THE BANK OF TOKYO-MITSUBISHI UFJ, LTD., as Lender

By: /s/ Thomas Danielson Name: Thomas Danielson Title: Authorized Signatory

By: <u>Name:</u> Title:

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S8 -

U.S. BANK NATIONAL ASSOCIATION, CANADA BRANCH, as Lender

By: /s/ Joseph Rauhala Name: Joseph Rauhala Title: Principal Officer

By: <u>Name:</u> Title:

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S9 -

U.S. BANK, NATIONAL ASSOCIATION, as Lender

By: /s/ Patrick Engel Name: Patrick Engel Title: Vice President

By: <u>Name:</u> Title:

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S10 -

MIZUHO CORPORATE BANK, LTD., as Lender

By: /s/ Leon Mo Name: Leon Mo Title: Authorized Signatory

By: <u>Name:</u> Title:

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S11 -

BANK OF AMERICA, N.A., CANADA BRANCH, as Lender

By: /s/ Medina Sales de Andrade Name: Medina Sales de Andrade Title: Vice President

By: <u>Name:</u> Title:

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S12 -

JPMORGAN CHASE BANK, N.A., TORONTO BRANCH, as Lender

By: /s/ Aized A. Rabbani Name: Aized A. Rabbani Title: Vice President

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S13 -

PNC BANK CANADA BRANCH, as Lender

By: /s/ Nazmin Adatia Name: Nazmin Adatia Title: Senior Vice President

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S14 -

THE BANK OF NOVA SCOTIA, as Agent

By: /s/ Jim Beninger Name: Jim Beninger Title: Director

By: /s/ Nikhil Shah Name: Nikhil Shah Title: Analyst

[signature page for Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S15 -

SCHEDULE A FORM OF NOTICE OF ADVANCE OR PAYMENT

[see reference in Section 9.6]

TO:	The Bank of Nova Scotia GWS – Loan Administration & Agency Operations 720 King Street West 2nd Floor Global Wholesale Services Toronto, ON M5V 2T3 Attention: Managing Director Eacsimile: (416) 866-5901	c.c.	The Bank of Nova Scotia West Metro Commercial Banking Centre 2 Robert Speck Parkway Mississauga, ON L4Z 1H8 Attention: Unit Head Facsimile: (905) 276-4920	c.c.	The Bank of Nova Scotia Corporate Banking – Loan Syndications 62nd Floor, Scotia Plaza 40 King Street West Toronto, ON M5W 2X6 Attention: Unit Head Facsimile: (416) 866-3329
	Attention: Managing Director Facsimile: (416) 866-5991				Facsimile: (416) 866-3329

We refer to the credit agreement dated as of 7 November 2012 between Waste Management of Canada Corporation and WM Québec Inc., as Borrowers, others, as Guarantors, The Bank of Nova Scotia, as Administrative Agent and the Lenders named therein, as amended, supplemented, restated or replaced from time to time (the "Credit Agreement"). All terms used in this certificate and that are defined in the Credit Agreement will have the meanings defined in the Credit Agreement.

1. Request for Advance

Notice is hereby given pursuant to Section 9.6 of the Credit Agreement that the undersigned hereby irrevocably requests as follows:

(a)) that an Advance be made under the following Credit:		
	Revolving Credit	())
	Term Credit	())
(b)) the requested Advance represents the following [check one or more]:		
	initial Advance under the Credit	())
	increase in an Advance under the Credit	())
	rollover of an existing Advance under the Credit	(\cdot))
	conversion of an existing Advance to another type of Advance	())
(c)) the Drawdown Date shall be:		

(d)	the Advance shall be in the form of [check one or more and complet	e details]:
	Prime Rate Advance	()
	Amount:	\$
	Base Rate Advance	()
	Amount	\$
	B/As	()
	Face Amount:	\$
	Term:	
	L/C	
	Face Amount:	\$
	Expiry:	

(e) the proceeds of the Advance shall be deposited in [specify Designated Account].

2. The undersigned hereby confirms as follows:

- (a) the representations and warranties made in Section 5.1 of the Credit Agreement, other than those expressly stated to be made as of a specific date or otherwise expressly modified pursuant to the provisions of Section 5.2 of the Credit Agreement, are true and correct on and as of the date hereof with the same force and effect as if such representations and warranties had been made on and as of the date hereof, but subject to the same qualifications as are contained in Section 5.2 of the Credit Agreement;
- (b) no Event of Default or Pending Event of Default has occurred and is continuing on the date hereof or will result from the Advance(s) requested herein;
- (c) after due inquiry, there is no reasonable expectation that the Borrower will not be in compliance with all covenants contained in Section 6.2 of the Credit Agreement at the end of its current fiscal quarter and was not in compliance with those covenants at the end of its immediately preceding fiscal quarter if it has not yet delivered its Compliance Certificate for that quarter;
- (d) the undersigned will immediately notify you if it becomes aware of the occurrence of any event which would mean that the statements in the immediately preceding paragraphs (a), (b) and (c) would not be true if made on the Drawdown Date; and

(e) all other conditions precedent set out in Sections [4.1 and 4.3 as applicable] of the Credit Agreement have been fulfilled.

CREDIT AGREEMENT

- A2 -

Form 10-Q

Pursua	nt to Section 9.6 of the Credit Agreement, the undersigned hereby irrevocably notify you of the following:	
(a)	that a payment will be made under the following Credit:	
	Revolving Credit	
	Term Credit	
(b)	the payment represents the following [check one or more]:	
redu	uction in Advances under the Credit	
pay	ment of existing Advances which will be rolled over as the same type of Advance under the Credit	
pay	ment of existing Advances which will be converted to another type of Advance under the Credit	
(c)	the payment date shall be;	
(d)	the Advance to be paid shall be in the form of [check one or more and complete details]:	
	Prime Rate Advance ()	
	Amount: \$	
	Base Rate Advance ()	
	Amount <u>§</u> ()	
	DAS () Face Amount:	
	Face Amount: \$	
	Expiry:	

DATED _____.

CREDIT AGREEMENT

- A3 -

WASTE MANAGEMENT OF CANADA CORPORATION

Name:			
Title:			
Name:			

WM QUÉBEC INC.



- A4 -

SCHEDULE B FORM OF COMPLIANCE CERTIFICATE

[see references in Section 6.3(1)(c)]

TO: THE LENDERS (as defined in the Credit Agreement referred to below)

AND TO: THE BANK OF NOVA SCOTIA, as Agent

We refer to Section 6.3(1)(c) of the credit agreement dated as of 7 November 2012 between Waste Management of Canada Corporation and WM Québec, as Borrowers, Waste Management, Inc. and others, as Guarantors, The Bank of Nova Scotia, as Administrative Agent and the Lenders named therein, as amended, supplemented, restated or replaced from time to time (the "**Credit Agreement**"). All terms used in this certificate that are defined in the Credit Agreement will have the meanings defined in the Credit Agreement.

The undersigned hereby certify that:

I, _____, [Chief Financial Officer] [Chief Accounting Officer] [Corporate Treasurer] of **WASTE MANAGEMENT, INC.** certify that no Pending Event of Default or Event of Default exists and that the Obligors are in compliance with Sections 6.1, 6.2 and 6.4 of the Credit Agreement, [as of the end of the quarter ended _____]. Computations to evidence compliance with the financial covenants are detailed below.

6.1(a) Interest Coverage Ratio

Consolidated Net Income (or Deficit) <u>Plus</u> (without duplication): interest expense equity in losses (earnings) of unconsolidated entities income tax expense non-cash writedowns or writeoffs of assets <u>Minus</u> non-cash extraordinary gains on the sale of assets	\$ \$ \$ \$	(i) (ii) (iv) (v) (v) (vi)
EBIT (sum of (i) through (vi))	\$	_(a)
Consolidated Net Income of Acquired Businesses <u>Plus</u> (without duplication): interest expense equity in losses (earnings) of unconsolidated entities income tax expense non-cash writedowns or write-offs of assets non-recurring extraordinary charges	\$ \$ \$ \$ \$	(i) (ii) (iv) (v) (v) (vi)

Form 10-Q

EBIT of Acquired Businesses (sum of (i) through (vi))	\$(b)
Sum of (a) <u>plus</u> (b)	\$(c)
Consolidated Total Interest Expense	\$(d)
Ratio of (c) to (d)	;
Minimum ratio	2.75:1
6.1(b) Total Debt to EBITDA	
EBIT (from 6.1(a) item (c) above)	\$(i)
Plus: Depreciation expense Amortization expense	\$(ii) \$(iii)
EBITDA (sum of (i) through (iii))	\$(iv)
The sum of the following (calculated on a consolidated basis for Waste Management Inc. and its Subsidiaries): Indebtedness for borrowed money Obligations for deferred purchase price of property or services (other than trade payables) Obligations evidenced by debt instruments Obligations, under conditional sales Obligations, liabilities and indebtedness under Capitalized Leases Obligations, liabilities and indebtedness under bonding arrangements (to the extent that a surety has been called upon to make payment on a bond) Guarantees of the Debt of others Debt secured by liens or encumbrances on property Reimbursement obligations with respect to letters of credit	\$(v) \$(vi) \$(vii) \$(xi) \$(xi) \$(xii) \$(xii) \$(xii)
Total Debt (sum of v - xiv)	\$(XII) \$ (xiv)
Ratio of (xv) to (iv)	3(XIV)
Maximum ratio:	3.50:1.00
Maximum facto.	5.50.1.00

DATED _____.

CREDIT AGREEMENT

- B2 -

WASTE MANAGEMENT, INC.

Name:		
Title:		

Name: Title:

By:

By:

WASTE MANAGEMENT OF CANADA CORPORATION

Name:		
Title:		

Name: Title:

WM QUÉBEC INC.

By: Name: Title: By:

Name: Title:

CREDIT AGREEMENT

- B3 -

SCHEDULE C FORM OF ASSIGNMENT AGREEMENT

[see references in Sections 1.1(9) and 10.2(1)(g)]

We refer to Section [6.3(1)(c)] of the credit agreement dated as of 7 November 2012 between Waste Management of Canada Corporation and WM Québec Inc., as Borrowers, Waste Management, Inc. and others, as Guarantors, The Bank of Nova Sectia, as Administrative Agent and the Lenders named therein, as amended, supplemented, restated or replaced from time to time (the "Credit Agreement"). All terms used in this certificate that are defined in the Credit Agreement will have the meanings defined in the Credit Agreement.

For value received, the "Assignor" and the "Assignee" named below hereby agree as follows:

- 1. The Assignor hereby sells and assigns, without recourse, to the Assignee, and the Assignee hereby purchases and assumes from the Assignor, the Proportionate Share specified on Appendix 1 in and to the Assignor's rights and obligations under the Credit Agreement, the Security and all other Credit Documents.
- 2. The Assignor (a) represents and warrants that it is the legal and beneficial owner of the interest being assigned by it hereunder, that such interest is free and clear of any lien or security interest and that it is entitled to enter into this Assignment Agreement, (b) makes no representation or warranty, other than as provided in this Assignment Agreement and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with the Credit Agreement or the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Credit Agreement or any other Credit Document, and (c) makes no representation or warranty and assumes no responsibility with respect to the financial condition of any Obligor or the performance or observance by any Obligor of any of the obligations under the Credit Agreement or any other Credit Document.
- 3. The Assignee, for the benefit of the Borrower, the Guarantors, the other Obligors, the Agent and all Lenders from time to time, including the Assignor, (a) acknowledges receipt of any upfront fee payable by the Assignor, (b) confirms that it has received a copy of the Credit Agreement, together with such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment Agreement, (c) agrees that it will, independently and without reliance upon the Agent, the Assignor or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Credit Agreement, (d) appoints and authorizes the Agent to take such action on its behalf and to exercise such powers and discretion under the Credit Agreement as are delegated to the Agent and the Collateral Agent under the Credit Agreement, (f) agrees that it will perform in accordance with their terms all of the obligations that by the terms of the Credit Agreement are required to be performed by it as a Lender, (g) agrees to be bound by the terms of all Intercreditor Agreements, and (h) specifies as its address for notice and payments its office at the address set forth on Appendix 1 hereto.

- 4. Following the execution of this Assignment Agreement, it shall immediately be delivered to the Agent, together with the processing and recording fee specified in Section [10.2 (1)(s)] of the Credit Agreement if applicable, for approval and recording by the Agent, the Issuing Lender and the Borrower, if applicable. The Assignee's agreement to become a Lender, as constituted by this Assignment Agreement, is irrevocable, unless the Assignee is not approved by the Agent, the Issuing Lender or the Borrower, if applicable. The Assignee shall become a Lender, and shall be bound by the obligations and entitled to the benefits in the Credit Agreement, immediately upon this Assignment Agreement being approved and recorded by the Agent, the Issuing Lender and the Borrower, if applicable (the "Effective Date"). On the Effective Date, the Assignee (a) shall pay the Assignor an amount equal to the Assignee's Proportionate Share of Prime Rate Advances made by the Assignor as of the Effective Date, advanced by the Lenders.
- 5. If Advances made by the Assignee to the Borrower are for any reason less than the Assignee's Proportionate Share of the aggregate Advances made by all Lenders under the Credit Agreement, the Assignee shall, on demand, indemnify the Assignor in respect of the principal amount of the corresponding Advances made by the Assignor in excess of the Assignor's Proportionate Share. The Advances by the Assignor in respect of which the Assignee is bound to indemnify the Assignor are set out on Appendix 2 to this Assignment Agreement. The Assigner shall pay the Assignee is obliged to indemnify the Assignor. The fee shall be in the amount specified on Appendix 2 and shall be payable on the Effective Date in respect of Advances by way of B/As.
- 6. This Assignment Agreement shall be governed by, and construed in accordance with the laws of the Province of Ontario, Canada.

CREDIT AGREEMENT

7. This Assignment Agreement may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement. Delivery of an executed counterpart of this Assignment Agreement by telecopier shall be effective as delivery of a manually executed counterpart of this Assignment Agreement.

IN WITNESS WHEREOF, the Assignor and the Assignee have caused this Assignment Agreement to be executed by their duly authorized officers as of the dates specified below.

-		
By:		
	Name:	
	Title:	
Date:		
Assignee:		
By:		
	Name:	
	Title:	
Date:		
		CREDIT AGREEMEN

- C3 -

Approved on		[If applicable] Approved on
THE B	ANK OF NOVA SCOTIA, as Agent	WASTE MANAGEMENT OF CANADA CORPORATION
By:		By:
	Name: Title:	Name: Title:
By:		By:
	Name: Title:	Name: Title:
Effective Date:		Date:
		WM QUÉBEC INC.
		By:
		Name: Title:

By:

Date:

Name: Title:

CREDIT AGREEMENT

- C4 -

Form 10-Q

<u>APPENDIX 1</u> <u>TO</u> ASSIGNMENT AGREEMENT

Proportionate Share assigned by Assignor: Proportionate Share retained by Assignor:

Payment Details, including address of Assignee for notices:

Principal Amount of Advance

<u>APPENDIX 2</u> <u>TO</u> ASSIGNMENT AGREEMENT

Advances in respect of which the Assignee is to indemnify the Assignor, as of the Effective Date:

Type of Advance

Maturity Date of Advance

Indemnity fee:

Form 10-Q

SCHEDULE D ADDRESS FOR NOTICE

Waste Management of Canada Corporation/WM Québec Inc. c'o Waste Management, Inc. 1001 Fannin Street, Suite 4000 Houston, Texas 77002

Attention: Treasurer Fax No.: 713-942-1580

with a copy to:

Attention: General Counsel Fax No.: 713-942-1580

The Bank of Nova Scotia

West Metro Commercial Banking Centre 2 Robert Speck Parkway Mississauga, Ontario L4Z 1H8

Attention:Unit HeadFax No.:(905) 276-4920

with a copy to:

Corporate Banking – Loan Syndications 62nd Floor, Scotia Plaza 40 King Street West Toronto, Ontario M5W 2X6

Attention:Unit HeadFax No.:(416) 866-3329

Comerica Bank Brookhollow II

2900 North Loop West, 9th Floor Houston, Texas 77092

Attention: Vice President Fax No.: 713-507-2889

PNC Bank Canada Branch

130 King Street West, Suite 2140 P.O. Box 462, Toronto, Ontario M5X 1E4

Attention: Relationship Manager Fax No.: 416-361-0085

Form 10-Q

with a copy to:

Attention:Relationship ManagerFax No.:412-762-6484

U.S. Bank, National Association Canada Branch 120 Adelaide Street West, Suite 2300 Toronto, Ontario M5H 1T1

Attention: Principal Officer Fax No.: 416-306-3545

U.S. Bank, National Association 214 North Tryon Street, 30th Floor Charlotte, NC 28202 Attention: Vice President Fax No.: 704-335-2815

Mizuho Corporate Bank, Ltd. 1251 Avenue of the Americas New York, NY 10020

Attention: Vice President Fax No.: 212-282-4488

Sumitomo Mitsui Banking Corporation of Canada Suite 1400, Ernst & Young Tower Toronto-Dominion Centre

P.O. Box 172 222 Bay Street Toronto, Ontario M5K 1H6

Attention:Senior Vice PresidentFax No.:416-214-3606

The Bank of Tokyo-Mitsubishi UFJ, Ltd.

1251 Avenue of the Americas New York, NY 10020-1104

Attention: Benita Volid Fax No.: 312-696-4535

Bank of America, N.A., Canada Branch 181 Bay Street, 4th Floor Toronto, Ontario M5J 2V8

Attention:Maria MaiaFax No.:980-233-7700

CREDIT AGREEMENT

- D2 -

with a copy to:

Attention:Medina Sales de AndradeFax No.:312-453-0401

JPMorgan Chase Bank, N.A., Toronto Branch 383 Madison Avenue Floor 24 New York, NY 10179

Attention: Aized Rabbani Fax No.: 212-622-6642

CREDIT AGREEMENT

- D3 -

Form 10-Q

SCHEDULE E LENDERS AND THEIR APPLICABLE PERCENTAGES

[see references in Sections 1.1(9), 1.1(74), 1.1(102), 1.1(113) and 9.2]

The Bank of Nova Scotia (original total Commitment – \$130,000,000)

Revolving Credit – \$30,000,000 Swing Line Tranche – \$20,000,000 Term Credit – \$100,000,000 Applicable Percentage: 20%

JPMorgan Chase Bank, N.A.

(original total Commitment - \$100,000,000)

Revolving Credit – \$23,076,923.08 Swing Line Tranche – nil Term Credit – \$76,923,076.92 Applicable Percentage: 15.38461538%

Bank of America, N.A. (original total Commitment – \$100,000,000)

Revolving Credit – \$23,076,923.08 Swing Line Tranche – nil Term Credit – \$76,923,076.92 Applicable Percentage: 15.38461538%

PNC Bank, National Association (original total Commitment – \$100,000,000)

Revolving Credit – \$23,076,923.08 Swing Line Tranche – nil Term Credit – \$76,923,076.92 Applicable Percentage: 15.38461538%

Mizuho Corporate Bank, Ltd. (original total Commitment – \$50,000,000)

Revolving Credit – \$11,538,461.54 Swing Line Tranche – nil Term Credit – \$38,461,538.46 Applicable Percentage: 7.69230769%
The Bank of Tokyo-Mitsubishi UFJ, Ltd. (original total Commitment - \$50,000,000)

Revolving Credit - \$11,538,461.54 Swing Line Tranche – nil Term Credit – \$38,461,538.46 Applicable Percentage: 7.69230769%

U.S. Bank, National Association (original total Commitment - \$50,000,000)

Revolving Credit - \$11,538,461.54 Swing Line Tranche – nil Term Credit – \$38,461,538.46 Applicable Percentage: 7.69230769%

Sumitomo Mitsui Banking Corporation of Canada (original total Commitment - \$50,000,000)

Revolving Credit - \$11,538,461.54 Swing Line Tranche – nil Term Credit – \$38,461,538.46 Applicable Percentage: 7.69230769%

Comerica Bank (original total Commitment - \$20,000,000)

Revolving Credit - \$4,615,384.62 Swing Line Tranche – nil Term Credit – \$15,384,615.38 Applicable Percentage: 3.07692308%

- E2 -

EXHIBIT 10.3

FIRST AMENDMENT AGREEMENT dated as of 16 August 2013.

BETWEEN:

WASTE MANAGEMENT OF CANADA CORPORATION WM QUÉBEC INC. as Borrowers

- and -

WASTE MANAGEMENT, INC. WASTE MANAGEMENT HOLDINGS, INC. as Guarantors

- and -

THE LENDERS FROM TIME TO TIME PARTY TO THIS AGREEMENT as Lenders

- and -

THE BANK OF NOVA SCOTIA as Administrative Agent

RECITALS:

- The parties hereto have entered into a credit agreement dated as of 7 November 2012 (the "Existing Credit Agreement"). Α.
- B. The parties have agreed to enter into this First Amendment Agreement to amend the provisions of the Existing Credit Agreement on the terms and conditions set out in this First Amendment Agreement.
- C. Capitalized terms used in this First Amendment Agreement and not otherwise defined have the meanings defined in the Existing Credit Agreement.
- FOR VALUE RECEIVED, and intending to be legally bound by this First Amendment Agreement, the Parties agree as follows:

1. Definitions

(a) The following definition is added as new subsection 1.1(27)A immediately following existing subsection 1.1(27):

""Code" means The Internal Revenue Code of 1986, as amended and in effect from time to time."

(b) The definition of "GAAP" in subsection 1.1(59) of the Existing Credit Agreement which reads:

""GAAP" means, when used in this Agreement, whether directly or indirectly through reference to a capitalized term used therein, means (a) principles that are consistent with the principles promulgated or adopted by the Financial Accounting Standards Board (U.S.) and its predecessors, in effect for the fiscal year ended on the December 31, 2010, and (b) to the extent consistent with the principles, the accounting practice of WMI reflected in its financial statements for the year ended on December 31, 2010, provided, that, with respect to any financial statements prepared after the December 31, 2010, such meaning in each of clause (a) and (b) shall include the revised guidance associated with multiple-deliverable revenue arrangements effective per FASB ASC 605 on January 1, 2011; provided, further, that in each case referred to in this definition of "GAAP" a certified public accounting principles) as to financial statements in which such principles have been properly applied. Notwithstanding the foregoing, for purposes of determining compliance with any covenant (including the computation of any financial covenant) contained herein, "Dett" of WMI and its subsidiaries shall be deemed to be carried at 100% of the outstanding principal amount thereof, and the effects of FASB ASC 825 and FASB ASC 470-20 on financial liabilities shall be disregarded."

is deleted in its entirety and replaced with the following:

""GAAP" means, when used in this Agreement, whether directly or indirectly through reference to a capitalized term used therein, means principles that are consistent with the principles promulgated or adopted by the Financial Accounting Standards Board and its predecessors as in effect from time to time, except as otherwise specifically prescribed herein. If any "Accounting Change" (as defined below) occurs subsequent to July 26, 2013 (the effective date of the U.S. Credit Agreement), such change results in a material change in the method of calculation of financial covenants, standards or terms in this Agreement or any other Loan Document and any of WMI, the Agent or the Required Lenders shall so request, then (A) WMI, the Lenders and the Agent agree to enter into negotiations in good faith in order to amend such provisions of this Agreement or such other Loan Document so as to reflect equitably such Accounting Changes with the desired result that the criteria for evaluating the WMI's financial condition shall be the same after such Accounting Changes as if such Accounting Changes had not occurred and (is ubject to the approval of the Required Lenders, and (B) until such time as such an amendment shall have been executed and delivered by WMI, the Agent and the Required Lenders, (i) the financial covenants, standards and terms in this Agreement and the other Loan Documents impacted by such material change shall continue to be calculated or construed as if such Accounting Changes had not occurred and (ii) WMI shall provide to the Agent and the Lenders a reconciliation between the calculation of such impacted covenants, standards and terms before and after giving effect to such Accounting Changes. "Accounting Changes are frees to changes in accounting principles required by the promulgation of any rule, regulation, pronouncement or opinion by the Financial

- 2 -

Accounting Standards Board of the American Institute of Certified Public Accountants or, if applicable, the Securities and Exchange Commission. Notwithstanding the foregoing, for purposes of determining compliance with any covenant (including the computation of any financial covenant) contained herein, "Debt" of WMI and its Subsidiaries shall be deemed to be carried at 100% of the outstanding principal amount thereof, and the effects of FASB ASC 825 and FASB ASC 470-20 on financial liabilities shall be disregarded."

(c) The definition of "Other Guaranteed Obligations" in subsection 1.1(86) of the Existing Credit Agreement which reads:

""Other Guaranteed Obligations" means the present and future debts, liabilities and obligations of the Borrowers or either of them to any Lender or its affiliates that arise under or in connection with, (i) derivatives or other hedging arrangements, and (ii) cash management arrangements, entered into with such Lender or its affiliates that are permitted under this Agreement, which Other Guaranteed Obligations shall rank pari passu with the Credits."

is deleted in its entirety and replaced with the following:

""Other Guaranteed Obligations" means the present and future debts, liabilities and obligations of the Borrowers or either of them to any Lender or its affiliates that arise under or in connection with, (i) derivatives or other hedging arrangements, and (ii) cash management arrangements, entered into with such Lender or its affiliates, which Other Guaranteed Obligations shall rank pari passu with the Credits."

(d) Clause (f) of the definition of "Permitted Encubrances" in subsection 1.1(90) of the Existing Credit Agreement which reads:

"(f) the Encumbrances on Schedule 1.1 of the U.S. Credit Agreement (a copy of which is attached hereto as Annex 1.1) securing the obligations listed on such Schedule and any replacement Encumbrance securing any renewal, extension or refunding of such obligations if the amount secured by such renewal, extension or refunding Encumbrance shall not exceed the amount of the outstanding obligations secured by the Encumbrance being replaced at the time of such renewal, extension or refunding (plus transactions costs, including premiums and fees, related to such renewal, extension or refunding) and if such replacement Encumbrance shall be limited to substantially the same property that secured the Encumbrance scienced,"

is deleted in its entirety and replaced with the following:

"(f) the Encumbrances on Schedule 1.1 of the U.S. Credit Agreement (a copy of which is attached hereto as Annex 1.1) securing the obligations listed on such Schedule and any replacement Encumbrance securing any renewal, extension or refunding of such obligations provided, that (i) the aggregate principal amount of obligations secured by any renewal, extension or refunding Encumbrance

- 3 -

permitted by this clause (f) shall not exceed the aggregate outstanding principal amount of the obligations secured by the Encumbrance being replaced at the time of such renewal, extension or refunding (plus transactions costs, including premiums and fees, related thereto), and (ii) each replacement Encumbrance shall be limited to substantially the same property that secured the Encumbrance so replaced;"

(e) Clause (m) of the definition of "Permitted Encubrances" in subsection 1.1(90) of the Existing Credit Agreement which reads:

"(m) Encumbrances securing other Debt, provided that the aggregate amount of all liabilities, including any Debt, of WMI and its Subsidiaries secured by all Encumbrances permitted in subsections (k), (l) and (m), when added (without duplication) to the aggregate amount of Debt of WMI's Subsidiaries permitted under Section 6.4(1)(b) and Debt with respect to Permitted Receivables Transactions, shall not exceed 15% of Consolidated Tangible Assets at any time."

is deleted in its entirety and replaced with the following:

"(m) Encumbrances securing other Debt, provided that the aggregate amount of all liabilities, including any Debt, of WMI and its Subsidiaries secured by all Encumbrances permitted in subsections (k), (l) and (m), when added (without duplication) to the aggregate outstanding amount of Debt of WMI's Subsidiaries permitted under Section 6.4(1)(b) and Debt with respect to Permitted Receivables Transactions, shall not exceed 15% of Consolidated Tangible Assets at any time."

(f) The definition of "U.S. Credit Agreement" in subsection 1.1(116) of the Existing Credit Agreement which reads:

""U.S. Credit Agreement" means the U.S. \$2,000,000,000 amended and restated revolving credit agreement dated as of May 9, 2011 by and among WMI, as borrower, Waste Management Holdings, Inc., as guarantor, various banks party thereto from time to time, as lenders, Bank of America, N.A., as administrative agent and others."

is deleted in its entirety and replaced with the following:

""U.S. Credit Agreement" means the U.S. \$2,250,000,000 second amended and restated revolving credit agreement dated as of July 26, 2013 by and among WMI, as borrower, Waste Management Holdings, Inc., as guarantor, various banks party thereto from time to time, as lenders, Bank of America, N.A., as administrative agent and others."

- 4 -

2. Representations and Warranties

The first two sentences of subsection 5.1(14)(b) of the Existing Credit Agreement which read:

"Each contribution required to be made to a Guaranteed Pension Plan, whether required to be made to avoid the incurrence of an accumulated funding deficiency, the notice or lien provisions of \$303(k) of ERISA, or otherwise, has been timely made. No waiver of an accumulated funding deficiency or extension of amortization periods has been received with respect to any Guaranteed Pension Plan."

are deleted in their entirety and replaced with the following:

"Each contribution required to be made to a Guaranteed Pension Plan, whether required to be made to avoid a violation of the minimum funding standards under §§412 and 430 of the Code, the notice or lien provisions of §303(k) of ERISA, or otherwise, has been timely made. No waiver of the minimum funding standards under §§412 and 430 of the Code or extension of amortization periods has been received with respect to any Guaranteed Pension Plan."

3. Financial Covenants

Subsection 6.1(b) of the Existing Credit Agreement which reads:

"(b) Total Debt to EBITDA. As of the end of any fiscal quarter of WMI, WMI will not permit the ratio of (a) Total Debt to (b) EBITDA for the four fiscal quarters then ending to exceed 3.50:1.00."

is deleted in its entirety and replaced with the following:

"(b) Total Debt to EBITDA. As of the end of any fiscal quarter of WMI, WMI will not permit the ratio of (a) Total Debt to (b) EBITDA for the four fiscal quarters then ending to exceed (i) for each fiscal quarter of WMI ending before September 30, 2015, 3.75:1.00, and (ii) for each fiscal quarter of WMI ending on or after September 30, 2015, 3.50:1.00."

4. Negative Covenants

Subsections 6.4(1) (a) and (b) of the Existing Credit Agreement which read:

- "(a) Debt of WMI's Subsidiaries listed in Schedule 8.1(a) of the U.S. Credit Agreement (a copy of which is attached hereto as Annex 8.1(a)) and any extension, renewal or refinancing of such Debt, provided that the terms and conditions of any such extensions, renewals or refinancings do not increase the relative priority of the original Debt and provided, further, that such extended, renewed or refinanced Debt does not in the aggregate exceed the U.S. Dollar amount of the original Debt; and
- (b) other Debt of WMI's Subsidiaries (other than Waste Management Holdings, Inc.) provided that the aggregate amount of all such Debt under this Section 6.4(1)(b), when added (without duplication) to the aggregate outstanding amount of secured Debt of WMI and its Subsidiaries under (k), (l) and (m) of the definition of "Permitted Encumbrances" and Debt with respect to Permitted Receivables Transactions, shall not exceed 15% of Consolidated Tangible Assets."

^{- 5 -}

are deleted in their entirety and replaced with the following:

- "(a) Debt of WMI's Subsidiaries listed in Schedule 8.1(a) of the U.S. Credit Agreement (a copy of which is attached hereto as Annex 8.1(a)), any extension, renewal or refinancing of such Debt, and any additional bonds issued and Capital Leases entered into from time to time after July 26, 2013 (the effective date of the U.S. Credit Agreement); provided that (i) if such Debt is an extension, renewal or refinancing of existing Debt, the terms and conditions of any such extensions, renewals or refinancings shall not increase the relative priority of such Debt over the priority of the original Debt, and (ii) in on event shall the aggregate principal amount of Debt permitted by this Subsection 6.4(1)(a) exceed the aggregate principal amount of the original Debt listed on Schedule 8.1(a) of the U.S. Credit Agreement (plus transaction costs, including premiums and fees, related thereto); and
- (b) other Debt of WMI's Subsidiaries (other than Waste Management Holdings, Inc.) provided that the sum (without duplication) of (i) the aggregate outstanding principal amount of Debt permitted under this Subsection 6.4(1)(b), plus (ii) the aggregate outstanding principal amount of secured Debt of WMI and its Subsidiaries permitted under (k), (l) and (m) of the definition of "Permitted Encumbrances", plus (iii) the aggregate amount of Debt with respect to outstanding Permitted Receivables Transactions (determined in accordance with the proviso to the definition of "Debt"), shall not exceed 15% of Consolidated Tangible Assets at any time."

5. Direct Payments

Section 7.11 of the Existing Credit Agreement is amended by replacing each reference therein to "Other Secured Obligations" with reference to "Other Guaranteed Obligations" and each reference therein to "Derivatives" with reference to "derivative transactions".

6. Payments by the Borrowers

The first paragraph of Subsection 9.17(1) of the Existing Credit Agreement which reads:

"(1) Except as otherwise specified in this Agreement, all payments made by or on behalf of the Borrowers shall be made to and received by the Agent and shall be distributed by the Agent to the Lenders as soon as possible upon receipt by the Agent. Except as otherwise provided in this Agreement (including Section 9.18), the Agent shall distribute:"

- 6 -

is deleted in its entirety and replaced with the following:

"(1) Except as otherwise specified in this Agreement, all payments made by or on behalf of the Borrowers shall be made to and received by the Agent and shall be distributed by the Agent to the Lenders as soon as possible upon receipt by the Agent. Except as required to make payments in respect of Other Guaranteed Obligations to the Persons to whom such Other Guaranteed Obligations are owed, or as otherwise provided in this Agreement (including Section 9.18), the Agent shall distribute:"

7. Increased Costs, Etc.

Subsection 9.19(2) of the Existing Credit Agreement which reads:

"(2) If any Lender determines in its sole and absolute discretion that any Change in Law affecting the Lender or any lending office of the Lender or its holding company (or other Controlling Person), if any, regarding capital requirements has or would have the effect of reducing the rate of return on the Lender's capital or on the capital of its holding company (or other Controlling Person), if any, as a consequence of this Agreement, the Commitments of the Lender or any Advance made by the Lender, to a level below that which the Lender or its holding company (or other Controlling Person), if any, as a consequence of this Agreement, the Commitments of the Lender or any Advance made by the Lender, to a level below that which the Lender or its holding company (or other Controlling Person) could have achieved but for that Change in Law (taking into consideration the Lender's policies and the policies of its holding company (or other Controlling Person) with respect to capital adequacy, each from time to time), then from time to time the Borrowers will pay to that Lender an additional amount or amounts that will compensate Lender or its holding company (or other Controlling Person) for the reduction suffered."

is deleted in its entirety and replaced with the following:

"(2) If any Lender determines in its sole and absolute discretion that any Change in Law affecting the Lender or any lending office of the Lender or its holding company (or other Controlling Person), if any, regarding capital adequecy or liquidity requirements has or would have the effect of reducing the rate of return on the Lender's capital or on the capital of its holding company (or other Controlling Person), if any, as a consequence of this Agreement, the Commitments of the Lender or any Advance made by the Lender, to a level below that which the Lender or its holding company (or other Controlling Person) could have achieved but for that Change in Law (taking into consideration the Lender's policies and the policies of its holding company (or other Controlling Person) with respect to capital adequacy and liquidity requirements, each from time to time), then from time to time the Borrowers will pay to that Lender an additional amount or amounts that will compensate Lender or its holding company (or other Controlling Person) with respect to capital adequacy and liquidity requirements.

8. Representations of the Obligors

Each Obligor acknowledges that this First Amendment Agreement is a Loan Document and that all of the representations and warranties concerning Loan Documents that are contained in the Existing Credit Agreement apply to this First Amendment Agreement and are deemed to be repeated on the execution of this First Amendment Agreement as if set out in full in this First Amendment Agreement other than those which relate to a specific date.

- 7 -

9. Ratification and Confirmation; Restatement

On and after this date, each reference in the "Existing Credit Agreement" to "this Agreement" and each reference to the Existing Credit Agreement in the Loan Documents and any and all other agreements, documents and instruments delivered by the Agent, the Lenders, any Obligor or any other Person shall mean and be a reference to the Existing Credit Agreement as amended by this First Amendment Agreement.

The Existing Credit Agreement, as amended by this First Amendment Agreement, remains in full force and effect and is hereby ratified and confirmed. Without in any way limiting the terms of the Existing Credit Agreement or the other Loan Documents, each Obligor confirms that the existing Security shall continue to secure all the Obligations, including but not limited to any arising as a result of this First Amendment Agreement.

10. Counterparts and Facsimile

This First Amendment Agreement may be executed in any number of counterparts, each of which when executed and delivered shall be deemed to be an original, and such counterparts together shall constitute one and the same agreement. The delivery of a facsimile copy of an executed counterpart of this First Amendment Agreement shall be deemed to be valid execution and delivery of this First Amendment Agreement, but the party delivering a facsimile copy shall deliver an original copy of this First Amendment Agreement as soon as possible after delivering the facsimile copy.

[SIGNATURE PAGES FOLLOW]

- 8 -

IN WITNESS OF WHICH, the Parties have duly executed this First Amendment Agreement.

WASTE MANAGEMENT OF CANADA CORPORATION

By: /s/ Devina A. Rankin Devina A. Rankin Vice President & Treasurer

By: /s/ Don P. Carpenter Don P. Carpenter Vice President, Chief Financial Officer & Controller

[signature page for First Amendment Agreement to Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S1 -

WM QUÉBEC INC.

By: /s/ Devina A. Rankin Devina A. Rankin Vice President & Treasurer

By: /s/ Don P. Carpenter Don P. Carpenter Vice President, Chief Financial Officer & Controller

WASTE MANAGEMENT, INC.

By: /s/ Devina A. Rankin Devina A. Rankin Vice President & Treasurer

By: /s/ Don P. Carpenter Don P. Carpenter Vice President & Chief Accounting Officer

WASTE MANAGEMENT HOLDINGS, INC.

By: /s/ Devina A. Rankin Devina A. Rankin Vice President & Treasurer

By: /s/ Don P. Carpenter Don P. Carpenter Vice President, Chief Financial Officer & Controller

THE BANK OF NOVA SCOTIA, as Lender

By: /s/ Matt Macdonald Name: Matt Macdonald Title: Director, Commercial Banking

By: /s/ Phil Damecour

Name: Phil Damecour Title: Director, Credit Solutions Group

COMERICA BANK, as Lender

By: /s/ Lad Perenyi Name: Lad Perenyi Title: Vice President

SUMITOMO MITSUI BANKING CORPORATION OF CANADA, as Lender

By:

/s/ E.R. Langley Name: E.R. Langley Title: Senior Vice President

By: Name: Title:

[signature page for First Amendment Agreement to Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S7 -

THE BANK OF TOKYO-MITSUBISHI UFJ, LTD., as Lender

By:

/s/ Jason Krogh Name: Jason Krogh Title: Authorized Signatory

By: Name: Title:

[signature page for First Amendment Agreement to Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S8 -

U.S. BANK NATIONAL ASSOCIATION, CANADA BRANCH, as Lender

By:

Name: Title:

By: Name: Title:

[signature page for First Amendment Agreement to Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S9 -

Lend	BANK, NATIONAL ASSOCIATION, as er
By:	
	Name:
	Title:
By:	
	Name:
	Title:

[signature page for First Amendment Agreement to Credit Agreement relating to Wast Management of Canada Corporation and WM Quebec Inc. et al.]

- S10 -

MIZUHO BANK, LTD., as Lender

By: /s/ Leon Mo Name: Leon Mo Title: Authorized Signatory

[signature page for First Amendment Agreement to Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S11 -

BANK OF AMERICA, N.A., CANADA BRANCH, as Lender

By: /s/ Median Sales De Andrade Name: Medina Sales De Andrade Title: Vice President

By: Name: Title:

[signature page for First Amendment Agreement to Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S12 -

JPMORGAN CHASE BANK, N.A., TORONTO BRANCH, as Lender

By:

Name: Title:

[signature page for First Amendment Agreement to Credit Agreement relating to Waste Management of Canada Corporation and WM Quebec Inc. et al.]

- S13 -

PNC BANK CANADA BRANCH, as Lender

By: /s/ Nazmin Adatia Name: Nazmin Adatia Title: Senior Vice President

THE BANK OF NOVA SCOTIA, as Agent

By: /s/ Robert Boomhour Name: Robert Boomhour Title: Director

By: /s/ Clement Yu Name: Clement Yu Title: Associate Director

Form 10-Q

ANNEX 1.1

EXISTING LIENS

1. Various capital leases entered into by Subsidiaries in the ordinary course of business for operating equipment and facilities.

2. The note payable associated with the investment in federal low-income housing tax credits as described in Note 9, Income Taxes, and Note 20, Variable Interest Entities, in the Annual Report on Form 10-K for the year ended December 31, 2012.

ANNEX 8.1(a)

		AITILA 0.1(a)				
EXISTING INDEBTEDNESS						
Name	Principal	Maturity				
Waste Management Holdings Senior Notes:						
\$450,000,000 due 8/01/26	\$ 448,975,000	8/1/2026				
Total WM Holdings Senior Notes	\$ 448,975,000					
Tax Exempt Revenue Bonds:						
Amelia, Virginia due 4/1/27	26,800,000	4/1/2027				
Arkansas due 6/01/28	15,000,000	6/1/2028				
Brazoria County	12,000,000	5/1/2028				
Bucks County due 12/01/22	25,000,000	12/1/2022				
California CPCFA	35,700,000	11/1/2038				
California CPCFA 2005A	50,000,000	4/1/2025				
California CPCFA 2005B	50,000,000	4/1/2025				
California CPCFA 2005C	75,000,000	11/1/2023				
California CPCFA due 1/1/22	48,500,000	1/1/2022				
California CPCFA due 12/01/27	15,000,000	12/1/2027				
California CPCFA due 7/01/31	19,000,000	7/1/2031				
California CPCFA due 7/1/27	38,435,000	7/1/2027 9/1/2014				
California Municipal Finance Authority 2009 Learner	15,000,000	2/1/2014				
California Municipal Finance Authority - 2008 Issuance	33,900,000	2/1/2019				
California Municipal Finance Authority - 2009A	30,000,000					
Charles City (Virginia due 2/1/29) Charles City (Virginia)	30,000,000	2/1/2029 8/1/2027				
Charles City (Virginia) Charles City (Virginia) due 4/1/27	10,000,000 10,000,000	4/1/2027				
Charles City (Originia) due 5/1/27		5/1/2027				
City of Minor Lane Heights due 31/27	30,320,000 11,000,000	3/1/2021				
City of Mobile		10/1/2038				
Cobb County Series 2004A	4,175,000 10,000,000	4/1/2033				
Cobb County Series 2004A Cobb County Series 2004B	10,000,000	4/1/2033				
Colorado due 7/1/27	14,160,000	7/1/2027				
Colorado due 8/1/38	10,000,000	8/1/2038				
Colorado Geries 2004	10,840,000	7/1/2018				
Countryide (Lake County) due 4/1/21	5,670,000	4/1/2021				
Countryside (Lake County) due 9/121	4,320,000	9/1/2021				
County of Logan due 3/1/21	7,450,000	3/1/2021				
Denton (County (TX 2003B)	10,000,000	5/1/2021				
Gilliam County	15,000,000	7/1/2028				
Gillian County (2007)	25,000,000	10/1/2018				
Gilliam County due 07/01/29	25,000,000	7/1/2029				
Gilliam County due 08/01/25	15,900,000	8/1/2025				
Glouester (VA 2003A)	10,000,000	9/1/2038				
Gulf Coast Series 2004A	35,000,000	4/1/2019				
Hampton due 9/1/28	10,000,000	9/1/2028				
Harrison County (West Virginia due 4/1/24)	8,420,000	4/1/2024				
Ilinois due 10/1/2023	20,000,000	10/1/2023				
Illinois due 8/1/2029	30,000,000	8/1/2029				
	2 - , - 50,000					

Indiana due 10/01/31 King George due 6/1/23 King George due 9/1/21 (Garnet) Maine Maine Maricopa (Arizona) due 12/01/31 Maryland due 41/1/6 Massachusetts due 5/1/27 Miami Dade County Series 2004A Miami Dade County Series 2006 Miami Dade County Series 2007 Miami Dade County Series 2007 Miami Dade County Series 2008 Miami Dade County Series 2007 Miami Dade County Series 2007 Miami Dade County Series 2007 Miami Dade County Series 2011 Michigan due 12/1/2013 Michigan due 81/1/207 Mississippi due 31/127 Mississippi due 31/127 Mississippi due 31/127 Mississippi due 31/128 Nashville (Tennessee) due 8/01/31 Nebraska Nevada due 10/01/14 New Jersey due 6/01/15 New Jersey due 6/01/15 New York City due 12/1/17 New York Series 2004A New York Series 2012 North Carolina due 8/01/14 New Take Series 2004A New York Series 2004A New Jork Series 2004A Schuldkil/Pine Grove due 10/1/19 South Carolina	Illinois due 9/1/27 Indiana due 10/01/25 Indiana due 10/01/25
Maine Maricopa (Arizona) due 12/01/31 Maryland due 4/1/16 Massachusetts Massachusetts due 5/1/27 Miami Dade County Series 2004A Miami Dade County Series 2004B Miami Dade County Series 2006 Miami Dade County Series 2007 Miami Dade County Series 2008 Miami Dade County Series 2008 Miami Dade County Series 2011 Michigan due 12/1/2013 Michigan due 12/1/2017 Mississippi due 3/1/27 Mississippi due 3/1/27 Nessistipi due 3/1/27 Nessistipi due 3/1/27 New York City due 1/1/13 New Jersey due 6/01/15 New Jersey due 6/01/15 New York Series 2012 North Carolina due 8/01/14 North Sumter, AL Ohio WDA due 11/1/22 Ohio WDA due 11/1/24 Ohio MDA	King George due 6/1/23 King George due 9/1/21 (Garnet)
Massachusetts Massachusetts due 5/1/27 Miami Dade County Series 2004A Miami Dade County Series 2004B Miami Dade County Series 2006 Miami Dade County Series 2007 Miami Dade County Series 2008 Miami Dade County Series 2011 Michigan due 12/1/2013 Michigan due 12/1/2013 Michigan due 12/1/2013 Mission, TX Series 2006 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 3/1/28 Nashville (Tennessee) due 8/01/31 Nevada due 10/01/14 New Jersey due 6/01/15 New Jersey due 6/01/15 New Jersey due 6/01/15 New Jersey due 6/01/15 New York City due 12/1/17 New York Series 2004A New York Guty due 1/1/22 Ohio WDA due 11/1/22 Ohio WDA due 11/1/24 Okeechobee due 8/1/24 Okeechobee Series 2004A Oklahoma Pennsylvania Series 2004A Nichland (SC) due 6/1/15 Savannah Series 2004A	Maine Maricopa (Arizona) due 12/01/31
Miami Dade County Series 2004B Miami Dade County Series 2006 Miami Dade County Series 2007 Miami Dade County Series 2008 Miami Dade County Series 2011 Michigan due 12/1/2013 Michigan due 12/1/2013 Mission, TX Series 2006 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 7/1/28 Nashville (Tennessee) due 8/01/31 Nebraska Nevada due 10/01/14 New Jersey due 6/01/15 New Jersey due 6/01/15 New Jersey due 6/01/15 New Jersey due 6/01/15 New York City due 12/1/17 New York City due 12/1/17 New York Series 2004A New York Series 2012 North Carolina due 8/01/14 North Sumter, AL Ohio WDA due 11/1/22 Ohio WDA due 11/1/24 Okeechobee due 8/1/24 Okeechobee due 8/1/24 Okeechobee Series 2004A New Yorkaia Pennsylvania Series 2004A Nichland (SC) due 6/1/15 Savannah Series 2004A	Massachusetts Massachusetts due 5/1/27
Miami Dade County Series 2008 Miami Dade County Series 2011 Michigan due 12/1/2013 Michigan due 12/1/2013 Mision, TX Series 2006 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 7/1/28 Nashville (Tennessee) due 8/01/31 Nebraska Nevada due 10/01/14 New Jersey due 6/01/15 New Jersey due 6/01/15 New Jersey due 6/01/15 New Jersey due 6/01/15 New York City due 12/1/17 New York City due 12/1/17 New York Series 2012 North Series 2012 North Carolina due 8/01/14 North Sumter, AL Ohio WDA due 11/1/22 Ohio WDA due 11/1/24 Okeechobee due 8/1/24 Okeechobee Series 2004A New York City due 2004A Pennsylvania Pennsylvan	Miami Dade County Series 2004B Miami Dade County Series 2006
Michigan due 8/1/2027 Michigan Strategic Fund Mission, TX Series 2006 Mississippi due 3/1/27 Mississippi due 3/1/27 Mississippi due 7/1/2017 Mississippi due 7/1/28 Nashville (Tennessee) due 8/01/31 Nebraska Nevada due 10/01/14 New Jersey due 6/01/15 New Jersey due 6/01/15 New Jersey due 6/01/15 New York City due 12/1/17 New York City due 12/1/17 New York City due 5/1/19 New York Series 2002A North Carolina due 8/01/14 North Sumter, AL Ohio WDA due 1/1/22 Ohio WDA due 1/1/24 Okeechobee due 8/1/24 Okeechobee due 8/1/24 Okeechobee due 8/1/24 Okeechobee Menter Pennsylvania Pe	Miami Dade County Series 2008 Miami Dade County Series 2011
Mississippi due 3/1/27 Mississippi due 3/1/29 Mississippi due 7/1/2017 Mississippi due 7/1/2017 Mississippi due 7/1/28 Nashville (Tennessee) due 8/01/31 Nebraska Nevada due 10/01/14 New Jersey due 6/01/15 New Jersey due 6/01/15 New York City due 12/1/17 New York City due 5/1/19 New York Series 2004A New York Series 2012 North Carolina due 8/01/14 North Sumter, AL Ohio WDA due 1/1/22 Ohio WDA due 7/1/21 Okeechobee due 8/1/24 Okeechobee Series 2004A Pennsylvania Pennsy	Michigan due 8/1/2027 Michigan Strategic Fund
Mississippi due 7/1/28 Nashville (Tennessee) due 8/01/31 Nebraska Nevada due 10/01/14 New Jersey due 6/01/15 New Jersey due 6/01/15 New York City due 5/1/19 New York City due 5/1/19 New York Series 2004A New York Series 2012 North Carolina due 8/01/14 North Sumter, AL Ohio WDA due 11/1/22 Ohio WDA due 7/1/21 Okeechobee due 8/1/24 Okeechobee Series 2004A Oklahoma Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Series 2004A Richland (SC) due 6/1/15 Savannah Series 2004A	Mississippi due 3/1/27 Mississippi due 3/1/29
Nevada due 10/01/14 New Jersey due 11/01/13 New Jersey due 6/01/15 New York City due 12/1/17 New York City due 5/1/19 New York Series 2004A New York Series 2012 North Carolina due 8/01/14 North Sumter, AL Ohio WDA due 11/1/22 Ohio WDA due 7/1/21 Okeechobee due 8/1/24 Okeechobee due 8/1/24 Okeechobee Series 2004A Oklahoma Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Sensylvania Sensylvania Pennsylvania Sensylvania Series 2009 Rhode Island Series 2004A Sichulkil/Pine Grove due 10/1/19	Mississippi due 7/1/28 Nashville (Tennessee) due 8/01/31
New Jerse'y due 6/01/15 New York City due 12/1/17 New York City due 12/1/17 New York Series 2004A New York Series 2012 North Carolina due 8/01/14 North Sumter, AL Ohio WDA due 11/1/22 Ohio WDA due 11/1/21 Okeechobee due 8/1/24 Okeechobee due 8/1/24 Okeechobee Series 2004A Oklahoma Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Sensylvania Sensylvania Series 2009 Rhode Island Series 2004A Richland (SC) due 6/1/15 Savannah Series 2004A	Nevada due 10/01/14 New Jersey due 11/01/13
New York Series 2012 North Carolina due 8/01/14 North Sumter, AL Ohio WDA due 11/1/22 Ohio WDA due 71/121 Okeechobee due 8/1/24 Okeechobee due 8/1/24 Okeechobee Series 2004A Oklahoma Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pensylvania Series 2009 Rhode Island Series 2009 Rhode Island Series 2004A Richland (SC) due 6/1/15 Savannah Series 2004A	New Jersey due 6/01/15 New York City due 12/1/17 New York City due 5/1/19
Ohio WDA due 11/1/22 Ohio WDA due 71/21 Okeechobee due 8/1/24 Okeechobee due 8/1/24 Okeachobee Series 2004A Oklahoma Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Series 2009 Rhode Island Series 2004A Richland (SC) due 6/1/15 Savannah Series 2004A Schuylkill/Pine Grove due 10/1/19	New York Series 2012
Okeechobee Series 2004A Oklahoma Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Series 2009 Rhode Island Series 2004A Richland (SC) due 6/1/15 Savannah Series 2004A Schuylkil/Pine Grove due 10/1/19	Ohio WDA due 11/1/22
Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Series 2009 Rhode Island Series 2004A Richland (SC) due 6/1/15 Savannah Series 2004A Schuylkil/Pine Grove due 10/1/19	Okeechobee Series 2004A
Pennsylvania Pennsylvania Pennsylvania Series 2009 Rhode Island Series 2004A Richland (SC) due 6/1/15 Savannah Series 2004A Schuylkill/Pine Grove due 10/1/19	Pennsylvania
Rhode Island Series 2004A Richland (SC) due 6/1/15 Savannah Series 2004A Schuylkill/Pine Grove due 10/1/19	Pennsylvania Pennsylvania
Schuylkill/Pine Grove due 10/1/19	Rhode Island Series 2004A
	Schuylkill/Pine Grove due 10/1/19

30,000,000	9/1/2027
14,000,000	10/1/2025
25,000,000	10/1/2025
10,000,000	10/1/2023
20,000,000	6/1/2023
19,890,000	9/1/2021
13,500,000	11/1/2015
30,000,000	2/1/2016
15,580,000	12/1/2031
10,200,000	4/1/2016
15,000,000	6/1/2014
15,000,000	5/1/2027
11,500,000	12/1/2018
11,500,000	12/1/2018
25,000,000	10/1/2018
25,000,000	9/1/2027
25,000,000	8/1/2023
20,000,000	11/1/2041
22,000,000	12/1/2013
35,000,000	8/1/2027
13,000,000	12/1/2013
41,750,000	12/1/2018
10,000,000	3/1/2027
10,000,000	3/1/2029
20,000,000	7/1/2017
10,000,000	7/1/2028
10,000,000	8/1/2031
10,000,000	11/1/2033
10,000,000	10/1/2014
20,000,000	11/1/2013
15,000,000	6/1/2015
10,000,000	6/1/2015
20,000,000	12/1/2017
25,000,000	5/1/2019
20,000,000	7/1/2017
25,000,000	5/1/2030
6,500,000	8/1/2014
4,350,000	10/1/2038
45,865,000	11/1/2022
15,000,000	7/1/2021
15,000,000	8/1/2024
15,970,000	7/1/2039
10,000,000	12/1/2021
40,000,000	9/1/2013
4,000,000	11/1/2021
20,000,000	11/1/2021
30,000,000	11/1/2021
14,000,000	10/1/2027
80,000,000	7/1/2021
100,000,000	12/1/2033
8,000,000	4/1/2016
10,000,000	6/1/2015
5,000,000	7/1/2015
11,700,000	10/1/2019
12,500,000	11/1/2016
12,500,000	11/1/2010

South Carolina 2008 Issue	15,000,000	
South Carolina Series 2003A	15,000,000	
State of New Hampshire	15,000,000	
State of New Hampshire due 5/1/27	20,000,000	
Sussex Co. Virginia	10,000,000	
Sussex County	10,000,000	
SW Illinois due 10/1/2027	4,700,000	
Tennessee - 2003	25,000,000	
Tennessee - 2012	18,000,000	
Texas due 8/1/20 (Mission EDC)	67,000,000	
Travis County (Texas 2003C)	12,000,000	
Washington due 10/1/25	13,650,000	
Washington due 10/1/25	13,650,000	
Washington due 10/1/27	20,000,000	
Washington due 11/1/2017	27,000,000	
Washington due 12/1/25	7,235,000	
Washington due 2/1/26	22,000,000	
Washington due 6/1/20	30,000,000	
Washington due 7/1/30	20,000,000	
Wisconsin Series 2003	50,000,000	
Wisconsin Series 2006A	30,000,000	
Wisconsin Series 2007A	20,000,000	
Wood County due 4/1/24	6,580,000	
Yavapai (Arizona) due 3/1/28	17,420,000	
Yavapai (Arizona) due 3/1/28	20,000,000	
Yavapai (Arizona) due 6/1/27	30,000,000	
Tax Exempt Revenue Bonds	\$ 2,511,630,000)
Tax Exempt Project Bonds:		
Concord Debt Series A	\$ 31,315,000	
Concord Debt Series B	4,925,000	
Gloucester Bonds	32,585,000	
Gloucester Bonds	6,930,000	
Massachusetts	10,000,000	
Tax Exempt Project Bonds	\$ 85,755,000)
Canada credit facility:		
Canada facility debt	\$ 23,770,000	
Canada Credit Facility	\$ 23,770,000)
Other		
Low Income Housing Tax Credit Investment Obligations	\$ 140,321,864	
King George Landfill Royalty Agreement	51,504,000	
	\$ 191,825,870	
Total Existing Indebtedness (a)	\$ 3,261,955,870)

(a) Excludes indebtedness incurred and scheduled payments made subsequent to June 30, 2013.

EXHIBIT 31.1

EXHIBIT 31.2

CERTIFICATION PURSUANT TO RULES 13a-14(a) AND 15d-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, David P. Steiner, certify that:

i. I have reviewed this report on Form 10-Q of Waste Management, Inc.;

ii. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

iii. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

iv. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a — 15(e) and 15d — 15 (e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a — 15(f) and 15d — 15(f)) for the registrant and have:

A. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

B. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

C. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

D. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

v. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):

A. All significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

B. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

By: /s/ DAVID P. STEINER

David P. Steiner President and Chief Executive Officer

Date: October 29, 2013

CERTIFICATION PURSUANT TO RULES 13a-14(a) AND 15d-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, James C. Fish, Jr., certify that:

i. I have reviewed this report on Form 10-Q of Waste Management, Inc.;

ii. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

iii. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

iv. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a — 15(e) and 15d — 15 (e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a — 15(f) and 15d — 15(f)) for the registrant and have:

A. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

B. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

C. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

D. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

v. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):

A. All significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

B. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting

/s/ JAMES C. FISH, JR. By: James C. Fish, Jr. Executive Vice President and Chief Financial Officer

Date: October 29, 2013

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Waste Management, Inc. (the "Company") on Form 10-Q for the period ended September 30, 2013 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, David P. Steiner, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that: (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ DAVID P. STEINER

David P. Steiner President and Chief Executive Officer

October 29, 2013

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350. AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Waste Management, Inc. (the "Company") on Form 10-Q for the period ended September 30, 2013 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, James C. Fish, Jr., Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

James C. Fish, Jr.

October 29, 2013

Mine Safety Disclosures

This exhibit contains certain specified disclosures regarding mine safety required by section 1503(a) of the Dodd-Frank Wall Street Reform and Consumer Protection Act and Item 104 of Regulation S-K. Certain of our subsidiaries have permits for surface mining operations that are incidental to excavation work for landfill development

During the quarter ended September 30, 2013, we did not receive any of the following; (a) a citation from the U.S. Mine Safety and Health Administration ("MSHA") for a violation of mandatory health or safety standards that could significantly and substantially contribute to the cause and effect of a coal or other mine safety or health hazard under section 104 of the Federal Mine Safety and Health Act of 1977 (the "Mine Safety Act"); (b) an order issued under section 104(b) of the Mine Safety Act; (c) a citation or order for unwarrantable failure of the mine operator to comply with mandatory health or safety standards under section 104(d) of the Mine Safety Act; (d) a flagrant violation under section 110(b)(2) of the Mine Safety Act; (e) an imminent danger order under section 107(a) of the Mine Safety Act; or (f) a proposed assessment from the MSHA.

In addition, during the quarter ended September 30, 2013, we had no mining-related fatalities, we had no pending legal actions before the Federal Mine Safety and Health Review Commission involving a coal or other mine, and we did not receive any written notice from the MSHA involving a pattern of violations, or the potential to have such a pattern, of mandatory health or safety standards that are of such nature as could have significantly and substantially contributed to the cause and effect of coal or other mine health or safety hazards under section 104(e) of the Mine Safety Act.

Acquisitions

Business Combinations [Abstract] Acquisitions

9. Acquisitions

Greenstar, LLC - On January 31, 2013, we paid \$170 million inclusive of certain adjustments, to acquire Greenstar, LLC ("Greenstar"). Pursuant to the sale and purchase agreement, up to an additional \$40 million is payable to the sellers during the period from 2014 to 2018; a portion of this consideration is contingent based on changes in certain recyclable commodity indexes. This additional consideration had a preliminary estimated fair value at closing of \$36 million. Greenstar was an operator of recycling and resource recovery facilities. This acquisition will provide the Company's customers with greater access to recycling solutions by supplementing the Company's extensive nationwide recycling network with the operations of one of the nation's largest private recyclers. For the three and nine months ended September 30, 2013, the Greenstar business recognized revenues of \$39 million and \$104 million, respectively, and net losses of \$2 million and \$9 million, respectively, which are included in our Condensed Consolidated Statements of Operations.

9 Months Ended

Sep. 30, 2013

http://markets.on.nytimes.com//research/modules/company_topic/drawFiling.asp?docKey... 11/25/2013

By: /s/ JAMES C. FISH, JR.

Executive Vice President and Chief Financial Officer

EXHIBIT 32.2

EXHIBIT 32.1

EXHIBIT 95

Goodwill of \$122 million was calculated as the excess of the consideration paid over the net assets recognized and represents the future economic benefits expected to arise from other assets acquired that could not be individually identified and separately recognized. Goodwill has been assigned predominantly to our Areas and, to a lesser extent, our recycling brokerage services, as they are expected to benefit from the synergies of the combination. Goodwill related to this acquisition is deductible for income tax purposes. Adjustments to the preliminary allocation of the purchase price of Greenstar were immaterial for the three and nine months ended September 30, 2013.

RCI Environnement, Inc. — On July 5, 2013, we paid C\$509 million, or \$481 million, to acquire substantially all of the assets of RCI Environnement, Inc. ("RCI"), the largest waste management company in Quebec, and certain related entities. Total consideration, inclusive of amounts for estimated working capital, was C\$515 million, or \$487 million. RCI provided collection, transfer, recycling and disposal operations throughout the Greater Montreal area. We acquired RCI to complement and expand the Company's existing assets and operations in Quebec. Since the acquisition date, the RCI business has recognized revenues of \$45 million and net income of \$7 million, which are included in our Condensed Consolidated Statement of Operations.

Goodwill of \$177 million was calculated as the excess of the consideration paid over the net assets recognized and represents the future economic benefits expected to arise from other assets acquired that could not be individually identified and separately recognized. Goodwill has been assigned to our Eastern Canada Area as it is expected to benefit from the synergies of the combination. A portion of goodwill related to this acquisition is deductible for income tax purposes in accordance with Canadian tax law.

The allocations of the purchase price for each acquisition are preliminary and subject to change based on the finalization of our detailed valuations. The preliminary purchase price allocations, as of the date acquired, were as follows (in millions):

	Greenstar	RCI
Accounts and other receivables	\$ 30	\$ 32
Part and supplies	4	_
Other current assets	2	_
Property and equipment	57	118
Goodwill	122	177
Other intangible assets	32	169
Accounts payable	(17)	_
Accrued liabilities	(12)	_
Deferred revenues	_	(4)
Landfill and environmental remediation liabilities	(2)	(1)
Deferred income taxes, net	_	_
Current portion of long-term debt	(4)	_
Long-term debt, less current portion	(1)	(4)
Other liabilities	(5)	
Total purchase price	\$ 206	\$487

The following table presents the preliminary allocations of the purchase price to other intangible assets (amounts in millions, except for amortization periods):

	0	Freenstar		RCI
		Weighted Average Amortization Periods		Weighted Average Amortization Periods
	Amount	(in Years)	Amount	(in Years)
Customer relationships	\$		\$ 162	15.0
Supplier relationships	31	10.0	_	_
Lease agreements	1	8.4	_	_
Trade name	_	-	7	5.0
Total other intangible assets subject to amortization	\$ 32	10.0	\$ 169	14.6

The following pro forma consolidated results of operations have been prepared as if the acquisitions of Greenstar and RCI occurred at January 1, 2012 (in millions, except per share amounts):

	Three	Three Months Ended		Aonths		
	En			ded		
	Septem	September 30,		tember 30, Septer		iber 30,
	2013	2012	2013	2012		
Operating revenues	\$3,623	\$3,548	\$10,585	\$10,493		
Net income attributable to Waste Management, Inc.	291	208	704	584		
Basic earnings per common share	0.62	0.45	1.51	1.26		
Diluted earnings per common share	0.62	0.45	1.50	1.26		

Other — During the nine months ended September 30, 2013, we also paid approximately \$47 million primarily to acquire ten businesses related to our collection and energy services operations. During the nine months ended September 30, 2012, we paid \$94 million for interests in oil and gas producing properties through two transactions. The purchase price was allocated primarily to "Property and equipment." Additionally, during the nine months ended September 30, 2012, we was acquired 21 other businesses related to our collection and recycling operations.

Restructuring - Additional		3 Months Ended				9 Months Ended			21 Months Ended
Information (Detail) (USD \$) In Millions, unless otherwise specified	Sep. 30, 2013	Sep. 30, 2012	Sep. 30, 2013	Sep. 30, 2012	E S	p. 30, 2013 Employee Severance Member]	Sep. 30, Emplo Severa [Memb	oyee Ince	Sep. 30, 2013 Employee Severance [Member]
Restructuring Cost and Reserve [Line Items]									
Pre-tax restructuring charges	\$3	\$44	\$13	\$ 51					
Pre-tax restructuring charges associated with this reorganization related to employee severance and benefit costs					6		46		
Employee severance and benefit costs from restructurings beginning in 2012								6	2
Employee severance and benefit cost payments through the balance sheet date								5	3
Remaining employee severance payable through our restructuring efforts					\$6			\$	6
Complection Separation activities								D	ec. 31, 2014
CONSOLIDATED		3 Mont	hs Ended	l	9 Mo	onths Ended			
STATEMENTS OF									
OPERATIONS (USD \$)	Ser	. 30. 201	3 Sep. 30.	2012 Sen	. 30. 2	013 Sep. 30, 2	012		
In Millions, except Per Share data, unless otherwise		,			,				
Income Stapacified[Abstract]									
Operating revenues	\$ 3	.621	\$ 3,461	\$ 10),483	\$ 10,215			
Costs and expenses:		, -	, .		- ,	, -			
Operating	2,3	25	2,229	6,84	45	6,655			
Selling, general and administrative	349)	335	1.09	92	1,116			
Depreciation and amortization	344	ł	331	1,00		971			
Restructuring	3		44	13		51			
(Income) expense from divestitures, asset impairments and unusual	items 23		22	38		55			

http://markets.on.nytimes.com//research/modules/company_topic/drawFiling.asp?docKey... 11/25/2013

Form 10-Q

Landfill and Environmental	9 Months Ended				
Cash dividends declared per common share	\$ 0.365	\$ 0.355	\$ 1.095	\$ 1.065	
Diluted earnings per common share	\$ 0.62	\$ 0.46	\$ 1.50	\$ 1.28	
Basic earnings per common share	\$ 0.62	\$ 0.46	\$ 1.50	\$ 1.28	
Net income attributable to Waste Management, Inc.	\$ 291	\$ 214	\$ 703	\$ 593	
Less: Net income attributable to noncontrolling interests	6	9	26	32	
Consolidated net income	297	223	729	625	
Provision for income taxes	155	125	368	329	
Income before income taxes	452	348	1,097	954	
Total other income (expense)	(125)	(152)	(392)	(413)	
Other, net	(3)	(14)	(12)	(16)	
Equity in net losses of unconsolidated entities	(3)	(17)	(19)	(35)	
Interest income		2	2	4	
Interest expense	(119)	(123)	(363)	(366)	
Other income (expense):					
Income from operations	577	500	1,489	1,367	
Total costs and expenses	3,044	2,961	8,994	8,848	

Remediation Liabilities

Basis of Presentation (Policies)

Accounting Policies [Abstract] Basis of Presentation

Text Block [Abstract]

Landfill and Environmental

Remediation Liabilities

2. Landfill and Environmental Remediation Liabilities

Liabilities for landfill and environmental remediation costs are presented in the table below (in millions):

		Septemb	oer 30, 2013			Decemb	er 31, 2012	
		Environmental				Envir	onmental	
	Landfill	Rem	ediation	Total	Landfill	Rem	ediation	Total
Current (in accrued liabilities)	\$ 99	\$	28	\$ 127	\$ 104	\$	28	\$ 132
Long-term	1,299		208	1,507	1,234		225	1,459
	\$1,398	\$	236	\$1,634	\$1,338	\$	253	\$1,591

Sep. 30, 2013

The changes to landfill and environmental remediation liabilities for the year ended December 31, 2012 and the nine months ended September 30, 2013 are reflected in the table below (in millions):

		Environmental
	Landfill	Remediation
December 31, 2011	\$1,292	\$ 273
Obligations incurred and capitalized	58	_
Obligations settled	(87)	(30)
Interest accretion	84	4
Revisions in cost estimates and interest rate assumptions	(8)	5
Acquisitions, divestitures and other adjustments	(1)	1
December 31, 2012	1,338	253
Obligations incurred and capitalized	45	_
Obligations settled	(53)	(15)
Interest accretion	65	3
Revisions in cost estimates and interest rate assumptions	—	(4)
Acquisitions, divestitures and other adjustments	3	(1)
September 30, 2013	\$1,398	\$ 236

At several of our landfills, we provide financial assurance by depositing cash into restricted trust funds or escrow accounts for purposes of settling final capping, closure, post-closure and environmental remediation obligations. Generally, these trust funds are established to comply with statutory requirements and operating agreements. See Note 14 for additional information related to these trusts.

9 Months Ended Sep. 30, 2013

The financial statements presented in this report represent the consolidation of Waste Management, Inc., a Delaware corporation; Waste Management's whollyowned and majority-owned subsidiaries; and certain variable interest entities for which Waste Management or its subsidiaries are the primary beneficiaries as described in Note 14. Waste Management is a holding company and all operations are conducted by its subsidiaries. When the terms "the Company," "we," "us" or "our" are used in this document, those terms refer to Waste Management, Inc., its consolidated subsidiaries and consolidated variable interest entities. When we use the term "WM," we are referring only to Waste Management, Inc., the parent holding company.

We are North America's leading provider of comprehensive waste management environmental services. We partner with our residential, commercial, industrial and municipal customers and the communities we serve to manage and reduce waste at each stage from collection to disposal, while recovering valuable resources and creating clean, renewable energy. Our "Solid Waste" business is operated and managed locally by our subsidiaries that focus on distinct geographic areas and provides collection, transfer, recycling and resource recovery, and disposal services. Through our subsidiaries, we are also a leading developer, operator and owner of waste-to-energy facilities in the United States.

We currently evaluate, oversee and manage the financial performance of our Solid Waste business subsidiaries through our 17 geographic Areas. Our Wheelabrator business provides waste-to-energy services and manages waste-to-energy facilities and independent power production plants. We also provide additional services that are not managed through our Solid Waste or Wheelabrator businesses, which are presented in this report as "Other." Additional information related to our segments can be found in Note 8.

The Condensed Consolidated Financial Statements as of September 30, 2013 and for the three and nine months ended September 30, 2013 and 2012 are unaudited. In the opinion of management, these financial statements include all adjustments, which, unless otherwise disclosed, are of a normal recurring nature, necessary for a fair presentation of the financial position, results of operations, comprehensive income, cash flows, and changes in equity for the periods presented. The results for interim periods are not necessarily indicative of results for the entire year. The financial statements presented herein should be read in connection with the financial statements included in our Annual Report on Form 10-K for the year ended December 31, 2012.

In preparing our financial statements, we make numerous estimates and assumptions that affect the accounting for and recognition and disclosure of assets, liabilities, equity, revenues and expenses. We must make these estimates and assumptions because certain information that we use is dependent on future events, cannot be calculated with precision from available data or simply cannot be calculated. In some cases, these estimates are difficult to determine, and we must exercise significant judgment. In preparing our financial statements, the most difficult, subjective and complex estimates and the assumptions that present the greatest amount of uncertainty relate to our accounting for landfills, environmental remediation liabilities, asset impairments, deferred income taxes and reserves associated with our insured and self-insured claims. Actual results could differ materially from the estimates and assumptions that we use in the preparation of our financial statements.

Comprehensive Income

Comprehensive Income — In February 2013, the Financial Accounting Standards Board issued amended authoritative guidance associated with comprehensive income, which requires companies to provide information about the amounts that are reclassified out of accumulated other comprehensive income by component. Additionally, companies are required to present significant amounts reclassified out of accumulated other comprehensive income by the respective line items of net

Operating revenues

Income from operations

income. The amendment to authoritative guidance associated with comprehensive income was effective for the Company on January 1, 2013. The adoption of this guidance did not have a material impact on our consolidated financial statements. We have presented the information required by this amendment in Note 12. Reclassifications Reclassifications When necessary, reclassifications have been made to our prior period consolidated financial information in order to conform to the current year presentation. Accumulated Other **Comprehensive Income -**3 Months Ended 9 Months Ended Other Comprehensive Income (Loss) Before **Reclassifications Associated** With Cash Flow Derivative Instruments (Detail) (USD \$) Sep. 30, 2013 Sep. 30, 2012 Sep. 30, 2013 Sep. 30, 2012 In Millions, unless otherwise specified Derivative Instruments and Hedging Activities Disclosures [Line Items] Other comprehensive income gain (loss) on derivatives before tax \$ (9) \$(21) \$16 \$ (41) Tax (expense) benefit 4 8 (6) 16 Net of tax (5)(13)10 (25)Interest Rate Swaps [Member] | Cash Flow Hedging [Member] Derivative Instruments and Hedging Activities Disclosures [Line Items] 11 Other comprehensive income gain (loss) on derivatives before tax (1)(3) (28)Foreign currency derivatives[Member] | Cash Flow Hedging [Member] Derivative Instruments and Hedging Activities Disclosures [Line Items] Other comprehensive income gain (loss) on derivatives before tax (15)11 (14)(8) Electricity commodity derivatives [Member] | Cash Flow Hedging [Member] Derivative Instruments and Hedging Activities Disclosures [Line Items] Other comprehensive income gain (loss) on derivatives before tax \$1 \$ (6) \$ (3) 9 Months Ended Restructuring Sep. 30, 2013 **Restructuring And Related** Activities [Abstract] Restructuring 10. Restructuring During the nine months ended September 30, 2013, we recognized a total of \$13 million of pre-tax restructuring charges, of which \$6 million was related to employee severance and benefit costs, including costs associated with our acquisition of Greenstar and our July 2012 restructuring. The remaining charges were primarily related to operating lease obligations for property that will no longer be utilized. During the nine months ended September 30, 2012, we recognized a total of \$51 million of pre-tax restructuring charges, of which \$46 million was related to employee severance and benefit restructuring charges, primarily attributable to our July 2012 restructuring and including other charges associated with the reorganization of Oakleaf and certain other actions taken by the Company in 2012. Through September 30, 2013, we had recognized charges of \$62 million related to employee severance and benefits associated with our restructuring efforts beginning in 2012 and we have paid approximately \$53 million of these costs. At September 30, 2013, we had approximately \$6 million of accrued employee severance related to our restructuring efforts, which will be paid through the end of 2014. Segment and Related 3 Months Ended 9 Months Ended Information - Reportable Segments (Detail) (USD \$) Sep. 30, 2013 Sep. 30, 2012 Sep. 30, 2013 Sep. 30, 2012 In Millions, unless otherwise Segment Reporting Information [Line Items] Operating revenues \$ 3,621 \$ 3,461 \$ 10,483 \$ 10.215 577 1,489 1,367 Income from operations 500 Solid Waste: Tier 1 [Member] Segment Reporting Information [Line Items] 758 716 2,215 2,145 Operating revenues 224 213 652 641 Income from operations Solid Waste: Tier 2 [Member] Segment Reporting Information [Line Items] Operating revenues 1.359 1.306 3.928 3.893 347 328 970 946 Income from operations Solid Waste: Tier 3 [Member] Segment Reporting Information [Line Items] Operating revenues 783 725 2,213 2,142 Income from operations 148 126 388 370 Wheelabrator [Member] Segment Reporting Information [Line Items] 185 188 551 539 Operating revenues Income from operations 39 46 52 68 Other [Member] Segment Reporting Information [Line Items] 1.576 1.496 Operating revenues 536 526 (31) (99) (167) (68)Income from operations Operating Group Total [Member] Segment Reporting Information [Line Items]

3.621

727

3 4 6 1

645

10.483

1 963

10.215

1 858

Corporate and Other [Member]							
Segment Reporting Information [Line Items]							
Income from operations	(150)	(145)	(474)	(491)			
Operating Segments [Member]							
Segment Reporting Information [Line Items]							
Operating revenues	4,283	4,083	12,379	11,948			
Operating Segments [Member] Solid Waste: Tier 1 [Member]							
Segment Reporting Information [Line Items]							
Operating revenues	898	854	2,632	2,531			
Operating Segments [Member] Solid Waste: Tier 2 [Member]							
Segment Reporting Information [Line Items]							
Operating revenues	1,670	1,596	4,827	4,698			
Operating Segments [Member] Solid Waste: Tier 3 [Member]							
Segment Reporting Information [Line Items]							
Operating revenues	940	862	2,637	2,523			
Operating Segments [Member] Wheelabrator [Member]							
Segment Reporting Information [Line Items]	214	21 0	(2)	(21			
Operating revenues	214	218	634	631			
Operating Segments [Member] Other [Member]							
Segment Reporting Information [Line Items]	561	553	1,649	1,565			
Operating revenues Operating Segments [Member] Operating Group Total [Member]	501	555	1,049	1,505			
Segment Reporting Information [Line Items]							
Operating revenues	4,283	4,083	12,379	11,948			
Intercompany Operating Revenues [Member]	4,205	4,005	12,577	11,940			
Segment Reporting Information [Line Items]							
Operating revenues	(662)	(622)	(1,896)	(1,733)			
Intercompany Operating Revenues [Member] Solid Waste: Tier 1 [Member]	()	()	(-,-,-)	(1,100)			
Segment Reporting Information [Line Items]							
Operating revenues	(140)	(138)	(417)	(386)			
Intercompany Operating Revenues [Member] Solid Waste: Tier 2 [Member]							
Segment Reporting Information [Line Items]							
Operating revenues	(311)	(290)	(899)	(805)			
Intercompany Operating Revenues [Member] Solid Waste: Tier 3 [Member]							
Segment Reporting Information [Line Items]							
Operating revenues	(157)	(137)	(424)	(381)			
Intercompany Operating Revenues [Member] Wheelabrator [Member]							
Segment Reporting Information [Line Items]							
Operating revenues	(29)	(30)	(83)	(92)			
Intercompany Operating Revenues [Member] Other [Member]							
Segment Reporting Information [Line Items]			(50)	(10)			
Operating revenues	(25)	(27)	(73)	(69)			
Intercompany Operating Revenues [Member] Operating Group Total [Member]	rj						
Segment Reporting Information [Line Items]	¢ (((2))	¢ ((22))	¢ (1.90C)	¢ (1 722)			
Operating revenues Accumulated Other	\$ (662)	\$ (622)	\$ (1,896)	\$ (1,733)			
Comprehensive Income - Reclassification of				3 Mont	hs Ended	9 Mont	hs Ended
Component of Accumulated							
Other Comprehensive				Sep. 30,	Sep. 30,	Sep. 30,	Sep. 30,
Income (Detail) (USD \$)				2013	2012	2013	2012
In Millions, unless otherwise							
Derivative Specified							
Interest expense				\$ (119)	\$ (123)	\$ (363)	\$ (366)
Other, net				3	14	12	16
Operating revenues Total real-positions for the period. Tey (expanse) hepefit				3,621	3,461	10,483	10,215
<u>Total reclassifications for the period, Tax (expense) benefit</u> Total reclassifications for the period, Net of tax				155	125	368	329
Amount Reclassified from Accumulated Other Comprehensive Income [Membe	-r]					(2)	
Derivative Instruments, Gain (Loss) [Line Items]	**1						
Total reclassifications for the period, before tax				(11)	(16)	(3)	(16)
Total reclassifications for the period, Tax (expense) benefit				4	6	1	6
Total reclassifications for the period, Net of tax				(7)	(10)	(2)	(10)
Interest Rate Swaps [Member] Amount Reclassified from Accumulated Other	Comprehensiv	e Income [Mei	mber] Cash	. /	. /	、 <i>′</i>	
· · · · ·							
Flow Hedging [Member]	comprenensi	-					
Flow Hedging [Member] Derivative Instruments, Gain (Loss) [Line Items]	Comprenensi	-					
	Comprehensiv	-		(2)	(2)	(6)	(2)
Derivative Instruments, Gain (Loss) [Line Items]		e Income [Men	nber] Cash	(2)	(2)	(6)	(2)

Form 10-Q

Derivative Instruments, Gain (Loss) [Line	Items]							
Interest expense	<u>rtems</u>					(2)	(1)	(6)
Foreign currency derivatives[Member] Amo [Member] Cash Flow Hedging [Member]	unt Reclassified f	rom Accumul	ated Other Comp	rehensive Income				
Derivative Instruments, Gain (Loss) [Line Other, net	<u>Items]</u>				(9)	(14)	10	(17)
Electricity commodity derivatives [Member] [Member] Cash Flow Hedging [Member]	Amount Reclass	ified from Ac	cumulated Other	Comprehensive Income		(1)	10	(17)
Derivative Instruments, Gain (Loss) [Line Operating revenues	Items]					\$ 2	\$ (6)	\$ 9
Debt - Components of Debt (Parenthetical) (Detail) (USD	9 Months En	ded				÷ –	+ (0)	Ŧ 2
\$) In Billions, unless otherwise specified	Sep. 30, 20	13 Jul. 31, 2	013 Dec. 31, 201	2				
Capital leases and other [Member] Debt Instrument [Line Items]								
Interest rate upper range	12.00%							
End period of maturity for debt instrument	Dec. 31, 2055							
Senior Notes [Member]	2001 21, 2000							
Debt Instrument [Line Items]								
Weighted average interest rate	5.70%		5.70%					
Interest rate lower range	2.60%							
Interest rate upper range	7.75%							
End period of maturity for debt instrument Nontaxable Municipal Bonds [Member]	Nov. 30, 2039)						
Debt Instrument [Line Items]								
Weighted average interest rate	2.40%		2.80%					
Interest rate lower range	0.10%							
Interest rate upper range	5.70%							
End period of maturity for debt instrument	Nov. 01, 2045	5						
Revolving credit facility [Member]								
Debt Instrument [Line Items]	¢ 2.25	¢ 0.05						
Revolving credit facility, aggregate capacity Weighted average interest rate	\$ 2.25 1.50%	\$ 2.25	1.40%					
Maturity date of credit facility	Jul. 26, 2018		1.40%					
Standby Letters of Credit [Member]	Jul. 20, 2010							
Debt Instrument [Line Items]								
End period of maturity for debt instruments	Jun. 30, 2015							
Canadian credit facility and term loan [Memb								
Debt Instrument [Line Items]								
Weighted average interest rate	2.70%		2.90%					
Maturity date of credit facility	Nov. 07, 2017	7						
Derivative Instruments and				9 Months Ended				
Hedging Activities (Tables)				Sep. 30, 2013				
Derivative Instruments And Hedging Activities Disclosure [Abstract]								
Fair Values of Derivative Instruments Recorded in Balance Sheet	The followin	g table summariz	tes the fair values of c	lerivative instruments recorded	in our Conder			
			edging Instruments	Balance Sheet Location	L	September 30, 2013	Decembe 2012	
		city commodity d		Current other assets		<u>\$ </u>	\$	1
		tal derivative as rate derivatives		Current accrued liabilities		<u>\$</u>	\$	
	Electric	city commodity d	lerivatives	Current accrued liabilities		3	-	5
		a currency deriva rate derivatives	tives	Current accrued liabilities Long-term other liabilities			_	11 42
		otal derivative lia	bilities			\$ 34	\$	58
Impact of Periodic Settlements of Active	Th. 6.11	a tabla arress '	and the imperies of the state	dia antilamento - Ct'		d the large of	f tomain - t - 1	
Swap Agreements and Impact of Terminated Swap Agreements on Results of Operations	on our results of op			odic settlements of active swap	agreements an	u uie impact o	i terminated sw	ap agreements
s map regreements on results of Operations					TI	ee Months	Nine M	onths
	Decrease to Inf	arest Expense Due t	o Hedge Accounting for 1	interact Rate Swane		September 30, 2012	Ended Sept	
	Periodic settl	ements of active	swap agreements(a)		\$	\$	\$	\$ 7
	Terminated s	wap agreements			5	7	15 \$ 15	$\frac{16}{\$ 23}$
					φ J	φ /		
	Our variable-	rate obligations	were based on a sprea	riable-rate interest obligations a ad from the three-month LIBOR	. These settler			
			p portfolio with a no	tional amount of \$1 billion in A Months Ended				
Debt (Tables)			,	Sep. 30, 2013				
Debt Disclosure [Abstract]								

	ving table summarizes the major components of d ory as of September 30, 2013:	ebt (in millions) at eac	h balance sheet date	e and provides the m	aturities and interest rate ranges of
				September 30,	December 31,
	\$2.25 billion revolving credit facility, maturin, interest rate of 1.5% at September 30, 2013 Letter of credit facilities, maturing through Jut Canadian credit facility and term loan, maturin average effective interest rate of 2.7% at Se	and 1.4% at December ne 2015 ng November 2017 (we	r 31, 2012) eighted	2013 \$ 140	\$ 400
	December 31, 2012) Senior notes maturing through 2039, interest r	ates ranging from 2.60	% to 7.75%	456	75
	 (weighted average interest rate of 5.7% at S December 31, 2012) Tax-exempt bonds, maturing through 2045, fi: ranging from 0.1% to 5.7% (weighted avera September 30, 2013 and 2.8% at December Capital leases and other, maturing through 205 Current portion of long-term debt 	ked and variable interest age interest rate of 2.4% 31, 2012)	st rates 6 at	$ \begin{array}{r} 6,292 \\ 2,719 \\ 452 \\ 10,059 \\ 568 \\ \underline{\$ 9,491} \\ 9 Month $	6,305 2,727 409 9,916 743 \$ 9,173 as Ended
Commitments and				Sep. 3	·
Contingencies - Additional information (Detail) (USD \$)				si Lan	te Afill
					owners
<u>Commitments And Contingencies [Line Iter</u> <u>Term of Guarantees of financial obligations</u>	<u>ms]</u>		Guarantees of u maturing through		ntities' financial obligations
Maximum potential future payments regarding Approximate number of homeowners' properti agreements guaranteeing market value			\$ 10,000,000 850		
Number of landfills adjacent to or near homeo value	wners' properties with agreements guara	nteeing market	21		
High end estimate of reasonably possible additional liabilities	tional losses associated with environment	tal remediation	190,000,000		
Environmental remediation liability			236,000,000		
Number of sites listed on the EPA's NPL for w	which we have been notified we are a PRI	2	78		
Number of owned sites listed on the EPA's NP			15		
<u>Number of non-owned sites listed on the EPA</u> Dollar threshold for environmental matters rec			63		
<u>S-K</u>		SEC 8 Regulation	100,000		
Approximate percentage of workforce covered Estimate of possible loss 2012 Audit [Member]	by collective bargaining agreements		20.00% \$ 40,000,000		
Commitments And Contingencies [Line Iter	<u>ns]</u>				
Number of months expected for IRS audit(s) to	o be completed		6 months		
2013 Audit [Member]	mel				
Commitments And Contingencies [Line Iter Number of months expected for IRS audit(s) to			18 months		
Basis of Presentation -		9 Months Ended	To monuto		
Additional Information (Detail)		Sep. 30, 2013			
Organization Consolidation And Presentati	on Of Financial Statements [Abstract]	Areas			
Number of Geographical areas	~~~	17			
Derivative Instruments and Hedging Activities - Fair Values of Derivative					
Instruments Recorded in Balance Sheet (Detail) (USD			Sep. 3	30, 2013 Dec. 31,	2012
\$) In Millions, unless otherwise specified					
Derivatives, Fair Value [Line Items]					
Electricity commodity derivatives				\$ 1	
Derivative assets designated as hedging instru- Interest rate derivatives	ments		31	1 42	
Electricity commodity derivatives			3	5	
Foreign currency derivatives				11	
Derivative liabilities designated as hedging ins		uu Uodaina D.C.	34	58	
Current other assets [Member] Electricity cor Derivatives, Fair Value [Line Items]	innounty derivatives [Member] Cash Flo	ow neuging [Mem]	ber]		
Electricity commodity derivatives				1	
Current accrued liabilities [Member] Electric Derivatives, Fair Value [Line Items]	ity commodity derivatives [Member] Ca	ash Flow Hedging	[Member]		
Electricity commodity derivatives			3	5	

http://markets.on.nytimes.com//research/modules/company_topic/drawFiling.asp?docKey... 11/25/2013

Form 10-Q

Current accrued liabilities [Member Derivatives, Fair Value [Line Iten		ate Swaps [M	[ember] Casl	n Flow Hedg	ing [Member]]			
Interest rate derivatives	1.115						31		
Current accrued liabilities [Member Derivatives, Fair Value [Line Iten Foreign currency derivatives		arrency deriv	atives[Membe	er] Cash Fio	w Hedging []	Memberj	11	L	
Long-term accrued liabilities [Mem Derivatives, Fair Value [Line Item		t Rate Swaps	[Member] C	Cash Flow He	dging [Mem]	ber]	1.		
Interest rate derivatives							\$	42	
	9	Months End	led	1 Months Ended	3 Months Ended	9 Months Ended	1 Mont	hs Ended	9 Months Ended
Acquisitions - Additional			Sep. 30,	Jan. 31,	Sep. 30,	Sep. 30,	Jul. 05, 2013	Jul. 05, 2013	Sep. 30, 2013
Information (Detail) In Millions, unless otherwise	Sep. 30, 2013	Sep. 30, 2012	2012 Oil and Gas	2013 Greenstar	2013 Greenstar	2013 Greenstar	RCI	RCI	RCI
specified	USD (\$)	USD (\$) Operations	Properties [Member]	LLC [Member]	LLC [Member]	LLC [Member]	Environnement Inc [Member] USD (\$)	Environnement Inc [Member] CAD	Environnement Inc [Member] USD (\$)
Business Acquisition [Line Items]			USD (\$)	USD (\$)	USD (\$)	USD (\$)			
Acquisition cost of acquired entity	\$ 698	\$ 178		\$ 170			\$ 481	509	
Additional consideration payable Estimated fair value of liability for				40					
additional cash payments related to acquisitions				36					
Revenue recognized since the acquisition date					39	104			45
Income (Loss) recognized since the acquisition date					(2)	(9)			7
Allocation of purchase price to Goodwill				122					177
Total consideration paid, inclusive of certain adjustment							487	515	
Number of collection and energy services operations acquired	10								
Acquisitions of interests in oil and gas properties			94						
Number of collection and recycling operations acquired		21							
Payments to acquire collection and	\$ 47								
energy services operations Accumulated Other	ψ 17					9 Months E	ndad		
Comprehensive Income						Sep. 30, 20			
(Tables) <u>Equity [Abstract]</u>						. /			
Component of Accumulated Other Comprehensive Income		of Waste Mana						net of tax, which is inc heses representing deb	
		r					Nine Mont	hs Ended September 30, 20	13(a)
							Unrealized Gains and		
						Los	ns and Losses on Available- ivative for-Sale	Foreign Currency Translation	Post- Retirement Benefit
		Balance, Decer	mber 31, 2012				uments Securities (74) \$ 4	Adjustments \$ 276	Plans Total \$ (13) \$193
			nprehensive inco reclassified from				10 1	(37)	— (26)
		income Net current per Balance, Septe	riod other compr	ehensive incom	ne (loss)		$\begin{array}{c} 2 \\ \hline 12 \\ \hline (62) \end{array} \qquad \begin{array}{c} - \\ \hline 1 \\ \hline 5 \\ \hline 5 \\ \end{array}$	(37) \$ 239	$\frac{-}{\frac{-}{13}}$ $\frac{2}{(24)}$
Other Comprehensive Income (Loss Reclassifications Associated With C			unts of other cor	nprehensive inc	come (loss) befo	ore reclassificati		r cash flow derivative i	
Derivative Instruments								nized in OCI (Effective Por	
		Derivative	s Designated as Cas	h Flow Hodges	2013	Three Months Ended September		Nine Mor Ended Septer 2013	nths nber 30, 2012
		Forward	starting interest currency derivati	rate swaps	\$	(1) (8)	\$ (3) (15)	\$ 11 11	\$ (28) (14)
		Electricit	y commodity de		<u></u>		(3)	(6)	1
		Tax	al before tax (expense) benef	fit	\$	(9)	$\frac{(21)}{8}$	\$ 16 (6)	\$ (41) <u>16</u> <u>2</u>
Reclassification of Component of A	ccumulated		of tax	reclassified out	of each compor	(5)	sted other comprehens	<u>\$ 10</u> ive income are as follow	\$ (25)
Other Comprehensive Income		i ne sigli	meant amounts i	icerassineu olli	or each compor	An	nount Reclassified from Acc Other Comprehensive Inco	umulated	as (m mmons).
		Details about Accu Income Componer	imulated Other Con hts	prehensive		Er	nded nber 30, Se	Ended eptember 30,	Statement of Operations Classification
		Forward	es on cash flow l -starting interest			\$ (2)	2012 2013 \$ (2) \$ (0	6) \$ (2) Int	erest expense
		Treasury	rate locks			_	(2) (1) (6) Int	erest expense

	Foreign currency derivatives Electricity commodity derivatives		(9) $\overline{{\$ (11)}}$	$ \begin{array}{r} (14) \\ \underline{2} \\ \$ (16) \\ 6 \end{array} $		(17) $\frac{9}{\$ (16)}$ 6	Other, net Operating revenues Total before tax Tax (expense) benefit
	Total reclassifications for the pe	eriod	<u><u></u> <u></u> <u></u> <u></u> (7)</u>	<u>\$ (10)</u>	<u><u></u>(2)</u>	<u>\$ (10</u>)	Net of tax
Condensed Consolidating	(a) Amounts in parentheses represe	nt debits to the s	statement of operati	ions classificati	on.		
Financial Statements - Condensed Consolidating		9 Mo	nths Ended				
Statements of Cash Flows (Detail) (USD \$)		Sep. 30, 20)13 Sep. 30, 201	2			
In Millions, unless otherwise		- /	- '				
Cash flowspecificherating activities:		\$ 720	\$ 625				
<u>Consolidated net income</u> Equity in earnings of subsidiaries, net of taxes		\$ 729	\$ 625				
Other adjustments		1,129	1,093				
Net cash provided by (used in) operating activities		1,858	1,718				
Cash flows from investing activities:		((09)	(179)				
Acquisitions of businesses, net of cash acquired Capital expenditures		(698) (824)	(178) (1,132)				
Proceeds from divestitures of businesses (net of cash	n divested) and other sales of asse		28				
Net receipts from restricted trust and escrow account		(23)	(79)				
Net cash provided by (used in) investing activities		(1,432)	(1,361)				
Cash flows from financing activities:		1 219	(95				
New borrowings Debt repayments		1,218 (1,244)	685 (473)				
Cash dividends		(512)	(493)				
Exercise of common stock options		116	39				
Distributions paid to noncontrolling interests and oth		(37)	23				
(Increase) decrease in intercompany and investment	<u>s, net</u>	(150)	(210)				
<u>Net cash provided by (used in) financing activities</u> Effect of exchange rate changes on cash and cash eq	uivalents	(459) (4)	(219) 2				
Increase (decrease) in cash and cash equivalents		(37)	140				
Cash and cash equivalents at beginning of period		194	258				
Cash and cash equivalents at end of period		157	398				
WM [Member]							
Cash flows from operating activities: Consolidated net income		703	593				
Equity in earnings of subsidiaries, net of taxes		(865)	(755)				
Other adjustments		(19)	75				
Net cash provided by (used in) operating activities		(181)	(87)				
Cash flows from investing activities: Acquisitions of businesses, net of cash acquired							
Capital expenditures							
Proceeds from divestitures of businesses (net of cash	n divested) and other sales of asse	ts					
Net receipts from restricted trust and escrow account	ts and other, net						
Net cash provided by (used in) investing activities Cash flows from financing activities:							
New borrowings		200	645				
Debt repayments		(460)	(335)				
Cash dividends		(512)	(493)				
Exercise of common stock options Distributions paid to noncontrolling interests and oth	20r	116 12	39 14				
(Increase) decrease in intercompany and investment:		765	14 368				
Net cash provided by (used in) financing activities		121	238				
Effect of exchange rate changes on cash and cash eq	uivalents						
Increase (decrease) in cash and cash equivalents		(60)	151				
Cash and cash equivalents at beginning of period Cash and cash equivalents at end of period		60	119 270				
WM Holdings [Member]			210				
Cash flows from operating activities:							
Consolidated net income		865	755				
Equity in earnings of subsidiaries, net of taxes Other adjustments		(881)	(770) (8)				
Net cash provided by (used in) operating activities		(8) (24)	(8)				
Cash flows from investing activities:		<u>(= ·)</u>	(==)				
Acquisitions of businesses, net of cash acquired							
Capital expenditures	1						
Proceeds from divestitures of businesses (net of cash	a divested) and other sales of asse	<u>ts</u>					

Net receipts from restricted trust and escrow accounts and other, net		
Net cash provided by (used in) investing activities		
Cash flows from financing activities:		
New borrowings		
Debt repayments		
Cash dividends		
Exercise of common stock options		
Distributions paid to noncontrolling interests and other		
(Increase) decrease in intercompany and investments, net	24	23
Net cash provided by (used in) financing activities	24	23
Effect of exchange rate changes on cash and cash equivalents		
Increase (decrease) in cash and cash equivalents		
Cash and cash equivalents at beginning of period		
Cash and cash equivalents at end of period		
Non-Guarantor Subsidiaries [Member]		
Cash flows from operating activities: Consolidated net income	907	802
Equity in earnings of subsidiaries, net of taxes	907	802
Other adjustments	1,156	1,026
Net cash provided by (used in) operating activities	2,063	1,828
Cash flows from investing activities:	2,000	1,020
Acquisitions of businesses, net of cash acquired	(698)	(178)
Capital expenditures	(824)	(1,132)
Proceeds from divestitures of businesses (net of cash divested) and other sales of a		28
Net receipts from restricted trust and escrow accounts and other, net	(23)	(79)
Net cash provided by (used in) investing activities	(1,432)	(1,361)
Cash flows from financing activities:		
New borrowings	1,018	40
Debt repayments	(784)	(138)
Cash dividends		
Exercise of common stock options		
Distributions paid to noncontrolling interests and other	(49)	9
(Increase) decrease in intercompany and investments, net	(789)	(391)
Net cash provided by (used in) financing activities	(604)	(480)
Effect of exchange rate changes on cash and cash equivalents	(4)	2
Increase (decrease) in cash and cash equivalents	23	(11)
Cash and cash equivalents at beginning of period	134	139
Cash and cash equivalents at end of period	157	128
Eliminations [Member] Cash flows from operating activities:		
Consolidated net income	(1,746)	(1,525)
Equity in earnings of subsidiaries, net of taxes	1,746	1,525
Other adjustments	1,7 10	1,020
Net cash provided by (used in) operating activities		
Cash flows from investing activities:		
Acquisitions of businesses, net of cash acquired		
Capital expenditures		
Proceeds from divestitures of businesses (net of cash divested) and other sales of a	assets	
Net receipts from restricted trust and escrow accounts and other, net		
Net cash provided by (used in) investing activities		
Cash flows from financing activities:		
New borrowings		
Debt repayments		
Cash dividends		
Exercise of common stock options		
Distributions paid to noncontrolling interests and other		
(Increase) decrease in intercompany and investments, net		
Net cash provided by (used in) financing activities		
Effect of exchange rate changes on cash and cash equivalents		
Increase (decrease) in cash and cash equivalents		
Cash and cash equivalents at end of period		
Cash and cash equivalents at end of period Condensed Consolidating 3 Man	ths Ended	9 Months Ended
		9 Months Ended Sep. 30, 2013 Sep. 30, 2012
Condensed Consolidating	5 Sep. 50, 2012 S	лр. 30, 2013 эср. 30, 2012
Statements of		
Comprehensive Income (Detail) (USD \$)		
$(2 \text{ cm}) (0 \text{ bb} \psi)$		

In Millions, unless otherwise

specified								
Condensed Financial Statements, Captions	[Line Items]							
Comprehensive income	<u> </u>	\$ 332	\$ 260	\$ 705	\$ 653			
Less: Comprehensive income attributable to n	oncontrolling interests		9	26	32			
Comprehensive income attributable to Waste	-	326	251	679	621			
WM [Member]								
Condensed Financial Statements, Captions	[Line Items]							
Comprehensive income	<u> </u>	292	215	714	581			
Less: Comprehensive income attributable to n	oncontrolling interests							
Comprehensive income attributable to Waste		292	215	714	581			
WM Holdings [Member]								
Condensed Financial Statements, Captions	[Line Items]							
Comprehensive income	<u> </u>	344	269	865	755			
Less: Comprehensive income attributable to n	oncontrolling interests							
Comprehensive income attributable to Waste		344	269	865	755			
Non-Guarantor Subsidiaries [Member]								
Condensed Financial Statements, Captions	[Line Items]							
Comprehensive income		389	319	872	842			
Less: Comprehensive income attributable to n	oncontrolling interests		9	26	32			
Comprehensive income attributable to Waste	-	383	310	20 846	810			
Eliminations [Member]			210	0.0	5.0			
Condensed Financial Statements, Captions	[Line Items]							
Comprehensive income		(693)	(543)	(1,746)	(1,525)			
Less: Comprehensive income attributable to n	oncontrolling interests	. ,	(0.0)	(1,710)	(1,0=0)			
Comprehensive income attributable to Waste		\$ (693)	\$ (543)	\$ (1,746)	\$ (1,525)			
Derivative Instruments and		+ (0) 0)	+ (0.10)	+ (-,)	+ (-,)			
Swap Agreements on Results of Operations (Parenthetical) (Detail) (USD \$) In Billions, unless otherwise								
specified								
Derivative Instruments And Hedging Activ	ities Disclosure [Abst							
Notional amount of interest rate swaps		\$ 1						
Landfill and Environmental Remediation Liabilities (Tables)				9 Month Sep. 3	ns Ended 0. 2013			
Text Block [Abstract]					,			
Liabilities for Landfill and Environmental								
Remediation Costs	Liabilities for la	indfill and enviro	onmental remedi	ation costs are pre-	sented in the table bel	ow (in millions):		
					September 30, 2013		December 31, 2012	
				Land	Environmental fill Remediation	Total Landfill	Environmental I Remediation	Total
	Current (in accrued li	abilities)		\$ 9	99 \$ 28	\$ 127 \$ 104	\$ 28	\$ 132
	Long-term			1,29		$\frac{1,507}{\$1,634}$ $\frac{1,234}{\$1,338}$		$\frac{1,459}{\$1,591}$
Changes to Londfill and Environmental				\$1,5		\$1,054 \$1,550	\$ <u>\$</u> <u>255</u>	<u>\$1,571</u>
Changes to Landfill and Environmental Remediation Liabilities	The changes to	landfill and envi	ronmental remed	diation liabilities for	or the year ended Dec	ember 31, 2012 and t	the nine months ended	d
Concentration Enconnects	September 30, 2013 a	re reflected in th	e table below (in	n millions):				
						E	Invironmental	
	Daa	ember 31, 2011					Remediation	
	Dec		urred and capital	lized		58		
		Obligations set				(87)	(30)	
		Interest accretic Revisions in co		interest rate assum	ptions	84 (8)	4 5	
			vestitures and o	ther adjustments		(1)	1	
	Dec	obligations inc	urred and capital	lized		1,338 45	253	
		Obligations set	led	lized		(53)	(15)	
		Interest accretic		interest rate essen	ntions	65	3	
				interest rate assum ther adjustments	ipuolis	3	(4) (1)	
CONDENSED	Sept	tember 30, 2013				\$1,398	236	
CONSOLIDATED			3	3 Months Ende	d 9 Me	onths Ended		
STATEMENTS OF			•					
COMPREHENSIVE INCOME (Parenthetical)			Son 1	10 2013 San 24	0 2012 San 20 2	013 Sen 20 2012	,	
(USD \$)			Sep. 2	10, 2013 Sep. 30	, 2012 Sep. 30, 2	013 Sep. 30, 2012	-	
In Millions, unless otherwise	The same field of the							
Statement Of Jacome And Comprehensive		in vol	ው / ለ ኑ	¢ (0)	¢ <	¢/10		
Tax effects on unrealized gains (losses) result			\$ (4)	\$ (8)	\$6 1	\$ (16)		
Tax effects on realized (gains) losses on deriv	auve instruments recla	issined into ea	<u>anngs</u> 4	6	1	6		

Tax effects on unrealized gains (losses) on av	ailable-	for-sale securit	ies	0	0	0	1	
Tax effects on change in funded status of pos	t-retiren	ent benefit ob	ligations	\$ 0	\$0	\$ 0	\$ 0	
CONDENSED CONSOLIDATED STATEMENT OF CHANGES IN EQUITY (USD \$) In Millions, except Share data in Thousands	Total	Common Stock [Member]	Additional Paid- in Capital [Member]	Retained Earnings [Member]	Com	umulated Other orehensive Income oss) [Member]	Treasury Stock [Member]	Noncontrolling Interests [Member]
Beginning balance at Dec. 31, 2012	\$ 6,675	\$ 6	\$ 4,549	\$ 6,879	\$ 193		\$ (5,273)	\$ 321
Beginning balance, shares at Dec. 31, 2012		630,282					(166,062)	
Consolidated net income	729			703				26
Other comprehensive income (loss), net of taxes	(24)				(24)			
Cash dividends declared	(512)			(512)				
Equity-based compensation transactions, including dividend equivalents, net of taxes	186		32	(4)			158	
Equity-based compensation transactions, including dividend equivalents, net of taxes, shares							4,956	
Distributions paid to noncontrolling interests	(40)							(40)
Other	1		1					
Other, shares							5	
Ending balance at Sep. 30, 2013	\$ 7,015	\$ 6	\$ 4,582	\$ 7,066	\$ 169		\$ (5,115)	\$ 307
Ending balance, shares at Sep. 30, 2013		630,282					(161,101)	
Debt				9 Months Sep. 30,				

Debt Disclosure [Abstract]

Debt

3. Debt

The following table summarizes the major components of debt (in millions) at each balance sheet date and provides the maturities and interest rate ranges of each major category as of September 30, 2013:

September 30,

December 31,

		2013	 2012
\$2.25 billion revolving credit facility, maturing July 2018 (weighted average interest rate of 1.5% at	-		
September 30, 2013 and 1.4% at December 31, 2012)	\$	140	\$ 400
Letter of credit facilities, maturing through June 2015		_	_
Canadian credit facility and term loan, maturing November 2017 (weighted average effective interest rate			
of 2.7% at September 30, 2013 and 2.9% at December 31, 2012)		456	75
Senior notes maturing through 2039, interest rates ranging from 2.60% to 7.75% (weighted average			
interest rate of 5.7% at September 30, 2013 and December 31, 2012)		6,292	6,305
Tax-exempt bonds, maturing through 2045, fixed and variable interest rates ranging from 0.1% to 5.7%			
(weighted average interest rate of 2.4% at September 30, 2013 and 2.8% at December 31, 2012)		2,719	2,727
Capital leases and other, maturing through 2055, interest rates up to 12%		452	409
		10,059	 9,916
Current portion of long-term debt		568	743
	\$	9,491	\$ 9.173

Debt Classification

As of September 30, 2013, we had (i) \$510 million of debt maturing within the next 12 months, including \$350 million of 5.0% senior notes that mature in March 2014 and \$92 million of tax-exempt bonds; (ii) short-term borrowings and advances outstanding under credit facilities with long-term maturities, including \$140 million of borrowings outstanding under the revolving credit facility and U.S.\$456 million of advances under our Canadian credit facility term loan; and (iii) \$857 million of tax-exempt borrowings subject to repricing within the next 12 months. Based on our intent and ability to refinance a portion of this debt on a long-term basis as of September 30, 2013, we have classified \$1.4 billion of this debt as long-term and the remaining \$568 million as current obligations.

Revolving Credit and Letter of Credit Facilities

As of September 30, 2013, we had an aggregate committed capacity of \$2.65 billion for letters of credit under various credit facilities. Our revolving credit facility is our primary source of letter of credit capacity. In July 2013, we amended and restated our revolving credit facility, increasing our total credit capacity to \$2.25 billion and extending the term through July 2018. As of September 30, 2013, we had an aggregate of \$1.3 billion of letters of credit outstanding under various credit facilities. Approximately \$942 million of these letters of credit have been issued under our revolving credit facility. We had \$140 million of outstanding borrowings under our revolving credit facility as of September 30, 2013, leaving \$1.2 billion of unused and available capacity.

Significant Changes in Debt Balances

Revolving credit facility - During the nine months ended September 30, 2013, we have had net repayments under our revolving credit facility of \$260 million.

Canadian credit facility — We repaid U.S.\$75 million of net advances under our Canadian credit facility during the nine months ended September 30, 2013. Additionally, we borrowed C\$500 million, or U.S.\$470 million, under a term loan to fund our acquisition of RCI Environmement, Inc., which is discussed further in Note 9. Under the terms and conditions of our Canadian credit facility, interest on the term loan is based on a 1.4% spread from the applicable Canadian Dealer Offered Rate (CDOR). Our outstanding CDOR-based advances, which have terms of one month and three months, may be renewed under the terms of the facility through its maturity. We repaid U.S.\$29 million of advances under our term loan during the nine months ended September 30, 2013. Accounting for changes in the Canadian currency translation rate and the recognition of interest expense increased the carrying value of our Canadian borrowings by U.S.\$11 million and U.S.\$4 million, respectively, during the nine months ended September 30, 2013.

Tax-exempt bonds — During the nine months ended September 30, 2013, we repaid \$107 million of our tax-exempt bonds with cash. We issued \$100 million of tax-exempt bonds in August 2013. The proceeds from the issuance of the bonds were deposited directly into a trust fund and may only be used for the specific purpose for which the money was raised, which is generally to finance expenditures for landfill and recycling facility construction and development. Accordingly, the restricted funds provided by these financing activities have not been included in "New Borrowings" in our Condensed Consolidated Statement of Cash Flows.

Other — We had increases in our other obligations during the nine months ended September 30, 2013, primarily related to the deferred purchase price of (i) land needed to support a landfill expansion and (ii) Greenstar, LLC, which is discussed further in Note 9.

9 Months Ended Sep. 30, 2013

Basis of Presentation

Accounting Policies [Abstract] Basis of Presentation

1. Basis of Presentation

The financial statements presented in this report represent the consolidation of Waste Management, Inc., a Delaware corporation; Waste Management's whollyowned and majority-owned subsidiaries; and certain variable interest entities for which Waste Management or its subsidiaries are the primary beneficiaries as described in Note 14. Waste Management is a holding company and all operations are conducted by its subsidiaries. When the terms "the Company," "we," "us" or "our" are used in this document, those terms refer to Waste Management, Inc., its consolidated subsidiaries and consolidated variable interest entities. When we use the term "WM," we are referring only to Waste Management, Inc., the parent holding company.

We are North America's leading provider of comprehensive waste management environmental services. We partner with our residential, commercial, industrial and municipal customers and the communities we serve to manage and reduce waste at each stage from collection to disposal, while recovering valuable resources and creating clean, renewable energy. Our "Solid Waste" business is operated and managed locally by our subsidiaries that focus on distinct geographic areas and provides collection, transfer, recycling and resource recovery, and disposal services. Through our subsidiaries, we are also a leading developer, operator and owner of waste-to-energy facilities in the United States.

We currently evaluate, oversee and manage the financial performance of our Solid Waste business subsidiaries through our 17 geographic Areas. Our Wheelabrator business provides waste-to-energy services and manages waste-to-energy facilities and independent power production plants. We also provide additional services that are not managed through our Solid Waste or Wheelabrator businesses, which are presented in this report as "Other." Additional information related to our segments can be found in Note 8.

The Condensed Consolidated Financial Statements as of September 30, 2013 and for the three and nine months ended September 30, 2013 and 2012 are unaudited. In the opinion of management, these financial statements include all adjustments, which, unless otherwise disclosed, are of a normal recurring nature, necessary for a fair presentation of the financial position, results of operations, comprehensive income, cash flows, and changes in equity for the periods presented. The results for interim periods are not necessarily indicative of results for the entire year. The financial statements presented herein should be read in connection with the financial statements included in our Annual Report on Form 10-K for the year ended December 31, 2012.

In preparing our financial statements, we make numerous estimates and assumptions that affect the accounting for and recognition and disclosure of assets, liabilities, equity, revenues and expenses. We must make these estimates and assumptions because certain information that we use is dependent on future events, cannot be calculated with precision from available data or simply cannot be calculated. In some cases, these estimates are difficult to determine, and we must exercise significant judgment. In preparing our financial statements, the most difficult, subjective and complex estimates and the assumptions that present the greatest amount of uncertainty relate to our accounting for landfills, environmental remediation liabilities, asset impairments, deferred income taxes and reserves associated with our insured and self-insured claims. Actual results could differ materially from the estimates and assumptions that we use in the preparation of our financial statements.

Adoption of New Accounting Pronouncement

.....

1 3 6 4

Comprehensive Income — In February 2013, the Financial Accounting Standards Board issued amended authoritative guidance associated with comprehensive income, which requires companies to provide information about the amounts that are reclassified out of accumulated other comprehensive income by component. Additionally, companies are required to present significant amounts reclassified out of accumulated other comprehensive income by the respective line items of net income. The amendment to authoritative guidance associated with comprehensive income was effective for the Company on January 1, 2013. The adoption of this guidance did not have a material impact on our consolidated financial statements. We have presented the information required by this amendment in Note 12.

Reclassifications

When necessary, reclassifications have been made to our prior period consolidated financial information in order to conform to the current year presentation.

4 0.14

				3 Months	1 Months				9 Months				
Derivative Instruments and Hedging Activities - Additional Information (Detail) In Millions, unless otherwise specified	30, 2013	Dec. 31, 2012 USD (\$)	Sep. 30, 2013 5.0% Senior Notes Due March 2014 [Member] USD (\$)	Ended Sep. 30, 2012 Cash Flow Hedging [Member]	Ended Apr. 30, 2012 Interest Rate Swaps [Member] Fair Value Hedges [Member] USD (\$)	Fair Value Hedges [Member] USD (\$)	Dec. 31, 2012 Interest Rate Swaps [Member] Fair Value Hedges [Member] USD (\$) Derivative	Starting Interest Rate Swap [Member] Cash Flow Hedging [Member]	Cash Flow Hedging	Starting Interest Rate Swap [Member Cash Flow Hedging [Member	Starting Interest Rate Swap [Member] Cash Flow Hedging [Member]	Hedging [Member] Long- term other liabilities	Sep. 30, 2013 Forward Starting Interest Rate Swap [Member Current accrued liabilitie [Member Cash Flow Hedging [Member USD (\$)
<u>Derivative Instruments,</u> Gain (Loss) [Line Items]												USD (\$)	
Number of hedges outstanding						0	0						
Notional amount of derivatives					\$ 1,000	0	0	\$ 200			\$ 525		
Cash received (paid) for					+ -,			+ = • •					
termination of interest rate					76			59					
swaps													
Increase in carrying value of debt instruments from fair													
value hedge accounting for						64	79						
interest rate swaps													
Maximum term of cash flow									10 years				
hedges									10 years				
Deferred losses, net of taxes, related to cash flow hedges													
included in accumulated other									35	39			
comprehensive income													
Deferred losses scheduled to													
be reclassified out of accumulated other													
comprehensive into interest									7				
expense over next 12 months,													
pre-tax													
Ineffectiveness recognized				The									
upon termination of forward starting swaps				ineffectiveness recognized									
starting swaps				upon									
				-									

Senior notes Total long-term debt

Senior notes	6,292 6,303	termination of these hedges was immaterial.				
Interest rate of senior notes	0,272 0,50	5.00%				
Fair value of forward-starting		210070				
swaps (interest rate derivatives) Merchant electricity sales hedged or expected to be hedged	\$31 \$42					\$42 \$31
Percentage of merchant electricity sales hedged Expected percentage of merchant electricity sales						
hedged						
E				9 Months En	ded	
Earnings Per Share (Tables)				Sep. 30, 20	13	
Earnings Per Share [Abstrac						
Common Share Data Used for Earnings Per Share	Computing	Basic and Diluted	Basic and diluted earnings per share were co	omputed using the	following common share d	lata (shares in millions):
Latings rei Share					Three Months	Nine Months
			Number of common shares outstanding at en Effect of using weighted average common s Weighted average basic common shares out Dilutive effect of equity-based compensation contingently issuable shares Weighted average diluted common shares on Potentially issuable shares Number of anti-dilutive potentially issuable diluted common shares outstanding	hares outstanding standing n awards and other utstanding shares excluded fr	$ \begin{array}{r} 1.9 \\ 471.2 \\ 12.9 \\ 16.5 \end{array} $	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
Fair Value Measurements			9 Months End			
(Tables)			Sep. 30, 201	3		
Fair Value Disclosures [Abst						
Fair Value of Assets and Liabil Measured on Recurring Basis	<u>ities</u>	Our assets and liabilities that are n	neasured at fair value on a recurring basis incl	ude the following	(in millions):	
					Fair Value Meas	urements at
					September 30, 2 Quoted Significant	t
					Prices in Other Active Observable Markets Inputs	Inputs
		Assets:		Total	(Level 1) (Level 2)	
		Money market funds Fixed-income securities		\$120 37	\$ 120 \$ — — 37	\$ — 7 —
		Redeemable preferred stock		25		25
		Total assets		\$182	\$ 120	<u>\$ 25</u>
		Liabilities: Interest rate derivatives Electricity commodity deriva	tives	\$ 31 3	\$ \$ 31 3	
		Foreign currency derivatives			<u> </u>	<u> </u>
		Total liabilities		\$ 34	<u>\$ </u>	
					Fair Value Meas	
		Assets		Total	December 31, 2 Quoted Significant Prices in Other Active Observable Markets Inputs (Level 1) (Level 2)	t Significant e Unobservable Inputs
		Assets: Money market funds		\$127	Quoted Significant Prices in Other Active Observable Markets Inputs (Level 1) (Level 2) \$ 127 \$ —	t Unobservable Inputs (Level 3)(a) \$ —
		Money market funds Fixed-income securities		\$127 37	Quoted Significant Prices in Other Active Observable Markets Inputs (Level 1) (Level 2)	t Significant unobservable Inputs (Level 3)(a) \$ — 7 —
		Money market funds Fixed-income securities Redeemable preferred stock Electricity commodity deriva	tives	\$127 37 25 1	Quoted Significant Other Prices in Observable Active Observable Markets Inputs Level 1) (Level 2) \$ 127 \$	t Significant Unobservable Inputs (Level 3)(a) 5 — 25 1 — —
		Money market funds Fixed-income securities Redeemable preferred stock Electricity commodity deriva Total assets	tives	\$127 37 25	Quoted Significant Prices in Other Active Observable Markets Inputs (Level 1) (Level 2) \$ 127 \$ 37	t Significant Unobservable Inputs (Level 3)(a) 5 — 25 1 — —
		Money market funds Fixed-income securities Redeemable preferred stock Electricity commodity deriva	tives	\$127 37 25 1	Quoted Significant Other Prices in Observable Active Observable Markets Inputs Level 1) (Level 2) \$ 127 \$	t Unobservable Imputs (Level 3)(a) 7 — 7 — 8 <u>\$</u> 8 <u>\$ 25</u>
		Money market funds Fixed-income securities Redeemable preferred stock Electricity commodity deriva Total assets Liabilities: Interest rate derivatives Foreign currency derivatives		$ \$127 \\ 37 \\ 25 \\ 1 \\ $190 $ 42 \\ 11 $	Quoted Significant Prices in Other Active Observable Inputs (Level 1) \$ 127 \$ -37 -1 \$ 127 \$ -1 \$ 127 \$ 38 -1 \$ 127 \$ 38 -1 \$ 127 \$ 38 -1 \$ 127 \$ 38 -1	t e Jupobservable Inputs (Level 3)(a) 7
		Money market funds Fixed-income securities Redeemable preferred stock Electricity commodity deriva Total assets Liabilities: Interest rate derivatives		127 37 25 <u>1</u> \$190 \$42	Quoted Significant Prices in Other Active Observable Markets Inputs Level I) (Level 2) \$ 127 \$ - -37 - -1 \$ 127 \$ 38 \$ -11 \$ 127 \$ 38 \$ -12 \$ 38	t Unobservable Imputs (Level 3)(a) 7 7 8 9
	(a)	Money market funds Fixed-income securities Redeemable preferred stock Electricity commodity deriva Total assets Liabilities: Interest rate derivatives Foreign currency derivatives Electricity commodity deriva Total liabilities Level 3 investments have been me evidence of fair value currently av valuation methodologies may incl		$ \begin{array}{c} & \\ & \\ & \\ & \\ & \\ & \\ & \\ & $	Quoted Significant Prices in Other Active Observable Markets Inputs [Level J) (Level 2) \$ 127 \$ <td>t Unobservable Inputs (Level 3)(a) S = - 2 = - 3 = - 2 = - 2 = - 3 = - 3</td>	t Unobservable Inputs (Level 3)(a) S = - 2 = - 3 = - 2 = - 2 = - 3
Debt - Components of Debt (Detail) (USD \$) In Millions, unless otherwise specified	(a)	Money market funds Fixed-income securities Redeemable preferred stock Electricity commodity deriva Total assets Liabilities: Interest rate derivatives Foreign currency derivatives Electricity commodity deriva Total liabilities Level 3 investments have been me evidence of fair value currently av valuation methodologies may incl	tives asured based on third-party investors' recent d ailable. When this evidence is not available, v ade transactions in similar instruments, discou ot been any significant changes in the fair valu	$ \begin{array}{c} & \\ & \\ & \\ & \\ & \\ & \\ & \\ & $	Quoted Significant Prices in Other Active Observable Markets Inputs [Level J) (Level 2) \$ 127 \$ <td>t Unobservable Inputs (Level 3)(a) S = - 2 = - 3 = - 2 = - 2 = - 3 = - 3</td>	t Unobservable Inputs (Level 3)(a) S = - 2 = - 3 = - 2 = - 2 = - 3

http://markets.on.nytimes.com//research/modules/company_topic/drawFiling.asp?docKey... 11/25/2013

\$ 6,292

10,059

\$ 6,305

9,916

Form 10-Q

Current portion of long-term debt	568	743			
Long-term debt, less current portion	9,491	9,173			
Capital leases and other [Member]					
Debt Instrument [Line Items]					
Capital leases and other	452	409			
Nontaxable Municipal Bonds [Member]					
Debt Instrument [Line Items]					
Capital leases and other	2,719	2,727			
Revolving credit facility [Member]					
Debt Instrument [Line Items]	1.40	100			
Outstanding borrowings under revolving credit facilit	<u>v</u> 140	400			
Letter of Credit Facilities [Member]					
Debt Instrument [Line Items]					
Letter of credit facilities Canadian credit facility and term loan [Member]					
Debt Instrument [Line Items]					
Outstanding borrowings under revolving credit facilit	v \$ 456	\$ 75			
Accumulated Other	<u>y</u> \$ 450	φ15			
Comprehensive Income -			3 Months Ended	0 N	Ionths Ended
Component of Accumulated			5 Months Ended	<i>,</i> 11.	iontilis Enucu
Other Comprehensive					
Income (Detail) (USD \$) In Millions, unless otherwise		Sep	. 30, 2013 Sep. 30, 20	12 Sep. 30.	2013 Sep. 30, 2012
specified			· · · , · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · · · · · · · ·
Accumulated Other Comprehensive Income (Loss) [Line Items]				
Accumulated Other Comprehensive Income (Loss), E	Beginning balan	<u>ce</u>		\$ 193	
Other comprehensive income (loss) before reclassific	ations			(26)	
Amounts reclassified from accumulated other compre	hensive income	<u>e</u>		2	
Other comprehensive income (loss), net of taxes		35	37	(24)	28
Accumulated Other Comprehensive Income (Loss), E	Ending balance	169		169	
Post - Retirement Benefit Plans [Member]					
Accumulated Other Comprehensive Income (Loss					
Accumulated Other Comprehensive Income (Loss), B		ce		(13)	
Other comprehensive income (loss) before reclassific					
Amounts reclassified from accumulated other compre	chensive income	<u>e</u>			
Other comprehensive income (loss), net of taxes		(10)		(10)	
Accumulated Other Comprehensive Income (Loss), E		(13)		(13)	
Foreign Currency Translation Adjustments [Member] Accumulated Other Comprehensive Income (Loss					
Accumulated Other Comprehensive Income (Loss)		00		276	
Other comprehensive income (loss) before reclassific				(37)	
Amounts reclassified from accumulated other compre-		_		(37)	
Other comprehensive income (loss), net of taxes	mensive meonic	<u> </u>		(37)	
Accumulated Other Comprehensive Income (Loss), E	Ending balance	239		239	
Unrealized Gain and Losses on Available-for-Sale Se	-			207	
Accumulated Other Comprehensive Income (Loss					
Accumulated Other Comprehensive Income (Loss), B		ce		4	
Other comprehensive income (loss) before reclassific	ations			1	
Amounts reclassified from accumulated other compre	hensive income	<u>e</u>			
Other comprehensive income (loss), net of taxes				1	
Accumulated Other Comprehensive Income (Loss), E	Inding balance	5		5	
Gains and Losses on Derivative Instruments [Member	r]				
Accumulated Other Comprehensive Income (Loss					
Accumulated Other Comprehensive Income (Loss), B		<u>ce</u>		(74)	
Other comprehensive income (loss) before reclassific				10	
Amounts reclassified from accumulated other compre-	hensive income	<u>e</u>		2	
Other comprehensive income (loss), net of taxes				12	
Accumulated Other Comprehensive Income (Loss), E	inding balance	\$ (6	2)	\$ (62)	